Mike Jenks · Colin Jones *Editors*



Dimensions of the Sustainable City

FUTURE CITY

Advisory Board
Jack Ahern, University of Massachusetts, Department of Landscape Architecture and
Regional Planning, Amherst, MA, USA
John Bolte, Oregon State University, Biological & Ecological Engineering Department,
Corvallis, OR, USA
Richard Dawson, University of Newcastle upon Tyne, School of Civil Engineering &
Geosciences, Newcastle upon Tyne, UK
Patrick Devine-Wright, University of Manchester, School of Environment and Development,
Manchester School of Architecture, Manchester, UK
Almo Farina, University of Urbino, Institute of Biomathematics, Faculty of Environmental
Sciences, Urbino, Italy
Raymond James Green, University of Melbourne, Faculty of Architecture, Building &
Planning, Parkville, VIC, Australia
Glenn R. Guntenspergen, National Resources Research Institute, US Geological Survey,
Duluth, MN, USA
Dagmar Haase, Helmholtz Centre for Environmental, Research GmbH – UFZ, Department
of Computational Landscape Ecology, Leipzig, Germany
Michael Jenks, Oxford Brookes University, Oxford Institute of Sustainable Development,
Department of Architecture, Oxford, UK
Cecil C. Konijnendijk, woodSCAPE Consult, Dragoer, Denmark
Joan Iverson Nassauer, University of Michigan, School of Natural Resources and Environment,
Landscape Ecology, Perception and Design Lab, Ann Arbor, MI, USA
Stephan Pauleit, University of Copenhagen, Centre for Forest, Landscape & Planning,
Frederiksberg C, Denmark
Steward T.A. Pickett, Cary Institute of Ecosystem Studies, Millbrook, NY, USA
Robert Vale, Victoria University of Wellington, School of Architecture and Design,
Wellington, New Zealand
Ken Yeang, Llewelyn Davies Yeang, London, UK
Makoto Yokohari, University of Tokyo, Graduate School of Sciences, Institute of
Environmental Studies, Department of Natural Environment, Kashiwa, Chiba, Japan

Future City Description

As of 2008, for the first time in human history, half of the world's population now live in cities. And with concerns about issues such as climate change, energy supply and environmental health receiving increasing political attention, interest in the sustainable development of our future cities has grown dramatically.

Yet despite a wealth of literature on green architecture, evidence-based design and sustainable planning, only a fraction of the current literature successfully integrates the necessary theory and practice from across the full range of relevant disciplines.

Springer's *Future City* series combines expertise from designers, and from natural and social scientists, to discuss the wide range of issues facing the architects, planners, developers and inhabitants of the world's future cities. Its aim is to encourage the integration of ecological theory into the aesthetic, social and practical realities of contemporary urban development.

For further volumes: http://www.springer.com/series/8178 Mike Jenks · Colin Jones Editors

Dimensions of the Sustainable City



Editors Prof. Mike Jenks Oxford Brookes University Oxford Inst. Sustainable Development Dept. Architecture Gipsy Lane Oxford Headington United Kingdom OX3 0BP mjenks@brookes.ac.uk

Prof. Colin Jones Heriot-Watt University School of the Built Environment Edinburgh United Kingdom EH14 4AS c.a.jones@hw.ac.uk

ISBN 978-1-4020-8646-5(HB) e-ISBN 978-1-4020-8647-2 ISBN 978-1-4020-8645-8(PB) DOI 10.1007/978-1-4020-8647-2 Springer Dordrecht Heidelberg London New York

Library of Congress Control Number: 2009938191

© Springer Science+Business Media B.V. 2010

No part of this work may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, microfilming, recording or otherwise, without written permission from the Publisher, with the exception of any material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work.

Cover illustrations: (front) Urban infill on brownfield land can intensify activity in the city and help towards a more sustainable environment (Oxford City centre, UK). Photo: Mike Jenks; (back) An efficient trans system can encourage people to use public transport and reduce traffic in the city (Sheffield City centre, UK). Photo: Mike Jenks.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

Preface

This book represents the overarching output from the *CityForm: Sustainable Urban Form Consortium* funded by the Engineering and Physical Sciences Research Council (EPSRC) under its Sustainable Urban Environments Programme (Grant number GR/520529/01). The consortium comprised a multi-disciplinary team based at the universities of De Montfort, Heriot-Watt, Oxford Brookes, Sheffield and Strathclyde. The consortium was supported by a large number of non-academic partners – Corus, the Environment Agency, Glasgow City Council, Groundwork UK, The Landscape Institute, Leicester City Council, Oxford City Council, The Prince's Foundation, Rogers Stirk Harbour + Partners, Scottish Executive, Sheffield City Council, Sheffield Wildlife Trust, Strathclyde Passenger Transport, and URBED.

The core of the research explored the relationship of five strands representing the environmental (energy and ecology), transport, and social and economic dimensions of urban sustainability to urban form. There were also three satellite projects that examined ways of adapting the city, whether sustainable developments lead to sustainable lifestyles and the ecological and psychological value of urban green space.

Many of the papers on which this book is based are to be found on the CityForm web page, www.cityform.org.uk. The editors of the book were the co-champions of the consortium.

Oxford, UK Edinburgh, UK Mike Jenks Colin Jones

Acknowledgements

Our thanks go to all those involved in the writing and production of this book. We are grateful to the CityForm team who contributed their chapters - there are so many and their names can all be found in list of authors. Their patience in responding to our many requests is greatly appreciated. The research, on which this book is based, owes its success to them, and to the way in which everyone collaborated with such good spirit.

We are indebted to the EPSRC for the grant support and for the wisdom of bringing us together in the CityForm research consortium. We could not have undertaken the research without them, and we gratefully acknowledge the role of Peter Hedges in bringing the EPSRC SUE (Sustainable Urban Environments) programme to fruition. Many others have also provided us with support for which we are grateful, and particularly our appreciation goes to Charlotte Phillips of the Ordnance Survey for helping us with access to Mastermap, and to our publishers Springer, in particular Catherine Cotton and Ria Kanters for all their help.

The tasks involved in preparing this book have been onerous. Our appreciation goes to colleagues in the Oxford Institute for Sustainable Development at Oxford Brookes University, UK, especially Carol Dair, Nicola Dempsey and Shibu Raman and in the School of the Built Environment at Heriot-Watt University with a special mention to Caroline Brown.

Our deepest gratitude goes to our families and friends for their support during the preparation of the book. In particular we would like to give our warmest thanks to Margaret Jenks and Fiona Jones for all their support and forbearance during the intensive work in bringing this book to fruition.

Contents

1	Issues and Concepts	1
2	Elements of Urban Form	21
3	Travel and Mobility	53
4	Environment and Biodiversity	75
5	Social Acceptability	105
6	Energy Use	129
7	Economic Viability	145
8	Adapting the CityHildebrand Frey and Samer Bagaeen	163
9	Neighbourhood Design and Sustainable Lifestyles	183
10	Ecological and Psychological Value of Urban Green Space Katherine N. Irvine, Richard A. Fuller, Patrick Devine-Wright, Jamie Tratalos, Sarah R. Payne, Philip H. Warren, Kevin J. Lomas and Kevin J. Gaston	215
11	Complementarities and Contradictions	239

Appendix A: MCA Spatial Analysis	257
Appendix B: Overview of Case Study Neighbourhoods	261
Index	267

Contributors

Samer Bagaeen School of Environment and Technology, University of Brighton; formerly at Department of Architecture, University of Strathclyde, samer.bagaeen@uclmail.net

Keith Baker Research Associate, Scottish Institute for Sustainable Technology (SISTECH); formerly PhD student, Institute of Energy and Sustainable Development, De Montfort University, keith.baker@sistech.co.uk

Glen Bramley Professor of Urban Studies and Director of Centre for Research into Socially Inclusive Services (CRSIS), School of the Built Environment, Heriot-Watt University, g.bramley@sbe.hw.ac.uk

Caroline Brown Research Associate, CRSIS, School of the Built Environment, Heriot-Watt University, C.J.Brown@hw.ac.uk

Carol Dair Research Fellow, Director of OISD (Oxford Institute for Sustainable Development):Cities, Oxford Brookes University, cmdair@brookes.ac.uk

Richard G. Davies School of Biological Sciences, University of East Anglia; formerly Department of Animal and Plant Sciences, University of Sheffield, richard.g.davies@uea.ac.uk

Nicola Dempsey I'DGO TOO Postdoctoral Researcher, CityForm Consortium Project Manager, Oxford Institute for Sustainable Development, Oxford Brookes University, ndempsey@brookes.ac.uk

Patrick Devine-Wright Reader, Manchester University, at the School of Environment and Development and Manchester School of Architecture, pdwright@manchester.ac.uk

Neil Ferguson Department of Civil Engineering, University of Strathclyde, n.s.ferguson@strath.ac.uk

Hildebrand Frey Consultant; formerly at Department of Architecture, University of Strathclyde, hildebrand.frey@orange.fr

Richard A. Fuller The Ecology Centre, University of Queensland, Australia; formerly Department of Animal and Plant Sciences, University of Sheffield, r.fuller@uq.edu.au

Kevin J. Gaston Professor of Biodiversity and Conservation, Biodiversity & Macroecology Group, Department of Animal and Plant Sciences, University of Sheffield, k.j.gaston@sheffield.ac.uk

Katherine N. Irvine Research Fellow, Institute of Energy and Sustainable Development, De Montfort University, KIrvine@dmu.ac.uk

Mike Jenks Professor Emeritus, Oxford Institute for Sustainable Development, Oxford Brookes University, mjenks@brookes.ac.uk

Colin Jones Professor of Estate Management, School of the Built Environment, Heriot-Watt University, C.A.Jones@sbe.hw.ac.uk

Chris Leishman Department of Urban Studies, University of Glasgow; formerly School of the Built Environment, Heriot-Watt University, c.leishman@lbss.gla.ac.uk

Morag Lindsay Research Assistant, Oxford Institute for Sustainable Development, Oxford Brookes University, dmlindsay@brookes.ac.uk

Kevin J. Lomas Professor, Department of Civil and Building Engineering, Loughborough University; formerly Director, Institute of Energy and Sustainable Development, De Montfort University, k.j.lomas@lboro.ac.uk

Charlie MacDonald Formerly Research Associate, School of Built Environment, Heriot-Watt University, charlotte.macdonald@scotland.gsi.gov.uk

Allison Orr Senior Lecturer, Department of Urban Studies, University of Glasgow, a.orr@lbss.gla.ac.uk

Sarah R. Payne Postgraduate Research Student, Manchester Architecture Research Centre, School of Environment and Development, University of Manchester; formerly Department of Animal and Plant Sciences, University of Sheffield, Sarah.Payne@postgrad.manchester.ac.uk

Aleksandra Pępkowska Institute of Nature Conservation, Krakow, Poland; formerly Department of Animal and Plant Sciences, University of Sheffield.

Sergio Porta Professor of Urban Design, UDSU – Urban Design Studies Unit, Dept. of Architecture, University of Strathclyde, Glasgow; Director of Human Space Lab, Politecnico di Milano, sergio.porta@strath.ac.uk

Sinead Power Senior Researcher, Children, Young People and Social Care Team, Scottish Government; formerly Research Associate, School of the Built Environment, Heriot-Watt University, sinead.power@scotland.gsi.gov.uk

Shibu Raman Research Fellow, Oxford Institute for Sustainable Development, Oxford Brookes University, skraman@brookes.ac.uk

Mark Rylatt Senior Research Fellow, Institute of Energy and Sustainable Development, De Montfort University, rylatt@dmu.ac.uk

Jamie Tratalos Senior Research Consultant, Dr Foster Research; formerly Department of Animal and Plant Sciences, University of Sheffield, jamie_tratalos@hotmail.com

Philip H. Warren Department of Animal and Plant Sciences, University of Sheffield, p.warren@sheffield.ac.uk

David Watkins Research Associate, CRSIS, School of the Built Environment, Heriot-Watt University, d.s.watkins@sbe.hw.ac.uk

Katie Williams Professor of Spatial Planning, Director of the Centre for Environment and Planning, Department of Planning and Architecture, University of the West of England, Katie4.Williams@uwe.ac.uk

Lee Woods Department of Civil Engineering, University of Portsmouth; formerly Department of Civil Engineering, University of Strathclyde, lee.woods@port.ac.uk

Chapter 1 Issues and Concepts

Mike Jenks and Colin Jones

What is Meant by the Sustainable City?

There has been a considerable amount of research that defines and characterises the form of the sustainable city, and which urban forms may most affect sustainability. It is a complex issue. The physical dimensions of urban form may include its size, shape, land uses, configuration and distribution of open space – a composite of a multitude of characteristics, including a city's transportation system and urban design features (e.g. Handy, 1996; Llewelyn-Davies, 2000). However, its sustainability depends on more abstract issues – environmental (including transport), social and economic. Research suggests that, not one, but a number of urban forms may be sustainable (Williams et al., 2000). Yet much of the debate about the sustainability of cities and urban forms has focused on increasing the density of development, ensuring a mix of uses, containing urban 'sprawl' and achieving social and economic diversity and vitality – often characterised as the concept of a 'compact city' (see Jenks et al., 1996; Jenks and Dempsey, 2005).

In the UK, government policy embodies such principles through its Urban White Paper (DETR, 2000a; DCLG, 2006a), mostly based on the report of the Urban Task Force (1999). Thus in the UK, a dominant paradigm is being implemented in many towns and cities. It is for more compact, high-density and mixed use urban forms, and the belief is that they will be sustainable. This book will take this type of urban form as its starting point and will test the claims made for it.

Urban Form and Claims to Sustainability

With the implementation of policy, practice has, to an extent, overtaken the knowledge and evidence needed to assure the success of sustainable urban forms. Notwithstanding examples of good practice, advocacy rather than research has often

M. Jenks (🖂)

Oxford Institute for Sustainable Development, Oxford Brookes University, Oxford, UK

characterised the debate, and has led to many positive claims for the influence and benefits of different urban forms.

The European Commission was an early and influential advocate of urban containment and more compact forms (CEC, 1990). The hypothesis was that compact urban forms would reduce urban sprawl, protect agricultural and amenity land, and lead to more efficient use of existing, previously developed urban land. With a mixture of uses in much closer proximity, alternative modes of travel would be encouraged, such as walking and cycling, and public transport use would also increase. This in turn would lead to environmental, social and economic benefits. Such ideas were taken up in UK government strategy (HM Government, 1994), strongly advocated as a basis for policy and implementation (Urban Task Force, 1999; DETR, 2000a) and in numerous government publications giving planning guidance (e.g. ODPM, 2004a, b, DCLG, 2006a, b), and policy guidance (e.g. CABE, 2005). Similar urban form concepts to achieve urban sustainability can also be found in, for example, the USA with New Urbanism and Smart Growth initiatives (Katz, 1994; Smart Growth, 2008). All tend to advocate urban forms that are higher than previous densities, with mixed use, which are contained in order to reduce travel distances and dependence on private transport, as well as being socially diverse and economically viable.

The effect of some of these proposed policies and 'solutions' has become evident. In the UK over the past five years there has been an increased take up of brownfield land, fuelled by government targets for building 60% of new homes on re-used urban land (DETR, 2000a, 2001). However, the intensive use of existing land means there is a potential loss of open space and amenity. Environmentally, less open space is likely to have adverse effects on biodiversity and the provision of environmental/ecosystem services (e.g. water and drainage), the consequences of which are largely unexplored. Socially, the impact of 'intensification' or compaction may affect the quality of life of users, and the effects may in some respects fall unevenly on the poor (Burton, 2000a).

Higher densities are strongly advocated, and on the ground there are examples of schemes built and being proposed at higher densities than recommended in guidance (CABE, 2005), of over 100 dwellings/ha (Owers and Oliver, 2001; Dawson, 2004). Theoretical studies have been undertaken on physical capacity, showing how reducing parking provision can increase density (Llewelyn-Davies, 1997; Stubbs and Walters, 2001), and examples of 'car free' housing have been built (Canmore Housing Association, 2008). An active Millennium Communities Programme (English Partnerships, 2007), emerging from the government's Millennium Villages initiatives, has identified seven potential developments, two of which (Greenwich Millennium Village, and Allerton Bywater) are being built. More recently the UK government has announced a competition to build a number of 'eco-towns' comprising zero carbon development as a move to support the UK's carbon reduction targets, and has short listed 15 potential sites for these developments (DCLG, 2007). The 'eco-towns' have been contested in sustainability terms, and are controversial, particularly where they are to be situated on green field sites (Revill and Davies, 2008).

Land use and built form	Environmental – energy conservation	Environmental – recycling and re-use	Communication and transport	
 Intensive use of urban land Networks of green corridors Community buildings, self-managed Mixture of land uses at relatively high density Affordable homes Local identity Sustainable building materials Flexible design and good space standards Improved noise insulation 	 Combined heat and power (CHP) - local power generation Micro power generation Renewable energy Reduced energy consumption and embodied energy High levels of insulation Intelligent lighting and integrated security, heating, and IT systems 'A' rated white goods Eco-rating e.g. BREEAM 'excellent' 	 'Grey' water systems Recycle water for gardening and car washing Reuse water and filter, to be directed to ecology parks or green spaces Waste recycling, and use for production of biogas Reduced domestic and construction waste Carbon-neutral lifestyle 	 Light transit routes, eco-friendly buses and bikeways Car clubs and cycle facilities Pedestrian-friendly infrastructure Restricted car parking Environmental advice – bus/transit times, energy and water monitoring IT enabled 	

Table 1.1 Aspects of a sustainable built environment (Sources draw upon, inter alia: CABE, 2008;English Partnerships, 2007)

The common features of these many initiatives that are claimed to contribute to sustainability are based on a general consensus of opinion, as well as some evidence, and are outlined in Table 1.1, above. These are largely the physical and environmental aspects of sustainability, and are those aspects that are probably easiest to incorporate into developments - not all of them relate to urban form those that do are highlighted in the table. Generally, only these relevant aspects have been included in the core research reported in this book. However, these have not all been tested to see if they work in practice, and little has been done to show how they interact or integrate as a whole. So while this more intense development may have benefits of providing a more viable model or public transport, it is not clear from experience to date that there has been the modal shift in transport use necessary to yield the claimed environmental benefits. And there are social sustainability arguments underlying these physical aspects, for example high densities and good public transport are linked to social benefits of ease of access to facilities. But more difficult to achieve are safe, inclusive and equitable environments - the link to urban form has received little empirical attention and may well be tenuous. Nor is there evidence to show a clear link between such forms and economic viability. These are therefore not included in Table 1.1: the social and economic aspects are a key to the research that follows and will be placed in context at the end of the book.

An issue of Global Significance

Despite the lack of empirical research to support such claims, it seems evident, at least in the UK, Europe and much of the western world, these policies and 'solutions' provide a useful model and are being implemented in practice. However they are untested, raising questions about their validity. For every positive claim there are also potentially negative impacts to consider. Higher densities may lead to overcrowding, more traffic, and may not be the favoured choice for residents wishing to purchase homes – populations trends still tend to favour rural or suburban locations (e.g. Breheny, 1995). Also, as indicated above, the social and economic impacts are hard to trace to the influence of urban form – many other factors are more important.

It is surprising, therefore, that similar ideas have resonated beyond the western world, and have been taken up by countries where the urban context is very different (e.g. Jenks and Burgess, 2000). Across the world there are some 60 metropolitan regions with populations of more than 5 million inhabitants. Of these regions 46% have populations in excess of 10 million, the largest being the Tokyo Metropolitan Region with more than 36 million people. Most of these large regions, some 62%, are to be found in Asia where the growth rate is fast (World Gazetteer, 2005). These vast urban regions, with either very large urban agglomerations or huge cities at their core, are a far cry from the comforting compact city model inspired by the historic cores of relatively small European cities. These cities often have not just one, but many centres, and these are frequently decentralised, disconnected and fragmented forms (e.g. Graham and Marvin, 2001; Kozak, 2008). In this global context it suggests that the idea of a 'traditional' compact city could be seen as a contradiction in terms.

Nevertheless, some policies and forms that could be appropriate in the world context where cities are experiencing rapid growth are beginning to emerge. In these contexts urban form needs to be considered at a larger and more strategic scale. For example, linking public transportation and development, which may take a number of different forms, including the intensification (or densification) of development around transport interchanges - what has been termed 'transit oriented development' (e.g. Boarnet and Crane, 2001; Calthorpe, 1993; Cervero, 1998; Low and Gleeson 2003). Major transport routes may also provide the opportunity for denser development along the routes, and this has been successfully undertaken in Curitiba in Brazil (Acioly, 2000; Curitiba websites) and in Bogota, Colombia with their 'Transmilenio' bus rapid transit system. When many transportation nodes at higher densities, linked by transportation routes combine into a city-wide strategy, a polycentric form may emerge. Such planned and controlled polycentric development may have the potential to achieve more sustainable forms, provided all the 'nodes' are linked with efficient public transport, although again, these forms have not been rigorously tested (e.g. Jenks, et al., 2008; Lambregts and Zonneveld, 2003; Urban Task Force, 1999).

Within these large urban contexts, fast-growing cities and mega-cities, there are also smaller scale attempts at creating the sustainable city. Indeed the claims for such developments, similar in concept but larger in scale, are often loud and assertive. In the Shanghai metropolitan region, the 'world's first sustainable city' has been proposed – Dongtan (Arup, 2008; Dongtan Development Co., 2008). This ticks most, if not all the boxes in the table above, yet in effect it is a relatively low density (for China) suburb of Shanghai. Even more surprising, in the United Arab Emirates, two eco-city initiatives are underway - Masdar in Abu Dhabi designed by Foster and Parnters (Basantani, 2008; Foster and Partners, 2008) and an eco-city in Ras Al Khaima, designed by Rem Koolhaas and OMA (Trotter, 2008; OMA, 2008). These are 'stand-alone' models, again ticking all the boxes and edging towards zero carbon development. There is a danger that these 'models' will become eco-theme parks, functioning in a similar way to carbon offset schemes, salving the conscience, and freeing the neighbouring cities to continue business as usual development. Obviously these eco-cities (like the eco-towns in the UK) will be a help if all new development is zero carbon, but most places and most cities are already existing, and often very large. Sustainable development and sustainable urban forms need to pervade the whole environment, and not simply be exemplars that are separated from every day, messy reality.

Understanding the dimensions of the sustainable city is a complex issue. Care needs to be exercised over the context within which the cities exist, their cultural background and regional and national differences. There will be significant differences in different parts of the world of the interpretation of the sustainable city; however there are common underlying and enduring themes that appear to inform both the debate about and claims for urban forms that promote sustainability. Overall, research indicates that there are unlikely to be single spatial or physical solutions, rather that there may be many forms that can achieve sustainability, depending on the context in which they are applied (e.g. Guy and Marvin, 2000; Jenks and Dempsey, 2005).

The Aim of the Book

This book is about experience in the UK, but it picks up on many of the common themes and issues experienced in cities worldwide and presents research that assesses and measures them. Thus, in order to make progress, the research reported here measures and characterises urban form so it can be related to environmental, social and economic sustainability, and comparatively analyses different forms. The research concentrates on the physical design of urban form with respect to: physical configuration and layout, including links to the wider urban system; its land uses and functions; the typology and density of built form and presence of open space. The claims made that more compact, high-density and mixed use urban forms are environmentally sound, efficient for transport, socially beneficial and economically viable will then be tested. Each of these aspects is considered below, the current state of knowledge is briefly reviewed and gaps in knowledge identified.

Researching the Dimensions of the Sustainable City

The book explores the relationships between urban forms and environmental, social and economic sustainability. The major research question addressed is: *To what extent and in what ways does urban form contribute to sustainability?* This question is important to tackle and addresses several gaps in knowledge identified from analysis and review of research in the field. The sections below indicate some of the key problems that led to this research question.

Environmental Sustainability

Environmental benefits are claimed to accrue from more compact urban forms where concentration of uses means less need to travel and therefore lower emissions from vehicles (see Transport below). In addition, claims about higher densities suggest benefits in energy savings through combined heat and power (CHP) provision, but that benefits might be outweighed by the loss of open space. In assessing aspects of environmental sustainability the research focussed on the different patterns of provision of open, and especially green, space. The environmental benefits of open green spaces include: reduced surface and air temperatures, due to solar shading, free radiation to the night sky and evapotranspiration from trees leading to improved summertime thermal comfort (Vu et al., 1998); a haven from urban pollution and noise (Tyrväinen, 1997); and, buffering against wind reducing wind chills (Lacy, 1977). Benefits extend to the surrounding buildings, e.g.: heightened market value (Savard et al., 2000); improved access to natural light, reducing lighting loads in non-domestic buildings (Crisp et al., 1988); provision of passive solar heat to dwellings (Yannis, 1994); and, a low-noise and low-pollution source of fresh air allowing natural, rather than mechanical ventilation. Finally, green spaces can ameliorate the urban heat island effect and provide 'free' cooling to buildings (Watkins et al., 2002) - thus reducing the use of air conditioning systems thereby lowering energy consumption and CO_2 emissions, running costs and further anthropogenic heat release into the city.

The ecological benefits suggested for open green space include: the provision of ecosystem/environmental services with consequences for such diverse issues as flood control, waste management, and pest control (Bolund and Hunhammar, 1999; Attwell, 2000; Pauleit and Duhme, 2000); the provision of habitat for biodiversity (Gilbert, 1989; Savard et al., 2000; Kinzig and Grove, 2001); and, heightened awareness of environmental issues among users (Cannon, 1999; Savard et al., 2000).

Claimed social benefits of access to green space encompass a range of quality of life dimensions, including: improved human health and well-being (Ulrich, 1981; Ulrich et al., 1991; Parsons et al., 1998); opportunities for social interaction and group activities and a possible reduction in crime (Whyte, 1980; Skjaeveland and Garling, 1997; Tinsley et al., 2002); strengthened feelings of neighbourhood attachment and local community (Bonaiuto et al., 1999; Langdon, 1994) and the

promotion of civic pride and sense of place (Duany and Plater-Zyberk, 1992) and providing opportunities for contact with nature (Burgess et al., 1988).

Despite this plethora of perceived benefits, there has been relatively little work done in UK cities to assess whether they materialise in practice. Little empirical research quantifies the magnitude of the benefit or determines who are the beneficiaries. There is little work on the effect which urban form has on the thermal comfort in open spaces and the energy and environmental implications for UK cities (except Regents Park, London (Watkins et al., 2002)). Neither have all the energy and emissions implications been studied simultaneously (i.e. heating of dwellings, cooling and lighting of non-domestic buildings) for particular combinations of building types, urban forms and open spaces.

While there is an extensive literature focusing on the social benefits, and usage, of managed urban green spaces (e.g. Tinsley et al., 2002; Faber et al., 1998; Payne et al., 2002; Burgess et al., 1988), there has been relatively little empirical investigation of the social benefits of non-managed public spaces and private gardens. Furthermore, very little is known about whether or how the perceived social benefits and usage of urban green spaces varies with the density of the built environment (investigated by only one empirical study by Syme et al., 2001), nor how frequently local green spaces are used and what benefits are perceived to be associated with them.

Transport

Key policy (DETR, 2000a) and planning guidelines in the UK (DTLR, 2001) suggest that planning can reduce both the need to travel and length of journeys, and give safer and easier access to facilities through more compact, higher density and mixed use forms. The benefit of these forms is inter alia claimed to reduce car use, and encourage a shift towards more sustainable modes of travel, such as walking, cycling, and through increased use of public transport. However, such transport benefits crucially depend on people changing their travel behaviour.

Links between urban form and travel behaviour were identified, for example, by Handy (1996) and Badoe and Miller (2000). Handy (1996) found that residents in traditional neighbourhoods, characterised by higher densities, better accessibility and pedestrian- friendly design exhibited more sustainable travel behaviour than residents of neighbourhoods with lower densities, poor accessibility and pedestrian unfriendly design. Urban form, as measured by density, diversity (i.e. mix) and design, exerted a *modest to moderate* influence on travel demand, and that compact, mixed-use, pedestrian friendly designs were associated with more sustainable travel patterns (e.g. Cervero, 1998). In the UK, Stead et al. (2000) found that socioeconomic characteristics typically explained around half of the variation in travel distance per person across different study areas, whilst land use characteristics explained around one third of the variation.

However, finding a significant association of this kind between urban form and travel demand does not in itself mean that urban form per se will bring about more sustainable travel behaviour. The potential for the 'compact city' to bring about this result was questioned by Simmonds and Coombe (2000). In a study using a transport model of the Bristol area, a number of urban form scenarios were tested and found to make little impact on both total travel and car travel when viewed against the trend of increasing car ownership and the low cost of travel by car. These factors have greatly weakened the influence of spatial segregation on travel behaviour, and have presented many individuals with a far wider choice of residential location, job opportunities, services and other activities than ever before. Thus, the policies of densification and mixed-use development are likely to have little effect on travel behaviour without appropriate supporting transport policies.

The gaps in research suggest the need to seek a better understanding of the impact of urban form on travel behaviour to enable more accurate predictions of the travel patterns resulting from changes in urban form to be made. Many previous studies that sought to do this can be criticised on either methodological or theoretical grounds (see Badoe and Miller (2000) for recent critiques).

A common failing is to give inadequate consideration (or none at all) to the effects of factors representing the availability and quality of transport options, such as travel times and journey costs. Many previous studies have used data aggregated at the local level which serves to mask the effect of within-area variability. Disaggregated analyses, which use individual and household personal and travel/activity data and urban form measured at the local or the household level, are better suited to reveal the underlying complexities of the relationship between urban form and travel behaviour. Ultimately, the development of the most appropriate balance of urban form policies for the delivery of more sustainable travel patterns in a given context requires the application of sound theory derived from causal inferences. The research reported here addresses how (and why) individual, or sets of interrelated, elements of urban form influence travel behaviour and in what way socioeconomic characteristics interact with urban form to influence travel behaviour.

Social Benefits

The claims about the influence of urban form on social sustainability are complex and embrace issues of both quality of life and social equity. Higher densities and mixed use urban forms will, it has been argued, lead a better quality of life due to more social interaction, community spirit and cultural vitality (e.g. Rudlin and Falk, 1999), in part due to 'proximity to work, shops and basic social, educational and leisure facilities' (Urban Task Force, 1999: 64). Having a variety of uses and the means to access them nearby is also seen as a key to achieving social equity (ibid: 45), especially for the more disadvantaged in society who may not have the resources (and for those who do not wish) to own a car. Burton (2000b) identifies a number of claims that positively link compact urban form to social equity. These claimed benefits include better access to facilities and jobs, better public transport and opportunities for walking and cycling, lower levels of social segregation and less crime. But alongside the benefits, there are claims that compaction leads to negative impacts such as poorer access to green spaces, poorer health, reduced living space and less affordable housing (ibid).

The evidence about these many claims raises a number of key questions for the research reported here. One clear message is that there is no single answer, as benefits (or costs) depend on the type of urban form and its social context. For example, while inner city mixed use areas might achieve benefits of more social interaction and vitality, and better access to facilities, they also could suffer from social tensions, crime or fear of crime and bad neighbour effects (Williams, 2000). In more suburban residential areas, guality of life may be enhanced by access to greenery, stronger social contacts and better safety and security, but poorer access to facilities (Masnavi, 2000). Questions are also raised about the acceptability of living in such urban forms (Jenks, 2000) and whether there is a social capacity beyond which environments begin to be unsustainable (Williams et al., 1999). Given the continuing counter-urbanisation trends (Breheny, 1995) and a powerful anti-urban ethos in the UK (Crookston et al., 1996), further questions are raised about peoples' attitudes to, and the extent to which they may wish to live in more 'compact' environments (notwithstanding the claimed benefits). When people do choose to live in more 'sustainable' urban forms (e.g. flagship or demonstration projects such as BedZed in the UK or Bo01 in Malmo, Sweden), the question remains whether or not they change to follow more sustainable lifestyles. Although the sustainability of certain physical aspects of the built environment such as density, compactness and design have been the subject of research (van Diepen, 2000; Williams, 2000; Williams et al., 2000; Carmona et al., 2001), in places these studies cast doubts on the link between built form and end user behaviour. However, so far no comprehensive studies have concentrated on behaviour or lifestyles associated with sustainable urban developments.

Economic Viability

It has been suggested that higher density forms support more diverse local service provision, by making local businesses and units more viable, while strengthening local supply chains. Higher density mixed used central areas are claimed to encourage more interaction and networking which promote innovation and creativity and hence greater endogenous growth including the formation of economic 'clusters'. Urban consolidation reduces infrastructure costs through scale and network economies and the re-use of existing capacity, whilst raising land values and so making (re-)development more viable, so reinforcing the spatial strategy.

Research has shown that range and quality of local services tends to be greater in high density areas, especially central/nodal places, but trends in the economics and

technology of some sectors is still leading to rationalisation into larger units, which may seek non-central locations. Service viability depends on income as well as density, and deprived urban neighbourhoods may lack services even when densities are high. Local supply chains are weak in many sectors, and may be weakening over time (Simmie et al., 2000; Simmie, 2001; Bramley et al., 2001). While some knowledge-based and cultural sectors are attracted to central cities, the evidence for clustering is weak in many cases and relevant networks may be more regional, national or international (Begg, 2001; Simmie, 2001). Many businesses are still attracted to low-density developments with ample car-parking in edge city locations accessible to motorways. Urban containment raises housing costs and reduces space consumption (Cheshire and Sheppard, 1989; Evans, 1991; Bramley, 1999, 2002; Bramley et al., 1995; Bramley and Watkins, 1996; Bramley, 2002), while high land values may militate against diversity of local services. Higher demand renders brownfield development viable but this is problematic for 'low demand' city regions (Monk and Whitehead, 2000; Bramley, 2002).

There were many gaps in economic research for this book to fill. There is a lack of systematic data on trends in the range and quality of local services and its relationship with urban form, and similarly with respect to supply chains and their key determinants. Systematic data on infrastructure costs of new or re-provision, and on their incidence, are conspicuously absent in the UK, unlike some other countries (US, Australia), although there is some literature on qualitative changes in provision (Guy et al., 1997). Data on property market performance is improving but has not been linked to urban form, and there are issues about the perceptions held by key property investors of different locations and forms of development including issues of risk (Heneberry and Guy, 1999). How susceptible such perceptions are to change based on experience and demonstration effects is important, and the same comment applies to business location preferences relating to congestion, access, parking and space constraints on expansion. Detailed urban form and design options, within a general high density framework, may significantly impact on the viability of residential developments, via values, costs, risks or timing.

Researching the Sustainable City

It is apparent that few of the claimed benefits of sustainable cities and features of these new 'sustainable' urban forms have been tested systematically, and the evidence to date is inconclusive. While it is possible to identify particular gaps in knowledge, there exists a deeper issue that needs to be addressed. Even where successes have been identified, and positive claims made, there is a considerable lack of evidence-based explanation, prediction and theory about the extent to which urban form as a whole contributes to sustainability. Many of the issues in this complex field interact and conflict. There may be many trade-offs and compromises to achieve advances in sustainability and to satisfy users and residents. In order to address these issues, and fill some of the gaps in knowledge, the research needed to deploy a range of robust and sophisticated methodologies.

Methodology

The methodology aimed to consider and integrate the range of aspects of urban form identified as influencing sustainability. It involves the measurement of urban forms, including building typologies, digital map footprints and configurations which are mapped on GIS, and analysed using methods including SPSS, measures of accessibility and spatial measures through Multiple Centrality Analysis (MCA). At the core of the research, three case study areas in each of five UK cities – Edinburgh, Glasgow, Leicester, Oxford and Sheffield - were selected. These areas represented a central, suburban, and 'in between' location in each city, representing a virtual 'slice' through a city. The research was carried out at a number of scales, including the city, the case study area (or neighbourhood), sub areas, the street and individual dwellings. Figure 1.1 below indicates the range of methods used at the city and neighbourhood scales of the investigation.

The methodologies employed to measure sustainability include:

- *Environmental*: environmental modelling, site surveys, biodiversity, mapped on GIS
- Social: questionnaires, focus groups, neighbourhood statistics
- Economic: interviews, land and property market data and models
- *Transport:* activity diaries, field surveys, accessibility assessments, transportation mode

These complex sets of data are analysed to find the urban forms which are beneficial, those which are problematic and those where conflicts between and within the sustainability impacts arise. The research was carried out by a number



Fig. 1.1 Research methods used at city and neighbourhood scales



of specialist groups within the consortium. Two common methods were used by all in order to provide a basis for the specialist research carried out in each aspect of sustainability (Fig. 1.2). These overall measurements and the specialist ones, were related to the measurements of the urban form of the 15 case studies.

The Structure of the Book

The book attempts to determine the relationships between different urban forms in a range of urban contexts, and: environmental sustainability; transport; social sustainability; and, economic sustainability. It provides guidance on sustainable urban forms, which are acceptable to users, and appropriate for the future. The book follows the structure of the research which was organised into a large, integrated, core project, with three related but separate projects feeding into it (Fig. 1.3).

Thus the book is organised in chapters that relate to the key claims about urban sustainability, and report on the research undertaken.¹ It addresses each major set of claims in separate chapters, and concludes at the end with an initial integration of the research findings that provides some answer to the key question as noted earlier – to what extent and in what ways does urban form contribute to sustainability, thus illuminating the dimensions to the sustainable city.

The first chapter presents an introduction to and justification for the book. In doing so it offers an overview of different urban forms and their claims to sustainability, assesses the state of the debate and introduces the key pillars of sustainability – environmental, transport, social benefits, and economic viability.

¹This forms part of the output from the *CityForm: Sustainable Urban Form Consortium* funded by the Engineering and Physical Sciences Research Council (EPSRC) under its Sustainable Urban Environments Programme (Grant number GR/520529/01). Further details of the research and publications can be found at http://www.city-form.org/



Fig. 1.3 Structure of the CityForm research

A platform for the research is presented in the second chapter. It begins with a review of measures of urban form recognising the role of different scales and outlines the methods of measurement used in the project. It profiles the five cities and fifteen case study areas within these cities that are examined in detail in later chapters. Case study areas represent a 'slice' through each city – a suburban area, a central area and an area 'in between'. These profiles encompass the urban physical form and the socio-economic characteristics by reference to Census statistics. The chapter concludes with a clear exposition of the elements that make up urban form and how they might integrate together.

The effect of urban form on mobility and activity, focusing particularly on the role of neighbourhood is assessed in the third chapter which reviews the underlying issues and current planning policy in the UK that aims to promote more sustainable mobility within cities. Multivariate analysis is used to explore the relationships between residential neighbourhood characteristics, travel behaviour and the type, duration and frequency of activity participation, whilst taking into account the effects of employment location, the wider urban structure and personal characteristics. This empirical research is based on a combination of national data sets and in-depth travel surveys. The chapter concludes with an appraisal of these relationships.

Increasing densities is a key aspect of contemporary urban development, but the ecological consequences of denser urban forms, which Chapter Four examines, are poorly understood. It is suggested that the major determinant of ecosystem performance in urban areas is the amount of available green space. While both green space availability and ecosystem performance are negatively related to urban density, there is considerable complexity within these relationships. This suggests that at any given density, there is substantial scope for maximising ecological performance. For example, in residential areas, the amount, quality and subdivision of green space are strongly influenced by housing morphology. Nevertheless, different aspects of ecosystem performance have different optima with respect to the configuration of urban green space. The implications for ecological optimisation of urban form are discussed alongside the empirical evidence derived from the study.

The social dimension of sustainability is widely accepted but exactly what this means has not been clearly defined and agreed. Similarly, claims about the social sustainability of certain urban forms, such as the 'compact city', have not been adequately tested. Chapter Five addresses both of these challenges. It begins with a discussion of social sustainability, identifying two main dimensions to the concept, related to equity of access and the sustainability/quality of community. The second part of the chapter concentrates on the relationships between social sustainability and urban form, presenting new evidence about the relationships between these two sets of measures based on quantitative measures and taking account of other socio-demographic influences. Finally qualitative evidence is drawn together to gain further insight into the relationships found.

In strictly physical terms the influence of urban form, and particularly built form on building energy use is well understood. At certain levels of description however there remain gaps to be filled if our understanding of these factors is to be well related to other aspects of energy use, particularly those determined by behaviour to improve building energy efficiency. Chapter Six describes work based on recently available UK energy consumption data that aims to improve knowledge in these areas, with the goal of improving energy models that underlie UK home energy rating schemes. Results are presented from the statistical analysis of consumption and questionnaire survey data from respondents living in different built form types in the study areas.

Chapter Seven initially focuses on the nature of the economic debate surrounding sustainable urban forms. From this base it considers the underlying forces that shape the elements of urban form and their impact in turn on a sustainable urban economy. The roles of transport infrastructure and spatial real estate markets in particular are highlighted in the examination of urban form. The chapter suggests an alternative formulation of the approach to urban sustainability that requires, as a necessary

condition, viable real estate sectors with sustainable markets. The implications of this viability condition are considered alongside the empirical evidence of individual intra-urban land use patterns in the five cities including 'viability' maps of the housing market and retailing.

Chapter Eight describes the development and application of a tool that allows the systematic comparison of the physical, social and economic characteristics of urban areas with commonly agreed indicators and target or threshold values of sustainable development. Establishing the difference between existing and target values allows the objective assessment of adaptation in urban areas in order to make them work in a more socially and economically balanced way. The applicability of the tool for the planning and the programming of urban areas is illustrated in a number of case studies in Glasgow.

Chapter Nine presents the results of research investigating the question: Do sustainable urban environments engender sustainable behaviour and lifestyles, and if not why not? The research findings review the behavioural impacts on households living in sustainable housing developments. The research tests the lifestyle impacts of 13 housing schemes built according to sustainability principles. These schemes include a number of design features associated with sustainability, such as energy-efficient homes and infrastructure which promotes walking and cycling. The chapter presents findings on the extent to which elements of neighbourhood design (drawn from a checklist of over 100 features) support eight key sustainable behaviours. The chapter concludes with comments on the relevance of the findings for designers and policy makers.

Urban green spaces are valuable resources in cities. There is a growing realisation that urban areas form significant components of regional and national biodiversity conservation networks, and the importance of urban nature for the well-being of human urban residents has been the focus of distinct bodies of research. With half of the world's human population living in urban areas and a continued decline of biodiversity in the wider landscape, urban green space plays an increasingly important role in creating sustainable cities. Chapter Ten considers the links between the distribution of people and biodiversity in the city, the relationship between biodiversity and human well-being, and the potential trade-offs between management of urban green space for people and nature based on empirical research undertaken.

The final chapter addresses the key question: to what extent and in what ways does urban form affect sustainability? It challenges theories about the sustainability of urban form and the oft repeated mantras so prevalent in policy. It summarises the relationships between urban form elements and the dimensions – social acceptability, energy use, travel and mobility, ecology and biodiversity and economic viability – set out in earlier chapters. Some key results that integrate the various components on the research are presented, thus going some way towards answering the key research question raised. The chapter aims to provide policy guidance based on the evidence presented and the key urban form trade offs are identified. The chapter also examines the potential impact of building individual sustainable developments, the use of open space and adapting the city.

References

- Acioly, C. (2000) 'Can Urban Management Deliver the Sustainable City?' In Compact Cities: Sustainable Urban Forms for Developing Countries, (eds. M.Jenks and R.Burgess), Spon Press, London.
- ARUP (2008) http://www.arup.com/eastasia/project.cfm?pageid=7047
- Attwell, K. (2000) Urban land resources and urban planting case studies from Denmark. Landscape & Urban Planning, **52**, pp.145–163.
- Badoe, D.A. and Miller, E.J. (2000) Transportation Land-use Interaction: Empirical Findings in North America, and Their Implications for Modeling, Transportation Research Part D, 5, pp.235–263.
- Basantani, M. (2008) Plans for Foster's Masdar Carbon Neutral City Debut, http://www.inhabitat.com/2008/02/06/plans-unveiled-for-worlds-first-zero-carbon-zerowaste-city/
- Begg, I. (2001) Urban Competitiveness: Policies for dynamic cities, Policy Press, Bristol.
- Boarnet, M, and Crane, R. (2001) *Travel by Design: The influence of urban form on travel*, Oxford University Press, New York.
- Bolund, P. & Hunhammar, S. (1999) Ecosystem services in urban areas. *Ecological Economics*, **29**, pp.293–301.
- Bonaiuto, M., Aiello, A., Perugini, M., Bonnes, M., and Ercolani, A.P. (1999) Multidimensional Perception of Residential Environment Quality and Neighbourhood Attachment in the Urban Environment. *Journal of Environmental Psychology*, **19**, pp.331–352.
- Bramley, G., Bartlett, W., and C. Lambert (1995) *Planning, the Market and Private Housebuilding*, UCL Press, London.
- Bramley, G. & Watkins, C. (1996) *Steering the Housing Market: new building and the changing planning system,* The Policy Press, Bristol.
- Bramley, G. (1999) Housing market adjustment and land supply constraints. *Environment & Planning A*, **31**(7), pp.1169–1188.
- Bramley, G, Kirk, K. & Russell, J. (2001) Planning Central Scotland: The Role Of Infrastructure, Urban Form And New Development In Promoting Competitiveness And Cohesion. ESRC Cities Central Scotland Integrative Cities Study, Policy Discussion Paper (Revised), University of Glasgow, Dept of Urban Studies (website).
- Bramley, G. & Morgan, J. (2002) Building Future Living Environments: The Role Of New Housing In Competitiveness And Cohesion. ESRC Cities Central Scotland Research Policy Paper for Communities Scotland and the Scottish Executive.
- Bramley, G. (2002) Housing supply and land use regulation. In *Housing Economics and Public Policy*, (eds. K.Gibb and A. O'Sullivan), Routledge, London.
- Breheny, M. (1995) Compact cities and transport energy consumption. *Transactions of the Institute* of British Geographers NS, **20**(1), pp. 81–101.
- Burgess, J., Harrison, C. and Limb, M. (1988) People, Parks and the Urban Green: A Study of Popular Meanings and Values for Open Spaces in the City. *Urban Studies*, 25, pp.455–473.
- Burton, E (2000a) The compact city: just or just compact? A preliminary analysis, *Urban Studies* **37**(11), pp.1969–2006.
- Burton, E. (2000b) The potential of the compact city in promoting social equity, in *Achieving Sustainable Urban Form* (eds. K. Williams, E. Burton and M. Jenks) E & FN Spon, London, pp 19–29.
- Commission for Architecture & the Built Environment (CABE) (2005) *Better Neighbourhoods: Making higher densities work*, CABE & the Corporation of London, London.
- Commission for Architecture & the Built Environment (CABE) (2008) www.buildingforlife.org
- Calthorpe, P. (1993) *The Next American Metropolis: Ecology, community and the American dream,* Princeton Architectural Press, New York.
- Canmore Housing Association (2008) *Slateford Green Housing, Edinburgh*, www.edinburgharchitecture.co.uk/slateford_green_housing.htm

- Cannon, A. (1999) The significance of private gardens for bird conservation. Bird Conservation International, 9, pp.287–297.
- Carmona, M., de Magalhaes, C., Edwards, M., Awour, B. and Aminossehe, S. (2001) *The Value of Urban Design*, Report by the Bartlett School of Planning for CABE and DETR, Thomas Telford, London.

Cervero, R. (1998) The Transit Metropolis: A global inquiry, Island Press, Washington D.C.

- Cheshire, P. and Sheppard, S. (1989) British planning policy and access to housing: some empirical estimates. *Urban Studies*, **26**, 469–85.
- Commission of the European Communities (CEC) (1990) *Green Paper on the Urban Environment*, European Commission, Brussels.
- Crisp, V., Littlefair, P., Cooper, I. and McKennan, G. (1988). BR129: Daylighting as a Passive Solar Energy Option; an Assessment of its Potential in Non-Domestic Buildings, BRE, Garston.
- Crookston, M., Clarke, P. and Averley, J. (1996) The compact city and the quality of life. In *The Compact City:* A Sustainable Urban Form? (eds. M. Jenks, E. Burton and K. Williams,) E & FN Spon, London, p.135.
- Curitiba websites:

http://solstice.crest.org/sustainable/curitiba/

http://www.solutions-site.org/artman/publish/article_62.shtml

- Dawson, R. (2004) *Towards Good Practice in Sustainable urban Land Use*, Bristol City Council, Bristol.
- Department for Communities and Local Government (DCLG) (2006a) *Planning Policy Statement* 3: *Housing*, The Stationary Office, London.
- Department for Communities and Local Government (DCLG) (2006b) Sustainable Communities: Building for the Future, www.communities.gov.uk/publications/communities/ sustainablecommunitiesbuilding
- Department for Communities and Local Government (DCLG) (2007) New eco-towns could help tackle climate change www.communities.gov.uk
- Department of the Environment, Transport and the Regions (DETR) (2000a) *Our Towns and Cities: The Future Delivering and Urban Renaissance*, Cm 4911 Urban White Paper, HMSO, London.
- Department of the Environment, Transport and the Regions (DETR) (2001) *Planning Policy Statement 13: Transport*, HMSO, London.
- Dongtan Development Co (2008) http://www.dongtan.biz/english/zhdt/plan.php
- Duany, A. and Plater-Zyberk, E. (1992) The second coming of the American small town. Wilson Quarterly, 16, pp.3-51.
- English Partnerships (2007) *Millennium Communities Programme*, ENG0049, English Partnerships, Warrington, UK.
- Evans, A. W. (1991) Rabbit hutches on postage stamps: planning, development and political economy. *Urban Studies*, **28**(6), pp.853-70.
- Faber Taylor, A., Wiley, A., Kuo, F. and Sullivan, W. (1998) Growing up in the inner city: Green Spaces as Places to Grow. *Environment and Behavior*, **30**, pp.3-27.
- Foster and Partners (2008) http://www.fosterandpartners.com/Projects/1515/Default.aspx
- Gilbert, O.L. (1989) The Ecology of Urban Habitats. Chapman & Hall, London.
- Graham, S. and Marvin, S. (2001) Splintering Urbanism: Networked infrastructures, technological mobilities and the urban condition, Routledge, London.
- Guy, S., Graham, S. & Marvin, S. (1997) Splintering networks: cities and technical networks in 1990s Britain. Urban Studies, 34(2), pp.191-216.
- Guy, S. and Marvin, S. (2000) Models and Pathways: The Diversity of Sustainable Futures. In Achieving Sustainable Urban Form, (eds. K. Williams, E. Burton and M. Jenks) E & FN Spon, London.
- Handy, S. (1996) Methodologies for exploring the link between urban form and travel behaviour, *Transportation Research* -D, **1**(2), pp.151-165.
- Henneberry, J. and Guy, S. (1999) Paper presented at 'Property' Colloquium of ESRC Cities Research Programme, Reading, May 1999.

HM Government (1994) Sustainable Development: The UK Strategy, HMSO, London.

- Jenks, M., Burton, E. and Wiliams, K.(eds) (1996) *The Compact City: A Sustainable Urban Form?*, E & FN Spon, London.
- Jenks, M. (2000) The acceptability of urban intensification, in *Achieving Sustainable Urban Form* (eds. K. Williams, E. Burton and M. Jenks) E & FN Spon, London, pp 242-250.
- Jenks, M. and Burgess, R. eds (2000) Compact Cities: Sustainable Urban Forms for Developing Countries, Spon Press, London.
- Jenks, M. and Dempsey, N. eds (2005) Future Forms and Design for Sustainable Cities, Architectural Press, Oxford.
- Jenks, M., Kozak, D. and Takkanon, P. eds (2008) World Cities and Urban Form: Fragmented, Polycentric, Sustainable?, Routledge, Oxford.
- Katz, P. (1994) *The New Urbanism: Toward an Architecture of Community*, McGraw-Hill, New York
- Kinzig, A.P. & Grove, J.M. (2001) Urban-suburban ecology. *Encyclopedia of Biodiversity*, Vol. 5 (ed. S.A. Levin), Academic Press, San Diego pp. 733-745.
- Kozak, D. (2008) Assessing Urban Fragmentation: The emergence of new typologies in central Buenos Aires. In World Cities and Urban Form: Fragmented, Polycentric, Sustainable? (eds. M. Jenks, D. Kozak. and P. Takkanon), Routledge, Oxford.
- Lacy, R. (1977) Climate and building in Britain. A review of meteorological information suitable for use in the planning, design, construction and operation of buildings, Building Research Establishment Report, BRE, Garston.
- Lambregts, B and Zonneveld, W. (2003) *Polynuclear Urban Regions and the Transnational Dimension of Spatial Planning*, Delft University Press, Delft.
- Langdon, P. (1994) A Better Place to Live: Reshaping the American Suburb, University of Massachusetts Press, Amherst: MA.
- Llewelyn-Davies (1997) *Sustainable Residential Quality: New Approaches to Urban Living*, report for the GOL, DETR and LPAC, LPAC, London.
- Llewelyn-Davies (2000) Urban Design Compendium, English Partnerships and The Housing Corporation, London.
- Low, N. and Gleeson, B. eds (2003) *Making Urban Transport Sustainable*, Palgrave Macmillan, Basingstoke.
- Masnavi, M.-R. (2000) The new millennium and the new urban paradigm: the compact city in practice, in *Achieving Sustainable Urban Form* (eds. K. Williams, E. Burton and M. Jenks) E & FN Spon, London, pp 64-73.
- Monk, S. and Whitehead, C. (2000) *Restructuring Housing Systems: from social to affordable housing*. York Publishing Services.
- Office of the Deputy Prime Minister (ODPM) (2004a) *Planning Policy Guidance 3: Housing*, ODPM, London.
- Office of the Deputy Prime Minister (ODPM) (2004b) *Planning Policy Guidance 13: Transport*, ODPM, London.
- OMA (2008) www.oma.eu
- Owers, R. and Oliver, G. (2001) 'Urban healing', *EcoTech Sustainable Architecture Today*, 4, pp.28-32.
- Parsons, R., Tassinary, L.G., Ulrich, R.S., Hebi, M.R. and Grossman-Alexander, M. (1998) The view from the road: implications for stress recovery and immunization. *Journal of Environmental Psychology*, 18, pp.113-140.
- Pauleit, S. and Duhme, F. (2000) Assessing the environmental performance of land cover types for urban planning. *Landscape & Urban Planning*, 52, pp.1-20.
- Payne, L., Mowen, A. and Orsega-Smith, E. (2002) An Examination of Park Preferences and Behaviours Among Urban Residents: The Role of Residential Location, Race and Age. *Leisure Sciences*, 24, pp.181-198.
- Revill, J. and Davies, C. (2008) 'Secret' eco-town plans spark protest. *The Observer*, February 10, 2008 online: www.guardian.co.uk/society/2008/feb/10/communities. planning1?gusrc=rss&feed=networkfront

- Rudlin, D. and Falk, N. (1999) *Building the 21st Century Home: The Sustainable Neighborhood*, Architectural Press, London.
- Savard, J-P.L., Clergeau, P. & Mennechez, G. (2000) Biodiversity concepts and urban ecosystems. Landscape and Urban Planning 48, pp.131-142.
- Simmie, J. (ed) (2001) Innovative Cities, Spon Press, London.
- Simmie, J., Wood, P. and Sennett, J. (2000) Innovation and clustering in the London Metropolitan Region. Paper presented to ESRC Symposium on 'Urban Competitiveness', South Bank University, April 2000, and reproduced in I. Begg (ed) (2001).
- Simmonds, D. and Coombe, D. (2000) The transport implications of alternative urban forms, in Achieving Sustainable Urban Form (eds. K. Williams, E. Burton and M. Jenks) E & FN Spon, London, pp 121-130.
- Skjaeveland, O. and Garling, T. (1997) Effects of Interactional Space on Neighbouring. *Journal of Environmental Psychology*, 17, pp.181-198.
- Smart Growth (2008) www.smartgrowth.org
- Stead, D., Williams, J. and Titheridge, H. (2000) Land use, transport and people: identifying the connections. In *Achieving Sustainable Urban Form* (eds. K. Williams, E. Burton and M. Jenks) E & FN Spon, London, pp 174-186.
- Stubbs, M. and Walters, S. (2001) Car Parking in Residential Development: Assessing the viability of design and sustainability in parking policy and layout. In RICS Foundation, *The Cutting Edge 2001*, RICS, London.
- Syme, G., Fenton, D. and Coakes, S. (2001) Lot size, garden satisfaction and local park and wetland visitation. *Landscape and Urban Planning*, **56**, pp.161-170.
- Tinsley, H.E.A., Tinsley, D.J. & Croskeys, C.E. (2002) Park Usage, Social Milieu, and Psychosocial benefits of Park Use Reported by Older Urban Park Users from Four Ethnic Groups. *Leisure Sciences*, 24, pp.199-218.
- Trotter, C. (2008) Rem Koolhaas' Ras Al Khaimah's Eco City to rival Masdar http://arquitectura.pt/forum/f11/rem-koolhaas-ras-al-khaimah-s-eco-city-to-rival-masdar-9816.html
- Tyrväinen, (1997). The amenity value of the urban forest: an application of the hedonic pricing method. *Landscape and Urban Planning*, **37**(3-4), pp.211-222.
- Ulrich, R.S. (1981) Natural versus Urban Scenes: Some Psychophysiological Effects. *Environment and Behavior*, 13, pp.523-556.
- Ulrich, R.S, Simons, R.F., Losito, B.D., Fiorito, E., Miles, M.A. & Zelson, M. (1991) Stress recovery during exposure to natural and urban environments. *Journal of Environmental Psychology*, **11**, pp.201-230.
- Urban Task Force (1999) Towards an Urban Renaissance, E & FN Spon, London.
- van Diepen, A. (2000) Households and their Spatial-Energetic Practices: Searching for Sustainable Urban Forms, Netherlands Geographical Studies, Groningen, Netherlands.
- Vu, T., Asaeda, T. and Abu, E. (1998) Reductions in air conditioning energy caused by a nearby park. *Energy and Buildings*, 29(1), pp.83-92.
- Watkins, R., Palmer, J., Kolokotroni, M. and Littlefair, P. (2002) The balance of the annual heating and cooing demand within the London urban heat island. *Proceedings of Climate Change and* the Built Environment, International Conf, CIB Task Group 21, Climate Data for Building Services, Manchester.
- Whyte, W.H. (1980) The Social Life of Small Urban Spaces, Projects for Public Spaces, New York.
- Williams, K., Burton, E. and Jenks, M. (Eds.) (2000) Achieving Sustainable Urban Form, E & FN Spon, London.
- Williams, K., Jenks, M and Burton, E. (1999) How much is too much? Urban intensification, social capacity and sustainable development. *Open House International*, 24(1), pp. 17-26.
- World Gazetteer (2005) http://www.world-gazetteer.com
- Yannis, S. (1994) Solar Energy and Housing Design, Vol. 1: principles, objectives, guidelines, Architectural Association, London.

Chapter 2 Elements of Urban Form

Nicola Dempsey, Caroline Brown, Shibu Raman, Sergio Porta, Mike Jenks, Colin Jones and Glen Bramley

Introduction

This chapter provides a common platform for the research presented in this book and is divided into two parts. The first section examines the elements of urban form identified for the purposes of the research and explains how they were measured. The second section profiles the five case study cities and fifteen case neighbourhoods which were the focus for the empirical research discussed in later chapters. These profiles provide an outline of both the urban form and socio-economic characteristics of the areas studied. The chapter concludes with a review of the urban form features of the case study cities and neighbourhoods, and shows how the different physical elements integrate together with socio-economic characteristics.

Elements of Urban Form

The term 'urban form' can be used simply to describe a city's physical characteristics. At the broad city or regional scale, urban form has been defined as the spatial configuration of fixed elements (Anderson et al., 1996). Features of urban form at this scale would include urban settlement type, such as a market town, central business district or suburbs. However, urban form is closely related to scale and has been described as the 'morphological attributes of an urban area at all scales' (Williams et al., 2000). Characteristics therefore range from, at a very localized scale, features such as building materials, façades and fenestration, to, at a broader scale, housing type, street type and their spatial arrangement, or layout.

It should be noted that urban form does not simply relate to physical features, but also encompasses non-physical aspects. One can see this in the example of density. Simply put, density is used as a measure of the number of people living in a

N. Dempsey (⊠)

Oxford Institute for Sustainable Development, Oxford Brookes University, Oxford, UK

given area: it is not just a physical, tangible element. Density is also closely linked with the configuration of the social environment and interaction within residential neighbourhoods: flats and apartments are examples of high-density housing whereas detached and semi-detached properties tend to be of lower densities. There are therefore non-physical economic, social and political processes in place which are physically manifested in housing, schools, parks and other services and facilities.

The scales at which urban form can be considered or measured include the individual building, street, urban block, neighbourhood and city. These levels of spatial disaggregation influence how urban form is measured, analyzed and ultimately understood. The issue of scale is discussed throughout this chapter (and the book) as it constitutes an underlying dimension of any examination of urban form.

Urban form generally encompasses a number of physical features and nonphysical characteristics including size, shape, scale, density, land uses, building types, urban block layout and distribution of green space. These are categorised here as five broad and inter-related elements that make up urban form in a given city (Fig. 2.1).





These elements of urban form have been identified on the basis that they are claimed to influence sustainability and human behaviour. They are considered in more detail below. These elements relate to developed, and not developing, countries. For this reason, infrastructure (e.g. water, roads, gas etc.) is not discussed here as an element of urban form; however, it is acknowledged that infrastructure would form an important part of examinations of urban form in developing countries.

Density

Density is a deceptively complex concept with a number of inter-related dimensions. While it may provide an objective, spatially-based, measure of the number of people (living) in a given area, it is also assessed subjectively; it is a social interpretation dependent on individual characteristics and so may differ from resident to resident (Churchman, 1999). For example, while the density of Trafalgar Square in London may be reported as low (density usually being a measure of residential occupancy), the perceived density, and extent of crowding, may be very high (after Rapoport, 1975).

Density entered the consciousness of UK policy makers in the nineteenth century when urban areas were growing rapidly and overcrowding and appalling living conditions were prevalent among the poor (Jenks and Dempsey, 2005). There is also a more cultural dimension to density, where the densities at which people live may be considered as relative. Current English housing policy states that new residential building should be at a minimum of 30 dwellings/ha which for some may be an unacceptably high density (DCLG, 2006). In Hong Kong however, a minimum of ten times that density would be considered low (after Jenks, 2000; Breheny, 1997; Jenks and Dempsey, 2005).

Density is also closely associated with other elements of urban form, such as land use and access to services – for example, for a service or facility to be viable, it needs to serve a population of a particular size. Density on the one hand can be seen as an outcome of the competition between land uses within a given urban transport infrastructure and its associated pattern of accessibility. On the other hand it is a policy goal as it is also an input into the quality of urban life through the viability of services provision and availability of public and private space. Density has therefore been used as a tool to measure the viability of public transport infrastructure and other service, in urban design and construction. At what point density becomes high (or too high) is unclear, but in recent years planning policy and practices in many countries have been attempting to increase the average density of new development.

Land Use

Broadly speaking, the term land use is used to describe the different functions of the environment. Within the urban context, the dominant land use tends to be residential but a functional urban area requires industrial, retail, offices, infrastructure and other uses. The spatial (micro) pattern of land uses is crucial to the arguments about the efficiency of a city and potential 'sustainable' urban forms in influencing urban travel patterns and the quality of life, for example through the existence of green space. There are also certain 'locally-unwanted land uses' such as prisons (Grant, 2002), airports, or landfill sites claimed to be undesirable in residential mixed-use

areas (Healey, 1997). Planners have traditionally attempted to separate land uses because of potential undesirable externalities but are now in favour of mixed use developments. For example current UK policy promotes easily accessible services and facilities for residents (DCLG, 2006); both 'horizontally' – at ground floor level – and, increasingly in new city centre developments, 'vertically' – within the same building (DETR, 2000). However, land use patterns are dynamic rather than static phenomena and are subject to real estate market forces.

A key component of local land use is the availability of local neighbourhood services. The provision of services and facilities is dependent on the resident population's requirements so a particular land use mix therefore differs from neighbourhood to neighbourhood (Urban Task Force, 1999). The local urban context and the requirements of the population are therefore important in this matter. It is not however clear which services and facilities can and should be provided at which spatial scale. An 'everyday eight' local neighbourhood services and facilities identified by Winter and Farthing in the UK context includes post office, supermarket, primary school, newsagent and open space (1997, p. 127). Other services to which residents need local access, albeit on a less frequent basis, include a doctor's surgery (Barton et al., 2003; Urban Task Force, 1999), chemist; bank (Burton, 1997); and community centre (Aldous, 1992). There is extensive prescriptive UK guidance on what those land uses should be for a given neighbourhood however there is no consensus (Dempsey, 2008).

Accessibility and Transport Infrastructure

Transport infrastructure is closely associated with accessibility as it determines the ease with which buildings, spaces and places can be reached. The level of accessibility describes the area residents and users are able to reach, as well as the extent to which they have the means to access places, services and facilities that are outside their local area (after Talen, 2003).

Accessibility is actually a layered concept and is not simply proximity as distance is just one contributor. It is dependent on a number of factors including the location of potential destinations relative to an individual's starting point, how well the transport system connects to spatially distributed locations, how the individual uses the transport system, and the characteristics of, for example, the services and facilities that the individual plans to use (Liu and Zhu, 2004). A key accessibility relationship is between home and the city centre. Different aspects of the concept encompass access in terms of what is available within walking distance of home (sometimes referred to as 'pedshed'), or access in terms of the means to get to, for example, services and facilities which are located further afield (Barton et al., 2003; Schoon, 2001). It is therefore closely linked to land use and layout: the services, facilities, open space, how they are arranged within a city or neighbourhood and the means of getting to them all contribute to how accessible a place or service might be described.

Urban Layout

Layout describes the spatial arrangement and configuration of elements of streets, blocks and buildings, often referred to at the street scale, such as grid or tree-like (cul-de-sac) street patterns. Layout has an important influence on pedestrian movement and the way in which different places and spaces are connected to each other (ODPM, 2005; CABE and DETR, 2000). The layout, whether or not it is 'permeable' and easy to find the way, controls access and movement for pedestrians, and could influence other aspects of urban form such as land use or density (Hillier and Hanson, 1984; Hillier, 1996).

The layouts of today's cities are largely artifacts of their historical development and planning and building regulations. The configuration of the street network, in terms of its urban block sizes, their overall location within the city, pedestrian and vehicular connectivity, can affect the functioning of a city by, for example, influencing the location intensity of activities (Penn et al., 1998; Porta et al., 2008).

The connectedness and permeability of urban layouts are claimed to determine the nature and extent of routes between and through spaces which in turn has an influence on how lively and well-used a space is (Cowan, 1997). Streets which are well-connected to services, facilities etc. and/or the means for the pedestrian of reaching them, are argued to be more frequently used than deserted or quiet options (Gehl, 2001; Gehl et al., 2004).

Housing and Building Characteristics

The characteristics of housing and other buildings in urban settlements can have an important bearing on everyday living: it has already been noted that residents living in low-density detached dwellings with large gardens will have a distinct experience of the urban environment from high-rise city centre apartment dwellers. However, the influence of building characteristics extends beyond the density of urban living. Factors such as building type, height and age may have an effect on a number of issues. These might include a building's orientation and exposure to sunlight and daylight (Mardaljevic, 2005) and the potential for modifications, such as changes to living space to work space or individual room conversion to continue accommodating an ageing resident as in the 'lifetime homes' model (Holmes, 2007).

Other factors such as the amount of living space in dwellings, number and types of particular rooms and lowest level of living space may also have significant influences on the efficiency of buildings in terms of its embodied, operating and life cycle energy (Newton et al., 2000).

Integrated Elements

While it is useful to examine these elements separately, it is also clear that they are inter-connected and interdependent. For example, the accessibility within an urban settlement is very closely linked to its density and the layout of, and extent of mixed
uses within. A neighbourhood is not accessible without viable services and facilities available for residents to use. Neither is it accessible without pedestrian, cycling and public transport networks through which the neighbourhood is connected both to its own services and to services outside. When planning and constructing new residential areas, housing type and size may be dictated by the proposed density of a site, which for example, in line with recent UK policy, will also provide a range of land uses (including different services and facilities and open space) and a connected and permeable urban layout. This message is endorsed by a series of US studies that have sought to develop measures of urban sprawl (Ewing et al., 2002; Cutsinger et al., 2004).

The interrelated associations between these elements have wide-reaching implications for the research. Firstly, there is a need for compatibility in how the different elements of urban form are measured. This is to ensure that, secondly, the statistical (and other) analyses conducted can account for the individual effect that each element may have on a particular aspect of sustainability, as well as the collective influence of the elements of urban form. In this way, the main research question which looks to explain how urban form affects sustainability can be answered in as robust and reliable a manner possible. The next section examines outlines the methodological approach used in the research to measure these elements of urban form.

Measuring Urban Form

A largely quantitative approach was adopted in this research, although qualitative methods were used in parts of the project and are detailed in the relevant chapters. To measure robustly the features of urban form outlined above, a two-pronged process of data collection was followed. Firstly, existing datasets such as the 2001 Census, the Valuation Roll and Ordnance Survey data were examined to provide information on, for example, initial density measures, non-domestic properties and their location. Secondly, after ascertaining gaps in the data, e.g. building heights and information on transport infrastructure (e.g. bus stop location and type of car parking), a site survey was conducted in the study neighbourhoods on a street-by-street basis.

The site surveys were undertaken by researchers using an innovative method employing PDAs (personal digital assistants) with GPS (global positioning system) modules which allowed geocoded survey data to be directly downloaded into a GIS (geographic information system) platform. A useful innovation was the production of prototype software to automatically link separately sourced data tables from Ordnance Survey and Valuation Data. This allowed site surveyors easily to identify, locate and check business and mixed use premises. The data collected in this way included details about: buildings such as condition and height; land uses; the presence of litter and graffiti; position of bus stops and shelters.

The research design was a cross-sectional one, where data are collected at one point in time providing a 'snapshot' approach (Gray, 2004). While the project did not detail the extent to which a changing urban form can contribute to sustainability

over time, it offers new and valuable data on how and to what extent current urban forms in UK cities can be described as sustainable. The following sections outline how each element of urban form was measured in the research.

Measuring Density

A wide range of different measurements have been used to calculate the density of a given area, such as persons per hectare (pph), dwellings per hectare (dph), bed spaces per hectare and habitable rooms per hectare (Woodford et al., 1976). Employing a number of density measures has been argued to be more robust than using one single density indicator which cannot accurately measure the density of a given area (Jenks and Dempsey, 2005).

A range of density indicators was selected to provide as complete a picture as possible of the overall density of the case studies, while accounting for the different scales of urban form (the city, neighbourhood, 'sub-area' and street). Examples of these indicators are presented in Table 2.1. It should be noted that the indicators of net residential density used here are based on a definition of residential which includes outdoor space such as gardens, but *excludes* streets and footpaths. This method of calculation results in density figures which are higher than those usually reported.

It is clear from the table that the indicators measure physical density and not perceived density. Aspects of perceived density are measured in Chapters 9.

Measuring Land Use

In order to measure the extent of mixed land uses in the case study neighbourhoods, a number of appropriate land uses were selected, excluding land uses which are not relevant for the purposes of the research, e.g. telecommunications, energy and waste infrastructure. Data on a number of particular services and facilities are not always specified in secondary data sources (e.g. name of supermarket, newsagent and children's nursery), indicating a need to conduct primary data collection.

It is necessary to account for any 'edge effects', where residents may be using local services and facilities outside the case study boundaries identified for the research purposes. A 'buffer zone' of approximately 400m (approx. 5 min walking distance) is applied around each case study area to account for any 'edge effect'. Households living on the edge of a chosen case study area might be closer to facilities just outside the boundary and therefore may choose to use those rather than the ones initially identified in the research. By applying a buffer zone around the neighbourhood, the researchers are able to capture and investigate usage of particular services and facilities including food shops, post office and GP surgeries. This land use information is then mapped using a GIS-based platform (Fig. 2.2). Examples of these land use indicators used in the research are presented in Table 2.2. The categories used are based on the National Land Use Database (NLUD) developed by the then ODPM (2003).

Measurement	Description	Examples of aspects/features measured	Sources of information
Gross Density (City)	The ratio of persons, households, or dwelling units to the entire area of the city regardless of land use.	 Total city population No. of households No. of dwellings City area 	– Census data – Local authorities
Gross Density (Neighbourhood)	Number of persons, households, or dwelling units per hectare of the total neighbourhood area.	 Total population No. of households No. of dwellings Case study area 	 Census data Local authorities Valuation Roll Ordnance Survey
Gross Residential Density (Sub-area)	Number of persons, households, or dwelling units per hectare of the total sub-area area.	 Total population No. of households No. of dwellings Sub-area 	Census dataValuation RollOrdnance Survey
Net Residential Density (Neighbourhood)	Number of persons, households or dwellings per hectare of the total land area devoted to residential land use.	 Total population No. of households No. of dwellings Total residential land area 	 Census data Ordnance Survey
Net Residential Density (Sub-area)	Number of persons, households or dwellings per hectare of the total land area devoted to residential land use within the sub-area.	 Total population No. of households No. of dwellings Total residential land area 	 Census data Ordnance Survey Valuation Roll
Net Residential Density (Street & Plot)	Number of dwellings per plot.	 No. of dwellings per plot Plot area 	- Ordnance Survey
Floor Area Ratio (Neighbourhood & Sub-area)	Ratio of floor area to site area.	 Floor area (of each building) No. of storeys Site area (of each plot) 	Ordnance SurveySite survey
Coverage Ratio (Neighbourhood & Sub-area)	Ratio of building footprint to site area.	Building footprint (each building)Site area (of each plot)	– Ordnance Survey

 Table 2.1
 Indicators of density

Measuring Accessibility and Transport Infrastructure

As indicated earlier, measures of accessibility can refer to different aspects of the concept. The indicators used to measure accessibility here cover transport infrastructure for pedestrians and cyclists as well as public and private transport. Table 2.3 shows some of the indicators used to measure accessibility which include



Fig. 2.2 Example of GIS-generated land use map (Oxford) (© Ordnance Survey)

characteristics of public transport infrastructure and journey times and distances. These indicators (and others such as socio-economic characteristics of residents and employment location) were included in a transport model (discussed in detail in Chapter 3) to provide as accurate a picture of accessibility in the case study neighbourhoods.

Measuring Housing/Building Characteristics

It is not possible or desirable to measure the characteristics of every building in the case study neighbourhoods. An efficient method of measuring the characteristics involves the identification of predominant housing types per street, and highlighting where there were exceptions to this. It is also useful to make use of the household questionnaire in measuring these characteristics. While the focus of the questionnaire is to measure aspects of sustainability, it proved to be a useful tool to collect urban form indicators, such as housing type, lowest level of living accommodation and a household's access to a garden/residential outdoor space.

The characteristics of non-domestic buildings are collected which understandably overlap with the indicators measuring land use. Indicators of maintenance are included in this category of indicators, such as the condition of buildings (where considered to be poor relative to other buildings in the street) and levels of litter and instances of graffiti and vandalism (Table 2.4).

		Examples of aspects/features	Sources of
Measurement	Description	measured	information
Residential land use (Individual dwellings)	Residential, institutional and communal accommodation	 Sheltered accommodation Care homes University halls of residence 	– Ordnance Survey – Site survey
Commercial and retail land use (Individual buildings)	Properties housing all commercial uses	 Retails & Supermarkets Shops Storage & Warehouses Restaurants/cafés 	Ordnance SurveyValuation RollSite survey
Offices (Individual buildings)	Office space	 Business parks Banks and building societies Other offices 	Ordnance SurveyValuation RollSite survey
Industrial (Individual buildings)	Industrial properties including industrial storage and warehouses	 Factories/Workshops Industrial storage facilities (depots etc.) 	Ordnance SurveyValuation RollSite survey
Community Buildings (Individual buildings)	Buildings used for community purposes including: – educational – health – community services	 Primary schools Health centres and GPs Hospitals Community centres Places of worship Police stations 	– Ordnance Survey – Site survey
Leisure and recreational Buildings (Individual buildings)	Buildings used for leisure and recreational purposes	– Museums – Libraries – Cinemas – Indoor sports facilities	Ordnance SurveyValuation RollSite survey
Outdoor Recreation (Individual spaces)	Outdoor amenity and open spaces	 Football pitches Golf courses Sports grounds Allotments 	Ordnance SurveySite survey
Other public green space (Individual spaces)	Spaces of grassland, woodland etc.	– Woodland – Heathland	Ordnance SurveySite survey
Previously developed land (Individual spaces)	Previously developed land which is or was occupied by a building or other permanent structure	– Derelict land – Vacant land	Ordnance SurveyValuation RollSite survey
Mixed use (Individual buildings)	Buildings with multiple land uses	 Vertical mixed uses (flats above shops/offices above commercial etc.) 	Ordnance SurveyValuation RollSite survey

 Table 2.2
 Indicators of land use

Measurement	Description	Examples of aspects/features measured	Sources of information
Public transport infrastructure (Street)	Location of public transport features	 Location of bus/tram stops Bus/tram routes Frequency of services 	 Ordnance Survey maps Site survey Public transport companies Local authorities
Private transport infrastructure (Street)	Location of private transport features (i.e. parking)	 Location of off-street parking and types Location of on-street parking and types 	 Ordnance Survey maps Site survey
Pedestrian/cycling infrastructure (Street)	Location of (cycle) paths/alleyways/ underpasses etc.	 Location of routes inaccessible to motorized transport 	 Ordnance Survey maps Site survey
Road management (Street)	Route management	 One-way systems Traffic management Speed restrictions 	 Ordnance Survey maps Site survey
Journey time/distance (Individual buildings)	Journey to work/other services etc. in terms of time and distance	Trip originTrip destination	 Ordnance Survey maps Site survey Transport modelling

ity

Measurement	Description	Examples of aspects/features measured	Sources of information
Housing type (Individual buildings)	Predominant housing type per street with exceptions marked	 Detached housing Semi-detached housing Terraced housing Tenements Flats/apartments 	 Ordnance Survey Site survey Questionnaire
Housing characteristics (Individual buildings)	Characteristics of individual dwellings	 Lowest level of living accommodation Access to garden Number of bedrooms Condition of building 	– Questionnaire
Building type (Individual buildings)	Building type according to land use categories	Commercial buildingsOfficesCommunity buildings	Ordnance SurveySite survey
Street characteristics (Street)	Level of maintenance	 Extent of litter Instances of graffiti Instances of vandalism Instances of no street lighting 	– Site survey

 Table 2.4
 Indicators of housing/building characteristics

Measuring Layout

Urban layouts are difficult to quantify. Spatial network analysis is one of the ways in which spatial layouts can be objectively described and quantified to identify similarities or differences. Typically in a spatial network analysis, relationships between spaces in a city/settlement/building are represented as relational graphs similar to social network graphs. These graphs can then be analyzed to identify patterns and to quantify the relationships between spaces.

For this research, Multiple Centrality Assessment (MCA) is employed to measure layout (see Porta et al., 2006). MCA operates on a standard (or primal) graph representation based on street network systems which accounts for metric distances while analyzing the relationship between spaces. The final output provides a set of simple and compound measures of centrality numerically expressing the relative importance of a space in relationship to other spaces in the city. The output also includes a network map that shows the location and clustering patterns of spatial centrality (Fig. 2.3). A more detailed description of MCA and a table of the indicators employed to measure urban layout can be found in Appendix 1.

Measuring Overview

The elements of urban form link the constituent and distinct parts of this research together. This is the first time empirical research has examined the effect that urban form has on sustainability in a holistic manner. Indicators employed to measure urban form include both simple and complex measures collected or derived from secondary sources, primary data collection or detailed computer modelling. The range of indicators described above allows us to determine the relative influence that differing elements of urban form – land use, density, accessibility, housing/building characteristics and layout – have on economic, environmental and social sustainability.

The relationships that the elements of urban form, both individually and as a whole, have on the different aspects of sustainability are analyzed and outlined later in the book. The next section describes the case studies in detail and outlines the features of urban form in each area.

Case Study Areas: Profiles

The research in this book is based on empirical analysis in five British cities and a small number of neighbourhoods within each of these cities chosen for more in depth study. This part of the chapter describes these places in some detail, providing information about their urban form, housing, socio-demographic characteristics, history and economic profile. The case study cities are described first, followed by the profiles of the case study neighbourhoods.



Fig. 2.3 Example of an MCA-generated map of Leicester

Case Study Cities

Five provincial British cities provide the focus for the CityForm research: Leicester, Oxford and Sheffield in England and Glasgow and Edinburgh in Scotland (Fig. 2.4). The cities are all university towns and cover a variety of geographical and economic situations. It is important to note, that while the cities are varied in their economic, demographic and physical make-up, they cannot be claimed to represent UK cities as a whole. As a result, care must be taken when interpreting the results of this research.



Fig. 2.4 Location of the case study cities

General Characteristics of the Five Cities

Glasgow and Sheffield are the largest of the cities, with populations of more than 500,000. They are both traditional industrial cities that grew dramatically in the nineteenth century and now have important commercial roles. Edinburgh is the next largest with a total population of approximately 450,000 and is one of the few large provincial cities in the UK whose population is growing. It is the capital of Scotland and an important administrative and financial services centre. Both Leicester and Oxford are significantly smaller; Leicester with a population of 280,000 and Oxford with a population of 134,000. Leicester has quite a diverse economic structure and a thriving ethnic minority community that accounts for more than a third of its population. Oxford is most well-known as a university city, but is also a thriving tourist and business centre.

Demographically, the five cities tend to have a younger than average population with a high proportion of people in their twenties. This may be linked to their status as university towns, with the effects most noticeable in Oxford where one quarter of local people are aged between 20 and 29 and the average age of Oxford residents is 35 years 4 months. The most affluent of the five cities are Edinburgh and Oxford and they also have the highest proportion of young adults. Glasgow and Sheffield residents tend to be slightly older (38 and 39 years respectively), while Leicester has the youngest population profile with an average age of 35 years – more than 3 years younger than the UK average.

All of the case study cities – with the exception of Edinburgh – have a lower proportion of owner occupiers than the UK average (70%). Sheffield has the lowest proportion of private renters (10%), while in Oxford almost a quarter of households are private renters. Social renting on the other hand is most prevalent in Glasgow, where 40% of households have a social landlord, while just over one in six households in Edinburgh are in this situation.

Driving to work by car or van is the most popular mode of travel in all five cities with more than half of commuters using a car in Leicester and Sheffield. Glasgow is the only one to have an underground system and 4% use it while Sheffield has a tram network which attracts 3% of commuters. More than a quarter use the bus in Edinburgh but the city with the highest proportion of commuters using (all types of) public transport is Glasgow. Another notable difference is the high percentage, 15%, of people who travel to work on bicycle in Oxford, which is much higher than the other four cities (3% in Edinburgh).

Some Physical Characteristics of the Five Cities

Density provides a standard international indicator of urban form and enables both a simple comparison of individual cities and a worldwide perspective on the five case study cities. The figures presented in Table 2.5 reveal a significant variation in population density and dwellings per hectare. Sheffield is the least densely

	Density	ensity		Housing types (%)			
City	People per hectare	Dwellings per hectare	Detached	Semi- detached	Terraced	Flats/ apartments	
Edinburgh	17.1	8.2	11	14	15	60	
Glasgow	33.1	16.1	3	13	13	70	
Leicester	38.3	15.8	10	37	36	17	
Oxford	29.5	11.5	10	32	31	27	
Sheffield	14.0	6.1	14	37	30	18	

Table 2.5 Density and housing types: case study cities

Source: Census 2001

populated city of the five, both in persons per hectare and dwellings per hectare. Leicester has the greatest number of persons per hectare and Glasgow has the greatest number of dwellings per hectare. Edinburgh has a comparably low density, similar to that of Sheffield. These statistics also show that there is no clear divide between England and Scotland despite the distinctive tenemented housing stock prevalent in both Scottish cities. Edinburgh and Glasgow both have much higher numbers of flats than the English cities with around 60% and 70% of households living in flats in these two cities. However, there are differences between the English and Scottish cities studied in the number of rooms per dwelling. The English cities all had an average of more than 5 compared with 4 in Glasgow and 4.6 in Edinburgh.

As Table 2.5 shows the densities of these cities range from 29 to 38 persons per hectare which places them at the lower end of the international spectrum of urban densities but much higher than most North American cities. The densities are at the low end of the range of densities of European cities and substantially below the densities of Asian cities that are typically over 200 people per hectare (Bertaud, 2003).

Case Study Neighbourhoods

The starting point for the empirical research is the neighbourhood and fifteen case studies were chosen to represent inner, between and outer neighbourhoods within the five cities. The case study neighbourhoods are chosen to provide a slice through each of the cities and to represent a wide range of neighbourhoods. The overview of these neighbourhood characteristics also provides a useful insight into the spatial socio/demographic and housing stock structure of (British) cities.

Each case study neighbourhood includes at least 2000 households, a mixture of land uses, a range of housing types and street patterns, nearby public transport and households with a range of socio-economic backgrounds. These neighbourhoods are used as the principal spatial unit of study and where appropriate are also split into sub-areas. Sub-areas are defined using maps and local knowledge, to identify natural physical sub-divisions respecting obvious major boundary features and to

reflect relative homogeneity of urban form within their boundaries. Each case study neighbourhood is divided into 6–7 sub-neighbourhoods, giving a set of 97 in total (a few of which contain only a small number of households, because they are dominated by non-residential land uses).

Characteristics of Neighbourhoods

A range of information was collected about the case study neighbourhoods. This includes a site survey capturing information about the built environment and a questionnaire survey providing information about household characteristics, behaviour, feelings about the neighbourhood and use of local facilities (e.g. parks, shops, public transport). An overview of the 15 neighbourhoods is profiled at the end of this chapter (Table 2.6), and more details on individual localities are included in Appendix 2.

At one level these areas represent a set of diverse neighbourhoods but a number of regularities can also be discerned. Gross population densities follow a negative gradient spatial structure from inner to outer neighbourhood. There are also consistent spatial demographic patterns: younger people with few children living in inner areas and older households and families predominating in outer areas. Between areas are more diverse. Private rented housing is focused in the inner areas and outer areas are almost exclusively owner occupied. Social housing in British cities is spatially concentrated, and this is reflected in our case study areas, being located mainly in a few inner areas. These findings show the spatial structure of the five cities, as given by the characteristics of the case study areas, conform to an urban system diffusing from a central core, and that the different physical urban form elements integrate together with socio-economic characteristics.

Layouts of Neighbourhoods

The spatial analysis is carried out at the city level, neighbourhood, sub-area and street level. Most of the comparative analysis to determine the performance of various types of urban form is at the neighbourhood and sub-area level. The spatial characteristics of the neighbourhoods reveal a range of layout types ranging from the predominantly gridded to those that are largely culs-de-sac. The characteristics are quantified using MCA analysis as it permits the calculation of many indices that measure street networks and allow systematic comparison of the case study neighbourhoods (see Appendix 1 for details). There are measurable differences between the neighbourhoods and none can be said to represent one layout type or another, as all have a complex mixture of layout forms (Fig. 2.5).

The MCA analysis also identified the density of street intersections, the relative degree of complexity of the street networks, their interconnectedness, and how efficient the networks are – related to actual distances between intersections. In addition, the degree of compactness or sprawl is measured by considering the 'fractal' dimension of street patterns in the case study neighbourhoods. The fractal



Fig. 2.5 Shows the spectrum of complexity index scores across the fifteen neighbourhoods. Lower values indicate higher choice-grid like pattern that are predominantly in inner neighbourhoods and higher values indicate low choice-tree like pattern in the suburbs. (*light blue* = suburban; *medium blue* = between; *dark blue* = inner city)

dimension index ranges from 1 to 2 - a score of 1 would be a perfect linear system where intersections lie on a straight line, while a score of 2 would be a system where intersections are distributed evenly throughout the space covering the whole neighbourhood. The range in the neighbourhoods from a more compact form in Sheffield inner area to the sprawling layout of Glasgow outer has a relatively narrow spread of values is again due to the fact that many case study areas have both compact and spread areas within them (Fig. 2.6).



Fig. 2.6 Examples of compact and sprawling neighbourhoods

Summary and Conclusions

At one level the elements of urban form are relatively simple – land use, density, accessibility defined by transport infrastructure, characteristics of the built environment and layout. Although there is an expanding literature that seeks

to quantify urban form using complicated measures and advanced statistical techniques the findings of these studies have emphasised the overlapping nature of the elements and the results have not justified the complexity. The study has chosen in general to apply as simple measures as possible as the best way to elucidate the issues on sustainability. The analysis reported in this book is based on five cities in the UK. The urban form characteristics of these cities in terms of population density are very similar within the spectrum of world cities. These densities are at the low end of the range of densities of European cities and substantially below the densities of Asian cities.

The fifteen case study neighbourhoods selected for deeper study and located at inner, middle and outer points to represent slices through each of the five cities. There are a number of pointers that arise from the various measures of the elements of urban form of these neighbourhoods:

- The inner neighbourhoods tend to be well connected and complex, with predominately grid-like structures;
- The inner neighbourhoods tend to have more compact layouts in comparison with suburban neighbourhoods;
- The suburban neighbourhoods have tree-like structures with a single or limited number of main roads acting as a spine or trunk with culs-de-sac;
- The gross population densities also follow a similar spatial structure; a broadly consistent negative gradient from the city centre;
- There are consistent spatial demographic patterns: younger people with few children living in inner neighbourhoods and older households and families predominating in outer neighbourhoods. Between neighbourhoods are more diverse.
- Private rented housing is focused in the inner neighbourhoods and outer neighbourhoods are almost exclusively owner occupied.
- Social housing in British cities is spatially concentrated, and this is reflected in our case study areas, being located mainly in a few inner neighbourhoods.

These findings show the spatial structure of the five cities, as given by the characteristics of the case study neighbourhoods, conform to an urban area diffusing from a central core, and that the different physical urban form elements integrate together with socio-economic characteristics. The following chapters investigate the sustainability of different dimensions of this urban system drawing on more detailed study of these case study neighbourhoods.

CASE STUDY CITY: EDINBURGH INNER



MIDDLE



OUTER



INNER

General description

- A high density neighbourhood dominated by traditional tenements and modern flats. The area has very little green space and few residential gardens. There is a busy high street with shops and cafes. The area also includes a large hotel, multiplex cinema and associated leisure uses.
- Layout is compact with grid and cul-de-sac form.

Social characteristics and housing

- The area is home to many single person households, young couples and homes in multiple-occupation (together about 75% of households). Around one-fifth of residents are retired, and there are relatively few families in the area. More than 90% of households live in flats.
- Just over half of homes here are owner occupied and one quarter privately rented. 20% of local stock is social rented.

MIDDLE

General description

Much of the area is characterized by post-war housing: detached and semi-detached homes with gardens. Flats, some of them built recently, occur in parts of the neighbourhood. The area lies to the north of the city centre and is bisected by a major road and railway line. A large supermarket, car park, cemetery and school dominate one part of the area.

Layout is predominantly gridded, not orthogonal.

OUTER

General description

A largely residential area, dominated by detached and semi-detached houses with private gardens. The area includes part of a university campus and is bisected by a major arterial road. This road functions as a local high street with a variety of shops and services. It is extremely busy with vehicular traffic, but provides easy access to bus services.

Layout is a compact super grid.

Social characteristics and housing Almost 40% of residents in this area are retired, and around 15% of households include children. Half of homes in this area are flats, and the other half are a fairly even mixture of detached, semi-detached and terraced houses.Two-thirds of these homes are owner occupied and less than one-tenth are privately rented. Social renting accounts for around 17% of the stock.

Social characteristics and housing

A large proportion of residents in this area are retired (40%) and almost one-quarter of households include children. Around one-third of homes are detached and one-third are semi-detached. Flats account for 17% of stock and terraced houses 11% of local homes. Almost all homes in this area are owner-occupied (97%). CASE STUDY CITY: GLASGOW INNER



MIDDLE



OUTER



INNER

General description

A very diverse area, including city centre shops, bar, restaurants, cultural facilities and a least one large open space. Residential uses are spread unevenly across the area, with pockets of high density flats – including student accommodation – in some parts. Not all parts of the area are doing well, and there are derelict buildings and underused sites. Residents here have very little private green space.

Layout is deformed compact grid.

MIDDLE

General description

- The area has two distinct parts. One part is very low density, with large houses (some subdivided into flats) set in large gardens and leafy streets. The other part has tenement flats, shops and a greater mix of uses. A local park contributes to the leafy feel.
- Layout is a deformed grid with some compact grids.

OUTER

General description

Close to the edge of the city, this area is bordered by farmland and has a river running through it. Many of the houses are semi-detached with private gardens and many families live in this area. Part of the area includes an industrial estate and a school.

Layout is clustered dispersed culs-de-sac.

Social characteristics and housing

- The area is home to many single person households, young couples and homes in multiple-occupation (together about 87% of households). Very few families live in this part of the city, and one-tenth of residents are retired. More than 90% of households live in flats.
- Almost 60% of homes here are owner occupied and one quarter privately rented. 16% of local stock is social rented.
- Social characteristics and housing Around one third of residents in this area are retired and almost one-quarter of households include children. Around 80% of households are living in flats. Most of the other homes in the area are terraced or semi-detached.
- 86% of homes here are owner occupied and only a very small proportion of housing is privately rented. Social renting accounts for around 6% of the stock.

Social characteristics and housing Just over one quarter of residents in this area are retired and around one-third of households include children. Around 40% of homes are semi-detached and the rest are evenly divided between flats, terraced houses and detached homes.

Around 70% of homes in this area are owner-occupied with the remainder let through social landlords. There is no private rented accommodation here. CASE STUDY CITY: LEICESTER INNER



MIDDLE



OUTER



INNER

General description

A very diverse area, including the city hospital, museum, art gallery and university campus. Residential uses are unevenly spread, with pockets of high-density flats and student accommodation. There is little private green space here, although there are some public open spaces. A significant proportion of residents are non-white; and there is a high number of single person households.

In terms of layout, the area has a deformed wheel, radial pattern.

MIDDLE

General description

This part of the city is dominated by terraced housing. It is close to the university and has a significant proportion of private rented properties. Some parts of the area are slightly lower density with semi-detached houses, and there are some flats. A significant proportion of local residents are non-white.

In terms of layout, the area has a deformed grid.

Social characteristics and housing

The area is home to many single person households, young couples and homes in multiple occupation (together about 80% of households). There are relatively few families in the area and around 15% of residents are retired. This area is notable for its high level of unemployment (13%). Just under 90% of households live in flats.

Only one-fifth of homes here are owner occupied and one quarter privately rented. More than half of homes in this area are social rented.

Social characteristics and housing Around two thirds of homes in this area are terraced, and just over 10% are flats.

74% of homes here are owner-occupied and one quarter are privately rented.

OUTER

General description

- Residential density is low in this part of the city. Detached and semi-detached houses with gardens dominate, although there are a few flats in places. There is some public open space, but most residents have access to private gardens.
- In terms of layout, the area has a very deformed grid with culs-de-sac

Social characteristics and housing

In this area around one third of households are older (one or more adults aged 60+), but only 20% of residents are retired. In most other case study areas the proportion of retired and older households are roughly equal. A quarter of households in this area include children. This area is dominated by semi-detached homes, which account for more than 60% of local housing stock. There are few flats or terraced houses here and almost all the other homes in the area are detached houses.

Owner-occupation rates are very high, and there is very little rented accommodation (around 5%). CASE STUDY CITY: OXFORD INNER



MIDDLE



OUTER



INNER

General description

The area includes a high number of non-residential buildings including university colleges, a library and museum. A significant proportion of residents are single and young, although some families and older people live in the southern part of the area. The proportion of private and public open space varies enormously, with very little private open space in the most central parts.

Layout is in crucifix form, with small blocks at the centre

MIDDLE

General description

- The area includes a mixture of housing types. Open spaces include allotments, a cricket ground, a river and private gardens. It is bound to the west by the railway line and a canal runs through the area. A number of non-residential uses include a community centre, hospital and numerous cafes and restaurants. A number of the residential properties are owned by the university.
- Layout is an elongated deformed grid with compact grid within.

OUTER

General description

This area is a large housing estate, laid out with culs-de- sac and a mixture of detached, semi-detached and terraced houses. Density is high for an outer case study area. The area includes a number of public buildings, e.g. school, stadium and business park. There is a mixture of public and private open space, with many of the houses having access to private gardens.

Layout is predominantly culs-de-sac.

Social characteristics and housing Around one quarter of residents are retired, and 15% of households include children.

Around one half of homes are flats, and most of the remaining housing stock is terraced, with a very small proportion of detached and semi-detached homes.

Around half of homes here are owner-occupied and one-third are privately rented.

Social characteristics and housing Around one third of residents in this area are retired and almost one-fifth of households include children. Just under half of homes are terraced and one quarter of households live in flats. Semi-detached homes account for almost all the other houses in the area.

Roughly 60% of homes are owner-occupied and one third are privately rented. Social renting accounts for less than 10% of stock.

Social characteristics and housing

In this area one-fifth of residents are older, and more than one-third of residents are retired. In most other case study areas the proportion of retired and older households are roughly equal. One third of households in this area include children. Just under half of homes in this area are terraced, and the rest are split equally between semi-detached homes and flats.

Around 40% of homes here are owner-occupied and a similar proportion is social rented. Around 10% of homes in this area are in shared ownership. CASE STUDY CITY: SHEFFIELD INNER



MIDDLE



OUTER



INNER

General description

This part of the city includes a number of non-residential buildings including: a sports stadium, university buildings, a school, cafes and restaurants. Residential buildings include student accommodation and both high and low-rise blocks of flats. There are also a number of open spaces, including parks and sports fields.

Layout is a deformed compact grid.

MIDDLE

General description

A largely residential area with the majority of residents living in terraced houses. Public open space is concentrated in specific areas, although most houses have some form of private garden. Sheffield's topography is particularly relevant in this area, with the main road from the city centre lying at the top of a steep hill.

Layout is a deformed compact grid.

Social characteristics and housing

- Around one quarter of residents are retired and about 15% of households include children. Unemployment is slightly higher here than in other areas (5%).
- This area is dominated by flats around 80% of households living in flats. Most of the other homes in the area are terraced or semi-detached. Just over one-quarter of homes here are owner occupied and almost one-fifth are privately rented. More than half of homes in this area are social rented

Social characteristics and housing Just over one-fifth of residents are retired and 15% of households include children. Around 60% of homes in this area are terraced and the rest are a fairly even mixture of detached, semi-detached and terraced houses.

Three-quarters of homes here are owner-occupied and just under 20% are privately rented. Social renting accounts for around 12% of housing stock.

OUTER

General description

This area is on the edge of the city close to farmland and open countryside. Residential density is fairly low, with many detached and semi-detached houses with large gardens. Non-residential uses are limited and concentrated in specific areas along the bus route. There is a housing estate to the north of the area made up of flats and houses which has access to shared open space.

Layout is curvilinear with culs-de-sac.

Social characteristics and housing Just over 40% of households in this area are older, and one-quarter include children. Just over half of homes in this area are semi-detached and around 30% are detached. The rest of the housing stock is almost all made up of flats.

A large proportion of homes (86%) are owner-occupied, and 11% are social rented.

References

- Aldous, T. (1992) Urban Villages: a concept for creating mixed-use urban developments on a sustainable scale. Urban Villages Group, London.
- Anderson, W. P., Kanargoglou, P. S. and Miller, E. (1996) Urban Form, Energy and the Environment: A Review of Issues, Evidence and Policy. *Urban Studies*, 33, pp. 17–35
- Barton, H. (2000) Sustainable Communities: the potential for eco-neighbourhoods, Earthscan, London.
- Barton, H., Grant, M. and Guise, R. (2003) *Shaping Neighbourhoods: a guide for health, sustainability and vitality, Spon Press, London.*
- Bertauld, A. (2003) *Metropolitan Structures around the World*, http://alain-bertaud.com/ #B.ComparativeUrbanStructures
- Breheny, M. (1997) Urban compaction: feasible and acceptable? Cities, 14(4), pp. 209-217
- Burton, E. (1997) *The Compact City: just or just compact?* Unpublished PhD thesis, Oxford Brookes University, Oxford.
- Churchman, A. (1999) Disentangling the Concept of Density. *Journal of Planning Literature*, **13**(4), pp. 389–411.
- Commission for Architecture and the Built Environment and Department of the Environment Transport and the Regions (2000) *By Design: urban design in the planning system: towards better practice*, Thomas Telford, London.
- Cowan, R. (1997) *The Connected City: a new approach to making cities work*, Urban Initiatives, London.
- Cutsinger, J., Galster, G., Wolman, H., Hanson, R. and Towns, D. (2004) Verifying the Multi-dimensional Nature of Metropolitan Land Use: Advancing the Understanding and Measurement of Sprawl, Wayne State University, Detroit, Michigan.
- Dempsey, N. (2008) Quality of the built environment in urban neighbourhoods, *Planning Practice and Research*, 23(2), pp. 249–264.
- Department for Communities and Local Government (DCLG) (2006) *Planning Policy Statement 3* (*PPS3*): *Housing*, DCLG, London.
- Department of the Environment Transport and the Regions (DETR) (2000) *Planning Policy Guidance Note 3: housing*, DETR, London.
- Ewing, R., Pendall, R. and Chen, D. (2002) *Measuring Sprawl and its Impact*, Smart Growth America, Washington, D.C.
- Gehl, J. (2001) Life Between Buildings: using public space, Arkitektens Forlag, Copenhagen.
- Gehl, J., Gehl, A., Transport for London and Central London Partnership (2004) Towards a Fine City for People: public spaces and public life - London 2004, Gehl Architects, Copenhagen.
- Grant, J. (2002) Mixed Use in Theory and Practice: Canadian experience with implementing a planning principle, *Journal of American Planning Association*, **68**(1), pp. 71–84
- Gray, D. E. (2004) Doing Research in the Real World, Sage, London.
- Healey, P. (1997) *Collaborative Planning: Shaping Places in Fragmented Societies*, MacMillan Press Ltd, Basingstoke.
- Hillier, B. (1996) Space is the Machine: a configurational theory of architecture, Cambridge University Press, Cambridge.
- Hillier, B. and Hanson, J. (1984) *The Social Logic of Space*, Cambridge University Press, Cambridge.
- Holmes, C. (2007) New Opportunities for an Ageing Population, *Town and Country Planning*, **76**(11), pp. 382–384
- Jenks, M. (2000) The Acceptability of Urban Intensification. In. *Achieving Sustainable Urban Form*, (eds. K. Williams, E. Burton and M. Jenks), E & FN Spon, London, pp. 242–250.
- Jenks, M. and Dempsey, N. (2005) The Language and Meaning of Density. In *Future Forms and Design for Sustainable Cities*, (eds. M. Jenks and N. Dempsey), Architectural Press, Oxford, pp. 287–309.

- Liu, S. and Zhu, X. (2004) Accessibility analyst: an integrated GIS tool for accessibility analysis in urban transportation planning. *Environment and Planning B: Planning and Design*, **31**(1), pp. 105–124.
- Mardaljevic, J. (2005) Quantification of Urban Solar Access. In Future Forms and Design for Sustainable Cities, (eds. M. Jenks and N. Dempsey), Architectural Press, Oxford, pp. 371–391.
- Newton, P., Tucker, S. and Ambrose, M. (2000) Housing Form, Energy Use and Greenhouse Gas Emissions In Achieving Sustainable Urban Form, (eds. K. Williams, E. Burton and M. Jenks), E & FN Spon, London, pp. 74–83.
- Office of the Deputy Prime Minister (ODPM) (2003) National Land Use Database (NLUD): Land Use and Land Cover Classification, HMSO, London.
- Office of the Deputy Prime Minister (ODPM) (2005) Sustainable Communities: homes for all, The Stationery Office, Norwich.
- Penn, A., Hillier, B., Banister, D. and Xu, J. (1998) Configurational modelling of urban movement networks, *Environment and Planning B: Planning and Design*, 25(1), pp. 59–84.
- Porta, S., Crucitti, P. and Latora, V. (2006) The Network Analysis of Urban Streets: a primal approach, *Environment and Planning B: Planning and Design*, 33(5), pp. 705–725.
- Porta, S., Latora, V., Wang, F., Strano, E., Cardillo, A., Scellato, S., Iacoviello, V. and Messora, R. (2008) Street centrality and densities of retails and services in Bologna, Italy. *Environment and Planning B: Planning and Design*, 36(3) pp.450–465.
- Rapoport, A. (1975) Toward a Redefinition of Density, *Environment and Behavior*, 7(2), pp. 133– 158.
- Schoon, N. (2001) The Chosen City, Spon Press, London.
- Talen, E. (2003) Neighbourhoods as Service Providers: a methodology for evaluating pedestrian access, *Environment and Planning B*, 30, pp. 181–200
- Urban Task Force (1999) Towards an Urban Renaissance, E & F Spon, London.
- Williams, K., Burton, E. and Jenks, M. (eds.) (2000) Achieving Sustainable Urban Form, E & FN Spon, London.
- Winter, J. and Farthing, S. (1997) Coordinating Facility Provision and New Housing Development: impacts on car and local facility use. In *Evaluating Local Environmental Policy* (ed. S. M. Farthing), Avebury, Aldershot, pp. 159–179.
- Woodford, G., Williams, K. and Hill, N. (1976) Research Report 6: The Value of Standards for the External Residential Environment, Department of the Environment, London.

Chapter 3 Travel and Mobility

Neil Ferguson and Lee Woods

Introduction

Combating the undesirable effects of mobility in cities caused by the use of the private car has become a key issue in the development of sustainable urban policy. Associated with car use are a number of well-documented problems including rising levels of energy consumption, road congestion, greenhouse gas emissions and pollution as well as road safety, health and severance effects (European Commission, 2007a). Underlying these problems is a complex process which involves interaction between rising levels of car ownership, the development of road transport provision and the location decisions of individuals and businesses in and around cities. The process has resulted in the emergence of new urban forms typified by the decentralisation of activities and the unstructured expansion of urban areas into the surrounding countryside otherwise known as urban sprawl (European Environment Agency, 2006). New suburban residential neighbourhoods, characterised by low density, single-use development, reinforce the dominance of the car as the principal, or sometimes sole, form of transport to access everyday activities.

The dominant role of car travel is exemplified by the fact that in the European Union (EU), passenger car use rose by 18% between 1995 and 2004 (European Environment Agency, 2008). In 2004, travel by car was responsible for 74% of all passenger transport, including air and sea transport, and constituted an average of around 27 km/EU inhabitant per day (European Commission, 2007b). In the future, worldwide travel is forecast to increase by 48% between 2000 and 2050. Whilst the largest increases are forecast in China, India and non-EU Europe, significant increases are also forecast in the EU and North America (European Environment Agency, 2007).

Around 75% of the population of the EU currently live in urban areas (European Environment Agency, 2006). It is therefore clear that the transport undertaken

N. Ferguson (⊠)

Department of Civil Engineering, University of Strathclyde, Glasgow, UK

by those living in urban areas constitutes a significant proportion of all personal travel. Moreover, the average number of car trips per person per day in urban areas increased by 10% in the EU in the 1990s, principally at the expense of public transport, walking and cycling. During the same time period, the length of the average car journey increased by 20% (European Commission, 2006).

In recent years, a return to traditional urban forms has been advocated by New Urbanists in the United States (Duany et al., 2001) and the Urban Villages Campaign (Thompson-Fawcett, 2000) and the Urban Task Force (DETR, 1999) in the United Kingdom. Although some differences in approach can be detected at the strategic level reflecting the distinctive planning contexts in the two countries, strong parallels can be detected at the neighbourhood level (Hall, 2000). The neighbourhood urban forms promoted by these groups are characterised by moderately high densities of housing, mixed land-uses, proximity to public transport and grid-pattern road layouts based around the traditional block (Panerai et al., 2004) with provision for walking and cycling. It is claimed that these urban forms will reduce travel by car by encouraging walking and cycling to local amenities and the use of public transport for longer journeys. In so doing, these urban forms will address, at least in part, the economic, environmental and social problems referred to above.

At the heart of these traditional urban forms lies the aim of providing a good level of local accessibility which meets the needs and wishes of the neighbourhood population and thereby supports more sustainable mobility patterns. However, the success of traditional urban forms in reducing car use can be questioned on four grounds.

First, for households with access to a car, the flexibility offered and the ease with which spatial separation can be overcome by car can provide a strong incentive to use a car even when viable transport alternatives exist. This has two potential implications (a) the car may remain the favoured mode to undertake a journey even when it would be possible to walk, cycle or use public transport, particularly in the absence of any measures of car restraint and (b) the car offers a more extensive set of potential destinations than the restricted set open to those without access to a car. In a comparison of the relative costs and benefits of travelling to potential destinations, a destination which realistically can only be reached by car may be favoured over a destination which can be reached by other means. For example, it may be the case that driving to an edge-of-city shopping centre might prove more attractive than walking a short distance to local shops.

Second, as discussed by Boarnet and Crane (2001), any time and cost savings gained by using locally available amenities may simply lead to the generation of additional travel for other purposes. Thus, all travel (local and non-local) must be taken into account when comparing the travel patterns of residents of more than one type of neighbourhood.

Third, when it is argued that traditional urban forms will reduce travel by car it is implied that travel, and the activities this travel serves, are organised with reference to the residential location. While this may be true of a certain proportion of travel, it is also the case that constraints which require an individual to be at a specific location at a particular time can reduce the influence of local neighbourhood design on travel choices. For example, the requirement to be at work for a large part of the day may lead to an individual organising some activities with reference to the work location or the journey to/from work. This requires a more complex perspective of accessibility (also known as constraints-based accessibility) which relates to how easily urban form affords access to activity locations given the characteristics of individuals, the activities in which they participate and the constraints which these activities place upon them (Pirie, 1979; Kwan, 1998; Miller and Shaw, 2001).

Fourth, implicit in the vast majority of studies undertaken to date is the assumption that the direction of causality runs from urban form to travel behaviour. That is, a change in urban form is assumed to engender a change in travel behaviour. However, this assumption overlooks how longer term decisions relating to residential location and employment may be influenced by the travel preferences of decision-makers. For example, individuals who have strong preference towards car use may be more likely to favour low density residential locations that support this preference. Thus, the observed relationship between urban form and travel behaviour may be attributable, at least in part, to underlying travel preferences that influence residential location choices. This is known as self-selection bias and has been addressed in a handful of studies undertaken in the United States (Boarnet and Crane, 2001; Krizek, 2003; Handy et al., 2005; Bhat and Gio, 2007). To date, there have been no studies in Europe which have examined this specific issue.

Given the potential limitations outlined above, this chapter seeks to investigate the effectiveness of urban form as a policy tool which can be utilised to induce more sustainable travel patterns. There is already a large body of empirical research which has explored the relationship between urban form and travel behaviour. Various dimensions of urban form (e.g. density, mix of uses, road layout and location) and travel behaviour (e.g. trip frequency, trip length, choice of mode and travel time) have been examined. This, in turn, has generated a number of syntheses of recent work (Crane, 2000; Ewing and Cervero, 2001; Badoe and Miller, 2000; Banister, 2005). In seeking to build upon this previous work, this chapter contributes to our understanding of the influence of urban form and travel in the following ways. First of all, it explores the extent to which the availability of cars in a household is influenced by urban form. As argued above, car availability may undermine any potential reductions in car use attributable to urban form. Understanding this relationship is therefore of central importance to the development of a fuller understanding of the causal relationship between urban form and travel behaviour. Following this, the trip-making behaviour of individuals in different urban settings is examined. This includes analyses of the influence of urban form on the number of trips undertaken by car and the frequency of using local amenities. Finally, the relationship between urban form and the total distance travelled by individuals in the course of their daily activities is investigated. Clearly, this measure of travel behaviour takes into account any additional travel generated from savings gained by using local amenities in traditional urban neighbourhoods.

Research Approach

Data

The research is based on evidence from Glasgow and Edinburgh. The Scottish Household Survey (SHS) was the principal dataset used. Households interviewed between 2003 and 2006 were selected for this research (Hope, 2007). A travel questionnaire survey was also carried out in the six case study areas located in the two cities based on self completion. This questionnaire combined questions about travel behaviour and transport resources. One of the objectives of this travel survey was to examine changes in travel behaviour before and after moving home, especially home owners.

Measures of Urban Form

With reference to Chapter 2, measures of the following elements of urban form – density, land use mix, accessibility of place of residence in relation to the city centre and access to public transport – are selected. Population density was chosen to represent intensity of development. Various options were explored to represent land use mix using workplace employment data. The ratio of retail jobs to population was selected as it has relatively low correlations with other urban form measures. The accessibility of place of residence in relation to the city centre was represented by the road network distance from home to city centre. Finally, a dummy variable was created which reflected the quality of access to public transport. If a SHS respondent lived within 6 min of a bus stop and the frequency of service was at least one bus every 10 min then the respondent was categorised as having access to a high quality bus service.

A prime consideration in the selection of scale was the geographical resolution of the SHS dataset. A specialised SHS dataset was obtained which contained the home zone of respondents. Each home zone has a population of between 2,500 and 6,000 people and there are 133 and 101 such Intermediate Geography zones in Glasgow and Edinburgh respectively (Scottish Government, 2005). In consequence, the urban form measures assigned to each respondent were those based on the Intermediate Geography zone system, with the exception of the access to public transport measure which was determined directly from variables within the SHS. The home postal address of respondents to the travel and household questionnaires was known and so the selection of geographical scale on which urban form was measured was not constrained in the same way as the SHS dataset. Where appropriate, urban form measures are determined at the Intermediate Geography level and also using a zoning system with smaller geographical units known as Datazones (Scottish Government, 2005) which have residential populations of between 500 and 1,000.

Analysis

The following sections report the results of the research undertaken. In the next section the influence of urban form on household car availability is examined. This analysis considers both cross-sectional data (that is, data which relates to household car availability at a single point in time) and longitudinal data which incorporates the change in household car availability following a change in residential location. The principle advantage offered by longitudinal data is that it allows greater insight into the causal relationship between urban form and car availability than that offered by cross-sectional data which is limited to providing evidence of co-variation (Finkel, 1995). This is followed by sections that investigate the relationship between tripmaking and urban form and travel distance and urban form, with the latter section considering both cross-sectional and longitudinal datasets.

Multivariate statistical models were estimated in order to examine the influence of the various elements of urban form on car availability and travel behaviour. The effects of socio-economic characteristics were taken into account by including variables such as age, gender and income as explanatory variables alongside urban form variables. For the analysis of travel behaviour the effect of urban form on workdays was distinguished from travel on non-workdays. In so doing, the effect of work-related constraints on travel behaviour was taken into account. In general, explanatory variables were taken to be statistically significant if there was found to be less than a 5% chance that they had no effect on the measure of travel behaviour under consideration.

Household Car Availability

The availability of cars in a household is a key factor in determining the level of accessibility enjoyed by household members. Between 1996/1998 and 2006, car ownership in urban areas of Great Britain with a population of more than 250,000 grew from 0.97 to 1.15 cars per household (DfT, 2002; DfT, 2007b). As discussed above car availability can also be regarded as an intervening variable which indirectly links urban form and travel behaviour. Although most multivariate analyses of travel behaviour have used car ownership as an explanatory variable alongside urban form and other socio-economic factors, this approach ignores the effect of urban form on car ownership itself. In so doing, the causal link between urban form and travel behaviour is overlooked and the overall effect of urban form on travel behaviour may be underestimated.

A few studies have specifically examined the relationship between urban form and car ownership. In the UK, Stead (1999, cited in Stead, Williams and Titheridge, 2000) found that car ownership was lower in areas close to good quality public transport. Research in the United States has found car ownership levels to be lower in mixed land use developments (Chu, 2002; Hess and Ong, 2002), in neo-traditional developments (Shay and Khattak, 2005) and in areas with higher household and employment densities and transit availability (Bhat and Gio, 2007). The number of cars normally available for use by households is predicted using an ordered logit model. This is an appropriate model to use in situations where the outcome variable – the number of available cars – is both discrete and ordered (Borooah, 2002). The estimated model predicts the probability of a household having zero, one, two or three or more available cars for a given set of urban form attributes and socio-economic characteristics.

The results of the model are shown in Table 3.1 where the unit of analysis is the intermediate geography zones in Edinburgh and Glasgow. The size and direction of influence of explanatory variables are indicated by the parameter estimates shown in this table. As might be expected, the strongest effect on available cars is household income – for example, households with an income greater than £40,000 per annum have a greater probability of having available three or more cars than a household with an annual income of between £20,000 and £30,000. With regard to the urban form variables tested, car ownership was found to increase with decreasing population density and land use mix and increasing distance from the city centre.

In addition to the cross-sectional model estimated from the SHS, the effect of a change in urban form was examined by constructing a longitudinal model of the effect of moving home on household car availability. This approach is useful because it allows us to study one of the mechanisms by which the development

	Parameter estimate
Small adult	0.70**
Single parent	-0.28^{*}
Small family	0.83**
Large family	1.22**
Large adult	1.83**
Older smaller	0.85**
Single pensioner	-0.35**
Single adult (Reference)	
Glasgow resident	-0.54^{**}
Edinburgh resident (Reference)	
No access to high quality bus service	0.16**
Access to high quality bus service (Reference)	
Income £0–£10 k	-2.14^{**}
Income £10–£20 k	-1.40^{**}
Income £20–£30 k (Reference)	
Income £40 k plus	0.92**
Distance from city centre (km)	0.07**
Retail employment to population ratio	-0.72^{*}
Population density	-0.05^{**}
Number of observations	8254
Log likelihood at 0	14198.66
Log likelihood at convergence	10344.86
McFadden pseudo R-square	0.23

Table 3.1 Ordered logit model for cars available to households

* indicates statistically significant at the 5% level

** statistically significant at the 1% level

of neighbourhoods which follow more traditional urban forms may influence car availability. (The other mechanism is to re-engineer existing neighbourhoods and explore the effect on the behaviour of inhabitants which lies outside the scope of this study). The number of cars available after moving house was estimated using an ordered logit model in which change in population density, the ratio of retail employment to population and change in the distance to the city centre were included as explanatory variables. Table 3.2 shows the model results where the unit of analysis this time is data zones, representing 500 to 1000 people. It can be seen that the estimated parameters for change in population density and change in retail employment: population ratio are both negative. Change in network distance to the city centre is not statistically significant and neither is change in distance to work. This suggests that, after controlling for change in household income and change in distance to work, moving to a neighbourhood with higher density and greater degree of land use mix will result in a reduction in the number of available cars.

The results of the analysis described in this section demonstrate that the number of cars available to a household is lower when the characteristics of the residential neighbourhood conform to a traditional urban form typology. This is an important finding in the sense that it follows that households in more traditional neighbourhoods have lower levels of car accessibility and thus may be more likely to walk or cycle to local amenities or use public transport for longer journeys. However, in order to find out if this is the case the direct effect of urban form on travel behaviour must be examined alongside the indirect effect of urban form on travel behaviour through car availability. This is undertaken in the following two sections.

	Parameter estimate
Change in population density	-0.06
Change in distance to work (kilometres)	0.003
Change in distance to city centre (kilometres) \times 1000	0.01
Change in retail employment to population ratio	-2.36*
Change in income – a lot less	-4.70**
Change in income – a little less	-0.28
Change in income – about the same	-0.08
Change in income – a little more	-0.56
Change in income – a lot more (Reference)	
0 cars before move	-7.30**
1 car before move	-1.41
2 cars before move	1.90
3 or more cars before move (Reference)	
Number of observations	111
Log likelihood at 0	226.70
Log likelihood at convergence	112.20
McFadden pseudo R-square	0.51

 Table 3.2
 Ordered logit model for change in available cars on moving house

* indicates statistically significant at the 5% level

** statistically significant at the 1% level

Trip Generation

Drawing on transport planning terminology, a trip can be defined as a one-way journey from an origin to a destination in order to undertake some activity at the destination; the type of activity undertaken is used to assign a purpose to the trip. The process by which an individual elects to undertake a trip is known as trip generation and the number of trips provides some indication of the amount of out-of-home activity in which an individual participates.

All other things being equal, and given the claims made in support of traditional urban forms, we would expect the number of trips made by car by those living in traditional neighbourhoods to be lower than those living in lower density, single-use neighbourhoods. We would also expect greater use to be made of local amenities. Previous studies give some supporting evidence. Næss (2006) observed higher weekday trip frequencies for residents in peripheral areas of the Copenhagen metropolitan area than in central areas. At the weekend, trip frequency was found to increase with proximity to the centre of the city after controlling for the effects of density (which itself was found to be to be inversely related to trip frequency). Drawing on a number of studies undertaken in the UK Banister (2005) concluded that trip frequency by car reduces as residential density increases. On the other hand, in a study conducted in the United States Boarnet and Sarmiento (1998) found no evidence that population density or functional mix influenced the number of non-work car trips.

In order to test the aforesaid claims, trip-making in different urban settings was examined and the results are given below. The analysis is divided into two parts. In the first part, the effect of urban form on trip frequency is examined. The second part examines the extent to which proximity to certain classes of local amenity influences the frequency of use of that class of amenity.

Trip Frequency

The SHS Travel Diary which forms part of the SHS was used to examine the effect of urban form on trip frequency. This diary records the travel behaviour of a randomly selected adult in each surveyed household over the course of a single day. All trips which are over a quarter of a mile or more than five minutes on foot are recorded. Again surveyed adults living in households residing within Glasgow and Edinburgh are selected for analysis and the spatial unit of analysis is intermediate geography zones of 2,500 to 6,000 people. Respondents for whom no travel was reported on the day in question are included in the dataset.

Ordered logit models are used to predict the total number of trips, the number of car trips and the number of single occupant car trips and the results are shown in Table 3.3. Factors representing urban form and socio-economics were included in these models as well as a dummy variable to indicate day type; that is, whether or not the day on which travel was recorded was a work day. Over and above the simple

3 Travel and Mobility

	Parameter estimates Total trips	Car trips	Car trips – Single occupancy
Male	-0.07	-0.10	0.24**
Female (Reference)			
Age 16–24 years	0.07	-0.24*	-0.64**
Age 25–34 years	0.03	0.04	-0.19
Age 35–44 years (Reference)			
Age 45–59 years	0.05	0.17*	0.18
Age 60–75 years	0.10	-0.04	-0.16
Age 75 years plus	-0.31*	-0.17	-0.66*
Income £0–£10 k	-0.28**	-0.47**	-0.20
Income £10–£20 k	-0.09	-0.16*	-0.06
Income £20–£30 k (Reference)			
Income £40 k plus	0.15*	-0.07	-0.12
Cars available $= 0$	-0.37**	-2.07**	-4.65**
Cars available $= 1$ (Reference)			
Cars available $= 2$	0.25**	0.68**	1.07**
Cars available $= 3$ plus	0.13	0.49**	1.00^{**}
Small adult	-0.24**	-0.32**	-1.00^{**}
Single parent	0.22*	0.15	-0.57**
Small family	-0.18*	-0.13	-1.06**
Large family	-0.27**	-0.25*	-1.18**
Large adult	-0.34**	-0.57**	-1.37**
Older smaller	-0.23*	-0.18	-0.90**
Single pensioner	0.05	0.22	0.51*
Single adult (Reference)	0100	0.22	0101
Non-workday	-2 26**	-1.00**	-1 18**
Workday (Reference)		1100	1110
Edinburgh resident	0.30**	0.11*	0.04
Glasgow resident (Reference)	0.00	0.11	0.01
No access to high quality bus	-0.82**	0.17**	0.20**
Access to high quality bus service			
(Reference)			
Employed or full time study	-0.01	-0.09	0.50
Not employed or full time study (Reference)	-0.70**		
Population density (persons per hectare) \blacklozenge	-0.01	-0.06**	-0.05**
Retail employment to population	0.80**	-0.95**	-0.93
Distance to centre (kilometres)		0.02	-0.01
Non-workday * Employed or full		0.10	-0.55
time study			
Non-workday * Not employed or			
full time study (Reference)			
Workday * Employed or full time study (Reference)			

 Table 3.3
 Ordered logit model for trip frequency

	Parameter estimates Total trips	Car trips	Car trips – Single occupancy
Workday * Not employed or full time study (Reference)			
Nonworkday * Population density $(\times 1000)$	-0.04**	-	-
Workday * Population density (Reference)		-	-
Number of observations	9657	8254	8254
Log likelihood at 0	26115.30	17211.64	10737.64
Log likelihood at convergence	23814.80	14560.42	7803.55
McFadden pseudo R-square	0.09	0.15	0.27

 Table 3.3 (continued)

* indicates statistically significant at the 5% level

** statistically significant at the 1% level

additive effects of each factor on the number of trips, the question as to whether socio-economic influences and/or day type moderate the effect of urban form factors was also explored by including interaction terms in the models. For example, it might be the case that the influence of density on trip frequency varies according to household income. A number of interaction terms were examined, but the only one which was found to be statistically significant was the interaction between day type and population density for the model of the total number of trips.

The overall model fit for total number of trips is quite low (as indicated by a McFadden pseudo R-squared value of less than 0.1) which suggests that trip frequency is not influenced to any great extent by the explanatory variables in the model. Nonetheless, the parameter estimate for the ratio of retail jobs to residential population is negative and statistically significant indicating that the total number of trips decreases as land use mix increases. On work days population density and distance from home to city centre are not found to be statistically significant. On the other hand, the total number of trips is found to decrease with increasing population density on non work days. This suggests that the residential neighbourhood context exerts a greater influence on trip making when the effect of travel to work is absent and is consistent with the finding of Næss (2006).

With regard to car trips only population density and the ratio of retail employment to population are both found to be statistically significant predictors of the number of such trips and the number of single occupant trips. In both cases the models suggest that the number of trips decreases with increasing values of these attributes of urban form. Home to city centre distance is not found to be statistically significant in either model.

Household income is found to have a statistically significant effect on trip-making behaviour. Households with an income of less than £10,000 are found to make fewer trips overall and trips by car than households with an income of between £20,000 and £30,000. High income households (more than £40,000) are
found to make more trips overall but no more car trips than households with an income between $\pounds 20,000$ and $\pounds 30,000$. This suggests that the number of car trips increases with income up to a certain point after which increasing income has no discernible effect.

The effect of the number of available cars on trip making is also investigated. It will be recalled from earlier that population density, land use mix and distance to city centre are found to be associated with the number of cars available to a household. It can also be seen from Table 3.3 that the parameter estimates for two or more cars is positive and statistically significant. This indicates that, relative to a household with only one car, the number of trips is higher for those living in households with two or more cars. Overall, this analysis highlights both the direct effects of urban form on trip frequency and also the indirect effect that elements of urban form have on trip making through car availability.

Use of Local Amenities

A complementary perspective on local trip-making can be obtained by examining the use of neighbourhood amenities. More specifically, the distance between home and a particular class of amenity can be thought of as a proxy for mixed use development. This section examines whether distance to an amenity and any constraints placed by work influence the frequency of use of that amenity.

The data used in this analysis is drawn from three sources. First, the household survey of case study neighbourhoods collected information on respondents' use of specified local amenities. Permissible responses are (1) most days, (2) at least once a week, (3) at least once a month, (4) occasionally, (5) don't use. Second, the locations of the following classes of amenity are mapped within a Geographical Information System (GIS): post office, library, bars, restaurants/cafes and convenience store. Third, the road network distance from each respondent's home address to specified categories of services and facilities is determined using GIS. This enables the distance to the nearest amenity in each class and the number of amenities in a class which were located within 500 m of each respondent's home to be determined.

Respondents are further categorised according to work status and those who are categorised as full-time workers, retired and looking after family/home were selected for analysis. This enabled the effect of employment on the frequency of use of local amenities to be tested. Full-time workers are sub-divided into those working locally (defined as those working within the same postcode sector as their place of residence) and those not.

The frequency of use of each class of amenity was estimated using an ordered logit model and the results are presented in Table 3.4. Positive parameter estimates indicate an increase in the likelihood of the "don't use" response whilst negative parameter estimates indicate an increase in the "most days" response. The distance to the nearest amenity is found to be a statistically significant predictor of the frequency of use of post office, library and convenience store after controlling for socio-economic and demographic factors. The parameter estimates are positive

		Post office	Restaurant/ café	Bar/pub	Library	Convenience store
Access	Distance to centre (km)	-0.03	0.03	-0.001	-0.03	0.05
	Nearest Service/Facility (km)	0.76**	-	-	0.58**	2.05**
	Number of Service/Facility	-	-0.02**	-0.05**	-	-
	Population Density	-0.39	1.15	0.07	0.63	-2.00*
	Min Distance to Supermarket (km)	-	-	_	-	-1.08**
City	Edinburgh	0.22	-0.01	0.31	-0.32	0.15
City	Glasgow	0.07	-0.67**	0.15	-0.41^{*}	(Reference)
	Leicester	_0.07	_0.34	_0.68**	_0.82**	(Reference)
	Oxford	-0.10	-0.34	-0.08	-0.82	_
	Shaffiald (Pafaranca)	0.00	-0.02	-0.18	-0.32	_
Gandar	Male	0.04	0.004	0 56**	0.06	0.20
Genuer	Eamala (Deference)	-0.04	-0.004	-0.50	0.00	-0.20
1	16 24 years	0.06**	1 52**	0 51**	0.26	0.74
Age	10–24 years	0.90	-1.35	-2.34	0.50	-0.74
	25-34 years	1.11	-1.20	-1.0/	0.30	-0.27
	55–44 years	1.01	-1.24	-1.54	0.04	-0.65
	45–54 years	0.79**	-0.97**	-1.34	-0.05	-0.37
	55–64 years	0.40*	-0.64	-1.11	0.06	-0.38
	65 years + (Reference)	1.00**	0.00	0.0.1**	0.50*	
Status	Employed full-time – work locally	1.02**	-0.33	-0.84**	0.72*	-0.29
	Employed full-time – not work locally	1.19**	-0.17	-0.80**	1.02**	0.06
	Retired	0.96**	-0.15	-0.81^{**}	0.62*	-0.32
	Looking after family/home (Reference)					
Income	<£10 k	-0.51*	1.45**	0.13	-0.12	-0.50
	£10–£20 k	-0.19	1.19**	0.05	-0.27	0.39
	£20–£30 k	-0.13	1.01**	-0.04	-0.28	-0.04
	£30–£40 k	-0.13	0.59**	-0.03	-0.25	-0.16
	£50–£80 k >£80 k (Reference)	-0.19	0.31	-0.28	-0.20	-0.12
Cars	No cars	0.08	-0.11	0.09	-0.24	-0.42
	1 car	0.11	-0.27	0.19	-0.08	-0.13
	2 or more cars (Reference)					
	Number of observations	2049	1642	1675	1856	858
	Log likelihood at 0	5434.52	4733.09	4757.46	3957.90	2387.32
	Log likelihood at convergence	5184.88	4292.37	4295.52	3834.95	2288.91
	McFadden pseudo R-Square	0.05	0.09	0.10	0.03	0.04

 Table 3.4
 Frequency of use of various classes of local amenity

* indicates statistically significant at the 5% level ** statistically significant at the 1% level

which indicate that frequency of use decreases as proximity to the specific local service/facility increases. For convenience shops, the effect of competition with supermarkets is also revealed – the greater the distance to the nearest supermarket the more frequent the use of a local convenience shop. Likewise, the number of restaurants/cafes and pubs within 500 m of a respondent's home are statistically significant predictors of frequency of use. The parameter estimates are negative indicating that an increase in the provision of this amenity class is associated with an increase in the frequency of use.

Distance from home to the city centre is found to be a statistically significant predictor of the frequency of use of local restaurants and pubs but not for the other categories of services/facilities considered. The analysis suggests that the further a person lives from the city centre, the less frequent his or her use will be of local restaurants and pubs. This result probably reflects the preponderance of opportunities for eating and drinking within city centres in comparison with other locations within the city and the fact that a threshold of 500 m was used to determine the number of local eating and drinking opportunities. For residents located in or near to city centres there are likely to be a significant number of eating and drinking opportunities which are located just beyond this threshold. Since this effect on frequency of use is not captured in the model by the local access variable it may be captured instead by the distance to the city centre variable.

To test the hypothesis that respondents with formal commitments outside their residential neighbourhood make less frequent use of amenities within their neighbourhood, respondents were divided into four work status categories as described above. The reference category is those looking after family and home. Relative to this group, employed respondents were estimated to use the local post office and library less frequently. The magnitude of the estimated parameters also suggests that those employed locally use local amenities more frequently than those employed outside the local area. No statistically significant differences are found between the employment categories for the frequency of use of restaurants or convenience stores. On the other hand, the frequency of use of local pubs is found to be higher for employed and retired respondents in comparison with those looking after family and home. The link between employment location and the use of local amenity is again evident on inspection of the magnitude of the parameter estimates for use of pubs – those working locally are more likely to use a local pub than those working outside the local area.

Trip Generation: Summary

The principal conclusion that can be reached from the analysis of trip frequency is that increasing land use mix and density reduces the frequency of car trips and single occupancy car trips. No difference was detected between work days and nonwork days. Increasing car availability was also found to increase trip frequency by car which, recalling the earlier results, indicates an indirect pathway linking urban form and trip frequency.

It will be remembered that only car trips which were over a quarter of a mile in length were recorded in the SHS Travel Diary and so it is possible that car trips undertaken by residents in neighbourhoods which have good local access to people and amenities (that is, high density, mixed-use neighbourhoods) are underestimated. However, it does not seem likely that the number of car trips that fall below this threshold would be sufficiently high to have a large influence on the results.

The frequency of use of a number of key local amenities is found to decline with distance from home. Economic status and employment location are also found to influence the frequency of use of some amenities which is consistent with a constraints-based accessibility perspective. A key limitation of this work is the extent to which frequency of use is correlated with local access. It is possible that demand is uniform across a neighbourhood, all other things being equal, and those with lower levels of access organise their use of amenities more efficiently than those living in close proximity to local services and thus undertake fewer trips to satisfy demand. This may provide an explanation for the use of essential services such as post offices, but does not satisfactorily explain the use of discretionary amenities such as restaurants and bars. Assuming uniform demand, it may also be the case that those living further from local services are more likely to use nonlocal services, such as those located close to the workplace. Even if demand is nonuniform and is influenced by proximity/intensity, the direction of causality may not be straightforward. It is possible that those with a propensity to use certain types of local services (such as bars and restaurants) may choose to live in locations with good access to these services.

Trip frequency is only one measure of travel behaviour and reflects the number of spatially separated activities carried out in the course of a day. It does not however capture the total amount of travel undertaken which has particular relevance to the question of the efficiency of resource use in transportation and urban form. This question is examined in the following section.

Distance Travelled

The total amount of travel undertaken is an appropriate measure to assess the overall effects of urban form on travel behaviour. It encompasses any variations in the pattern of trips which may exist between residents of different neighbourhood urban form typologies. This includes substitution effects whereby travel time savings made when undertaking an activity close to home may be used for additional travel. It also takes into account all travel that may be associated with specific activities such as work which are fixed in time and space. Previous work has noted that population density is negatively associated with travel distance (Stead, 2001) and positively associated with distance from home to the city centre (Næss, 2006: 104). Travel distance to the city centre (Næss, 2006).

3 Travel and Mobility

The analysis presented in this section commences with an examination of the effect of urban form on the distance travelled by all modes and the distance travelled by car using the SHS. The Travel Diary records an origin and a destination for each recorded trip. The network distance travelled for each trip was estimated using GIS and then aggregated for each respondent living within the cities of Glasgow and Edinburgh to give two measures of travel behaviour – total distance travelled by all modes and total distance travelled by car. As before, respondents for whom no travel was reported were included in the analysis. The areal unit of analysis is the Intermediate Geography zone.

Multiple linear regression analyses were undertaken in which total distance travelled and total car distance travelled were regressed on to number of explanatory variables reflecting urban form, socio-economic characteristics and day type (that is, work day or non work day). Interaction terms between day type and the urban form variables are tested, but only the interaction term between day type and distance to city centre is found to be statistically significant in both models. This suggests that, with this exception, the influence of urban form on distance travelled is the same for work and non-work days.

To avoid endogeneity bias in the estimation of distance travelled, a two-stage estimation procedure was used (Train, 1986). The expected number of cars available to a household was first estimated from the SHS for each respondent using the car ownership model described above. The estimated number of cars available to each household was then included as a predictor of distance travelled.

With reference to Table 3.5, total distance travelled decreases with increasing population density and increasing ratio of retail employment to population. Proximity to a high quality bus route is found to reduce the amount of car travel whereas household income is not shown to have a statistically significant effect. For work days, distance between home and the city centre is positively associated with total distance travelled. However, non-work days reduce the effect of this factor; that is, the location of home relative to the city centre is found to have a greater effect on total distance travelled on work days than it has on non-work days. This is consistent with the work of Næss (2006) who found the effect of distance between home and city centre on total distance travelled to be lower at the weekend than on weekdays.

A similar picture emerges for total distance travelled by car. The principal difference is that the ratio of retail employment to population is not statistically significant. This suggests that, all other things being equal, living in a mixed use neighbourhood does not reduce the amount of travel undertaken by car. Also, no difference is detected in the amount of travel undertaken on work days and non-work days.

Both total travel distance and travel distance by car are found to increase with increasing car availability, which highlights the indirect influence that urban form has on these measures of travel behaviour. Maps showing the model results applied to Glasgow and Edinburgh for workdays and non-workdays are shown in Figures 3.1, 3.2, 3.3, and 3.4. It can be seen that distance travelled by car tends to increase with distance from the centres of the two cities reflecting the predominately

	Parameter estimates		
	Distance Travelled Total	Distance Travelled by Car	
Constant	2.00**	0.47*	
Population density (persons per hectare) \blacklozenge	-0.02^{**}	-0.03**	
Distance to centre (km)	0.05**	0.08^{**}	
Retail jobs to population ratio \blacklozenge	-0.39	-0.22	
City – Edinburgh	0.12**	0.06	
High Quality Bus Route	-0.09^{*}	-0.07^{*}	
Car ownership	0.35**	0.87**	
Gender – female	-0.02	-0.04	
Age – 16–24	0.02	-0.19^{**}	
Age – 25–34	0.02	-0.02	
Age - 45-59	-0.07	0.01	
Age - 60-74	0.05	0.04	
Age – 75 plus	-0.15	-0.09	
Income – £0–£10 k	-0.10	-0.10	
Income – $\pounds 10$ – $\pounds 20 \text{ k}$	-0.02	-0.03	
Income £30 k plus	0.05	-0.07	
Household – small adult	-0.09	-0.12	
Household – single parent	-0.05	0.07	
Household – small family	-0.13	-0.11	
Household – large family	-0.26**	-0.24^{*}	
Household – large adult	-0.38**	-0.52^{**}	
Household – older smaller	-0.17	-0.05	
Household – single pensioner	0.04	0.12	
Employed	0.30	0.24	
Day type – non-work	-0.86^{**}	-0.15	
Non-work * Employed	-0.24	-0.01	
Non-work * Distance to centre	-0.03	-0.07^{**}	
Number of observations	5240	5240	
Adjusted R-square	0.213	0.184	

 Table 3.5
 Multiple regression models of distance travelled

* indicates statistically significant at the 5% level

** statistically significant at the 1% level

mono-centric structure of both cities and the decline in population density from the middle to the outer city. From a statistical perspective, this also serves to highlight the correlation between these two measures of urban form and the difficulty in completely separating the effects of density and location in an observational study of this kind.

Further evidence of the influence of urban form on car use is available from the longitudinal data set collected amongst home-movers. Respondents were asked to compare their current total amount of car travel as a driver and passenger with their car use before moving home. Permissible responses were (a) a lot less, (b) a little less, (c) about the same, (d) a little more and (e) a lot more. The determinants of the change in car use is estimated using an ordered logit model in which changes

3 Travel and Mobility



Fig. 3.1 Distance travelled by car in Glasgow (work day)



Fig. 3.2 Distance travelled by car in Glasgow (non-work day)



Fig. 3.3 Distance travelled by car in Edinburgh (work day)



Fig. 3.4 Distance travelled by car in Edinburgh (non-work day)

in population density, the ratio of retail employment to population and distance to the city centre are included as explanatory variables. The results are presented in Table 3.6 based on datazones. It can be seen that a change in population density brought about by moving home is negatively associated with a change in car use. The other urban form attributes are not found to be statistically significant, although this may be partly attributable to the relatively small sample size used in this analysis.

It should also be noted that a change in the number of cars available to a household after moving home is found to be positively associated with the amount of car travel. This specific result should be read in conjunction with the longitudinal model of household car availability described above in which changes in population density and retail employment: population mix following household re-location are found to be negatively associated with car availability. This highlights the indirect effect that urban form has on car travel through car availability which can

	Parameter estimate
Change in population density (persons per hectare) \times 1000	-0.07**
Change in distance to work (kilometres)	0.02
Change in distance to city centre (kilometres) \times 1000	0.05
Change in retail employment to population ratio	-0.77
Change in available cars	0.72*
Age 16–24 years	0.77
Age 25–34 years	1.02
Age 35–44 years	-0.13
Age 45–54 years	-0.09
Age 55–64 years	-0.15
Age 65 years plus (Reference)	
Male	0.59
Female (Reference)	
Change in income – a lot less	0.28
Change in income – a little less	0.45
Change in income – about the same	-0.25
Change in income – a little more	0.87
Change in income – a lot more (Reference)	
Attitudinal cluster 1 – non-car driver	-0.54
Attitudinal cluster 2 – non-car driver	-1.30
Attitudinal cluster 3 – car driver	-0.70
Attitudinal cluster 4 – car driver	0.01
Attitudinal cluster 5 – car driver	0.01
Attitudinal cluster 6 – car driver	
Number of observations	96
Log likelihood at 0	286.7
Log likelihood at convergence	244.2
McFadden pseudo R-square	0.15

 Table 3.6
 Ordered logit model for change in car travel

* indicates statistically significant at the 5% level

** statistically significant at the 1% level

be regarded as additional to the direct effect of urban form factors on car travel discussed in the previous paragraph.

The extent to which residential self-selection explains change in car travel is also explored. The travel questionnaire asked respondents the degree to which they agreed with a number of statements on car travel. Cluster analysis was undertaken and respondents were then assigned to one of six groups reflecting attitude towards car travel. Four of these groups related to car drivers and two related to those who either did not have access to a car or did not have a driving licence. Dummy variables are included in the model to represent group membership. Following Handy et al. (2005), evidence of self selection would be detected if the attitudinal dummy variables that are statistically significant predictors of change in car travel remain constant over the period of residential re-location. With reference to Table 3.6 it can be seen that none of the attitudinal dummy variables is statistically significant. This suggests that self-selection of residential location on the basis of travel preferences is not a major influence on car travel. Several caveats must however be added. First, the sample is relatively small. Second, the sample of home-movers is itself selfselected and hence subject to potential bias. Finally, in many households, the choice of residential location is a joint decision whereas the dataset analysed here uses attitudinal data relating to only a single respondent from each household.

In summary, this section has shown that neighbourhoods corresponding to traditional urban form typologies are associated with lower levels of distance travelled by all modes and distance travelled by car. Urban form is shown to influence distance travelled directly and also indirectly through household car availability. The possibility that this association between urban form and car travel arises from households choosing neighbourhoods that support inherent travel preferences is also explored but no evidence is established in support of this premise.

Conclusions

This chapter has sought to examine the extent to which urban form, in particular the residential neighbourhood context, influences certain key aspects of travel-related behaviour. In recent years arguments have been advanced in support of traditional urban forms, partly on the basis that they will induce more sustainable travel patterns. Three measures of travel-related behaviour are explored, namely car availability, trip generation and distance travelled, and the effects of population density, functional mix and distance from the city centre on these measures are tested using multivariate statistical models which allows socio-economic effects to be controlled.

From the work presented in this chapter several conclusions can be drawn that are reinforced by the broad agreement between the results of the cross-sectional analyses of car ownership and car use and the outcomes of elective urban form change brought about by moving home. First, traditional urban forms are associated with lower levels of car availability which in turn are associated with lower trip

3 Travel and Mobility

frequencies and shorter travel distances. The indirect influence of urban form on travel behaviour through the intermediate variable of car availability is in addition to the direct effect of urban form on travel behaviour. Second, traditional urban forms are associated with lower car trip frequencies, total distance travelled and distance travelled by car. There are a number of reasons why this may be the case. Areas of higher population density areas are able to support a greater number and variety of local services and facilities and have good public transport connectivity which leads to a lower household demand for access to cars. In the highest density areas, limited parking supply and regulatory control can also be expected to play a role in reducing car demand. Third, household income has a strong influence on car ownership and also influences trip making behaviour but has no measurable effect on distance travelled.

References

- Badoe, D.A. and Miller, E.J. (2000) Transportation-Land-Use Interaction: Empirical Findings in North America and their Implications for Modeling. *Transportation Research Part D*, 5, pp. 235–263.
- Banister, D. (2005). Unsustainable Transport: City Transport on the New Century, Routledge, London.
- Bhat, C., and Gio, J.Y. (2007) A Comprehensive Analysis of Built Environment Characteristics on Household Residential Choice and Auto Ownership Levels, *Transportation Research Part B*, 41, pp. 506–526.
- Boarnet, M.G. and Crane, R. (2001) *Travel by Design: The Influence of Urban Form on Travel*, Oxford University Press, Oxford.
- Boarnet, M.G and Sarmiento, S. (1998) Can Land-use Policy Really Affect Travel Behaviour? A Study on the Link between Non-work Travel and Land-use Characteristics. *Urban Studies*, **35**(7), pp. 115–1169.
- Borooah, V. K. (2002) Logit and Ordered: Ordered and Multinomial Models, Sage Publications, London.
- Chu, Y-L (2002) Automobile Ownership Analysis Using Ordered Probit Models. *Transportation Research Record: Journal of the Transportation Research Board*. Issue Number 1805, pp. 60–67.
- Crane, R. (2000) The Influence of Urban Form on Travel: An Interpretive Review. *Journal of Planning Literature*, **15**(1), pp. 3–23.
- Department of the Environment, Transport and the Regions (DETR) (1999) *Towards an Urban Renaissance. Final Report of the Urban Task Force*, E and FN Spon, London.
- Department for Transport (DfT) (2002) Revised National Travel Survey Data for Urban and Rural Areas, DfT, London.
- Department for Transport (DfT) (2007) National Travel Survey: 2006, DfT, London.
- Duany, A., Plater-Zyberk, E. and Speck, J. (2001) Suburban Nation: The Rise of Sprawl and the Decline of the American Dream, North Point Press, New York.
- European Commission (2006) Commission Staff Working Document, Annex to the Communication from the Commission to the Council and the European Parliament on the Thematic Strategy on the Urban Environment: EIA, Com(2005) 718 Final, Commission of the European Communities, Brussels.
- European Commission (2007a) Sustainable Urban Transport Plans: Preparatory Document in Relation to the Follow-up of the Thematic Strategy on the Urban Environment, Commission of the European Communities, Brussels.
- European Commission (2007b) *Panorama of Transport, 2007 Edition*, Eurostat Statistical Books, Commission of the European Communities, Brussels.

- European Environment Agency (2006) Urban Sprawl in Europe: the Ignored Challenge. EEA Report 10/2006, European Environment Agency, Copenhagen.
- European Environment Agency (2007) *The Pan-European Environment Glimpses into an Uncertain Future.* EEA Report 4/2007, European Environment Agency, Copenhagen.
- European Environment Agency (2008) *Climate for a Transport Change. TERM 2007: Indicators Tracking Transport and Environment in European Union.* EEA Report 1/2008, European Environment Agency, Copenhagen.
- Ewing, R. and Cervero, R. (2001) Travel and the Built Environment: A Synthesis. Transportation Research Record: Journal of the Transportation Research Board. Issue Number 1780 pp. 87– 114.
- Finkel, S. E. (1995) Causal Analysis with Panel Data. Sage Publications, London.
- Handy, S., Cao, X, and Mokhtarian, P (2005) Correlation or Causality between the Built Environment and Travel Behaviour: Evidence from Northern California. *Transportation Research Part D*, pp. 427–444.
- Hall, P. (2000) Urban Renaissance/New Urbanism: Two Sides of the Same Coin? *Journal of the American Planning Association*, **66**(4), pp. 359–360.
- Hess, D.B and Ong, P.M. (2002) Traditional Neighbourhoods and Automobile Ownership. *Transportation Research Record: The Journal of the Transportation Research Board*, Issue Number 1805, pp. 35–44.
- Hope, S. (2007) Scotland's People: Scottish Household Survey Methodology 2006–06. Scottish Executive.
- Krizek, K.J., 2003. Residential relocation and changes in urban travel Does neighborhood-scale urban form matter? *Journal of The American Planning Association* 69(3), pp. 265–281.
- Kwan, M-P. (1998) Space-Time and Integral Measures of Accessibility. *Geographical Analysis*, 30, pp. 191–216.
- Næss, P. (2006) Accessibility, activity Participation and Location of Activities: Exploring the Links between Residential Location and Travel Behaviour. *Urban Studies*, 43(3), pp. 627–652.
- Miller, H.J. and Shaw, S-H. (2001) Geographic Information Systems for Transportation: Principles and Applications, Oxford University Press, Oxford.
- Panerai, P., Castex, J., Depaule, J.C. and Samuels, I. (2004) Urban Forms: The Death and Life of the Urban Block. Architectural Press, Oxford.
- Pirie, G. H. (1979) Measuring Accessibility: A Review and a Proposal. *Environment and Planning A*, **11**(3), pp. 299–312.
- Scottish Government (2005) Scottish Neighbourhood Statistics: Intermediate Geography Background Information. http://www.scotland.gov.uk/Publications/2005/02/20732/53086 Web publication date: February 18th 2005, modified 7th April 2006.
- Shay, E. and Khattak, A.J. (2005) Automobile Ownership and Use in Neotraditional and Conventional Neighbourhoods. *Transportation Research Record: Journal of the Transportation Research Board*, Issue Number 1902, pp. 18–25.
- Stead, D. (1999) Planning for Less Travel: Identifying Land Use Characteristics Associated with More Sustainable Travel Patterns, unpublished PhD thesis, Bartlett School of Planning, University College London, London.
- Stead, D. (2001) Relationships between Land Use, Socio-economic Factors and Travel Patterns in Britain. *Environment and Planning B*, 28(4), pp. 499–528.
- Stead, D. Williams, J. and Titheridge, H. (2000) Land Use, Transport and People, Identifying the Connections. In *Achieving Sustainable Urban Form* (eds. K. Williams, E. Burton and M. Jenks). E and FN Spon, London.
- Thompson-Fawcett, M (2000) The Contribution of Urban Villages to Sustainable Development, In *Achieving Sustainable Urban Form* (eds. K. Williams, E. Burton and M. Jenks). E and FN Spon, London.
- Train, K. (1986) Qualitative Choice Analysis. The MIT Press, Cambridge, MA.

Chapter 4 Environment and Biodiversity

Richard A. Fuller, Jamie Tratalos, Philip H. Warren, Richard G. Davies, Aleksandra Pępkowska and Kevin J. Gaston

Introduction

Research over many decades has documented numerous environmental effects of urbanization, ranging from the loss and reconfiguration of green space to dramatic changes in ecosystems and biodiversity. Rather less is known about how urban form, in particular the density of urban development, alters environmental patterns and processes *within* cities. Investigation of the relationships between urban form and environmental structure and performance is an important issue in the urban sustainability debate and here we use that work to illustrate some of the key ideas in this newly emerging field. After outlining the general effects of urbanization on environment and biodiversity, we then consider in turn the relationships between urban form and patterns of green space, the degree to which urban environments can provide useful ecosystem services to human populations, and finally the responses of biodiversity to urban development.

Urbanization transforms the ecology of an area. Such transformation can include: (i) the alteration of habitat, such as the loss and fragmentation of natural vegetation, and the creation of novel habitat types (Niemelä, 1999; Pickett et al., 2001; McKinney, 2002; Johnson and Klemens, 2005); (ii) the alteration of ecosystem services (e.g. air, water and climate regulation, pollination), and other resource flows, including reduction in net primary production, increase in regional temperature, and degradation of air and water quality (Henry and Dicks, 1987; Rebele, 1994); (iii) the alteration of disturbance regimes, typically an increase in disturbance frequencies (Rebele, 1994); and (iv) disruption of species occurrence and abundance patterns, commonly including the local extinction of many species that are habitat specialists, require large habitat patches, utilise the interiors rather than the edges of patches, or are associated with complex vegetation structures (Pickett et al., 2001; McKinney, 2002; Chace and Walsh, 2006). The extent and intensity of these effects depend largely on the extent, composition and management

R.A. Fuller (⊠)

The Ecology Centre, University of Queensland, Brisbane, QLD, Australia

of green spaces in urban areas. However, both variation in these characteristics and the details of their interactions particularly with ecosystem services and biodiversity remain rather poorly understood. This is perhaps surprising, given that both the characteristics of green space and its ecological correlates also bear on a number of other important issues. In particular, the extent, composition and management of green space has been shown to have significant effects on urban economies, through a diverse array of impacts, including on house prices, the costs of heating and cooling buildings, and the ease of attracting businesses and employees to areas (e.g. CABE Space, 2004), and on human physical, mental and social well-being (Dunnett and Quasim, 2000; Kuo and Sullivan, 2001; de Vries et al., 2003; Fuller et al., 2007a).

A useful step in developing an improved understanding of the structure of green space and some of the benefits that it brings within urban areas is to determine how these depend on urban form (the structure of the built environment). At the most simplistic level, the structure of urban green space might be viewed as the converse of urban form. Indeed, both measures of green space and of the built environment have been identified as key environmental indicators in urban areas (e.g. Pauleit and Duhme, 2000; Whitford et al., 2001). However, for a given ratio of green space to coverage by impervious surface, the nature and density of the built form can vary widely, such that there is a great deal of scatter in the relationship between measures of green space cover and urban density (Tratalos et al., 2007a). Further, the spatial configuration of habitat patches, rather than simply the degree of coverage by green space, can be important in determining biological processes for certain groups of organisms and types of ecosystem function (Bastin and Thomas, 1999). Links between coverage by green space and the nature of urban form have not been well explored (Pickett and Cadenasso, 2006).

There is considerable variation in the structure of cities, with local and regional factors heavily influencing urban form. For example, Sheffield, one of the case study cities, has a long history as a market town with a distinctly urban character (Hey, 2005). The population grew from 10,000 inhabitants in 1736-83,000 in 1851, and 90,000 by 1901. In the second half of the nineteenth century steel manufacture became the major industry and remained so for nearly one hundred years. Access to water for this industry dictated much of the pattern of early urbanization, and prevailing winds from the west meant that affluent residential areas were typically sited in the western suburbs with cleaner air, leading to a westward sprawl of the city boundary and increased population growth (currently 513,000). During the 1970s and 1980s, manufacturing industry began to shrink drastically, the economy diversified, and by the mid-1990s two-thirds of jobs were in the service sector. Substantial areas were redeveloped under regeneration programmes that replaced many industrial sites with housing or office blocks (Hey, 2005). This history has, in part, shaped the configuration of urban Sheffield today, with the most heavily developed areas in the river valleys, forming a y-shaped pattern. Large green spaces are restricted to the zones of intermediate and low levels of urban development, and a band of high density housing is noticeable between the centre and the outer suburbs (Fig. 4.1). In the context of the environment and biodiversity, urban form is perhaps best measured in terms of the density of various elements of urbanization,



Fig. 4.1 Satellite image of Sheffield. Image of Sheffield acquired June 2005, 17° off nadir angle, cloud cover 1%. The area shown is all Ordnance Survey 1 km² national grid squares, inside the administrative boundary which comprise 10% or more impervious surface (buildings, roads or other sealed surface)

the patterns of coverage of different land use types, and the degree to which different patches of land cover are connected to each other. Insofar as the road network alters the configuration of green spaces by dividing them into smaller fragments, and might form a barrier (or a conduit) for dispersal of animals and plants, the physical structure of the road network is an important component of urban form as it impacts environmental performance. Although they may remain of interest, the effectiveness or efficiency of other measures of connectivity, as reflected for example in social cohesion and the ease with which people can move between areas using transport networks, are of less direct relevance, because they do not bear directly on the amount and configuration of patches of usable habitat.

In this chapter, we review aspects of the relationships between urban form and green space extent, ecosystem service provision and biodiversity. In so doing, we use examples drawn principally, although not exclusively from empirical studies of these issues conducted within the five case study cities.

Green Space and Urban Form

Broadly defined to include most soft rather than hard or impervious surfaces, local government statistics often indicate that green space covers a substantial proportion of most cities in the UK. However, values based on the administrative boundaries of cities are difficult to interpret as these often extend well beyond the actual limits of urbanization and may include substantial rural areas. For example, urban areas form only about one-third of the area within Sheffield's administrative boundary, while another third is agriculture, and the remaining area forms part of the Peak District National Park (Beer, 2005). To generate more comparable data, we define the urban area simply as all the 1 km^2 grid squares inside the administrative boundary that comprise 10% or more sealed surface (buildings, roads or other human-made surface). Within this urban area, green space covers 60–70% if gardens are included, and 30–45% if gardens are excluded (Table 4.1). The data in Table 4.1 make two interesting points. First, the figures are remarkably consistent across the five study cities, despite their very different histories and topographies. Secondly, the figures are perhaps surprisingly high; a substantial majority of the surface within urban areas comprises green space, if gardens are included (see also Pauleit and Duhme, 2000). In this context it is important to note that these analysis are based on high resolution vector mapping, which includes even very small areas of green space such as road verges and amenity plantings. Given that small patches of green space make a disproportionately large contribution to overall levels of urban green space (see below), analyses of green space coverage that use mapping at a coarser scale might substantially underestimate its extent. In addition, these figures treat gardens as purely green space, whereas a proportion of the coverage of most gardens comprises impervious surfaces (e.g. in Sheffield, about one-third of garden area comprises impervious surfaces such as paths, patios, etc.; Tratalos et al., 2007a).

City	Total urban area (km ²)	Total green space (km ²)	Gardens (km ²)	Non garden green space (km ²)	% coverage by green space incl. gardens	% coverage by green space excl. gardens
Edinburgh	124.02	80.75	27.63	53.12	65.1	42.8
Glasgow	197.60	120.27	32.47	87.80	60.9	44.4
Leicester	70.09	41.84	18.88	22.96	59.7	32.8
Oxford	37.28	25.06	8.18	16.88	67.2	45.3
Sheffield	158.93	104.61	39.56	65.05	65.8	40.9

 Table 4.1
 Total Green Space Coverage within the Five Case Study Cities

Green space is defined as any land parcel classified as "natural surface" by Ordnance Survey's MasterMap dataset (Ordnance Survey 2006), while gardens are those parcels classified as "multiple". The urban area is defined as that area inside the city's administrative boundary intersecting Ordnance Survey 1 km² national grid squares with 10% or more of their area comprising impervious surfaces (buildings, roads or human made surface).

4 Environment and Biodiversity

A detailed grid-based analysis of green space across the five case study cities (Fig. 4.2) reveals that in each case coverage by green space is highly heterogeneous, with a long left tail of areas with very low levels of green space coverage, and a significant number of cells with almost complete coverage, representing large parks, well-wooded remnants and encapsulated patches of undeveloped land. The general pattern is remarkably consistent across the five cities, with the peak in proportional coverage generally occurring around 0.6. The frequency of cells with very high coverage varies markedly though, with Edinburgh and Oxford showing distinct peaks, indicating large patches of internal green space within those cities,



Fig. 4.2 Proportions of green space across case study cities at 250×250 m cell resolution (a) Edinburgh, (b) Glasgow, (c) Leicester, (d) Oxford and (e) Sheffield. Green Space is vegetated surface including domestic gardens



and Glasgow and Leicester showing a much less pronounced peak indicating that few large patches of green space exist within the boundaries of those cities.

A high proportion of urban green space in UK cities is composed of small patches, such as domestic gardens and roadside verges, rather than large patches, such as public parks and playing fields. For example, there are 326,147 separate parcels of green space within the Sheffield urban area, of which 93% are front or rear garden. When arranged in rank order of increasing area, the contribution to overall cumulative area of green space decelerates strongly as parcel size increases, a pattern that persists even when gardens are excluded (Fig. 4.3). Across urban Sheffield, 50% of green space comprises the parcels with an area less than 0.59 ha. Excluding gardens, the equivalent figure is 9.9 ha. This is significant, as much strategic planning of urban green space is focussed on the large green spaces, and ignores the smaller ones. In practice, however, the vast network of small patches may make substantial contributions to the net environmental benefits provided by urban green space, and may be vital to maintaining the contributions of some of the larger patches (e.g. through influences on overall habitat availability and on the degree of connectivity among larger patches). Without the smaller patches the larger ones would genuinely resemble the isolated habitat islands that strategic planning often portrays them to be.

The size of a patch of green space is an important variable in ecological terms, because larger patches tend to support more species and individuals, buffer those populations from local extinctions, and allow the organisms within them to disperse across the landscape (Bastin and Thomas, 1999). As well as overall green space, average patch size tends to decline with increasing building/housing density, with some evidence that the relationship is markedly non-linear, being steeper at lower densities (Tratalos et al., 2007a). This effect may be exacerbated by the larger numbers of land parcels in areas of higher density. Crucially though, the ecological impact of changes in habitat patch sizes depends on whether habitat characteristics of patches also vary with their size, the minimum patch areas that particular species

can tolerate, and the ability of species to disperse across the landscape to other patches (Watson et al., 2005; Bierwagen, 2007).

Green space is not distributed evenly within cities (Fig. 4.4a, b). Yet, despite variation in geographic settings, socio-economic and developmental histories, topographies and sizes, there are often marked similarities between cities in this distribution, and some common correlates with key measures of urban form. For example, across the five case study cities the overall extent of green space tends to increase with elevation and topographic slope, and decline with number of road nodes, length of roads, and the density and area of housing and other buildings (Davies et al., unpublished data). The slopes of these relationships are often broadly similar between these cities. The relationships between overall green space and building/housing density result in large part from the net trade-off between two processes, the decline and the increase respectively of non-garden and garden green space with increasing housing density. Indeed, this is arguably a key driver of patterns of green space in urban areas. This is particularly apparent for relationships between green space coverage and the density of given housing types (flats, terraced, semi-detached, detached; Fig. 4.4i-l), with the net balance between the two processes shifting, such that for flats it is dominated by the loss of non-garden green space whilst for detached houses it is dominated by the gain of garden green space.

Whilst interesting patterns are evident in the distribution of green space treated as a whole, it is also clearly the case that all green space is not the same. One marked difference is in the occurrence of trees. Trees can contribute to some, though not all, of the key advantages of green space disproportionately to the area they occupy on the ground because of their complex structure that includes a canopy. This may be important in urban areas where ground space is at a premium. Therefore the numbers and distribution of trees are potentially important aspects of greenspace provision in urban areas. In what seems likely to be a general pattern, across Sheffield there is an approximately triangular relationship between tree cover and extent of green space, such that whilst the maximum amount of tree cover increases with the extent of green space, below this limit the full range of variation in tree cover is exhibited (Davies et al., 2008). There are also marked declines in tree cover with increasing housing density (Iverson and Cook, 2000). This is a particular concern at the present time given that the occurrence of trees in individual private gardens declines as those gardens become smaller (Smith et al., 2005), and that decreases in garden size are occurring in urban areas of the UK as a consequence of increased densification both in the form of high density new developments and infilling development.

Such densification, both in the UK and elsewhere, also results in general reductions of green space (e.g. Pauleit and Duhme, 2000; Pauleit et al., 2005; Yli-Pelkonen and Niemelä, 2005). This occurs both as a consequence of infill development in areas which historically have had lower densities of buildings, and through development in those areas which recently have had higher levels of brownfield land. For Sheffield, a trend of increasing urban density is apparent from a generally positive relationship between the date of urbanization and current levels of green space (Fig. 4.5a). However, a flattening of the relationship since the



Fig. 4.4 Environmental Surfaces for Sheffield at 250×250 m cell resolution: (**a**) extent of green space (m²); (**b**) vegetation cover (Normalized Difference Vegtation Index: NDVI); (**c**) tree-cover (m²); (**d**) elevation (m); (**e**) degree angle of slope; (**f**) buildings area (m²); (**g**) road length (m); (**h**) road nodes (junctions); (**i**) density of households in blocks of flats; (**j**) terraced housing density; (**k**) semi-detached housing density; (**l**) detached housing density; and (**m**) total housing density. All housing densities were measured as households ha⁻¹



1960s could indicate that recently developed areas have suffered a disproportionate loss of green space. An alternative explanation for this pattern is that urbanization of an area results initially in the loss of a certain proportion of green space, which is then gradually further eroded over time. Tree cover shows a much less pronounced, and more variable relationship with date urbanized (Fig. 4.5b). This suggests that, while more recently developed areas have similar levels of tree cover, a greater proportion of it is over impervious surface. In Sheffield, a tendency for larger trees in the western part of the city partly results from their being planted in the nineteenth century during planting initiatives associated with creation of parks and avenues. Tree cover as seen in aerial imagery is therefore probably associated with age of urbanisation, since a mature species will have a large crown area.

Ecosystem Services and Urban Form

Recent emphasis on assessing the value of the environment in terms of the benefits that ecosystems provide to humans (ecosystem services), has resulted in increasing interest in evaluating the role of urban areas in providing ecosystem services (e.g. Bolund and Hunhammar, 1999; Pataki et al., 2006; Tratalos et al., 2007a). This is partly because local provision of some of these services is valuable to human

communities in those areas, for example services such as temperature and water regulation, pest control, pollination, and recreation, all have implications for urban economies and human well-being. It is also partly because although urban systems have traditionally been accorded rather low value where environmental metrics focus on naturalness and lack of human influence, this does not necessarily mean their contribution to other types of ecosystem service such as carbon sequestration, is also low.

In order to examine the relationship between ecosystem service provision and urban form within and between cities we largely take an approach based on simple general models of the relationships between variables on the ground that can be measured across entire cities, and specific services. This allows patterns to be explored, but may not pick up all the local, and city-specific, detail that direct measurement of the services themselves would allow. However the data demands of the latter at the scales of comparison we are interested in here make the general approach the only one practical at the present time for most services.

Temperature Regulation

Urban areas experience heat island effects, in which temperatures are elevated compared with surrounding landscapes, particularly at night and in cold weather (Pickett et al., 2001; Baker et al., 2002). Although they are a function of several factors, across urban areas both air and land surface temperatures tend to increase with proportional coverage by impervious surfaces and to decrease as proportional coverage of green surfaces increases (Chen and Wong 2006; Jenerette et al., 2007). Levels of tree cover in particular can have a marked influence on temperatures, in major part by generating energy loss, and therefore cooling, through water loss to the atmosphere (Stone and Rodgers, 2001).

As well as wholesale changes in temperature regimes wrought by urbanization, variation in urban form within a built up area can have significant thermal impacts. Data from an array of temperature loggers placed across urban Sheffield revealed that elevated temperatures in both summer and winter were associated largely with the city centre, declining toward the less built up suburbs (Fig. 4.6a, b). This effect was still apparent at a local scale in the environs around each temperature logger; there was a positive relationship between temperature and impervious surface coverage within 100 m of a location, in both summer and winter (Fig. 4.7a, b).

Diurnal temperature range also varied markedly, with areas near the city centre showing a smaller difference between minimum and maximum daily temperature (Fig. 4.6c, d), although local coverage by impervious surface was not related to daily temperature range (Fig. 4.7c, d). These general patterns demonstrate marked buffering of temperatures in heavily developed areas, and show that the effect persists throughout the year. It is crucial to note, however, that while the amount of heat energy emitted per unit area might be higher in places dominated by impervious surfaces, the per household contribution to these emissions depends



Fig. 4.6 Temperature surfaces across Sheffield (a) Mean summer temperature, (b) mean winter temperature, and mean diurnal temperature range in (c) summer and (d) winter. $^{\circ}$ C interpolated from hourly data collected by an array of 50 temperature loggers arranged in five concentric rings of 10 loggers each centred on the city centre. Loggers were buried in soil at a depth of 20 mm, and data were downloaded every three months

on the absolute coverage by impervious surface within the housing parcel (Stone and Rodgers, 2001). Although smaller housing parcels tend to have a higher proportional coverage by impervious surface (Smith et al., 2005), this is unlikely to cancel out the reduction in absolute coverage by impervious surface as housing parcel size declines. Multi-dwelling buildings will further reduce the per household contribution to urban heating. Consequently, although higher density residential urban forms might generate higher levels of heating per unit area, the total amount of heat energy released from such developments is likely to be less than that produced by a low density neighbourhood comprising the same number of households.

Temperature changes through urbanization can have a variety of ecological consequences, some of which extend beyond the urban area itself, including changes



Fig. 4.7 Relationships between Cover by Impervious Surface within a 100 m radius of each Temperature Logger and (a) Mean Summer Temperature (b) Mean Winter Temperature, and Mean Diurnal Temperature Range in (c) Summer (d) Winter. Based on linear regressions

relative to rural areas in the timing of germination, leaf flush, leaf drop, and flowering of plants, and in the breeding and survival of animals (Zhang et al., 2004a, 2004b; Partecke et al., 2005; Neil and Wu, 2006). It seems likely that similar changes may also occur within the boundaries of urban areas, and be particularly influenced by variation in coverage by green space, although disentangling the effects of green space per se and its influences on temperature may not be straightforward.

Water Regulation

Cities in the UK almost invariably draw the bulk of their water supplies from outside the urban area, and have rather low coverage by standing water. However, urban areas have to deal with water influx, either directly on the area, or in rivers and streams that pass through it. Flood control and reduction of storm-water runoff are therefore key components in the ecosystem service of water regulation. Water runs directly over impervious surfaces, increasing the frequency and severity of urban flooding. In recognition of this, much research has focussed on the relationship between impervious surface coverage and water regulation (Arnold and Gibbons, 1996). However, the converse of this is that increasing the extent of green spaces are not heavily compacted. Green spaces also increase water loss

from the ground through evapotranspiration (the transfer by soil and plants of water to the atmosphere as water vapour), with additional beneficial effects on climatic conditions in urban areas.

The distribution of green space within cities therefore becomes crucial for explaining variation in water regulation across the urban landscape. Although urbanization dramatically reduces green space coverage, significant levels (60–70%) remain in many typical urban areas (Table 4.1). It has been estimated that 59% of the surface of urban Manchester is evapotranspiring (Gill et al., 2007). For the case study neighbourhoods across the five cities these reveal non-linear relationships with the extent of green space: runoff increases as greenspace declines, but at low levels of greenspace changes in the remaining greenspace have little effect (Tratalos et al., 2007a). This is a consequence of the relatively low infiltration rates of even many of the non-sealed surfaces when building densities become high.

As well as problems associated with the quantity of run-off generated in urban environments, transport of pollutants can occur as storm water washes over impervious surfaces such that suspended particulate matter within the run-off might include anthropogenically derived materials. Such contaminants can represent a significant non-point source of pollution in urban areas (Characklis and Wiesner 1997; Bibby and Webster-Brown, 2005).

Carbon Sequestration

It is becomingly increasingly apparent that, at least in some regions, carbon sequestration in urban areas may not be as trivial a consideration as some have suggested (Jo and McPherson 1995, Golubiewski 2006; Pataki et al., 2006). Indeed, whilst obviously small compared with carbon emissions the per unit area and the gross sizes of urban carbon pools (reservoirs of stored carbon) can nonetheless be substantial (Nowak and Crane 2002; Kaye et al., 2005; Lorenz et al., 2006). There are two major natural carbon pools in urban green spaces, comprising respectively vegetation and soils.

The levels of carbon sequestration by trees in urban areas, resulting from the fixation of carbon during photosynthesis and its storage as biomass, tend to be estimated using simple functions of tree cover (e.g. Rowntree and Nowak, 1991; Whitford et al., 2001; Tratalos et al., 2007a). Detailed mapping in Sheffield has shown that the surface of the city is about 15% tree-covered (see also Fig. 4.4c). This is notably lower than in some other cities in the northern hemisphere (e.g. Nowak and Crane, 2002). An analysis of carbon sequestration across the five case study cities indicated that more densely urbanized areas are associated with a lower predicted rate of carbon sequestration (Tratalos et al., 2007a). Because calculations for carbon sequestration were based on a linear relationship with tree cover (see Rowntree and Nowak, 1991), results for carbon sequestration can typically be interpreted as matching those for tree cover. However, while carbon sequestration will generally increase with tree cover, in practice the relationship will depend,

among other things, on the demographic structure and species composition of the urban forest. This problem is compounded by the fact that tree cover will itself vary with the age of the properties which make up the urban form, and tree growth rates will vary according to soil compaction, pollution, impervious surface area under tree crown and water potential (Quigley, 2004).

A sample of 2170 trees from Sheffield indicates clear systematic patterns in tree species richness and size (Fig. 4.8). Broadly speaking, tree richness increased with distance from the city centre, along with an associated increase in tree size. The sample comprised 96 tree species, although the ten most abundant species accounted for 64% of all individuals recorded. Trees nearer the city centre tended to be taller relative to their girth than those on the margins of the city, although there was substantial variation in this (Fig. 4.8d). While some very mature trees were present in the sample, with girths of 3–4 m, the size distribution of the trees was strongly right-skewed (Fig. 4.9), with 50% of individuals having a girth <0.5 m. Continual



Fig. 4.8 Surfaces for Sheffield of (**a**) Tree Species Richness, (**b**) Average Tree Circumference (m), (**c**) Average Tree Height (m) and (**d**) Height/Circumference. Data are derived from field measurements of 2170 trees (the five trees nearest to 434 randomly-chosen locations across the city)



Fig. 4.9 Distribution of Circumferences of Trees across Urban Sheffield. Data are derived from field measurements of 1924 trees selected as the five trees nearest to 434 randomly-chosen locations across the city (circumference could not be measured for 246 individuals because of access difficulties)

replacement of mature urban trees with smaller species that are more manageable, and less prone to cause damage through deadfall, windthrow and root intrusion has recently been documented in London (London Assembly Environment Committee, 2007).

Management of the urban forest is further complicated by the influence of land ownership. Recent data from Sheffield indicate that, within a 13 km² study area, approximately 69% of tree cover occurred on privately owned land (Dennis, unpublished data). About 73% of the land was privately owned, indicating that proportional levels of tree cover on private and public lands were broadly similar. These data highlight the very limited extent to which adequate management of the urban forest can emerge simply from policies focusing on land under direct public control. There is a need to recognise that much urban land is under dispersed ownership, with small private parcels representing domestic gardens being managed in markedly different ways (Gaston et al., 2005). Furthermore, smaller private gardens are less likely to contain trees, and there are concerns that continuing densification of urban areas through infilling development might lead indirectly to further reductions in tree cover through this route (Smith et al., 2005). In public lands, street trees may be removed for public safety reasons, or as a result of subsidence claims by insurance companies. Despite no change in overall tree numbers in a five year study period, data from London indicate a rapid turnover of street trees for these reasons, and an ongoing disproportionate loss of mature native trees (London Assembly Environment Committee, 2007).

The other significant carbon pool in urban areas is that contained within soils. The sizes of such carbon pools have been surprisingly little explored, in large part because urban soils (i) are typically extremely heterogeneous both spatially and temporally, comprising a mix of islands of apparently natural soils within a matrix of highly human-altered soils; (ii) have almost invariably been poorly mapped; and (iii) often have altered processes of decomposition and nutrient cycling, for a range of reasons including the urban heat island effect (Effland and Pouyat, 1997; Pouyat et al., 1997; Carreiro et al., 1999; Pickett et al., 2001). Where soil carbon pools have been estimated in urban areas, this has typically involved extrapolation from data collated from just a handful of cores and tiny quantities of soil (e.g. Pouyat et al., 2006). Nonetheless, such work suggests that these pools may be substantial. Legal protection of the ecological functions of urban soils in Germany attests to this importance, despite the rather rudimentary current understanding of the properties of urban soil (Lorenz et al., 2006).

The significance of carbon sequestration in urban areas depends fundamentally on how vegetation and soils are managed. Carbon emissions associated with management (e.g. from chain saws, chippers, lawn mowers and transport of cut vegetation) could, for example, negate any positive sequestration effects, although the extent to which this is a problem can be influenced by the choice of management approaches and the fate of vegetation that is removed (e.g. landfill, bio-fuels). However, even where this is the case, urban trees in particular may typically bring a number of other advantages (e.g. control of storm water, reducing energy use in buildings and human health benefits), and a decided net benefit (Akbari, 2002; McPherson et al., 2005).

Pollination

Many plant species are pollinated by insects, which transfer pollen from one plant to another. Some plant species cannot produce fertile seeds without pollination by insects, and the yield of most plants is improved where insect pollination occurs. Concern has been expressed about regional and local reductions in the numbers of pollinators in a variety of areas of the world (Biesmeijer et al., 2006; Klein et al., 2007). As well as impacting on agricultural crop production (Allen-Wardell et al., 1998), declines in pollinators adversely affect the functioning of natural and anthropogenically-perturbed ecosystems (Kearns et al., 1998; Cheptou and Avendaño, 2006; but see Ghazoul, 2005). Additionally, some people with private domestic gardens rely on, and derive economic benefit from, pollination of garden plants (Nabhan and Buchmann, 1997). Although several studies have documented declines in insect pollinators in response to urbanization, others have found elevated bee species richness in urban conditions, attributed to increased temperatures, reduced exposure to agricultural chemicals, and a wider variety of microhabitats being present in urban landscapes (Eremeeva and Sushchev, 2005 and references therein).

Pollinator faunas can also change systematically within urban areas. Across private gardens in Sheffield, the number of species of an important group of pollinators, bumblebees, is influenced both by immediate local and larger scale factors, increasing with the habitat diversity within the gardens and with the area of green space in the environs of the gardens (Smith et al., 2006a). However, no such relationships were apparent for the abundance of this group (Smith et al., 2006b).

As well as changes to pollinator faunas, impacts of urbanization on the distribution and population dynamics of plant species can affect plant-pollinator relationships. Fragmentation of the natural environment through urbanization frequently leads to small, isolated populations of plants, and diminished population densities, which can result in few pollinators finding a particular patch of plants. Cheptou and Avendaño (2006) studied an urban plant (*Crepis sancta*: Asteraceae), showing that the number of pollinating insects visiting a patch of plants increased as the number of plants in the patch increased. This suggests that where urbanization leads to a plant becoming rarer, visits by pollinators might also decline, potentially compounding the effect of urbanization. This will be particularly important for those plant species that are wholly dependent on insects for pollination, and cannot reproduce unless they are visited by pollinators. *Crepis sancta* is able to self-fertilize, but it was not able to increase its levels of selfing in response to a low number of visits by pollinators. More research is needed on the adaptations of plants to urbanization.

Provision of Pest Control

Effective functioning of ecosystems depends on the maintenance of interactions between species, including competition, predation, parasitism and mutualism. By differentially altering the abundance of particular species as areas are progressively developed, urbanization often results in new combinations of species that have not previously interacted. There is little information on how species interact in such new combinations, and how these interactions vary across urban landscapes. In one well studied example, urbanization of Phoenix, Arizona, led to a dramatic increase in available water, a limiting resource in the surrounding desert landscape. This has resulted in a shift in species interactions, including a greater top-down influence of predators on the system (Faeth et al., 2005). Urbanization can also lead to asymmetric changes in the abundances of particular species, such as rapid increases in species that might be regarded as "pests" (Alberti et al., 2003).

To assess the potential for such changes in response to increases in urban density, recent work in Sheffield examined rates of mortiality in a widespread herbivorous insect, the holly leaf-miner *Phytomyza ilicis* Curtis (Diptera, Agromyzidae). The holly leaf-miner is the most common insect herbivore of European holly *Ilex aquifolium*, feeding inside the leaves with the large blotch mines visible on the upper surface of occupied leaves. A pest population of holly leaf-miners was successfully controlled by introducing a parasitic wasp in Canada (Clausen, 1978), highlighting the potential economic importance of understanding such species interactions. While in the leaf, the larva is subject to a number of potential mortalities, including miscellaneous death during the larval or pupal stages, e.g. starvation caused by low plant quality, parasitism (always fatal) by various species of wasp, and bird predation (Cameron, 1939).

Sampling across Sheffield indicated only rather weak effects of impervious surface cover, housing and tree cover on the abundance and mortality of the holly

Urban form variable	Prop. mined leaves	Misc. larval death	Larval parasitism	Bird predation	Pupal parasitism	Misc. pupal death	Successful emergence
Impervious surface cover Housing density Tree cover	-0.1 0.14* -0.15*	0.04 -0.11 0.21**	0.20** 0.01 -0.002	-0.12 -0.19** 0.23***	-0.06 0.14* -0.20**	-0.19* -0.04 -0.10	-0.12 0.09 -0.22**

 Table 4.2
 Holly Leaf-Miner Demographics in relation to Urban Form in Sheffield

Of 460 sampling locations across the city, holly plants were found within the survey area (approx 1 ha) at 276. The proportion of mined leaves on each plant was estimated by haphazardly sampling The table reports Spearman rank correlation coefficients of the relationships of these demographic outcomes with three urban form variables, measured within a 100 m radius from the holly plant location.

 * indicates statistically significant at the 5% level ** statistically significant at the 1% level

*** statistically significant at 0.01% level

leaf-miner (Table 4.2). The strongest relationships ($r_s > 0.2$) were those with tree cover. Both miscellaneous larval death and bird predation increased with tree cover, while pupal parasitism declined. This translated into an overall decline in successful emergence with increasing urban tree cover. The other measures of urban form (impervious surface cover and housing density) were only weakly related to rates of mortality and parasitism, and were not associated with any overall change in successful emergence. These results show that at least some complex interactions among species can remain apparently largely intact even within highly developed sites.

Recreation

For many people, green spaces in urban areas provide their primary contact with biodiversity and the "natural" environment (Jorgensen et al., 2002), may influence their physical and mental well-being (Ulrich et al., 1991; Jackson, 2003), and, in the case of public green space, can offer broader social benefits as meeting places that give a shared focus to diverse communities and neighbourhoods (Germann-Chiari and Seeland, 2004; Martin et al., 2004). In consequence, regulatory and advisory agencies have made various recommendations for the minimum provision of urban green space, usually expressed as the walking distance or time to access the resource (e.g. Stanners and Bourdeau, 1995).

Across Sheffield, there is enormous variation in the distances through the road network that separate households from their nearest accessible public green space (Barbosa et al., 2007). Many households do not enjoy the levels of access recommended by governmental agencies, with the distribution of distances being strongly right-skewed such that for some households these distances are particularly large. The mean level of access varies significantly across different sectors of

society. As these sectors tend to occupy areas characterised by different urban form, levels of access are also likely to vary systematically with urban form, although this has not explicitly been tested. These distance-based measures of access could usefully be refined to include travel constraints, such as physical and psychological barriers to pedestrian movement (Handy, 1996).

Even if green space is locally accessible, its quality is extremely variable. Work in Sheffield has demonstrated that benefits to psychological well-being of visitors to urban parks are positively correlated with the species richness of those spaces, suggesting that the biological complexity of urban green space is important in enhancing human well-being, as well as for the conservation of biodiversity itself (Fuller et al., 2007a; see Chapter 10).

The private domestic garden has long been considered an important part of human health and well-being (see e.g. Gerlach-Spriggs et al., 1998). Access to a garden reduced self-reported sensitivity to stress (Stigsdotter and Grahn, 2004), while lack of access has been associated with increased self-reported levels of depression and anxiety (Macintyre et al., 2003). When asked to identify the contribution of spending time in the garden to overall well-being, 57% of householders in Perth, Australia indicated it was very important or the most important factor (ARCWIS, 2002). Interactions between people and nature frequently happen in private gardens, with wildlife-friendly gardening practices now receiving greater attention, and a large proportion of the population specifically providing food and shelter for birds (Lepczyk et al., 2004; Fuller et al., in press). Because of the diverse array of benefits to human well-being of gardens on the one hand, and public green spaces on the other, it is unclear to what extent they are substitutable in urban planning (Barbosa et al., 2007). This is especially important given that increasing housing density reduces non-garden green space and expands overall coverage by gardens, although individual gardens become smaller.

Biodiversity and Urban Form

Biodiversity can be thought of as the variety of life, at all levels of organisation from genetic through species diversity to ecolosystem diversity. Biodiversity has variously been considered to be a contributor to some ecosystem services, to be an ecosystem service itself, and to be a product of some ecosystem services (Gaston and Spicer, 2004; Millennium Ecosystem Assessment, 2000). Doubtless, in some senses it is all three. However, in the context of urban systems, biodiversity has typically been considered in its own right, and largely with respect to how it relates to the physical level of urbanization, with little consideration of its relevance to ecosystem services or how it is influenced by variation in human activity across the urban landscape.

A large number of studies has now been conducted examining the relationships between the species richness of selected taxonomic groups (species richness is a key measure of biodiversity) and levels of urban development, usually along a rural-urban gradient. Generally, studies have shown declines in overall species richness at very high levels of urbanization, but a mixed response at low and intermediate levels of urban development, where the observed pattern depends strongly on taxonomic group (Marzluff, 2001; Chace and Walsh, 2006; McKinney, 2008). Some studies have shown no simple pattern with increased urban development (e.g. Roy et al., 1999; Niemelä et al., 2002; Mason, 2006). A decline in richness as urbanization intensifies is usually attributed to the loss of suitable habitat and resources. A peak in richness at intermediate levels is often associated with a greater number of land use types in areas of intermediate levels of land that leads to variation in management styles. Increases in richness with urban development seem usually, at least in part, to occur because of the relatively high numbers of alien/introduced species in more heavily developed areas (e.g. Kowarik, 1990; Roy et al., 1999; Marzluff, 2001; Wittig, 2004).

Across Sheffield, both plant and bird diversity shows strong heterogeneity (Fig. 4.10). Species richness of breeding birds is concentrated around the edges of the city, with generally low values in the centre (Fig. 4.10a). Breeding bird density shows a superficially similar pattern (Fig. 4.10b), although values appear to peak inside the margin of the city, an effect particularly evident along the south-eastern and northern fringes of the city. This pattern suggests that breeding bird densities peak some distance inside the edge of the city. Patterns of plant species richness are less clear. There are high values of native plant richness near the edge of city, particularly in the south and east (Fig. 4.10c), although there is substantial heterogeneity in the distribution. Alien plants show very low species richness around the edge of the city, peaking at intermediate levels of urban development, and declining again toward the city centre (Fig. 4.10d).

Despite the large number of studies documenting the responses of biodiversity to urbanization, understanding of the relationship between biodiversity and urban form per se remains poor. The different components of the rural-urban gradient are often not well differentiated in analyses, and comparison tends to be focussed on the difference between rural and urban areas rather than between urban areas of differing structure. This is particularly significant at the present time, when on the one hand urban areas in the UK are becoming havens for some species that have undergone marked declines in the wider countryside (e.g. blackbird *Turdus merula*, song thrush *T. philomelos*; Gregory and Baillie, 1998; Mason, 2000), largely as a consequence of intensive agriculture, and on the other hand some previously common and widespread species are undergoing marked declines in urban areas (e.g. starling *Sturnus vulgaris*, house sparrow *Passer domesticus*; Cannon et al., 2005). There are strong suggestions that both of these trends are more apparent in some urban forms than in others, but empirical evidence remains scant.

In Sheffield, housing density was strongly negatively associated with breeding bird richness (Fig. 4.11a), while the relationship with breeding bird abundance appeared hump-shaped, with bird abundance peaking at intermediate levels of housing density (Fig. 4.11b). This is presumably related to the overall loss of green space with increasing urban development, as well as a reduction in the range



Fig. 4.10 Surfaces for Sheffield of (**a**) Breeding Bird Species Richness, (**b**) Breeding Bird Density (individuals km⁻²), (**c**) Native Plant Species Richness and (**d**) Non-native Plant Species richness. Bird data are derived from 5-minute point transects carried out at 640 locations across Sheffield (a point randomly located within each 500×500 m grid cell across the city). Plant data are derived from 1000 quadrats (1 m²) placed across the city

of available habitat types. A range of processes may contribute to such changes in abundance, including natural food availability, the availability of artificiallyprovided supplementary food, nest site availability and quality, predation pressure and interspecific competition (Clergeau et al., 1998; Thorington and Bowman, 2003; Shochat, 2004; Faeth et al., 2005). The ratios of different types of habitat available are likely to change systematically with increasing urban development. For example, as its overall coverage declines, green space changes markedly in composition, with a greater proportion being made up of gardens rather than other forms of vegetated surface.

The diversity of plants was more weakly related to housing density (as measured by the number of addresses per unit area). Raw plots show a negative relationship between housing density and native plant species richness (Fig. 4.11c), and a



Fig. 4.11 Relationships between Housing Density and (a) Breeding Bird Species Richness, (b) Breeding Bird Density, (c) Native Plant Species Richness and (d) Non-Native Plant Species Richness. Housing density is log address density within a 100 m buffer around each sampling location. Bird density is individuals km⁻². Error bars are 95% confidence intervals

positive relationship between housing density and alien plant species richness (Fig. 4.11d). Many urban alien plants in northern Europe have natural ranges in regions with Mediterranean climates and dry soils, rather akin to those found within highly urbanized environments (Sukopp and Wurzel, 2002). Moreover, propagule pressure resulting from introductions and escapes of alien plants from gardens and amenity plantings is likely to be much higher in towns and cities than in rural areas, and disturbed urban habitats may promote the establishment of non-native weedy species (Smith et al., 2006c; Dehnen-Schmutz et al., 2007).

Work in Sheffield has also revealed an intriguing behavioural response to urban noise (arguably either a measure, or a close correlate, of urban form). Sites where European robins *Erithacus rubecula* sang nocturnally tended also to be those places with high noise levels during the day, suggesting that the birds were singing at night to avoid acoustic competition with daytime urban noise (Fuller et al., 2007b). As noise levels are closely related to certain components of urban form, such as where transport networks are located, the times at which birds choose to sing may in turn be affected by urban design. (Warren et al., 2006). Daytime noise at a particular urban location is strongly positively correlated with the proportion of impervious surface in the surrounding area (Fig. 4.12). Integrative studies that can disentangle these multiple ecological and behavioural effects at a city-wide scale are needed.



Fig. 4.12 Relationship between Noise Levels and Impervious Surfaces. Noise levels are the mean of 10 measurements taken with a handheld digital sound meter at 30 s intervals at 628 locations across Sheffield and the proportion of impervious surface within a 100 m radius around these points (r = 0.48, n = 628, p < 0.001)

Looking more widely across Britain as a whole, the species richness and abundance of breeding birds responds systematically to variation in housing density (Tratalos et al., 2007b). At a resolution of 1×1 km squares, avian species richness at first increases strongly with housing density, but then declines rapidly at higher housing densities. A similar pattern is seen in the richness of 27 urban specialist species used as urban health indicators by the UK government (DEFRA, 2002, 2003), suggesting that even those species best able to exploit urban environments are impacted at high urban densities. Standardised abundances of all species, including the urban indicator species, are positively associated with housing density, but decline at very high urban densities (Tratalos et al., 2007b).

Conclusions

The analysis has shown that the density of urban development, a key measure of urban form, is strongly associated with a reduction in total green space coverage, and changes to the connectivity of vegetated patches within the urban landscape. This has ramifications for the ecosystem services that are mediated by green space, including the regulation of water and temperature regimes, carbon sequestration and the provision of pest control and pollinators across the urban landscape. Moreover, changes to the amount and quality of green space will have significant consequences for recreation within urban areas, access to an experience of nature, and for human

quality of life more generally. Levels of biodiversity generally increased initially with housing density, but then declined sharply at highly developed sites.

An understanding of the distribution of ecosystem services and the associated pattern of biodiversity across the urban landscape is crucial for predicting the consequences of the increases in urban density required by current UK legislation. Although many of these responses point to a decline in ecosystem function and biodiversity potential with increasing urban density, there is substantial scatter around many of these relationships, suggesting scope for maximizing the environmental and ecological performance of urban areas for any given level of urban density. Perhaps rather more troubling are the observed declines in abundance at high urban densities even of those species most able to exploit urban environments. This suggests a difficult trade-off between on the one hand, minimizing the conversion of land for new development, and on the other maintaining meaningful levels of biodiversity and ecosystem function around the places where most of us live. There is a need for longitudinal studies documenting past changes in the spatial configuration of urban landscapes in response to urbanization, and using that information to predict the future consequences of alternative modes of development and regeneration at regional and even national scales. Spatially explicit area selection exercises, of the kind that have been used extensively in the conservation planning literature (e.g. Pressey et al., 1997; Cabeza and Moilanen, 2001) will help identify areas that are crucial to maintaining effective ecosystem function, and those that might efficiently be used for high density residential developments.

With more than half the world's population now living in urban areas, changes in how we plan, manage and develop such areas has potentially profound impacts. A better understanding of the best strategies for managing the trade-offs among environmental functions and urban form is urgently required if we are to ensure that increasing urban density and concomitant declines in green space and biodiversity are not to lead to impaired ecosystem function, reduced provision of ecosystem services and the degradation of human experiences of nature.

Acknowledgements We are grateful for generous assistance, discussion and/or comments from N. Burke, C. Dennis, N. Doar, I. Fishburn, C. Gascoigne, J. Glasscock, K. Irvine, P. Johnson, C. Jones, D. Knight, D. Lewis, and J. Vulliamy. Ordnance Survey kindly provided MasterMap data under license to the CityForm consortium. K.J.G. holds a Royal Society-Wolfson Research Merit Award.

References

- Akbari, H. (2002) Shade trees reduce building energy use and CO₂ emissions from power plants. *Environmental Pollution*, **116** (Suppl. 1), S119–S126.
- Alberti, M., Marzluff, J.M., Shulenberger, E., Bradley, G., Ryan, C. and Zumbrunnen, C. (2003) Integrating humans into ecology: Opportunities and challenges for studying urban ecosystems. *BioScience*, **53**(12), pp. 1169–1179.
- Allen-Wardell, G., Bernhardt, P., Bitner, R., Burquez, A., Buchmann, S., Cane, J., Cox, P.A., Dalton, V., Feinsinger, P., Ingram, M., Inouye, D., Jones, C.E., Kennedy, K., Kevan, P.,
Koopowitz, H., Medellin, R., Medellin-Morales, S., Nabhan, G.P., Pavlik, B., Tepedino, V., Torchio, P. and Walker, S. (1998) The potential consequences of pollinator declines on the conservation of biodiversity and stability of food crop yields. *Conservation Biology*, **12**(1), pp. 8–17.

- ARCWIS (2002) Perth Domestic Water-Use Study: Household Appliance Ownership and Community Attitudinal Analysis 1999 2000, Australian Research Centre for Water in Society, CSRIO, Perth, Australia.
- Arnold, C.L. and Gibbons, J. (1996) Impervious surface coverage: The emergence of a key environmental indicator. *Journal of the American Planning Association* 62(2), pp. 243–258
- Baker, L.A., Brazel, A.J., Selover, N., Martin, C., McIntyre, N., Steiner, F.R., Nelson, A. and Musacchio, L. (2002) Urbanization and warming of Phoenix (Arizona, USA): Impacts, feedbacks and mitigation. *Urban Ecosystems*, 6(3), pp. 183–203.
- Barbosa, O., Tratalos, J.A., Armsworth, P.R., Davies, R.G., Fuller, R.A., Johnson, P. and Gaston, K.J. (2007) Who benefits from access to green space? A case study from Sheffield, UK. *Landscape and Urban Planning*, 83(2-3), pp. 187–195.
- Bastin, L. and Thomas, C.D. (1999) The distribution of plant species in urban vegetation fragments. Landscape Ecology, 14(5), pp. 493–507.
- Beer, A. (2005) The green structure of Sheffield, in *Green structure and urban planning. Final report of COST Action C1*, (eds. A.C. Werquin, B. Duhem, G. Lindholm, B. Oppermann, S. Pauleit and S. Tjallingii), COST Programme, Brussels.
- Bibby R. and Webster-Brown J.G. (2005) Characterisation of urban suspended particulate matter (Auckland Region, New Zealand); a comparison with non-urban SPM. *Science of the Total Environment*, **343**(1-3), pp. 177–197.
- Bierwagen, B.G. (2007) Connectivity in urbanizing landscapes: The importance of habitat configuration, urban area size, and dispersal. *Urban Ecosystems*, **10**(1), pp. 29–42.
- Biesmeijer, J.C., Roberts, S.P.M., Reemer, M., Ohlemuller, R., Edwards, M., Peeters, T., Schaffers, A.P., Potts, S.G., Kleukers, R., Thomas, C.D., Settele, J. and Kunin, W.E. (2006) Parallel declines in pollinators and insect-pollinated plants in Britain and the Netherlands. *Science*, 313(5785), pp. 351–354.
- Bolund, P. and Hunhammar, S. (1999) Ecosystem services in urban areas. *Ecological Economics*, 29(2), pp. 293–301.
- Commission for Architecture and the Built Environment (CABE) (2004) *The value of public space: How high quality parks and public spaces create economic, social and environmental value.* CABE Space, London.
- Cabeza, M. and Moilanen, A. (2001) Design of reserve networks and the persistence of biodiversity. *Trends in Ecology and Evolution*, 16(5), pp. 242–248.
- Cameron, E. (1939) The holly leaf miner (*Phytomyza ilicis*, Curtis) and its parasites. *Bulletin of Entomological Research*, **30**, pp. 173–208.
- Cannon, A., Chamerlain, D., Toms, M., Hatchwell, B. and Gaston, K. (2005) Trends in the use of private gardens by wild birds in Great Britain 1995–2002, *Journal of Applied Ecology*, 42(4), pp. 659–671.
- Carreiro, M.M., Howe, K., Parkhurst, D.F. and Pouyat, R.V. (1999) Variation in quality and decomposability of red oak leaf litter along an urban-rural gradient. *Biology and Fertility of Soils*, **30**(3), pp. 258–268.
- Chace, J.F. and Walsh, J.J. (2006) Urban effects on native avifauna: a review. *Landscape and Urban Planning*, **74**(1), pp. 46–69.
- Characklis, G.W. and Wiesner, M.R. (1997) Particles, metals and water quality in runoff from a large urban watershed. *Journal of Environmental Engineering*, **123**(8), pp. 753–759.
- Chen, Y. and Wong, N.H. (2006) Thermal benefits of city parks. *Energy and Buildings*, **38**(2), pp. 105–120.
- Cheptou, P.-O. and Avendaño V, L.G. (2006) Pollination processes and the Allee effect in highly fragmented populations: consequences for the mating system in urban environments. *New Phytologist*, **172**(4), pp. 774–783.

- Clausen, C.P. (ed.) (1978) Introduced Parasites and Predators of Arthropod Pests and Weeds: A World Review., Agricultural Handbook No. 480, USDA Agricultural Research Service, Washington, D.C.
- Clergeau, P., Savard, J.-P.L., Mennechez, G. and Falardeau, G. (1998) Bird abundance and diversity along an urban-rural gradient: A comparative study between two cities on different continents. *Condor*, **100**(3), pp. 413–425.
- Davies, R., Barbosa, O., Fuller R., Tratalos, J., Burke, N., Lewis, D., Warren, P and Gaston, K. (2008), City-wide relationships between green spaces, urban land use and topography. *Urban Ecosystems*, **11**(3), pp. 269–287
- de Vries, S., Verheij, R.A., Groenewegen, P.P. and Spreeuwenberg, P. (2003) Natural environments–healthy environments? An exploratory analysis of the relationship between greenspace and health. *Environment and Planning A*, **35**(10), pp. 1717–1731.
- Department of the Environment, Food and Rural Affairs (DEFRA) (2002) *Working with the Grain of Nature*, Defra Publications, London.
- Department of the Environment, Food and Rural Affairs (DEFRA) (2003) *Measuring Progress: Baseline Assessment*, Defra Publications, London.
- Dehnen-Schmutz, K., Touza, J., Perrings, C. and Williamson, M. (2007) The horticultural trade and ornamental plant invasions in Britain. *Conservation Biology*, 21(1), pp. 224–231.
- Dunnett, N. and Quasim, M. (2000) Perceived benefits to human well-being of urban gardens. *HortTechnology*, **10**(1), pp. 40–45.
- Effland, W.R. and Pouyat, R.V. (1997) The genesis, classification, and mapping of soils in urban areas. *Urban Ecosystems*, 1(4), pp. 217–228.
- Eremeeva, N.I. and Sushchev, D.V. (2005) Structural changes in the fauna of pollinating insects in urban landscapes. *Russian Journal of Ecology*, **36**(4) pp. 259–265.
- Faeth, S.H., Warren, P.S., Stochat, E. and Marussich, W.A. (2005) Trophic dynamics in urban communities. *BioScience*, 55(5), pp. 399–407.
- Fuller, R.A., Irvine, K.N., Davies, Z.G., Armsworth, P.R. and Gaston, K.J. (in press) Interactions between people and birds in urban landscapes. *Studies in Avian Biology*.
- Fuller, R.A., Irvine, K.N., Devine-Wright, P., Warren, P.H. and Gaston, K.J. (2007a) Psychological benefits of greenspace increase with biodiversity. *Biology Letters*, 3(4), pp. 390–394.
- Fuller, R.A., Warren, P.H. and Gaston, K.J. (2007b) Daytime noise predicts nocturnal singing in urban robins. *Biology Letters*, **3**(4), pp. 368–370.
- Gaston, K.J., Smith, R.M., Thompson, K. and Warren, P.H. (2005) Urban domestic gardens (II): Experimental tests of methods for increasing biodiversity. *Biodiversity and Conservation*, **14**(2), pp. 395–413.
- Gaston, K.J. and Spicer, J.I. (2004) *Biodiversity: An Introduction*. 2nd edition, Blackwell Publishing, Oxford.
- Gerlach-Spriggs, N., Kaufman R. E. and Warner, S.B.Jr. (1998), *Restorative Gardens: The Healing Landscape*, Yale University Press, New Haven.
- Germann-Chiari, C. and Seeland, K. (2004) Are urban green spaces optimally distributed to act as places for social integration? Results of a geographical information system (GIS) approach for urban forestry research. *Forest Policy and Economics*, **6**(1), pp. 3–13.
- Ghazoul, J. (2005) Buzziness as usual? Questioning the global pollination crisis. Trends in Ecology and Evolution, 20(7), pp. 367–373.
- Gill, S.E., Handley, J.F., Ennos, A.R. and Pauleit, S. (2007) Adapting cities for climate change: The role of the green infrastructure. *Built Environment*, **33**(1), pp. 115–133.
- Golubiewski, N.E. (2006) Urbanization increases grassland carbon pools: Effects of landscaping in Colorado's front range. *Ecological Applications*, 16(2), pp. 555–571.
- Gregory, R.D. and Baillie, S.R. (1998) Large-scale habitat use of some declining British birds. Journal of Applied Ecology, 35(5), pp. 785–799.
- Handy, S.L. (1996) Urban form and pedestrian choices: Study of Austin neighbors. *Transportation Research Record*, 1552, pp. 135–144.

- Henry, J.A. and Dicks, S.E. (1987) Association of urban temperatures with land-use and surface materials. *Landscape and Urban Planning*, 14(1), pp. 21–29.
- Hey, D. (2005) A History of Sheffield, Carnegie Publishing Limited, Lancaster.
- Iverson, L.R. and Cook, E.A. (2000) Urban forest cover of the Chicago region and its relation to household density and income. Urban Ecosystems, 4(2), pp. 105–124.
- Jackson, L.E. (2003) The relationship of urban design to human health and condition. *Landscape* and Urban Planning, **64**(4), pp. 191–200.
- Jenerette, G.D., Harlan, S.L., Brazel, A., Jones, N., Larsen, L. and Stefanov, W.L. (2007) Regional relationships between surface temperature, vegetation, and human settlement in a rapidly urbanizing ecosystem. *Landscape Ecology*, 22(3), pp. 353–365.
- Jo, H.-K. and McPherson, E.G. (1995) Carbon storage and flux in urban residential greenspace. *Journal of Environmental Management*, 45(2), pp. 109–133.
- Johnson, E.A. and Klemens, M.W. (2005) The impacts of sprawl on biodiversity, in *Nature in Fragments: The Legacy of Sprawl*, (eds. E.A. Johnson and M.W. Klemens), Columbia University Press, New York.
- Jorgensen, A., Hitchmough, J. and Calvert, T. (2002) Woodland spaces and edges: Their impact on perception of safety and preference. *Landscape and Urban Planning*, 60(3), pp. 135–150.
- Kaye, J.P., McCulley, R.L. and Burke, I.C. (2005) Carbon fluxes, nitrogen cycling, and soil microbial communities in adjacent urban, native and agricultural ecosystems. *Global Change Biology*, **11**(4), pp. 575–587.
- Kearns, C.A., Inouye, D.W. and Waser, N.M. (1998) Endangered mutualisms: The conservation of plant-pollinator interactions. *Annual Review of Ecology and Systematics*, **29**, pp. 83–112.
- Klein, A.-M., Vaissière, B.E., Cane, J.H., Steffan-Dewenter, I., Cunningham, S.A., Kremen, C. and Tscharntke, T. (2007) Importance of pollinators in changing landscapes for world crops. *Proceedings of the Royal Society B: Biological Sciences*, 274(1608), pp. 303–313.
- Kowarik, I. (1990) Some responses of flora and vegetation to urbanization in Central Europe, in Urban Ecology: Plants and Plant Communities in Urban Environments, (eds. H. Sukopp, S. Hejný and I. Kowarik), SPB Academic Publishing, The Hague.
- Kuo, F.E. and Sullivan, W.C. (2001) Environment and crime in the inner city. Does vegetation reduce crime? *Environment and Behavior*, **33**(3), pp. 343–367.
- Lepczyk, C.A., Mertig, A.G. and Liu, J. (2004) Assessing landowner activities related to birds across rural-to-urban landscapes. *Environmental Management*, 33(1), pp. 110–125.
- London Assembly Environment Committee (2007) Chainsaw massacre: A review of London's street trees, Greater London Authority, London.
- Lorenz, K., Preston, C.M. and Kandeler, E. (2006) Soil organic matter in urban soils: Estimation of elemental carbon by thermal oxidation and characterization of organic matter by solid-state 13C nuclear magnetic resonance (NMR) spectroscopy. *Geoderma*, **130**(3-4), pp. 312–323.
- Macintyre, S., Ellaway, A., Hiscock, R., Kearns, A., Der, G. and McKay, L. (2003) What features of the home and the area might help to explain observed relationships between housing tenure and health? Evidence from the West of Scotland. *Health and Place*, **9**(3), pp. 207–218.
- Martin, C.A., Warren, P.S. and Kinzig, A.P. (2004) Neighborhood socioeconomic status is a useful predictor of perennial landscape vegetation in residential neighborhoods and embedded small parks of Phoenix, AZ. *Landscape and Urban Planning*, **69**(4), pp. 355–368.
- Marzluff, J.M. (2001) Worldwide urbanization and its effects on birds, in Avian Ecology and Conservation in an Urbanizing World, (eds M. Marzluff, R. Bowman and R. Donnelly), Kluwer Academic Publishers, Boston.
- Mason, C.F. (2000) Thrushes now largely restricted to the built environment in eastern England. Diversity and Distributions, 6(4), pp. 189–194.
- Mason, C.F. (2006) Avian species richness and numbers in the built environment: can new housing developments be good for birds? *Biodiversity and Conservation*, **15**(8), pp. 2365–2378.
- McKinney, M.L. (2002) Urbanization, biodiversity, and conservation. *BioScience*, **52**(10), pp. 883–890.

- McKinney M.L. (2008) Effects of urbanization on species richness: A review of plants and animals, Urban Ecosystems, 11(2), pp. 161–176
- McPherson, G., Simpson, J.R., Peper, P.J., Maco, S.E. and Xiao, Q. (2005) Municipal forest benefits and costs in five US cities. *Journal of Forestry*, **103**(8), pp. 411–416.
- Millennium Ecosystem Assessment (2000) Ecosystems and Human Well Being: A Framework for Assessment, Island Press, Washington, D.C.
- Nabhan, G.P. and Buchmann, S.L. (1997) Services provided by pollinators, in *Nature's Services*, (ed. G. Daily), Island Press, Washington, D.C.
- Neil, K. and Wu, J. (2006) Effects of urbanization on plant flowering phenology: A review. Urban Ecosystems, 9(3), pp. 243–257.
- Niemelä, J. (1999) Ecology and urban planning. *Biodiversity and Conservation*, 8(1), pp. 119–131.
- Niemelä, J., Kotze, D.J., Venn, S., Penev, L., Stoyanov, I., Spence, J., Hartley, D. and De Oca, E.M. (2002) Carabid beetle assemblages (Coleoptera, Carabidae) across urban-rural gradients: An international comparison. *Landscape Ecology*, **17**(5), pp. 387–401.
- Nowak, D.J. and Crane, D.E. (2002) Carbon storage and sequestration by urban trees in the USA. *Environmental Pollution*, **116**(3), pp. 381–389.
- Ordnance Survey (2006) OS MasterMap User Guide. Version 6.1.1., Ordnance Survey, Southampton, UK.
- Partecke, J., Van't Hof, T.J. and Gwinner, E. (2005) Underlying physiological control of reproduction in urban and forest-dwelling European blackbirds *Turdus merula*. *Journal of Avian Biology*, **36**(4), pp. 295–305.
- Pataki, D.E., Alig, R.J., Fung, A.S., Golubiewski, N.E., Kennedy, C.A., McPherson, E.G., Nowak, D.J., Pouyat, R. V. and Romero Lankao, P. (2006) Urban ecosystems and the North American carbon cycle. *Global Change Biology*, **12**(11), pp. 2092–2102.
- Pauleit, S. and Duhme, F. (2000) Assessing the environmental performance of land cover types for urban planning. *Landscape and Urban Planning*, **52**(1), pp. 1–20.
- Pauleit, S., Ennos, R. and Golding, Y. (2005) Modeling the environmental impacts of urban land use and land cover change – a study in Merseyside, UK. *Landscape and Urban Planning*, 71(2-4), pp. 295–310.
- Pickett, S.T.A. and Cadenasso, M.L. (2006) Advancing urban ecological studies: Frameworks, concepts, and results from the Baltimore Ecosystem Study. *Austral Ecology*, **31**(2), pp. 114–125.
- Pickett, S.T.A., Cadenasso, M.L., Grove, J.M., Nilon, C.H., Pouyat, R.V., Zipperer, W.C. and Costanza, R. (2001) Urban ecological systems: linking terrestrial ecological, physical, and socioeconomic components of metropolitan areas. *Annual Reviews in Ecology and Systematics*, 32, pp. 127–157.
- Pouyat, R.V., McDonnell, M.J. and Pickett, S.T.A. (1997) Litter decomposition and nitrogen mineralization in oak stands along an urban-rural land use gradient. *Urban Ecosystems*, 1(2), pp. 117–131.
- Pouyat, R.V., Yesilonis, I. and Nowak, D.J. (2006) Carbon storage by urban soils in the USA. *Journal of Environmental Quality*, 35(4), pp. 1566–1575.
- Pressey, R.L. Possingham, H.P. and Day, J.R. (1997) Effectiveness of alternative heuristic algorithms for identifying indicative minimum requirements for conservation reserves. *Biological Conservation*, **80**(2), pp. 207–219.
- Quigley, M.F. (2004) Street trees and rural conspecifics: Will long-lived trees reach full size in urban conditions? *Urban Ecosystems*, **7**(1), pp. 29–39.
- Rebele, F. (1994) Urban ecology and special features of urban ecosystems. *Global Ecology and Biogeography Letters*, **4**(6), pp. 173–187.
- Rowntree, R.A. and Nowak, D. (1991) Quantifying the role of urban forests in removing atmospheric carbon dioxide. *Journal of Arboriculture*, **17**(1), pp. 269–275.
- Roy, D.B., Hill, M.O. and Rothery, P. (1999) Effects of urban land cover on the local species pool in Britain. *Ecography*, 22(5), pp. 507–515.

- Shochat, E. (2004) Credit or debit? Resource input changes population dynamics of city-slicker birds. *Oikos*, **106**(3), pp. 622–626.
- Smith, R.M., Gaston, K.J., Warren, P.H. and Thompson, K. (2005) Urban domestic gardens (V): relationships between landcover composition, housing and landscape. *Landscape Ecology*, 20(2), pp. 235–253.
- Smith, R.M., Warren, P.H., Thompson, K. and Gaston, K.J. (2006a) Urban domestic gardens (VI): environmental correlates of invertebrate species richness. *Biodiversity and Conservation*, 15(8), pp. 2415–2438.
- Smith, R.M., Gaston, K.J., Warren, P.H. and Thompson, K. (2006b) Urban domestic gardens (VIII): environmental correlates of invertebrate abundance. *Biodiversity and Conservation*, 15(8), pp. 2515–2545.
- Smith, R.M., Thompson, K., Hodgson, J.G., Warren, P.H. and Gaston, K.J. (2006c) Urban domestic gardens (IX): Composition and richness of the vascular plant flora, and implications for native biodiversity. *Biological Conservation*, **129**(3), pp. 312–322.
- Stanners, D. and Bourdeau, P. (1995) The Urban Environment, in *Europe's Environment: The Dobříš Assessment*, (eds. D. Stanners and P. Bourdeau), European Environment Agency, Copenhagen.
- Stigsdotter, U. A. & Grahn P. (2004) A Garden at your doorstep may reduce stress: Private gardens as restorative environments in the city. Proceedings of the Open Space: People Space Conference, 27-29 October 2004, Edinburgh, Scotland. Paper 00015.
- Stone, B. Jr. and Rodgers, M.O. (2001) Urban form and thermal efficiency. *Journal of the American Planning Association*, **67**(2), pp. 186–198.
- Sukopp, H. and Wurzel, A. (2002) The effects of climate change on the vegetation of central European cities. *Urban Habitats*, **1**, pp. 66–86.
- Thorington, K.K. and Bowman, R. (2003) Predation rate on artificial nests increases with human housing density in suburban habitats. *Ecography*, **26**(2), pp. 188–196.
- Tratalos, J., Fuller, R.A., Warren, P.H., Davies, R.G. and Gaston, K.J. (2007a) Urban form, biodiversity potential and ecosystem services. *Landscape and Urban Planning*, 83(4), pp. 308– 317.
- Tratalos, J., Fuller, R.A., Evans, K.L., Davies, R.G., Newson, S.E., Greenwood, J.J.D. and Gaston, K.J. (2007b) Bird densities are associated with household densities. *Global Change Biology*, 13(8), pp. 1685–1695.
- Ulrich, R.S., Simons, R.F., Losito, B.D., Fiorito, E., Miles, M.A. and Zelson, M. (1991) Stress recovery during exposure to natural and urban environments. Journal of *Environmental Psychology*, **11**(3), pp. 201–230.
- Warren, P.S., Katti, M., Ermann, M. and Brazel, A. (2006) Urban bioacoustics: It's not just noise. *Animal Behaviour*, 71(3), pp. 491–502.
- Watson, J.E.M., Whittaker, R.J. and Freudenberger, D. (2005) Bird community responses to habitat fragmentation: how consistent are they across landscapes? *Journal of Biogeography*, **32**(8), pp. 1353–1370.
- Whitford, V., Ennos, A.R. and Handley, J.F. (2001) "City form and natural process" indicators for the ecological performance of urban areas and their application to Merseyside, UK. *Landscape* and Urban Planning, 57(2), pp. 91–103.
- Wittig, R. (2004) The origin and development of the urban flora of Central Europe. Urban *Ecosystems*, **7**(4), pp. 323–339.
- Yli-Pelkonen, V. and Niemelä, J. (2005) Linking ecological and social systems in cities: Urban planning in Finland as a case. *Biodiversity and Conservation*, 14(8), pp. 1947–1967.
- Zhang, X.Y., Friedl, M.A., Schaaf, C.B. and Strahler, A.H. (2004a) Climate controls on vegetation phenological patterns in northern mid- and high latitudes inferred from MODIS data. *Global Change Biology*, **10**(7), pp. 1133–1145.
- Zhang, X., Friedl, M.A., Strahler, A.H. and Schneider, A. (2004b) The footprint of urban climates on vegetation phenology. *Geophysical Research Letters*, 31(12), L12209.

Chapter 5 Social Acceptability

Glen Bramley, Caroline Brown, Nicola Dempsey, Sinead Power and David Watkins

Introduction

Urban forms cannot be considered 'sustainable' in the full sense if they are not acceptable to people as places to live, work and interact. This chapter focuses on the relationship between urban form and *social sustainability*, and has four main aims. First, drawing upon a wide-ranging literature we advance and clarify an understanding of social sustainability that allows us to explore possible links between social sustainability and urban form. There has hitherto been a lack of clear and agreed definitions for this concept and we hope our approach contributes to a better shared understanding. The second aim is to address ways of measuring social sustainability and testing/quantifying some of the hypothesized relationships between selected dimensions of social sustainability and urban form. The measures used draw primarily on the household survey of case study neighbourhoods, linked to urban form measures as described in Chapter 2. The third aim of the chapter is to examine the empirical relationships between aspects of social sustainability and different aspects of urban form, particularly density, housing type and location. This analysis highlights the importance of controlling for exogenous and intervening variables, such as housing tenure and the social composition of neighbourhoods, in testing and calibrating these relationships. The final aim is to offer some more insights into how and why some urban forms may provide more beneficial social outcomes for different groups of people, based on qualitative focus group evidence.

Defining Social Sustainability

Debates within the wider sustainable development literature have long moved beyond considering sustainability solely as an environmental concern, to

G. Bramley (⊠)

Centre for Research into Socially Inclusive Services (CRSIS), School of the Built Environment, Heriot-Watt University, Edinburgh, UK

include economic and social dimensions. A significant proportion of sustainable development rhetoric now stresses the importance of social equity (Burton, 2000a cites: CEC, 1990; Sherlock, 1990; Yiftachel and Hedgcock, 1993; Elkin et al., 1991; Selman 1996). The importance of a social dimension of sustainability has also been underlined at policy level. DETR (1997) stressed the need to empower all sections of the community to participate in decision making and to consider the social and community impacts of decisions, while DETR (2001) sought to give social progress the same emphasis as economic and environmental objectives. UK policy parallels international sustainability development agreements, with poverty eradication seen as an essential requirement (UN, 2002).

Whilst the sustainable development agenda emphasizes the importance of 'social' aspects of sustainability there has been little agreement as to what this constitutes. Polese and Stren (2000: 15–16) have put forward the following definition of social sustainability:

Development (and/or growth) that is compatible with harmonious evolution of civil society, fostering an environment conducive to the compatible cohabitation of culturally and socially diverse groups while at the same time encouraging social integration, with improvements in the quality of life for all segments of the population.

The above definition discusses social sustainability in terms of the collective functioning of society as well as in terms of individual quality of life issues. Yiftachel and Hedgcock (1993:140) have further defined *urban* social sustainability as:

the continuing ability of a city to function as a long-term, viable setting for human interaction, communication and cultural development.

Arguing that the sustainability debate has largely ignored social aspects, they propose an analytical framework that delineates between three key dimensions of 'urban' social sustainability: equity; (sense of) community; and urbanity. Again, within this definition we see the emphasis on the need for a city to function as an interactive unit as well as on equity issues. However, the concept of 'urbanity' is not clearly defined.

The UK Government has made efforts to link sustainability to quality of life, particularly through the indicators known as 'Quality of Life Counts' (DETR, 1999b, 2001). While it is not self-evident that sustainability and quality of life are synonymous, nevertheless, quality of life may provide the bridge between the 'urbanity' mentioned above and the 'liveability' agenda which has emerged in UK urban policy (ODPM 2003; DCLG 2006). In practice, the DETR (2001) document seeks to reconcile these obvious tensions by adopting an extended economic concept of 'capital'. This identifies three broad classes of 'capital' – economic, social and environmental – and defines sustainable development as a process of growth/development in which total capital (the combination of these three kinds) is 'non-decreasing' – i.e. we are not squandering the proverbial family silver. The notion of husbanding different kinds of 'capital' is one way of expressing the Brundtland (WCED, 1987) philosophy of inter-generational equity.

5 Social Acceptability

That sustainability, including its social dimension, is central to current planning and urban policies in the UK is evidenced by the title of the overarching policy document known as the Sustainable Communities Plan (ODPM, 2003). The most recent policy statement by HM Government (2005, Annex A), subsequently adopted by the EU as 'The Bristol Accord', set out its policy definition of 'What is a Sustainable Community' in terms of eight headings including: 1. active, inclusive and safe; 2. well served; 3. well designed and built; 4. well run; 5. environmentally sensitive; 6. well connected; 7. thriving; and 8. fair for everyone. We would regard this all-embracing definition as encompassing all three main 'legs' of the broader sustainability concept, including the environmental dimension (in 5.), the economic dimension (in 7.), transport (in 6.), and matters of governance (in 4.) and planning/design (in 3.) that go beyond the usual outcome focus into means rather than ends. The aspects of social sustainability with which we are concerned are located particularly within headings 1, 2, 3 and 8. The first heading includes 'identity and belonging', 'tolerance of difference', 'friendly and cooperative communities', leisure and cultural opportunities, crime/anti-social behaviour, and 'a good quality of life'. The second heading identifies a range of accessible services including health, education and social care. The third reiterates issues around sense of place, friendliness, healthiness and safety of spaces, as well as accessibility by non-motorised transport and the issue of affordable housing. Although this appears as something of a long shopping list of desirables without a transparent logical coherence, if nothing else this document serves to underline the apparent policy priority given to the issues addressed in this chapter.

Social Capital, Social Cohesion and Social Inclusion

Whilst there is a relatively limited literature that focuses specifically on social sustainability, there is a much broader literature on the overlapping concepts of social capital, social cohesion and social exclusion. It is beyond the scope of this chapter to provide an adequate review of these concepts as they are discussed in this literature. We recognize that they are not very tightly or consistently defined and also that they are, in important respects, contested concepts. Table 5.1 simply identifies some of the key elements of each concept as exemplified in representative examples of recent literature reviewing them.

It may be argued that the underlying premise to all of these concepts is that individuals within society need to work together and interact in order for societies to be socially 'sustained'. *Social networks* clearly provide a common thread between the three concepts as set out here. These concepts recognize the importance of people being involved and having a vested interest in society, as well as individuals having equal access to societal benefits. However, beyond this it may be observed that the third concept (social exclusion) is more distinct, focusing more on access to economic opportunities and services, whilst the first and second concepts have more areas of overlap. In addition to social networks and association, one can also

Table 5.1 Comparing Concepts of Social Capital, Social Cohesion and Social Exclusion

Social capital

'Social capital refers to features of social organisation such as networks, norms and trust that facilitate co-ordination, and co-operation for mutual benefit.' (Putnam, 1993: 35)

Participation
Common purpose
Reciprocity
Trust
Belonging (Forrest and Kearns, 2001)

Social cohesion

'Social cohesion can emphasis the need for a shared sense of morality and common purpose; aspects of social control and social order; the threat to social solidarity of income and wealth inequalities between people, groups and places; the level of social interaction within communities or families; and a sense of belonging to place' (Forrest and Kearns, 2001: 2128).

Suggested elements Common values and civic culture Social order and social control Social solidarity and reductions in wealth disparities Social networks and social capital Territorial belonging (Kearns and Forrest, 2000)

Social exclusion

'Social exclusion is a process that deprives individuals and families, groups and neighbours of the resources required for participation in the social, economic and political activity of society as a whole. This process is primarily a consequence of poverty and low income, but other factors such as discrimination, low educational attainment and depleted living environments also underpin it. Through this process people are cut off for a significant period in their lives from institutions and services, social networks and development opportunities that the great majority of a society enjoys' (Pierson, 2002: 7).

Suggested elements Poverty and low income Lack of access to jobs Lack of social support and networks Effect of the local area Exclusion from services (Pierson, 2002)

discern common clusterings of concern with *norms*, *values and culture*, with *sense of belonging* (to place), and with *safety and trust* which may be seen as the positive side of social control and order.

A Working Definition of Social Sustainability

From the above review of the literature there are two recognisable, overarching concepts at the core of the notion of social sustainability within an area context. These are *social equity issues* (access to services, facilities and opportunities) and issues to do with the *sustainability of community itself*. Whilst social equity issues are powerful political and policy concerns, and centre upon a distributive notion of social justice – that is 'fairness in the apportionment of resources in society' (Burton, 2000a: 1970) – the more collective 'sustainability of community'

dimension may be seen as more nebulous. However, we would argue that this clearly maps onto the concerns of both government and academic writers, particularly those addressing issues of social capital and cohesion. In exploring social sustainability at the neighbourhood level both of these dimensions need to be covered.

With regard to the first dimension, we are particularly interested in access to local services, while recognising that a fuller account of the equity dimension would also encompass access to jobs and affordable housing. In practice, we focus in this chapter on a limited representative selection of services (e.g. convenience shopping, primary healthcare) although in our wider study we will examine a broader range (see for example Fisher and Bramley (2006) analysis of poverty and local services).

Turning to the second dimension, and drawing on the above review of social sustainability and related concepts in both academic and policy literature, we argue that the following aspects are likely to be significant in helping to sustain communities at neighbourhood level:

- Interaction with other residents/social networks.
- Participation in collective community activities.
- Pride/sense of place
- Residential stability (versus turnover).
- Security (lack of crime and disorder)

Individually these dimensions tap into a number of interesting debates within urban policy. The variable 'interaction in the community' may be related to the social mix agenda. This emphasises that it is not just achieving a mix of characteristics of population within an area that matters, but whether people *actually* personally interact with their neighbours (Atkinson and Kintrea, 2001). Work in this area distinguishes 'strong' and 'weak' social ties, but suggests that both forms are of positive value. Informal social ties may be distinguished from active participation in formal community activities/organisations, frequently used as an indicator of local social capital; we would argue that both are potentially significant.

We are also interested in whether people use facilities within their neighbourhood, and their attitudes towards these facilities. The social premise is that if people participate in activities within their local community then they will have stronger ties to the community (this is distinct from the environmental benefit of reduced travel). A similar argument applies to the inclusion of the concept of pride/sense of place, the idea being that, if people feel attached to the neighbourhood, they will want to stay living in the area and contribute to its continued development (Woolever, 1992). The fourth aspect of sustainability of community is residential stability; within some of the literature, areas of high turnover are perceived to be unsettled and undesirable areas, although this is not always be the case (Bramley et al., 2000; Bramley and Morgan, 2003, Bailey and Livingstone, forthcoming). High outflows of residents combined with low or no inflows can mean that an urban community will be literally unsustainable over time (Power and Mumford, 1999; Bramley et al., 2000). Community stability is often associated with higher levels of social cohesion and associated benefits such as lower crime (Hirschfield and Bowers, 1997).

With regard to the final dimension UK government policy is increasingly stressing the need and ability of communities to combat crime for themselves (ODPM, 2004; Atkinson and Flint, 2003). This issue has long been connected to urban form (through the 'design and crime' literature). Shaftoe (2000: 230) argues that,

community safety is an essential prerequisite for a stable and sustainable neighbourhood' with crime and fear of victimization being 'two of the top deleterious ingredients of urban living.

The Relationship Between Social Sustainability and Urban Form

Jenks et al. (1996, p.11) described the relationship between urban form and sustainability as one of the most hotly debated issues on the international environmental agenda. There has been a move away from focusing solely on environmental dimensions of sustainability to consider other dimensions including 'urban sustainability' and 'social sustainability' (CEC, 1990; Knight, 1996; Darlow, 1996; HM Government, 2005). What emerges from this review of the literature is that there are competing claims regarding the extent to which urban form influences social sustainability; claims and debates that have, to date, rarely been supported by empirical evidence (Jenks et al., 1996). Of the elements of urban form which might be considered, density is the one that has received the most attention in the literature with regard to its social impact. Much of this focus has been upon the policy question of whether we should contain the spatial extent of urban development by developing at higher densities or whether we should allow the spatial extension of urban areas and build at lower densities - the 'compact city' versus 'sprawl' debate (Breheny 1992a, 1992b; Ewing, 1997; DETR, 1999a; Barton 2000) and in the related 'new urbanism' literature (CNU, 2004; Katz, 1994; Calthorpe, 1993).

The density of urban development has the potential to impact upon all of the dimensions of social sustainability. For example, higher densities may make access to services and facilities both easier and more economically viable (Bunker, 1985; Collie, 1990; Haughton and Hunter, 1994; Burton, 2000b). Williams (2000) found that access may vary for different services. ODPM (2003) argues that particular densities are needed to support basic amenities in the neighbourhood and to minimize the use of resources such as land. Burton (2000a; 2000b) has produced (arguably) the most comprehensive work exploring the impact of urban form on social equity. Burton (2000b) found that nearly all of the 14 social equity effects that she identified are related in some way to urban compactness; job accessibility and wealth being the exceptions. For medium-sized English cities she found that higher urban densities may be positive for some aspects of social equity and negative for others. We revisit some of these equity-related issues of service access in the empirical work below.

Higher densities may also mean that people are more likely to meet each other on the street than in lower density areas (Talen, 1999; Duany and Plater-Zyberk, 2001). In contrast lower densities reduce the potential for spontaneous interaction and leads to an orientation towards car travel (TCRP, 1998). Glynn (1981) and Nasar and Julian (1995) both found 'sense of community' to be higher in neighbourhoods that facilitated face-to-face interaction. There are, however, alternative arguments that in higher density societies, people may withdraw from social contact. Wirth (1938) argued that high density living, along with the anonymity of city life, leads to an increase in stress and the severing of traditional ties that result in a decline in community or social ties. Bridge (2002:4) refers to Simmel's (1995) discussion of the 'psychic over-stimulation' of the city'. In this way higher densities may lead to weaker social ties. There is an argument that, whilst very low densities may undermine social ties, at some point further up the scale higher densities may start to have the same effect (Freeman, 2001). We return later to this hint of non-linear relationships. It is argued that in a compact city, that is a city with high-density and mixed uses, communities are likely to be more mixed, and that as such there is likely to be a lower level of social segregation. Suburban sprawl in particular has come to be associated with high levels of segregation and inner city decay (CEC, 1990; Burton, 2000a; Bramley and Morgan, 2003). However, it is not axiomatic that social mix correlates with density or use mix in this way; in our empirical work below we distinguish these factors.

The density of development may also affect the appearance and aesthetics of places, and hence people's sense of attachment to and pride in place. The TCRP (1998) review found that there is little evidence within the literature to suggest that Americans find sprawl less attractive than more compact forms of development, although they do cite work by Nelessen (1994), Shore (1995) and Diamond and Noonan (1996) which argues that lower density development is less aesthetically pleasing. There is also an argument that low density developments can be more attractive (Audirac and Zifou, 1989). Gordon and Richardson (1997) argue that given the choice people prefer low-density suburban living to high-density urban living. They note that many consumer preference surveys have shown a strong preference for suburban living (findings which are echoed, in some senses, in our own empirical findings reported below).

We can see from this brief review of the literature that the discussions on the relationship between density and social sustainability are quite complex, with at times contradictory hypotheses or findings. There are reasons to expect access to services to be better in denser urban forms, while quality of neighbourhood environment, community and social interaction may be less good in denser areas. The latter relationship is less clear-cut a priori from the literature, and could well vary contingent upon the social/demographic groups considered and interactions between urban form and social composition factors, including those associated with housing tenure. Further, there is a dearth of analysis at a small scale local level.

Measuring Social Sustainability

Our empirical investigation of social sustainability is mainly focused on the fifteen case study neighbourhoods described in Chapter 2, areas chosen to reflect a diversity

of typical British urban forms with varying ages, types and tenures of housing and socio-demographic profiles. In our view the principal source of evidence on the social acceptability of different urban forms should be people themselves, particularly those living in the areas in question, and so the primary data source is the household survey in these neighbourhoods. The household survey responses are re-weighted to reflect the underlying demographic profile of the population of each neighbourhood based on the Census, so countering possible differences in response rates between demographic groups.

In general we do not rely on a single question within the survey to provide evidence on a given aspect of social sustainability, but rather draw on responses to a cluster of questions. In grouping responses together for the main composite social outcome measures reported below, both their logical/linguistic interpretation ('face validity') are considered but also the patterns of correlations between the responses across our sample.

The survey generates eight composite measures capturing the different aspects of social sustainability as defined above:

- Pride and attachment
- Interaction
- Safety
- Environment
- Satisfaction with home
- Stability vs mobility
- Participation in collective/group activity
- Use neighbourhood facilities/services

Although the emphasis in this listing appears to be primarily on the 'sustainability of community' aspect, the last heading clearly relates to equity of access and can be subdivided into types of service/facility, for example everyday utility services vs. cultural and recreational services. It can also be argued that satisfaction with home and quality of local environment are aspects of equity.

With our focus here on the big picture we concentrate on summary composite measures based on answers to multiple questions. These outcome measures have been expressed in an index form which is subject to commonsense interpretation. Taking the example of 'social interaction', this is based on responses to 13 questions, such as whether they have friends in neighbourhood, see them frequently, know neighbours by name, look out for each other, chat, borrow things, etc. For each question, negative responses reduce the score from a neutral value, while positive responses increase it (with neutral responses being possible in all cases). If someone gave all neutral responses scores 200; if all responses were negative then the score is zero. The resulting index scores are numbers in the range 0–200, but are typically around or just above 100, for individuals. These are effectively continuous variables and we can therefore compare mean values and variations between and within groups, areas or area-types.

5 Social Acceptability

To establish relationships between these outcomes and urban form, we rely on linking the location of sample households (addresses) to information about those locations, particularly information about the small areas within which people reside. It should be noted at this point that the emphasis here is on urban form of the *residential neighbourhood* rather than, for example, the form of the place where people work, shop or carry out their recreation. We have a choice about the spatial *level* at which urban form characteristics might be attributed to individuals. For the analyses reported in this chapter we rely mainly on linkage at the level of the 'sub-area' (see Chapter 2).

Census and other neighbourhood data (including use of OS Mastermap) are apportioned to these spatial units. These data encompass a wide range of socio-demographic characteristics, physical urban form (density, dwelling type, storey height), proportions/ratios of land area attributable to different elements (e.g. residential buildings, gardens, greenspace, roads), other urban form and quality measures derived from the site survey, along with simple measures of access distance to city centres and more sophisticated network connectivity measures based on MCA. Other socio-economic data attached includes components from the Indices of Multiple Deprivation (IMD) and house prices.

In parallel with analyzing our own purpose-designed survey, we carried out analysis of certain large scale government surveys, which ask a more limited range of relevant questions, again linking sample locations to urban form and socio-demographic data for small neighbourhoods. This helps to benchmark our more selective case study results and check that these are not unrepresentative. We make brief reference below to results from this exercise based on the Survey of English Housing (see also Bramley and Power, 2005).

A key issue in interpreting evidence on the association between social outcomes and urban form is that of untangling 'real' and potentially 'causal' associations from what may be apparent, fortuitous or ambiguous relationships, given the complexities of urban life and the different kinds of relationships which may be at work. Simply showing that, in a two way table or correlation, there is an apparent (negative) relationship between, say, density and neighbourhood satisfaction/attachment, does not establish that there really is such a relationship, let alone that it is causal. It is essential to take account of ('control for') the influence of other relevant variables, i.e. other physical, social, economic or demographic factors which we have reason to believe may also influence neighbourhood satisfaction. For example, older people may tend to answer satisfaction surveys in more positive ways than younger people, and at the same time older people may be more likely to live in low density suburbs. To take another example, neighbourhoods with concentrations of poor people tend to exhibit a greater incidence of certain social problems which affect neighbourhood satisfaction; such neighbourhoods are also often higher density areas.

So although we do report simple descriptive tables showing patterns between different forms, we rely more in our main analysis and conclusions on statistical modeling. Multiple regression analysis provides a convenient and flexible tool for establishing the direction and strength of relationships while simultaneously taking account of the relationships with other variables.

G. Bramley et al.

Quantitative Relationships

Descriptive Patterns

Table 5.2 shows the simple pattern of scores on our eight composite outcome measures by broad location and density categories. For six of our eight outcomes there is a clear pattern of higher scores in outer areas and lower scores in inner areas. The exceptions are 'safety' where the between areas actually scored highest, and 'use of neighbourhood services/facilities' where inner areas scored highest.

The lower part of the table shows the pattern of scores by broad bands of gross residential density. For four of the outcome measures there is a clear pattern, whereby the highest score is for the lowest density band and the lowest score for the highest density band. This applies to pride/attachment, interaction, safety and home satisfaction. For environmental quality the highest score is for the second density band, and the same pattern applies to participation in groups. For the stability/mobility criterion, both the lower density bands have a similar higher score, while it is the medium-higher density band (40–70) which shows the lowest score. Not unexpectedly, use of neighbourhood services shows a different pattern with density, although this is not a simple reversal of the pattern shown for pride/attachment and the other social/community indicators. In fact the highest score is in the medium-higher density band, with lower scores in both the highest and the medium-low density.

Clearly, we could add further tables of this kind, looking at different aspects of urban form, for example house type, and subdividing our sample according to different demographic groups. However, this would take up a lot of space, and is in any case a less effective way of looking at the issue of different urban form effects and demographic effects than considering the results of multiple regression modeling.

Regression Modelling Results

As explained earlier, the key value of using statistical techniques such as multiple regression analysis is that they can identify relationships of outcomes with urban form factors, while taking account of other socio-demographic factors which also influence the measured outcomes and which may confound the apparent urban form effects. An additional advantage of the modelling is that it can also highlight the differential effect of different urban form features. Ordinary least squares regression is considered suitable for these data because the dependent variables (composite outcome scores) are continuous variables with approximately normal distributions.

The regression models typically 'explain' around 25% of the variance in individual scores, with a higher figure for 'stability/mobility' (36%) but a particularly low figure for participation in groups (8%). Relatively low proportions of variance explained are typical of models fitted to individual data such as these,

Sub-area type location	Pride & attachment	Interaction	Safety	Environment	Satisfaction home	Stable/mobile	Participation groups	Use NH services
Inner	92.5	97.3	123.4	100.4	117.3	109.7	94.2	108.5
Middle	115.7	122.5	133.0	120.2	130.9	127.6	101.5	103.0
Outer	118.4	129.1	126.9	129.8	139.7	147.1	104.6	88.0
Total	108.5	115.9	127.6	116.4	129.0	127.7	100.0	100.0
Density								
<20 DPH	116.3	118.4	134.4	120.6	139.0	132.4	105.4	102.4
20-40 DPH	112.7	116.4	131.4	121.1	131.3	132.1	106.0	98.2
40-70 DPH	109.0	117.3	127.2	113.9	125.7	121.0	97.5	105.7
>70 DPH	97.3	111.7	119.1	111.2	123.6	128.1	92.5	92.7
Total	108.5	115.9	127.6	116.4	129.0	127.7	100.0	6.66

Table 5.2 Social Sustainability Outcome by Location and Density

5 Social Acceptability

because there is a large element of unmeasured and random individual variation in preferences, behaviour and circumstances. Nevertheless, given the large samples the models capture general relationships reasonably well.

Table 5.3 shows the detailed pattern of impact of urban form and related variables on our eight outcomes. Individual socio-demographic control variables included in the models (typically between 12 and 20 such variables per model) are omitted from this table. These factors, including for example age, gender, ethnicity, household type, income and tenure, are expected to influence behaviour, experience and preferences, and we do not want these to confuse or confound the effects of urban form. The effects are shown by the regression coefficients, which show the impact of one unit increase in that particular variable on the outcome in question. Since our outcomes are all indexes centred on 100, most of these coefficients can be readily interpreted. For example, the first figure in the second column suggests that residents of terraced houses would have a 3.4 point higher interaction score, other things being equal. Figures shown in bold are coefficients which are statistically significant at the 10% level (ie they are 90% or more likely to be different from zero in the direction shown). Non-bold figures are coefficients which fall short of that level but where there are some signs, from a range of tests, of a relationship.

Urban form variables in the top half of the table are attributes of individual households/houses and generally take the dummy form (1 = yes, 0 = no). Variables below the line are attributes of our 97 sub-areas and are generally continuous measures, either percentages, logs of distances or density ratios.

We now comment briefly on some of the findings which this table reveals, starting with house types. Bungalows are good for home satisfaction but bad for use of local services. Detached houses are similarly good for home satisfaction but possibly less good for safety. Terraced homes appear better for interaction but less good for home satisfaction and environmental quality. Flats are associated with less stability/more mobility, unsurprisingly; slightly more surprising is the finding that people living above the fifth floor are more attached to their neighbourhood and feel safer.

'IMD Geog Access' basically measures more rural and isolated locations (within an essentially urban sample), and this is associated with more attachment and environmental quality but less use of local services. Living further from a major city centre is associated with lower scores on attachment, interaction, environment and service use. Net dwelling density has generally negative relationships with social outcomes and this is significant in four cases: attachment, safety, stability and local services (the last is slightly surprising, but perhaps suggests that location is more important for services than net density). The squared term in the model for stability indicates a non-linear relationship. The proportion of dwellings over 4 storeys is positively associated with interaction and participation, but negatively with environmental quality.

Having an individual private garden or yard is quite strongly positive across most outcomes including pride, interaction, environment and home satisfaction. This is somewhat reinforced by the positive effect of gardens as a percentage of land area on safety and of average garden size on pride and home satisfaction. At the same time other forms of greenspace are also positive for pride, interaction and participation.

	Table 5	3.3 Individual in	npacts of Urban for	rm variables o	n social sustainabi	lity outcomes		
Regression Coeffic's (signif 10% level)	Pride and attachment	Interaction	Safety (crime)	Environ quality	Home satisfaction	Stability	Group participation	Local services
Individual Level Bungalow					6.12			-7.28
Detached house			-5.27	-3.80	7.06	-4.39		2
Terrace	2.03	3.41		-4.67	-6.46			-3.23
Flat						-7.50		
floor5	10.77		8.27				-18.08	6.25
Garden	4.28	2.27		3.02	5.91			
no garden/yard/terrace	-8.85	-14.11	2.32	-5.05	-9.84	-3.71	-12.36	-2.94
above ground floor		-3.53			2.09		-7.23	-2.28
Sub-area level								
IMD geog access -	4.16			7.14			-1.86	-3.17
Log dist major centre	-5.27	-3.35		-6.66				-6.63
Gross dwelling								
density				-0.03				
Net dwelling density	-0.04	-0.01	-0.04		-0.01	-0.05	-0.02	-0.03
Net density squared						0.02		
Dwellings detached %	0.05		0.20					
Dwellings Terrace %	-0.05	0.13	-0.13		-0.08			-0.14
Over 4 storeys %	0.03	0.08		-0.13			0.15	-0.09
Gardens % area			0.24				0.38	-0.77
Average Garden size	389.88				307.32			
Greenspace %		0.12	0.10		0.05		0.44	-0.50
Greenspace absence	0.55							
Garden + Green %				0.05				
Non-Residential Mix	-0.05		-0.20	-0.29		-0.12	0.80	-0.14
Minutes to bus stop- Frequency of buses								-3.03 -4.39

Mixed uses appear to be positive for group participation but negative for safety and environment, without being significant for use of local services. Being further from the nearest bus stop reduces the use of local services, but so also does having more frequent buses (perhaps enabling more use of central city facilities).

Most of these individual effects make sense in terms of our prior expectations and reading of the previous literature. This gives further confidence that these models are capturing the main urban form effects, which may have a degree of complexity about them.

Summarizing the Urban form Relationships

The models as just described underline that a number of aspects of urban form vary between neighbourhoods and appear to have differing effects on different outcomes. Can we try to summarize these effects across the broad spectrum, from 'compact central city' to more 'dispersed suburban' areas, without doing injustice to that complexity? We believe we can do this, by generating a composite function, for each sample household, which measures all of the urban form related effects as revealed by the coefficients in Table 5.3. We then take density (net residential) as a convenient summary scale measure, and chart the relationship between each composite function and density. We also show for comparison the raw scores. Four of the eight outcome indicators are selected to illustrate the varying patterns found.

Figure 5.1 looks at raw scores and 'urban form effects' for neighbourhood pride and attachment, which is it should be noted is the best overall summary measure among the eight outcome indicators. The raw scores fall unsteadily, from about 150 to under 100, as densities rise to 150 dwellings per hectare (DPH), after which they fluctuate somewhat. The urban form effect falls from about 140 to a plateau of around 112 at 100 DPH, with a slight further fall to about 107 at 200 DPH. In other words, the modelled urban form effect is less strong than the apparent effect from the raw scores, but is clearly still present.



Fig. 5.1 Neighbourhood pride and attachment by net density – overall and Urban form effects



Fig. 5.2 Neighbourhood interaction by net density – overall and Urban form effects

Figure 5.2 looks at the neighbourhood interaction outcome score in the same way. This produces perhaps the most interesting finding of this investigation. Although the raw interaction scores fall as densities rise, over most of the range, the modelled urban form effect rises with increasing density (compactness) up to the level of around 120 DPH (net). This reflects the effect of variables such as terraced housing and storey height as reported above. It also makes sense intuitively and in terms of hypotheses derived from the literature. In sparser suburbs people are less likely to bump into each other, partly because they are more likely to get around by car. In terraced housing or lower flatted areas people are more likely to meet coming and going. In very high density, this is less likely to be the case, or the way in which people meet is less likely to encourage people to get to know each other.

Figure 5.3 looks at home satisfaction. This shows a stronger negative effect of increasing density, up to about 150 DPH. While the urban form effect is slightly less strong than the raw score pattern, it is still quite a strong relationship



Fig. 5.3 Satisfaction with home by net density – overall and Urban form effects



Fig. 5.4 Use of neighbourhood facilities by net density - overall and Urban form effects

Finally, in Fig. 5.4 we look at use of local services. Here both the raw scores, and more particularly the urban form effect, show the opposite pattern, as expected, of rising with density, at least up to the level of 150 DPH.

Other Area Effects

In understanding the difference between what we have termed urban form effects and the overall differences in social outcomes between different types of area, it is worth focusing briefly on some of the other factors which vary between neighbourhoods and which could affect these outcomes. It is convenient to illustrate this point by drawing on an analysis, similar to that just described, undertaken using a national survey dataset, the Survey of English Housing (SEH), linked to neighbourhood characteristics in a similar way. This also serves to illustrate the point that patterns found in our five case study cities are representative of wider national patterns. A more detailed treatment of this analysis may be found in Bramley and Power (2005).

Figure 5.5 reports raw scores for 'dissatisfaction with neighbourhood', essentially the inverse of our 'pride/attachment' factor, and decomposes variations into the parts which can be attributed (on the basis of a regression model) to demography, social factors, urban form and access (the latter treated separately here, rather than being combined as in Figs. 5.1, 5.2, 5.3, and 5.4). As with our survey results, area dissatisfaction increases with density up to the level of about 80 DPH (gross). What Fig. 5.5 shows is that all of the components contribute to the adverse satisfaction scores at higher densities. Even demography is not neutral, and the access-dissatisfaction component is slightly worse at higher density. The urban form effect (shown in red) is adverse up to 80 DPH, but the slope of this effect is much less than that of the raw score. The really striking finding



Fig. 5.5 Components of area dissatisfaction by density

is that, across England as a whole, the most important contributor to the higher dissatisfaction of higher density areas is the social component. This comprises IMD neighbourhood poverty and related indicators. The national analysis based on SEH. generally indicates that poverty is more important, for dissatisfaction and common neighbourhood problems, than is urban form per se.

This leads us on to a broader conclusion, which is that 'who lives where' may be as or more important than the physical form of the neighbourhood. Furthermore, this pattern of dissatisfaction may be related to choice, which in turn relates to poverty. Poorer households have less choice in the housing market, often reliant on social rented housing which has been traditionally rationed and allocated administratively.

Qualitative Findings: Use of Focus Groups

To complement the data collected in the household questionnaire survey, focus groups were held with residents in nine of the fifteen case study neighbourhoods. Their purpose was to gather in-depth qualitative data about the links between urban form and particular dimensions of social sustainability. The aim was to build up a clearer picture of how people use their local environment and to get a sense of what the local environment means to them. In doing so, it was hoped that the focus groups would help to explain some of the findings of the questionnaire survey. For example, what is it about a particular street or local environment which makes people feel unsafe? Why is it that particular people use services and facilities in a neighbourhood more than others? Why is it that people feel a stronger sense of pride in their local area than others? In addition, the focus groups were also a way of exploring those dimensions of social sustainability which were not addressed in the household questionnaire survey.

Crudely put, the questionnaire asks respondents what they do in the neighbourhood, and the focus group asks why they behave in such a way and how they feel while in the neighbourhood (Bryman, 2001; Krueger 1994). Since these two research methods complement one another (Miles and Huberman, 1994) it is possible to triangulate the resulting data to strengthen the research findings. Another issue was the ability of focus groups to capture *collective* experiences of the participants' neighbourhood.

Local Services and Spaces

The use of local services and facilities was found to be influenced by urban form in terms of their location in relation to residents, which also affected the method of transport used and frequency of use. Those services closer to home were more likely to be reached on foot or bike, and those further away by car. There was no indication that participants necessarily used those services (e.g. supermarket) closest to home as, for some, factors such as product quality were more important than convenience. Some participants also reported using services and facilities en route to and around the workplace (see Chapter 3). However, having accessible key services within the neighbourhood was highlighted as very important for different groups of residents such as the unemployed, older people and young families. Throughout all of the discussions, dissatisfaction was expressed at the closure of local services such as post offices and specialist shops such as butchers and greengrocers. These findings have implications for policy, highlighting the importance for residents of good quality, easily accessible services and facilities in the neighbourhood. A further finding, which should be the basis of future empirical research, shows that there may be potential for local supermarkets (with café) to act as a hub for social interaction, being used as a meeting place by older participants in one particular case study.

The reported *use of, and access to, open space* was, on the whole, high and satisfactory across the different demographic groups living in the different locations. Two important factors affected participants' use of public open space which related to perceived safety and maintenance. Respondents were less likely to report using open spaces if they perceived them to be unsafe. They were also less likely to feel comfortable using public open spaces if they were not well-maintained. The responsibility for maintaining open spaces was recognized to be two-fold, lying both with the user and the local authority, which is supported elsewhere (CABE Space, 2005a, 2005b).

The discussion groups also reported on access to, and use of, shared open spaces which were provided for residents in higher density housing types such as tenements and blocks of flats, but were often not maintained. The findings show that this was cited as a reason for non-use, along with a general perceived lack of comfort and, to some extent, privacy, when using the communal space. This finding has important implications for policy and requires further empirical examination to ascertain if shared open spaces are the most suitable use of space to provide residents with areas in which they feel safe and comfortable. A further finding indicates that formal arrangements for maintaining and managing shared open spaces are more successful than informal collective action on the part of residents. There is a need to empirically examine this further to determine if there is a need to, for example, engage more formally resident participation in the management of shared open space.

A number of features of urban form were found to influence *feelings of safety*. While the household questionnaire findings showed that respondents reported feeling safer the further they lived from the centre, this association did not emerge in the focus group discussions. The maintenance of open spaces was strongly highlighted as an important influence on perceived safety, indicating that secluded, overgrown and poorly maintained spaces were less likely to be used. Characteristics of streets were also influential on feelings of safety and alley-ways and streets that were not overlooked by residences made some participants feel less safe when moving around the neighbourhood. The speed and volume of road traffic also had negative effects on feelings of safety, particularly for the safety of children. A significant non-physical influence on participants' feelings of safety was anti-social behaviour, by children and teenagers among others. Participants were often quick to point out however that this may be resolved (to some degree) by giving young people a place to go and something to do other than hanging around on streets, a well-cited argument (Margo, 2007; Institute for Public Policy Research, 2006).

Community and Attachment

Community stability and *sense of place attachment* were found to be influenced by a number of physical features, including one's accommodation and its location in relation to services/facilities, public transport and the city centre. Other influences include feelings of satisfaction towards the neighbourhood. Older participants were less likely to report a desire to move house than younger participants and those with families, the latter groups citing a need for more space, a garden and a quieter place to live among their reasons. The main non-physical reason given for staying in an area was not being able to afford to move to more desired areas; however friendliness, organized activities and sense of community were also more positive reasons given. There was a sense throughout the case studies that, while some participants may not be living in the ideal place for them, the neighbourhoods functioned well, fulfilling residents' requirements to a considerable extent and therefore constituting a good compromise. The implications for policy here have therefore already been described, in terms of the provision of good services, facilities and open spaces in neighbourhoods.

The household questionnaire found that *social interaction* and *social networks* tended to be stronger the further away from the city centre respondents lived. The focus group findings did not reflect this tendency nor did social interaction seem to be stronger among any one demographic group. Physical features which positively supported social interaction included the physical layout of housing

(in tenements however this was reported as a barrier to interaction), services and facilities including schools and shops, and bus stops. Other, non-physical, influences included one's children, organized groups in the neighbourhood, friendliness in an area, the propensity of neighbours to interact socially, and housing tenure. This latter point echoes other findings in that longer-term residents, rather than more transient ones, are more likely to interact and forge social networks in the neighbourhood.

Conclusions

The social dimension of sustainability has become increasingly prominent but there is still a lack of coherence and shared understanding of the concept. Its main dimensions, we believe, are equity and community, and it embraces issues of social inclusion, social capital and social cohesion. In an urban context, social equity is particularly concerned with access to services and opportunities. A community which is sustainable displays higher levels of what some would term social capital and/or social cohesion – pride in and attachment to place, social interaction, safety/trust and stability – and is likely to offer its residents a good 'quality of life', with high levels of satisfaction with home and neighbourhood and an appreciation of the local environment.

So, we would argue that social sustainability is meaningful, policy-relevant and arguably important. Social sustainability represents both public/collective goods and some key drivers of individual private choice. The social 'bottom line' is that people will not choose to move to or remain in a neighbourhood which does not promote these qualities to a degree. We would also argue that social sustainability saves public costs, promotes happiness, and can contribute to the kind of urban vitality which underpins modern economic competitiveness.

For most aspects of social sustainability, (particularly pride/attachment, stability, neighbourhood and home satisfaction and perceived environmental quality) lower density suburbs appear 'best'; so the social perspective somewhat challenges 'compact city' orthodoxy. Some aspects are neutral (e.g. participation), some favour more compact forms, particularly access to services; social interaction is best at medium densities. In order to isolate the effects of urban form it is necessary, through statistical modelling or other means, to control for the effects of other socio-demographic factors. In general the disadvantages of compactness are more marginal once you control for these influences. Poverty is often more important than urban form – who lives where, and whether they choose, matters. Management of urban (public) space rather than design also emerges as an important issue.

References

Atkinson, R. and Flint, J. (2003) *Locating the local in informal processes of social control: the defended neighbourhood and informal crime management*, CNR Paper 10. Bristol University, School of Policy Studies, Centre for Neighbourhood Research.

- Atkinson, R. and Kintrea, K. (2001) Disentangling neighbourhood effects: evidence from deprived and non-deprived neighbourhoods. *Urban Studies* **38**(11), pp. 2277–98.
- Audirac, I. and Zifou, M. (1989) Urban development issues: What is controversial in urban sprawl? An annotated bibliography of often overlooked sources, Council of Planning Librarians, Monticello, IL.
- Bailey. N. and Livingstone, M. (2007) *Population Turnover and Area Deprivation*, Research Report to Joseph Rowntree Foundation, JRF, York.
- Barton, H. (ed) (2000) Sustainable Communities: the potential for Eco-Neighbourhoods, Earthscan, London.
- Barton, H., Grant, M. and Guise, R. (2003) *Shaping Neighbourhoods: a guide for health, sustainability and vitality, Spon Press, London.*
- Berg, B. L. (2004) Qualitative Research Methods for the Social Sciences, Pearson International, Boston, Massachusetts.
- Bloor, M., Frankland, J., Thomas, M. and Robson, K. (2001) *Focus Groups in Social Research*, SAGE Publications, London.
- Bramley, G., Brown, C., Dempsey, N. and Power, S. (2007) Social Sustainability and Urban Form: measuring and calibrating the relationship. Paper presented at *Planning Research Conference*, Heriot-Watt University, Edinburgh, April 2007.
- Bramley, G., Dempsey, N., Power, S. and Brown, C. (2006) What is 'social sustainability', and how do our existing urban forms perform in nurturing it? Paper presented at *Global Places, Local Spaces Planning Research Conference 2006*, Bartlett School of Planning, University College London, 5th-7th April 2006.
- Bramley, G. and Power, S. (2005) Urban Form and Social Sustainability: the role of density and housing type. Paper presented at *European Network for Housing Research*, Reykjavic, Iceland, June 2005.
- Bramley, G. and Morgan, J. (2003) Building competitiveness and cohesion: The role of new house building in central Scotland's cities. *Housing Studies*, 18(4), pp. 447–471.
- Bramley, G., Pawson, H. and Third, H. (2000) Low Demand Housing and Unpopular Neighbourhoods. DETR, London.
- Breheny, M. (1992a) Sustainable development and urban form, Pion, London.
- Breheny, M. (1992b) The contradictions of the compact city: a review. In Sustainable development and urban form, (ed. M. Breheny), Pion, London, pp. 138–159.
- Bridge, G. (2002) *The neighbourhood and social networks*, CNR, Paper 4, April 2002, www.neighbourhoodcentre.org.uk
- Brook Lyndhurst (2004) Sustainable Cities and the Ageing Society: the role of older people in an urban renaissance, Office of the Deputy Prime Minister, London.
- Bryman, A. (2001) Social Research Methods, Oxford University Press, Oxford.
- Bunker, R. (1985) Urban consolidation and Australian cities. Built Environment, 11, pp. 83–96
- Burton, E. (2000a) The compact city: Just or just compact? A preliminary analysis. *Urban Studies*, **37**(11), pp. 1969–2001.
- Burton, E. (2000b) The potential of the compact city for promoting social equity. In Achieving Sustainable Urban Form, (eds. K. Williams, L. Burton, and M. Jenks), E & FN Spon, London.
- Burton, E. (2003) Housing for an urban renaissance: implications for social equity. *Housing Studies*, **18**(4), pp. 537–562.
- Commission for Architecture and the Built Environment (CABE) (2005a) *Decent parks? Decent behaviour? the link between the quality of parks and user behaviour;* CABE Space, London.
- Commission for Architecture and the Built Environment (CABE) (2005b) Parks Need Parkforce: a report on the people who work in our urban parks and green spaces, CABE Space, London.
- Commission for Architecture and the Built Environment (CABE) (2005c) *Start With the Park: creating sustainable urban green spaces in areas of housing growth and renewal*, CABE Space, London.
- Calthorpe, P. (1993) *The Next American Metropolis: Ecology, Community and the American Dream*, Princeton Architectural Press, New York

Collie, M. (1990) The case for urban consolidation. Australian Planner, 28, pp. 26-33.

- Commission of the European Communities (CEC) (1990) *Green paper on the urban environment*, CEC, Brussels.
- Congress of the New Urbanism (CNU) (2004) Charter of New Urbanism, (http://www.cnu.org/about/index.cfm)
- Darlow, A. (1996) Cultural policy and urban sustainability: making a missing link? *Planning Practice and Research*, 11(3), pp. 291–301.
- Dempsey, N. (2006) *The Influence of the Quality of the Built Environment on Social Cohesion in English Neighbourhoods.* Unpublished PhD thesis, Oxford Brookes University, Oxford.
- Department for Communities and Local Government (DCLG) (2007) Survey of English Housing (SEH) Live Tables, DCLG, London.
- Department for Communities and Local Government (DCLG) (2006) *State of the English Cities*, Research Summary 21, DCLG, London.
- Department of the Environment, Transport and Regions (DETR) (1997) *Indicators of Sustainable Development*, DETR, London.
- Department of the Environment, Transport and Regions (DETR) (1999a) Towards an Urban Renaissance Final Report of the Urban Task Force, E and FN Spon, London.
- Department of the Environment, Transport and Regions (DETR) (1999b) A Better Quality of Life, DETR, London.
- Department of the Environment, Transport and Regions (DETR) (2001) Achieving a Better Quality of Life: Review of Progress Towards Sustainable Development, Government Annual Report 2000, HMSO, London.
- Diamond, H. L. and Noonan, P. F. (1996) Land Use in America, Island Press, Washington DC.
- Duany, A. and Plater-Zyberk, E. (2001) *The rise of sprawl and the decline of the American dream*, North Point Press, Florida.
- Elkin, T., McLaren, D. and Hillman, M. (1991) *Reviving the city: towards sustainable urban development*, Friends of the Earth, London.
- English Heritage, Sport England and The Countryside Agency (2003) *The Use of Public Parks in England 2003*, Sport England, London.
- Ewing, R. (1997) Is Los Angeles style sprawl desirable? *Journal of the American Planning Association*, **63**(1), pp. 107–126.
- Fisher, T. and Bramley, G. (2006) Poverty and local services. In *Poverty and Social Exclusion in Britain*, (eds. C. Pantazis, D. Gordon and R. Levitas), Policy Press, Bristol.
- Forrest, R. and Kearns, A. (2001) Social cohesion, social capital and the neighbourhood. *Urban Studies*, **38**(12), pp. 2125–2143.
- Freeman, L. (2001) The effects of sprawl on neighbourhood social ties. *Journal of the American Planning Association*, **67**(1), pp. 69–77.
- Glynn, T. (1981) Psychological sense of community: Measurement and application. *Human Relations*, **34**, pp. 789–818.
- Gordon, P. and Richardson, H. (1997) Are compact cities a desirable planning goal? *Journal of the American Planning Association*, **63**(1), pp. 95–106.
- Groves, R. M. and Couper, M. P. (1998) Nonresponse in household interview surveys, Wiley, New York.
- H M Government, (2005) Securing the Future, Cm 6467, The Stationary Office, London.
- Harvey, D. (1989) The Urban Experience, Blackwell, Oxford.
- Haughton, G. and Hunter, C. (1994) Sustainable Cities, Jessica Kingsley Publishers, London.
- Hills, J., Le Grand, J. and Piachaud, D. (2002) *Understanding Social Exclusion*. Oxford University Press, Oxford.
- Hirschfield, A. and Bowers, K. (1997) The effects of social cohesion on levels of recorded crime in disadvantaged areas. *Urban Studies*, 34(8), pp. 1275–1295.
- Institute for Public Policy Research (2006) Childhood is changing, but 'paedophobia' makes things worse. IPPR. [Online]. Retrieved on 26th July 2007 from: http://www.ippr.org.uk/ pressreleases/?id=2388

- Jenks, M. (2000) The Acceptability of Urban Intensification. In Achieving Sustainable Urban Form, (eds. K. Williams, E. Burton and M. Jenks), E & FN Spon, London, pp. 242–250.
- Jenks, M., Burton, E. and Williams, K. (1996) *The Compact City: a sustainable urban form?* E & FN Spon, London.
- Jenks, M. and Dempsey, N. (2007) Defining the Neighbourhood: challenges for empirical research. *Town Planning Review*, 78(2), pp. 153–177.
- Katz, P. (1994) The new urbanism: Toward an architecture of community, McGraw-Hill, New York.
- Kearns, A. and Forrest, R. (2000) Social cohesion and multilevel urban governance. Urban Studies, 37(5-6), pp. 995–1017.
- Kearns, A. and Turok, I. (2003) Sustainable communities: dimensions and challenges, Urban and Neighbourhood Studies Research Network.
- Knight, C. (1996) Economic and Social Issues In *The Compact City: A Sustainable Urban Form?* (eds. M. Jenks, E. Burton and K. Williams), E & FN Spon, London, pp. 114–121.
- Krueger, R. A. (1994) Focus Groups: a practical guide for applied research, SAGE Publications, Thousand Oaks, CA.
- Krueger, R. A. (1998) Analyzing and Reporting Focus Group Results, SAGE Publications, Thousand Oaks, CA.
- Laurie, H. (2006) Lessons for Census Enumeration from Survey Research. Paper presented at *Hard* to Count? Royal Statistical Society Seminar, London, 4th December 2006.
- MacDonald, C. and Bramley, G. (2006) Cityform Household Survey Methodology. Unpublished working paper, Heriot-Watt University, Edinburgh.
- Margo, J. (2007) Gordon's Plan to Keep the Kids Under Control. *The Sunday Times*, 8th July 2007, p. 27.
- Miles, M. B. and Huberman, A. M. (1994) *Qualitative Data Analysis*, SAGE Publications, Thousand Oaks, CA.
- Nasar, J. and Julian, D. (1995) The psychological sense of community in the neighbourhood. *Journal of the American Planning Association*, 61, pp. 178–184.
- Nelessen, A. C. (1994) Visions for a New American Dream: process, principles and an ordinance to plan and design small urban communities, American Planning Association, Chicago.
- Office of the Deputy Prime Minister (ODPM) (2003) Sustainable communities: building for the future, HMSO, London.
- Office of the Deputy Prime Minister (ODPM) (2004) Safer Places: The planning system and crime prevention, ODPM, London.
- Pierson, J. (2002) Tackling social exclusion, Routledge, London.
- Pantazis, C., Gordon, D. and Levitas, R. (2006) Poverty and Social Exclusion in Britain, Policy Press, Bristol.
- Polese, M. and Stren, R. (2000) *The social sustainability of cities: diversity and management of change*, University of Toronto Press, Toronto.
- Power, A. and Mumford, K. (1999) *The slow death of great cities? Urban abandonment or urban renaissance, JRF/York Publishing Services, York.*
- Putnam, R.D. (1993) Making Democracy Work: Civic traditions in modern Italy, Princeton University Press, Princeton, NJ.
- Rees, J. (1988) Social polarisation in shopping patterns: an example from Swansea. *Planning Practice and Research*, **6**, pp. 5–12.
- Royal Commission on Environmental Pollution (2007) *The Urban Environment*, The Stationery Office, London.
- Rydin, Y. (2003) Rationalities of planning: Development versus environment in planning for housing, Urban Studies, 40(10), pp. 2099–2101.
- Selman, P. (1996) *Local Sustainability: Managing and planning ecologically sound places*, Paul Chapman Publishing, London.
- Shaftoe, H. (2000) Community safety and actual neighbourhoods. In *Sustainable communities: The potential for eco-neighbourhoods*, (ed. H. Barton), Earthscan Publications, London.

- Sherlock, H. (1990) Cities are good for use: the case for close knit communities, local shops and public transport, Palodin, London.
- Shore, W. B. (1995) 'Recentralization': the single answer to more than a dozen United States problems and a major answer to poverty. *Journal of the American Planning Association*, **61**(4), pp. 496–503.
- Simmel, G. (1995) The metropolis and mental life. In *Metropolis: Center and Symbol of our times*, (ed. P Kasnitz), Macmillan, Basingstoke, pp. 30–45
- Talen, E. (1999) Sense of community and neighbourhood form: an assessment of the social doctrine of new urbanism. Urban Studies, 36(8), pp. 1361–1379.
- Transit Cooperative Research Program (TCRP) (1998) *The costs of sprawl revisited, Report 39,* Transportation Research Board, National Research Council.
- Tunstall, R and Lupton, R. (2003) Is targeting deprived areas an effective means to reach poor people? An assessment of one rationale for area-based funding programme, CASE Paper 70, June 2003, CASE, LSE.
- United Nations (2002) Report of the world summit on sustainable development, Johannesburg, 26th August-4th September 2002.
- Urban Task Force (1999) Towards an Urban Renaissance, E & F Spon, London.
- Whitehead, M. (2003) (Re) Analysing the sustainable city: Nature, urbanisation and the regulation of socio-environmental relations in the UK. Urban Studies, 40(7), pp. 1183–1206.
- Williams, K. (2000) Does Intensifying Cities Make Them More Sustainable? In Achieving Sustainable Urban Form, (eds. K. Williams, L. Burton, and M. Jenks), E & FN Spon, London.
- Wirth, L. (1938) Urbanism as a way of life. *American Journal of Sociology*, 44, pp. 1–24.
 Woolever, C. (1992) A contextual approach to neighbourhood attachment. *Urban Studies*, 29(1), pp. 99–116.
- World Commission on Environmental Development (1987) Our common future, Oxford University Press, Oxford.
- Yiftachel, O and Hedgcock, D. (1993) Urban social sustainability: The planning of an Australian city. *Cities*, **10**(2), pp. 139–157.

Chapter 6 Energy Use

Keith Baker, Kevin J. Lomas and Mark Rylatt

Introduction

A high proportion of the energy consumption of cities is linked to buildings. In the UK for example it is estimated that energy use in buildings is responsible for around 29% of all energy consumption compared with 37% for transport (Fig. 6.1). This fact is a major driver in sustainability policies in the UK with the government expecting the construction sector and the planning system to deliver all new housing by 2016 that produces zero net emissions of carbon dioxide from all energy use in the home.

There are a series of potential relationships between domestic energy consumption and the influence of urban form in terms of residential density, layout etc. Density is linked to the type of housing so if we all lived in high density terraced houses/flats what would be the impact on energy consumption or is it to do with house size? Urban form also has an impact on the lifestyles of households and this too could potentially influence energy consumption for example through home working.

This chapter focuses on energy consumption in dwellings, and seeks to examine the determinants of energy consumption by reference to the characteristics of the housing stock, urban form and lifestyle factors. The chapter begins by examining the relationship between urban form and energy use. It then considers the state of domestic energy modelling noting the insufficiency of the use of permanent building physical characteristics such as dwelling type and age, and data limitations on occupancy characteristics and behaviour relating to patterns of heating and appliance use and less permanent aspects of the building fabric. The next section sets out the research methods of the study and how these issues are addressed, and the main forms of statistical analysis involved are explained. In the concluding sections results from the analysis are presented with some observations on the significance of urban form.

K. Baker (⊠)

Scottish Institute for Sustainable Technology (SISTECH), Institute of Energy and Sustainable Development, De Montfort University, Leicester, UK



Energy Use and Urban Form

Unlike other topics in this book research into the relationships between building energy use and broad measures of urban form, such as density, have received limited attention. Usually such variables have been regarded as merely of minor statistical interest; hence the literature is slight. A rare recent example of a study that seeks evidence for more complex relationships is provided by Larivière and Lafrance (1999) but no significant relationship was found between different densities and per capita electricity consumption; indeed they recommend a cautious view of non-trivial urban energy use and population density relationship claims. Ratti et al. (2005) discern a relationship between overall energy consumption and global urban form characteristics but this is based on a city scale morphological categorisation using a novel form of image processing derived from digital elevation models rather than conventional measures and, as an isolated study, cannot be considered conclusive.

More generally, studies of urban form and energy prove to be at the level of what may be termed the local aspects of urban form: the physical form of buildings such as the size of their footprint and curtilage, the number of storeys, and their dwelling type, such as detached, semi-detached, terraced, etc. (e.g. Perkins, 2002). Such aspects do of course relate to measures of density and morphology and the energy consumption and efficiency of buildings is undoubtedly related to their physical form, given that buildings are never perfectly insulated. The influence on energy consumption due to fabric heat loss has long been understood and, *ceteris paribus*, is potentially a useful basis for prediction and comparison (Martin and March, 1972; Yannas, 1994).

The effects of solar gain on heating energy consumption and of overshading on lighting energy consumption are also related to built form and density but are much harder to model reliably. It is possible to crudely estimate the trade-off between gain and loss that might occur under different building density scenarios and to infer that there must be a point beyond which the advantages of reduced heat loss from a higher proportion of dwellings with party walls begin to be outweighed by the reductions in solar gain and illuminance resulting from over shading in closely massed developments (Steemers, 2003). However, crucially, as heat loss standards improve, the benefits of medium to high densities will be far less significant in this respect, while the disadvantages of very high densities and the lack of natural light, though mitigated, will remain.

In the UK the government's zero carbon policy for new build, if successful will mean very low heat loss for all types of buildings, including detached dwellings, and so issues of heat loss abatement related to density of new development will be minimised. For different reasons solar gain and overshading related to density are also unlikely to be a significant factor in new planning guidelines and regulations. This is because although non-statutory guidelines exist for site layouts that optimise passive solar gain (e.g. Littlefair, 1995) and, as such, these encourage medium to low density developments, the trend in this direction observable in the UK from 1970 onwards has since been reversed under Planning Policy Guidance 3 (DETR, 2000), which set a minimum density of 30 dwellings per hectare. It is worth adding that although these regulations have now been superseded by Planning Policy Statement 3 (DCLG, 2006) and the minimum density regulation will no longer be strictly enforced at the national level, it is unlikely that there will be a significant relaxation despite recent signs of green belt policy change to meet the UK housing crisis.

The determinants of energy use are not simply determined by the nature of the built form as it is recognised that life style and the dynamic effects of occupant behaviour, in particular appliance use, and of retrofitted energy efficiency measures, are likely to obscure static physical influences. These are discussed in the next section.

Modelling Domestic Energy Consumption

Domestic energy models used for prediction on a medium to large scale are subject to considerable uncertainty as they commonly operate on just a few data relating to permanent building physical characteristics such as dwelling type and age, and sets of default data and assumptions about occupancy characteristics and behaviour relating to patterns of heating and appliance use and less permanent aspects of the building fabric. Even where full datasets are available such models internally are commonly quite simplistic in their treatment of occupant behaviour related energy consumption. There has for example been criticism of the UK Building Research Establishment Domestic Energy Model (BREDEM) (see Anderson et al., 2002 for a description of the latest version) on these grounds. The detailed influences raised by these issues are now considered.

Fabric

Particularly for older dwellings, any differences in energy consumption that might be predicted on the basis of the building regulations in force at the time of construction, are likely to be masked by changes to the building fabric, which alter the dwelling's energy efficiency, e.g. installation of double glazing, draught proofing. This is particularly so in the UK where a large proportion of the building stock is over fifty years old. Much of this was originally relatively low grade accommodation built to house workers following the industrial revolution and is very likely to have been improved considerably at various times. These changes are for the most part not a matter of record.

Occupancy

Although the UK population increased by 4.1 million from 1970 to 2000, average household size decreased from 2.9 to 2.3 people and the total number of households increased by 6.4 million over the same period (Shorrock and Utley, 2003). Occupancy type is related to dwelling type which in turn is linked to tenure (Fig. 6.2) and so there are strong reasons for expecting a relationship with household energy consumption. Changing patterns of ownership and especially the targeting of energy efficiency schemes may also be very significant. As illustrated by Fig. 6.3, owner-occupancy has shown a steady increase since 1970, and although the number of dwellings rented from local authorities has declined some of this decrease is accounted for by ownership switching to registered social landlords and housing associations. The number of privately rented properties, which fell from 1970 to a low point in 1988, has subsequently shown to be increasing again. This



Fig. 6.2 UK dwellings by dwelling type and tenure, 2000 Source: Survey of English Housing, Department of the Environment, Transport and the Regions



Fig. 6.3 UK Housing trends by Tenure Source: Communities and Local Government

may well be a reflection of an increasingly mobile workforce and a greater number of households comprised of students and non-related young professionals.

It is to be expected that owner-occupiers, having a substantial investment in their properties will have more interest in and direct control over changes relating to energy efficiency. Local authority and housing association properties are likely to influence overall energy consumption to some degree by implementing energy efficiency improvements to comply with regulations and agreements such as Local Agenda 21 in the UK. However, privately rented homes are likely to benefit least from improvement schemes owing to the fragmented nature of this sector in the UK. Tenants are unlikely to invest in anything other than basic energy efficiency measures due to due to typically short tenancy periods, while landlords are currently not incentivised to make costly improvements just to save money for their tenants.

Use of Household Appliances

Changing patterns of appliance use represent an additional source of uncertainty. As shown in Fig. 6.4, the amount of energy used to light our homes and power appliances increased significantly between 1970 and 2001. The total amount of energy consumed by space heating has changed little over this period, with 2001 representing an unusual high point, and the overall proportion has remained fairly constant at between 60 and 65%. Energy used for water heating has followed a similar trend, averaging around 25% of total consumption over the period. However,



Fig. 6.4 Domestic energy consumption per household: by final use Source: National Statistics Online

over the same period, significant changes appear to have occurred in the amount of energy used for cooking and lighting and appliances. The amount of energy used for cooking has shown a consistent and steady decline, falling from 6% of total consumption in 1970 to under 3% in 2001, while energy used for lighting and appliances has more than doubled in real terms, rising from 7% to just under 13% of total consumption, despite a rapid rise in the use of energy efficient light bulbs for domestic lighting and the introduction and expansion of energy efficiency labelling for major appliances from 1995 onwards.

It is probable that the increase in the number of households with falling occupancy levels together with a trend towards more background and "mood" lighting has led to more light sources and appliances being shared by fewer individuals. Consumers are purchasing greater numbers of appliances, particularly high-tech goods, and whilst some, such as microwaves and LCD televisions, reduce the usage of other appliances or replace less efficient technologies, others will introduce additional energy demands. One example of the latter is the impact from digital, cable and satellite TV boxes being left on standby rather than being switched off because of poor design and/or unacceptable reboot times.

Research Method

To fully understand all these sources of variation in energy consumption between dwellings requires behavioural studies with intensive observation and recording, and care to avoid observer effects. Additionally, direct observation of attitudes and behaviour risks introduces propagation errors through inaccurate reporting and contextual bias attributable to respondents. The alternative approach adopted in this study factors was less invasive, using methods that focus on the manifested effects of behaviour attributable to ownership and use of types and numbers of lights and major domestic appliances and physical measures taken to reduce consumption such as improving dwelling insulation, rather than reported behaviour such as switching off lights and appliances when not in use.

Selection of Samples

Dwellings were selected to provide, as far as possible, homogeneous samples of dwelling types (e.g. neighbourhoods of exclusively detached houses). A geographical information system was used to identify the most promising areas for this approach within the fifteen case study areas. This would enable between subgroups studies while controlling for built form on a consistent basis. The response rate was expected to be low for a very detailed postal survey on energy issues as the necessary questions are generally unappealing to many potential respondents. The study was based originally on the selection for survey of 1674 terraced and semi-detached dwellings in Leicester (Fig. 6.5), 2083 detached or



Fig. 6.5 Map of Clarendon Park, Leicester. City Form study area is in colour, the sub-area selected for the energy study is in red


Fig. 6.6 Map of Fulwood, Sheffield case study area is in colour, the dwellings selected for the energy study are in pink (detached dwellings) and blue (semis)

semi-detached dwellings in Sheffield (Fig. 6.6) and tenements and flats in Glasgow which, for reason explained below could not in fact be analysed.

Design of Survey Instrument

Home energy efficiency questionnaires are conducted by pubic agencies in the UK to raise the householder awareness of inefficiencies in their energy consumption. These questionnaires request data for dwelling type and age, number of storeys and rooms, wall and roof type, loft insulation, window frame and glazing type, main and secondary heating type and extent, water heating and cylinder insulation size, cooker type and quantity of low energy lighting. They may also include dimensional data in the form of a floor plan. However in the study survey this was replaced by data acquired from a geographical information system (GIS), which, in combination with returned storey data, enabled reasonable estimates of floor area to be made automatically for all the dwellings. The scope of the study survey was, however, expanded to include more information on occupancy and household appliances to support the aim of investigating the principle sources of uncertainty described above.

A pilot survey was used to assess the responses both to individual questions and the survey as a whole. This confirmed that a very low response rate was likely and so modifications were made where questions were answered unsatisfactorily. Even so the final response rate with signed energy consumption data release permissions averaged only 6% in Leicester and Sheffield, which was considered just adequate for meaningful analysis. Unfortunately postal delivery and other problems in Glasgow produced a sample for tenements and flats well below the size considered acceptable and the study was therefore restricted to detached, semi-detached and terraced dwelling types.

Energy Consumption of Individual Properties

This data was enhanced by energy consumption for individual properties. Disaggregated energy consumption data has rarely been available in the UK to enable detailed investigation. Although this problem is less severe in some other countries, for example Canada, often the building stock is not comparable (Aydinalp-Koksal et al., 2008). The original intention was to approach individual suppliers, of which there are many in the UK, this proved futile. Fortunately the data was ultimately acquired from the Department of Business Enterprise and Regulatory Reform, a government department, which since 2004 has held energy consumption data for all UK households on a central database in order to provide statistics at various levels of aggregation.

Data Analysis Procedure

Cluster analysis is an exploratory statistical technique intended to reveal groupings in the data based on some measure of similarity. Such clusters may correspond to expected classes within the data and the statistically determined centres of the clusters can be used to investigate degrees of association with other aspects. A form of this analysis was applied to the combined gas and electricity consumption data. It was of particular interest to see whether any clusters would centre on dwelling type in the full data sample (i.e. all the responses from Sheffield and Leicester). Selected response variables, using appropriate banding for continuous data, were then cross tabulated to show the strength of any significant associations with the cluster centres.

Simple regression can also be used as an explanatory technique to investigate the strength of linear relationships between one independent variable with a dependent variable, such as energy consumption. This approach is used to explore the relationships between energy consumption and total floor area for smaller subsamples of data categorised by response type (for example the type of wall insulation or the presence of various forms of heating system controls).

Multiple regression analysis is then used to explain the effects of a number of independent variables on the dependent variable, expressed as r^2 , the coefficient of determination (e.g. by convention an r^2 of .25 indicates that the regression explains

25% of the variance observed around the mean). It is the basis for a confirmatory phase of analysis in which the variables initially identified are subject to the rigour of model building: with care, variables can be added to the regression to build a model with increasing explanatory power. The effects of interesting variables that showed relatively weak but still significant relationships in the exploratory phase are also tested.

The dataset contains categorical response variables that are mainly dichotomous and these were coded using dummy variables; multichotomous variables generally proved to have quasi-ordinal characteristics (for example single, double and triple glazing) and these were given integer representation (for example 1,2,3). After the addition of each new variable a record is kept of the change in the number of records and the associated change in key statistics (the significance of the F statistic change, and the r^2 , adjusted r^2 and *p*-values). These results are used to determine which variables produce the most statistically significant improvements to the relationships with electricity and gas consumption after carefully checking for any potentially problematic co-linearities (linear relationships between independent variables selected for prediction that might bias the results).

Results of Analysis

Three distinct energy consumption clusters corresponding to low medium and high energy consumption levels are identified from the cluster analysis of the combined electricity and gas consumption data for the whole sample. However, these are not centred on dwelling type as might have been expected. This gives an early indication that built form classification is likely to show at best only a very weak relationship with energy consumption. When the analysis is repeated with sub-samples by dwelling type only two distinct clusters are found. For the Leicester terraces subsample, local knowledge has confirmed that the clusters differentiate between smaller mid-terraces and end-terraces and larger mid-terraces. Cross tabulations based on the full sample do however show quite significant associations with total floor area, occupancy, dwelling age, number of rooms, number of bedrooms, and regular home working. Perhaps surprisingly, the number of bedrooms has the strongest association with the clusters in the full sample and in two of the dwelling type subsamples as Table 6.1 indicates.

The most significant associations found in the cluster analysis phase are confirmed by the regression analysis. Weak but still significant relationships between electricity consumption and storage heaters, TVs in use, showers per week, digiboxes in use, PCs in use, and portable electric heaters in use are reported in Table 6.2. However, the statistic for storage heaters appears to be leveraged by a small number of dwellings from the Leicester subsample with this form of heating. The relationship with the number of showers per week is not been observed in the subsample analyses but this is probably explained by differences in occupancy characteristics. The remaining variables are present in one or more tables for the

Response Variable	Statistic	Leicester	Sheffield Detached	Sheffield Semis	Combined dataset
Total Floor Area (20% bands)	ρ	-0.56	-0.24	-0.09	-0.49
	p-value	0.00	0.14	0.56	0.00
Occupancy	ρ	-0.57	-0.47	-0.28	-0.50
	p-value	0.00	0.002	0.06	0.00
Age	ρ	-0.10	0.33	0.27	-0.16
	p-value	0.56	0.04	0.07	0.08
No. Rooms	ρ	-0.55	-0.30	-0.30	-0.44
	p-value	0.00	0.06	0.04	0.00
No. Bedrooms	ρ	-0.63	-0.35	-0.55	-0.59
	p-value	0.00	0.03	0.00	0.00
Homeworking	ρ.	-0.37	-0.29	-0.25	-0.26
-	p-value	0.02	0.07	0.09	0.01

 Table 6.1 Correlations of key variables with the consumption clusters found within subsamples and the full sample

Table 6.2 Statistics for the variables found to be the best indicators of electricity consumption for the full sample

Response variables	No. of houses	r ²	Adjusted r ²	Significant. F Change
Total floor area, Total occupancy, Age, No. of Rooms, No. of Bedrooms	142	0.30	0.27	0.00
Homeworking	142	0.32	0.29	0.03
Main heating – Storage heaters	142	0.38	0.35	0.001
No. TVs in use	139	0.40	0.36	0.10
Showers per week	122	0.44	0.39	0.05
No. digiboxes in use	106	0.45	0.39	0.36
No. PCs in use	101	0.47	0.40	0.37
No. portable electic heaters	101	0.47	0.40	0.47

subsamples and their presence is consistent with the findings from the exploratory analyses. The supporting evidence given in Table 6.3 for these variables came from the correlations (the Spearman rank coefficient, ρ , was used as it does not require variables to be measured on ordinal scales) with the three energy consumption clusters discovered within the full sample.

Response variable	ρ	p-value
No. of TVs	-0.36	0.00
No. of digiboxes	-0.97	0.04
No. of portable electric heaters	-0.16	0.07
	Response variable No. of TVs No. of digiboxes No. of portable electric heaters	Response variableρNo. of TVs-0.36No. of digiboxes-0.97No. of portable electric heaters-0.16

Adjusted r ²	Sig. F Change
0.42	0.00
0.44	0.03
0.46	0.05
0.46	0.17
0.47	0.38
0.47	0.46
	Adjusted r ² 0.42 0.44 0.46 0.46 0.46 0.47 0.47

Table 6.4 Relative strengths and significances of the relationships between gas consumption and the numbers of rooms and bedrooms after the addition of total floor area and occupancy to the multiple regression

A smaller number of significant variables were found for gas consumption than that for electricity consumption as indicated in Table 6.4. These findings are consistent with those for the sub-samples and conform with the prior expectation that differences in gas consumption should relate to a smaller range of factors as its uses in dwellings are less diverse than for electricity. The finding for glazing type again appears to be due to leverage by small numbers of dwellings reporting single or triple glazing. The initial phase of the analyis produces evidence of several strong relationships between gas consumption and total floor area when the subsamples are split according to wall insulation type.

Similar but weaker relationships are found when the subsamples are split by boiler type. Thermostats are one of three variables relating to the level of respondents' control over their heating systems, the others being the presence of thermostatic radiator valves (TRVs) and the type of controls (none, clock/timers, and digital controls). It is notable that within the combined dataset a strong and statistically significant relationship is found when gas consumption was regressed against total floor area for the group of 27 households with the highest level of control over their heating systems, i.e. those with TRVs, digital controls and thermostats (shown in Fig. 6.7).

Discussion of Key Findings

The analysis identifies the number of bedrooms as a key indicator of energy demand. This is rare in the literature but has been reported by Colton (1998) as a control variable for changes in occupancy between the time of collection of consumption data and the time of the survey. Although bedrooms are of course a subset of the total number of rooms there is not a straightforward, linear relationship. Trivially of course, the number of bedrooms variable is related to total floor area, although, as would be expected, total number of rooms is a better predictor of this. The variable is also clearly related to occupancy and has been used, for example, as a proxy for hot water use (Cyber Business Centre, 2002).



Fig. 6.7 Gas consumption regressed against total floor area for dwellings with TRVs, digital heating controls and thermostats

The predictive power of bedrooms may reflect changes in household behaviour that affect heating regimes: for example, the trend towards smaller households may, firstly, imply that specific rooms rather than entire dwellings may be heated and, secondly, that greater numbers of appliances will be moved into rooms traditionally used only for sleeping. The first possibility finds weak support in other results from this study which show some correlation between the number of portable electric heaters in use and the number of bedrooms. For the second possibility there is at least anecdotal support in reported trends towards more solitary living and the increased freedom of many employees to work from home.

As the questionnaire specified a bedroom as a room in which a bed is permanently set up, rather than whether it was regularly used as such, and as a significant number of respondents reported regularly working from home, it is likely that bedrooms in use as or doubling as home offices have influenced these results. The results also show a significant relationship between electricity consumption and the numbers of PCs and digiboxes in a household. This is quite weak but even so is perhaps stronger than expected given the limitations of the dataset and may be further evidence of the effects of these changes on lifestyle. Bedrooms doubling as home offices can also be expected to be heated during their periods of business use.

After total floor area, occupancy, dwelling age and rooms and bedrooms, the distinction between whether or not respondents reported regularly working from home was found to be the next strongest indicator of differences in gas and electricity consumption within the combined dataset, although the results were weaker within the separate samples. It was also found to be a determining variable for the consumption clusters. The significant trend towards home working is of relatively recent origin in the UK (Energy Star for Homes Website, 2007). Although

home working conditions are unregulated, home workers obviously need adequate warmth and lighting and usually some electrical equipment for their work, which will lead to increases in energy consumption.

It is recognised that this issue is complicated by the tradeoffs between transport use (which is certainly more strongly related to measures of urban form) and the relative efficiencies of energy provision at home and at work. These effects on domestic energy consumption have received surprisingly little attention to date, especially in the UK (Boardman et al., 2005). This study contributes to the debate by at least showing that these effects can be observed in a survey on this limited scale and suggests the potential for larger, perhaps longitudinal studies. It is likely that future predictions of domestic energy demand will need to account for changes in occupancy patterns, both in terms of time spent at home and which types of room are occupied most frequently.

Conclusions

The initial review of the relationships between energy consumption in buildings and urban form suggests that the principal association is likely to be via building forms. However, the determinants of energy use are not simply determined by the nature of the built form and life style and occupier behaviour, but also important are the ownership and use of appliances and the fitting of energy efficiency measures.

This chapter has sought to examine these relationships by a study of energy consumption in two different types of neighbourhoods based on the analysis of responses to an extensive energy questionnaire survey and associated annual consumption data. The commonly used broad classifications of types of dwelling have been used as the basis both for controlling analysis of selected subsets of data, on the assumption that they would exert significant effects on energy consumption and on the expectation that these would be observable to some extent despite the anticipated effects of other variables. However, although subsample analysis showed significant differences in key variables, overall analysis did not suggest that these classifications have a very significant effect on actual energy consumption.

These results cannot be regarded as conclusive, which is to be expected as these studies represent path-finding work with a new and small dataset. That the results conform to broad expectations in most respects is a finding of some values in this context and represents a good platform for further work, hopefully with the benefit of a much larger sample size. But beyond this, cluster analysis also revealed an interesting relationship with the number of bedrooms, which is of significance for density metrics, and also effects attributable to regular home working on energy consumption and to the use of modern technology are found to be important. The latter may also be related to transport energy use, which would also be expected to have a relationship with urban form, but it was not possible to investigate this possibility further.

References

- Anderson B.R., Chapman, P.F., Cutland, N.G., Dickson, C.M., Henderson, G., Henderson, J.H., Iles, P.J., Kosmina, L., Shorrock, L.D. (2002) *BREDEM-12: model description (2001 update)*, Building Research Establishment, Watford.
- Aydinalp-Koksal, M.V., Ugursal, I.V. and Fung, A.S. (2008) Comparison of neural network, conditional demand analysis, and engineering approaches for modelling end-use energy consumption in the residential sector, *Applied Energy*, 85(4), pp. 271–296.
- Boardman B, Darby S, Killip G, Hinnells M, Jardine CN, Palmer J, et al. (2005) 40% house, Environmental Change Institute, Oxford.
- Colton, R.D. (1998) *Determining household energy consumption in Washington State in the absence of 12 months of usage data*, Fisher, Sheehan and Colton, Public Finance and General Economics, 34 Warwick Road, Belmont, MA, USA.
- Cyber Business Centre (2002) *Go to work in your pyjamas*, Cyber Business Centre, Nottingham University, 11/06/02. Available online at: https://www.nottingham.ac.uk/cyber/tw-flex.html [last cited 23/02/07].
- Department for Communities and Local Government (DCLG) (2006) *Planning Policy Statement* 3: *Housing (PPS3)*, The Stationary Office, London.
- Department of the Environment, Transport and the Regions (DETR) (2000) *Planning Policy Guidance Note 3: Housing (PPG3)*, The Stationary Office, London.
- Energy Star for Homes website (2007), Available at http://www.energystar.gov/index.cfm?c=new_homes.nh_features.
- Larivièr, I. and Lafrance, G. (1999) Modelling the electricity consumption of cities: effect of urban density. *Energy Economics*, **21**, pp. 53–66.
- Littlefair, P.J. (1995) Site layout planning for daylight and sunlight: a guide to good practice, Building Research Establishment, Watford.
- Martin, L. and March, L. (eds.) (1972) Urban space and structures, Cambridge University Press, Cambridge.
- Perkins, A. (2002) *The Influence of Urban Form on Life Cycle Transport and Housing Energy and Greenhouse Gas Emissions,* School of Geoinformatics, Planning and Building, University of South Australia, Adelaide, Australia.
- Ratti C., Baker N., Steemers K. (2005) Energy consumption and urban texture. *Energy and Buildings*, **37**, pp. 762–77
- Sheehan, K.B. (2002) Online research methodology: reflections and speculations, [online]. Journal of Interactive Advertising, 3(1). Available at: http://jiad.org/vol3/no1/sheehan/ [last cited 22/10/06].
- Shorrock, L.D., & Utley, J.I. (2003) Domestic energy factfile 2003, [online]. BRE Housing Centre publication, Watford, UK. Available at: http://www.bre.co.uk/housing/page.jsp?id=396 [last cited 17/11/06].
- Steemers K. (2003) Energy and the city: density, buildings and transport. *Energy and Buildings*, **35**, pp. 3–14
- Yannas, S. (1994) Solar Energy and Housing Design. Volume 2: Examples, Architectural Association Publications, London.

Chapter 7 Economic Viability

Colin Jones, Chris Leishman, Charlie MacDonald, Allison Orr and David Watkins

Introduction

The arguments for sustainable cities are primarily couched in environmental and social rather than economic terms. Nevertheless the economics of cities is a central aspect of sustainability and a number of authors have endeavoured to incorporate economic arguments into the debate about cities. Lynch (1981), for example, preempts the Brundtland definition of sustainable development in the context of the urban economy. He sees the sustainability problem being one of enabling the urban economy to exist long into the future, whilst keeping resource use within levels that allow the earth's finite resources to provide indefinitely, alongside social equity. Inevitably with economics there has not been a consensus.

This chapter seeks first to consider the economics of city form by examining the micro-forces that shape the primary constituent elements set out in Chapter 2. This analysis provides a base for reviewing studies of the relationships between economic performance, city form and sustainability. Economic viability of land use patterns is argued as a prerequisite if spatial structures of cities are to evolve to be more sustainable and the chapter then applies this criterion to different land use sectors. Finally the chapter assesses the relationship between infrastructure costs and city form and draws some policy conclusions.

Economics of City Form

Cities are dynamic entities and their physical form is constantly changing to reflect the underlying economics of land use markets. This section considers each of the primary elements of urban form in turn and assesses not just the underlying economics that establishes these phenomena but also property market and development processes that drive and influence urban change.

C. Jones (⊠)

School of the Built Environment, Heriot-Watt University, Edinburgh, UK

Land Use Patterns

Specific spatial distributions of land use are crucial to the arguments about potential 'sustainable' urban forms and these are influenced by both supply and demand factors. There are two underlying substantive land use demand factors. The first of these is the spatial pattern of revenues and costs (see Alonso, 1964; Dunse and Jones, 2005). The demand for different land uses will depend on the relevant cost of using certain locations, and the revenues they will provide. Land that is deemed to be more productive for one use may not be considered so for another use. This depends on the suitability of the location to generate large revenues, and if a certain area is thought to generate large revenues for one or more land uses then it will be more sought after. Higher demand for a location is likely to result in higher land values. Typically the location attributes preferred by residential users are different to industrial demands, and these will be different to retail and commercial demands. Second, there are agglomeration benefits which arise from firms and shops locating together in the same place (Parr, 2002). These benefits are known as agglomeration economies (see later).

The (changing) supply of land and property has a number of underlying influences. Land availability offers opportunities/constraints to the adaptation of the land use pattern. New development or redevelopment or the conversion of existing buildings requires not only that it is perceived as profitable to property companies but also that there are funds made available from private investors to do so. The property market is very imperfect and both property companies and investors can be constrained by attitudes toward the risks associated with particular land uses and potential locations if they do not conform to established patterns (Jones and Watkins, 1996). For example investors in the UK have been traditionally reluctant to invest in mixed use development. This in turn means that public sector investment can play a key role in pump-priming private development in the initial steps of the development of sustainable markets in non-traditional locations (Jones and Watkins, 1996).

Accessibility/Transport Infrastructure

The transport infrastructure of a city in terms of the quality of its roads, the scale and availability of public transport and location of transport nodes defines the ease by which people can reach buildings, spaces, and places. It provides a set of accessibility relationships within the urban area that can be seen in terms of distances, travel times and travel costs. Infrastructure in the form of transport networks has a direct impact on the scale of local market areas. For example the spatial extent of retail (and other services') catchment areas is partly a function of how easy it is for people to travel to shops, or put another way the costs of travel by customers. In fact a hierarchy of services provision/facilities exists determined by the transport network. This is most evident in retailing where shoppers are usually

prepared to travel further to large retail superstores than local supermarkets, which in turn attracts people located further away than corner shops.

From an economic perspective the transport infrastructure primarily determines travel costs. If costs are high then travel trips may be short and some households could be excluded from access or refrain from using services/facilities. These costs take the form of both financial and time costs. Accessibility has a key impact on land and property. There is likely to be a higher demand for a parcel of land or property that has good accessibility both to it, and to other services and infrastructure networks. The land use that benefits most from accessibility to a given location and makes the highest profit will outbid other potential users. This also means that land prices/rents are highest at the most accessible points in a town reflecting their potential for profits at these locations.

Density

The density of a city can be viewed in a number of ways as Table 2.1 examines. Both the city-wide average and the spatial pattern of population and employment densities within a city are usually the consequence of the competition between land uses. In a competitive land market the higher the price the more intensively the land is used for housing or for commercial space and hence the greater the density of utilisation. As a result typically the highest employment density is in the central business district (CBD).

In the housing market inner area locations have the lowest travel costs to the CBD, the main centre of employment, and competition leads to high land prices and high residential densities. In western economies the implications are that low incomes households consume small amounts of housing at high unit costs in inner high density locations and high incomes households consume the converse (Muth, 1969). This general pattern is distorted by the provision of social housing allocated on a non-market basis.

Characteristics of the Built Environment

Investment attitudes of banks and other financial institutions can influence the characteristics of the built environment (the building itself and its environment). While such investors are market led and will invest their funds where they will receive the greatest return, they tend to be risk averse with regard to new types of building form or the introduction of a building form new to an area. For example, they have been ambivalent in the UK until recently to green buildings. Further, banks in the UK are reluctant to lend on flats above a certain storey height. Similarly, house builders have taken a cautious approach to innovation for example in energy conservation, because introducing such measures increases the sale price. Demand factors can also influence the built environment through building type preferences, such as the desire of families with young children to occupy housing with gardens.

The level of building maintenance may also be important. High maintenance buildings, *ceteris paribus*, may not be desirable in terms of the costs they incur and so demand tends to be low.

This brief review has emphasised that city form is the outcome of the operation of real estate markets within a given transport infrastructure framework. Looking at the broader urban picture it can be seen that these processes in turn determine the spatial distribution of employment and population.

Urban Form, Economic Performance and Sustainability

A crucial question for this chapter is how urban form influences economic performance. Many of the economic arguments in favour of the compact city develop from agglomeration economies or benefits and the fundamental reasons why cities exist. Agglomeration economies include localisation economies such as the access to a pool of labour, availability of a range of auxiliary trades and specialised services plus knowledge spillovers from similar firms located near to each other. Urbanisation economies result from the common location of firms belonging to different and unrelated industries. These benefits include the availability of a range of municipal services, public utilities, transportation and communication facilities, the existence of a wide variety of business and commercial services and a complementary of labour supply. Households can also benefit from agglomeration economies or city living in the form of a wide range of shops, amenities and cultural facilities and firms and households from public services/infrastructure (Henderson, 1974). The benefits of agglomerations are linked to urban size. For example the larger the city the greater is the viability of specialist shops and hence the wider the choice to the consumer. Similarly a larger labour market also provides employers with a wider skills base, leading to an increased likelihood of skills-job matching and greater productivity.

The role of density is at the heart of this sustainable urban form debate. Higher employment densities are traditionally linked to potential agglomeration economies. There are knowledge spillovers between firms in an industry in close physical proximity which promote urban economic growth by reducing product costs and the costs associated with product development and innovation (Jacobs, 1969, Porter, 1990). This argument suggests that competitive pressures brought about by the geographical concentration in an industry will stimulate innovation. Prud'homme & Lee (1999) similarly claim that the closer people are to their place of work, the more productive the economy is because workers spend less time commuting. They conclude that the labour productivity is negatively related to sprawl.

High land use densities also have a number of implications for the demand for services and the costs of their provision as high land use density results in a greater concentration of demand, and associated consumer spending, which *ceteris paribus*, reduces the spatial extent of viable social and private services' catchment areas (including business services). This in turn suggests the potential for more consumer

choice and diversity in high density areas. Barton (2000) argues that mixed land use is the most sustainable type of urban use, in that it increases the viability of services and transport provision supported by high residential density. Mixed use is linked closely to ideas of ease of access and the provision of a greater choice. A further argument is that by having many types of land use in one development area there is a

critical mass and level of activity (created) which is greater than the sum of individual users, thereby making a critical contribution to location and character (ODPM, 2002, para 6.1.2).

These arguments tend to be normative, partial and emphasise the benefits of high density cities, and have not been subject to detailed scrutiny. Empirical studies are thin on the ground and suffer data limitations, examples include Camagni et al. (1998), Cevero (2001), and Ciccone and Hall (1996). They do not address the question of urban size or cities as dynamic entities. There is also a contradiction between the arguments that a compact city encourages economic growth and the implicit assumption that the compact city should be tightly defined. A high density compact city is less likely to be able to cope with significant population growth, as there is less potential for expansion if development is already at a high density (Anas et al., 2000). The ultimate logic of the proponents of compact city economics is that market forces would create compact cities yet the opposite is occurring. Urban systems are decentralising driven by market forces and the choices of firms and households. Cities and urban systems have been subject to decentralisation pressures as a result of a combination of the transport infrastructure in large cities unable to resolve congestion and increasingly improved inter-urban transport links fostering growth in medium sized towns. For these and other reasons such as the growth of information technologies arguments that agglomeration economies are becoming more diffuse have also recently emerged (Parr, 2002).

Along side this sustainable city literature and predating it is the related concept of 'optimal' city size that spawned a number of papers in the early 1970s including Alonso (1971), Evans (1972) and Richardson (1973). The logic derives from the argument that while agglomeration economies are the driving force for the growth of cities beyond a certain size negative agglomeration economies such as congestion and pollution occur. These theories suggest there may be an 'optimal' urban size at a population where the total benefits of size equate with the total costs (McCann, 2001, Capello and Camagni, 2000). Similarly Fujita & Thisse (2002) argue that cities can become too big to be sustainable.

Some authors have extended this concept to the sustainable city, defined by a set of sustainable criteria that establishes an appropriate optimum city size. Camagni et al. (1998) define a sustainable city where the three environments – physical, economic and social - interact in such a way that the sum of all the positive externalities is larger than the negative external effects. In a narrower paper Capello & Camagni (2000) argue that urban size influences location costs and benefits through creating greater potential for more mixed and higher level urban functions, specialisation and integration within the urban system. An optimal efficient city size is one that achieves economic sustainability. However, this efficient size depends on what is produced, how it is produced, and how the area in question operates within the urban economy.

These approaches tend to view a city simply as a static uniform entity rather than a dynamic changing place with neighbourhoods that have different densities and characteristics. The role of spatial structure on economic sustainability in these approaches is also ignored. Just like optimal city size models they do not appear to aid a practical solution. These complex theoretical models contrast with the simple sustainability argument underpinning the 'smart growth' or high density/urban containment policies followed by some states in the USA (Cervero, 2001). These policies are based on a study with a very limited evidence base that found the public sector costs of sprawl are much higher than for a compact city (Burchall, 2000).

Urban containment and compact city policies in turn create a number of concerns that follow from the operation of property markets. Limiting land supply by development constraints on greenfield sites, through for example green belts surrounding cities have potential implications for the local economy. In the long term market forces will lead to higher densities as the proponents of the compact city seek. However, this will happen because land values will be bid up and as result land is used more intensively (Cheshire and Sheppard, 1989). It has been argued that as a result economic growth will be siphoned away as increased land and housing costs are followed by demands for higher wages and reduced competitive advantage with firms moving to alternative cheaper locations (Cervero, 2001). The reality is likely to be quite complex and difficult to dissect as the countervailing forces of agglomeration and crowding out from land shortages vary with the sector of the economy.

In the housing market many households will be priced out of the immediate housing market in the sense that they will not be able to afford the type of housing they require and are prepared to commute a longer distance from another settlement. This process exacerbates a wider phenomenon of increasing real household incomes which drives the demand for more housing space usually in low density developments at peripheral locations (see Brueckner, 2000). Proponents of the decentralised solution to urban sustainability extrapolate this trend and stress either the benefits of a decentralised 'rural' or 'semi-rural' life style with low development costs or that it will happen anyway as unstoppable market forces will create dispersed communities with low energy consumption and congestion (Richardson and Gordon, 1993; Gordon and Richardson, 1997). However, to date the process has led to increased commuting distances.

This review of the economic sustainability of cities reveals that much of the theoretical analysis has simplicities, flaws and serious limitations, and lacks practical application. Much of the arguments incorporate empirical analyses that are not usually designed to address economic sustainability. For example agglomeration economies are normally viewed at city level rather than the neighbourhood. The discussion suggests that the problem can be expressed as maximising potential urban output or productivity that stem from agglomeration economies subject to a series of sustainability constraints. This can be viewed as a form of (non) linear programming problem where a function is maximised subject to constraints that encompass social, environmental and economic factors.

7 Economic Viability

From an economic viewpoint an all embracing constraint is the viability of the urban form that would in turn include the viability of sectors of the local economy. These sectors can be seen in terms of wealth creation, namely manufacturing and services, the labour market, transport, public administration and land use property markets. Real estate markets have to be 'sustainable' as defined in terms of prices being achieved without public subsidy and the ability of the market to sustain itself through downturns in the property cycle (Jones and Watkins, 1996). An additional constraint is an adequate supply of housing for the workforce and their families that can be called an affordability constraint. The upgrading of infrastructure can also viewed as a distinct cost constraint on adapting or shaping the expansion of a city for example through the provision of an efficient transport system.

The precise city form that this formulation of the problem generates is indeterminate as there are too many variables and in any case it lacks a dynamic dimension. The formulation is best seen as a set of guiding principles or economic conditions on making the existing urban form more sustainable.

The Economic Viability of Individual Land Uses

The starting point of the empirical research presented here is that, to make cities more sustainable, policies will need to adapt the existing urban forms. The sustainability of cities will not be driven by economic factors but desired urban forms will need to be subservient to economic viability otherwise market forces will mean they will be not be stable in the long run. A particular focus for the analysis is the viability of land use markets. Urban form, as noted earlier, is primarily determined by the operation of local real estate markets within a framework of transport costs (that determines accessibility relationships) which in turn is dependent on the transport infrastructure. Planning will also shape the market but not alter these fundamental relationships. The research examines different land uses in turn.

Offices

As noted earlier, the very existence of cities is based on agglomeration benefits. Historically the city centre was the preferred or optimum location for office users where these benefits were maximised at the point of greatest accessibility within a city. The nature of these agglomeration economies are a crucial viability constraint on reshaping urban form. It can be argued that over the last half century decentralisation pressures within the urban economy and the property market have been promoted by changes in transport technologies and advances in production and information and communication technologies which have weakened the agglomeration economies of central city locations. The consequence for the city has led to a change in its urban form as the pattern of commercial land use has become more spatially diffused.

Distance band	Edinburgh	Glasgow	Leicester	Oxford	Sheffield
Centre – 3 km 3–6 km 6–9 km 9–12 km 12 km +	63.1 16.5 12.0 6.9	84.8 7.1 6.6 1.5 0.0	69.4 19.8 6.6 1.2 2.9	27.6 24.9 33.8 13.4	50.6 25.0 15.1 5.2 4 1

Table 7.1 Spatial Pattern of Rateable Values of Offices in the Case Study Cities

Numbers are percentages of the respective column total

The research problem is that it is difficult to measure these agglomeration benefits. In this study the 'rateable value' of a property is used as an indicator. In the UK commercial and industrial properties are subject to a tax called 'rates'. A rateable value is attributed to each property, broadly defined as its (estimated) market rental value on a given date. In an efficient market the rent of a property is generated from the economic surplus it creates and so the aggregate rateable value of an area is a useful indicator of the agglomeration benefits of locating there. The spatial structure of rateable values for offices within rings from the centre of each of the five cities is given in Table 7.1. The distances applied are based on the actual road network, not "as the crow flies" and are therefore a better representation of true distance from the centre

The spatial patterns show a continuing agglomeration of offices in city centres but the degree of concentration within cities does not vary with city size. Oxford has a very flat dispersed pattern with the 6–9 km band having the highest concentration. At the other extreme Glasgow offices are most focused on the city centre which is partly a reflection of local planning policies. Overall while the city centre remains the dominant location for offices the analysis suggests that the office property market will not constrain different urban forms including decentralised urban systems. Our research also shows that this conclusion applies to the industrial property market too. There is one caveat in that it is probable that the decentralisation of offices is partly the consequence of administration functions moving from city centres.

Retailing

The viability of shops requires a close relationship between the location and catchment of retail centres and the spatial distribution of population, with the latter driving the former. The hierarchy of retailing and associated catchment areas is also driven by transport networks and hence the ease of travel to shop. The viability of individual retailing centres is also based on the expenditure potential of its catchment area. The relationships between urban form and the viability of retailing can be estimated using a 'gravity model'. The basic idea of a gravity model is that a shopping centre exercises a pull on potential customers based on its attractiveness relative to the attractiveness of other shopping centres and how far away they live. A model has been estimated for one of the case study cities, Edinburgh, and provides a

	Impact of Population Changes on Trip Flows %	Impact of Floorspace Changes on Trip Flows %
-10.0%	-11.4	-16.9
-7.5%	-8.6	-12.8
-5.0%	-5.7	-8.6
-2.5%	-2.9	-4.3
+2.5%	2.9	4.4
+5.0%	5.8	8.9
+7.5%	8.7	13.5
+10%	11.6	18.2

 Table 7.2 Changes to Number of Shopping Trips when Density and Size of Retail Centres Change

base for exploring the relationship between retailing and various elements of urban form (Orr et al., 2007).

The estimated model is applied to examine how sensitive the level of shopping trips are to changes in the density of residential areas, the size of shopping centres and accessibility relationships. Table 7.2 illustrates the impact of altering the population density in incremental steps on the scale of shopping flows while holding the travel times constant and assuming the number of shopping flows are unconstrained. As can be seen, a rise in the number of residents living within each neighbourhood will yield a higher rise in the number of shopping trips made between all the residential zones and the shopping centres whereas a fall in residential densities will reduce shopping flows. In reality if cities become more densely populated then the growth in some areas will be greater and faster than others and this will result in a more uneven effect on shopping centres. A further caveat is that the growth would also impact on travel times with the greater density resulting in greater travel times, which in turn results in an even greater impact on the shopping centres closer to the neighbourhoods with the greater population rise.

The changes in shopping flows are not the same magnitude for the same sized positive or negative change in density. This is due to the non-linear nature of the model and the existence of exponentials in the gravity model developed. Table 7.2 also illustrates the result of altering the size of the retail centres when the total numbers of shopping trips are not constrained. This reveals that the shopping trip flows are more sensitive to changes in the scale of the retail centres than the density of the population. When travel times are changed in the simulation exercise, the impact is substantially more variable and dependent on the size of shopping centre. Unsurprisingly, residents located in more remote locations are more sensitive to small changes in travel time than more centrally located shoppers.

Commercial and Retail Change

Adapting the provision of retailing and offices to a different urban form is not simply a matter of changing locations. This is illustrated by the changes to these sectors over the last twenty five years of decentralisation pressures within cities of the UK. The consequence for the city has led not only to a changed urban form by influencing the spatial patterns and density of land use but also new property forms such as flexible commercial space usually located in business parks and out of town shopping malls. An analysis by Jones (2010, forthcoming) of the gradual introduction of the new property forms of retail warehouses and office parks that emerged as part of this process demonstrated it took sometime for sustainable markets to develop. The evidence revealed that developers tended to take up these new property forms quickly, and the range of occupiers gradually expanded but financial institutions were slower to embrace them within their property investment portfolios. Changing UK planning policies toward retail warehouses were a particular influence, after initial ambivalence policy switched to restrict their development. This process of establishing investment markets for new property forms is not necessarily a smooth process and in the case of retail warehouses took up to two decades. The analysis suggests that policies to reshape the city form will require a consensual long term public policy framework to ensure confidence for private property investment.

Housing

The operation of urban housing markets is the major shaper of city form moderated by the planning system. Simple theoretical models highlight household commuting costs as central to household location decisions, and this gives rise to a house price gradient declining from the city centre. A key influence on a household's choice of location is income (see earlier). This basic model can be adapted by the spatial pattern of social housing and the effectiveness of planning constraints in the form of green belts. Taken together this means that spatial price structures of city housing markets (and hence urban form) are principally determined by the local distribution of income and the affluence of the city, social housing locations, planning constraints such as green belts and the cost and speed of local travel. This theoretical perspective is reinforced by a statistical analysis of the pattern of house prices within the five cities that demonstrates substantial variation between cities.

To examine the potential for adapting a local housing market to a more sustainable form by building more houses/flats for sale the research estimates the spatial patterns of local housing market viability within each of the five cities (Jones et al., 2009). The statistical problem has as its starting point data on individual house prices and data on construction costs for individual development projects. There are four steps to the analysis:

- 1. Estimate a statistical model of house prices for locations within each city.
- 2. Estimate a statistical model of construction costs.
- 3. Combine the two models of price and cost to estimate development 'viability'.
- 4. Assess the spatial pattern of development 'viability'.

7 Economic Viability

The model of house prices is estimated using multiple regression, a statistical technique that breaks down house prices into its constituent attribute prices. The attributes can be differentiated into housing, location and neighbourhood characteristics including density and are based on data from the Land Registry and the Census. The construction costs model is also estimated using statistical analysis based on data derived from planning applications. The output of these models is used to simulate the development value and costs of standardised new housing projects. Maps of viability within Glasgow and Oxford (the extreme cases) are then constructed and shown in Figures 7.1 and 7.2. Bright yellow represents strong viability and dark blue strong negative viability.

There are a number of striking points that emerge from these maps. First, the absolute level of viability in a city is clearly a function of the affluence of the city, an exogenous factor in the original conceptual framework. Second, the intra-urban



Key - Residual value expressed as a % of gross development value.

	50 to	80
Ħ	31 to	50
	10 to	31
	-21 to	10
	-66 to -	-21
	-161 to -	-66

Fig. 7.1 Spatial variation in mixed housing/flatted development viability within Glasgow Source: Jones et al., 2009



Fig. 7.2 Spatial variation in mixed housing/flatted development viability within Oxford Source: Jones et al., 2009

patterns of viability are primarily determined by the spatial structure of house prices, which is in turn linked to intra-urban accessibility and the tenure distribution within local neighbourhoods. Neighbourhood density is only a minor influence on viability but this may be partly due to the fact that densities in public sector housing estates are included in the correlation analysis. Densities are also a function of historic development patterns. These conclusions must be tempered by one caveat that while the analysis has abstracted the influence of the role of individual planning permissions the aggregate effect of the tightness of a local planning regime may have an impact on the absolute levels of viability in a city as a whole and in popular submarkets. This effect may contribute to the high levels of viability, for example in Oxford and parts of Glasgow. At the same time a very tight planning regime can constrain housing supply so that there is a serious affordability problem.

The results show a substantial difference between cities that can be attributed not to urban form per se but to socio-economic factors. This demonstrates that in practice it is impossible to divorce the physical structure of cities from their economic and social structure. Viability is also influenced strongly by public policy through the location of social housing and overall planning regime constraints. The research suggests that a major influence/constraint for development viability is the level of neighbourhood house prices. Large swathes of negative viability in some cities are found even without accounting for the additional costs of brownfield development suggesting that there are major constraints to the reconfiguration of housing markets in a piecemeal way.

To examine the underlying influences on these patterns of viability a study of owner occupiers of stated preferences to urban form characteristics in the six case study areas of Glasgow and Edinburgh was undertaken based on a questionnaire survey. Respondents were given a series of paired random 'choice tasks' that listed key characteristics of notional neighbourhoods and associated housing and were asked to decide which combination they preferred.

Analysis of the choice task results reveal general preference for low density areas, with the preference for lower density over high density neighbourhoods being stronger for households that already live in suburban areas. Public transport accessibility is important to households but it is not so important that this public transport is within easy walking distance or just within walking distance. It is more important to those already living in city centre locations to have public transport within easy walking distance. Households seem to want amenities to be either just within, or within easy walking distance at all. Green space is more important to households currently residing in suburban areas, and also for those in in-between areas, albeit slightly less important. Inner city residents have a less clear pattern of emphasis on the amount of green space.

The household characteristics of the respondents also reveal some interesting findings that could support a household life-cycle element to residential location choice. Younger and non-pensioner single households are more prevalent in the central city areas. The central study areas also have the smallest average household size, with the suburban areas having the largest average number of people in the household. The suburban areas also have the largest proportion of households that consist of a couple with one or more children. Initial findings show that as households grow older and expand, different factors become important in making a residential location choice so that the age structure of a city is an important influence.

Infrastructure Costs and City Form

The adaptation of urban form will be subject to infrastructure cost constraints. There are two types of urban infrastructure costs – social and physical. Social infrastructure costs include education, health, personal social services, environmental, protective and cultural services together with public goods such as police and the fire service. Some of these services are provided simply on a per capita basis. Small area data on these costs are not generally available. Research by Bramley and Evans (2000) of small areas in three local authorities in the UK focuses on the impact of deprivation or concentrations of poverty on increasing costs but also acknowledges other drivers of expenditure such as local population and socio-economic-demographic structure influencing demand and take up of

services as well as urban form features. These latter variables comprised population, density, road length per capita, distance from CBD and are generally not significant factors. However, multiple regression analysis finds that density and road length per capita are statistically significant negative and positive influences on cost of providing public goods (keeping other factors constant). Distance to the CBD tends to have a positive effect on different components on social infrastructure costs. However, there is a problem in the interpretation of these variables and Bramley and Evans suggest that these variables approximate to socio-economic/local geographic patterns of social housing influences not included in the model.

Physical infrastructure costs comprise the construction and maintenance of roads, sewers, water, gas, electric and telephone provision. These costs are influenced by local landscape and geology. Costs can also be dependent on the length of the distribution network. Water, for example, has to be transported from treatment plants and hence distribution costs vary geographically with distance from that location. It is useful to set aside these location specific influences as these costs do not vary with urban form and are in effect fixed costs. Similarly while pipe sizes will vary with distance from the treatment plant and density of development the differential costs of pipe size are small relative to the principal cost associated with the physical placement. It is possible therefore to simplify the empirical analysis by ignoring differential component costs. Similar points apply to the other utilities. Variations in physical infrastructure within cities can then be taken as a function of the number of addresses (density) and road length.

Given the discussion above, the empirical analysis here assumes that social infrastructure costs do not vary with urban form and so the focus is only on physical infrastructure costs. The variation in physical infrastructure costs is estimated by applying standard costs per road distance and per address. In this way the research compares the costs to build the physical infrastructure of nine 'between' and 'outer' case study neighbourhoods in the five cities. The ratios of road length per person to density, as a measure of relative road related infrastructure costs, reveal that broadly doubling density in the Sheffield and Edinburgh case studies results in these costs reducing by more than half and 70% respectively. A 14% reduction in density in Oxford is associated with a more dramatic 62% increase in costs. In Leicester a 40% reduction in density brings a three fold increase in these infrastructure costs. This commentary demonstrates that there is clearly some variation in this inverse relationship and that there are some local influences. Overall Table 7.3 shows while density in the case study areas varies by almost a factor of three, road infrastructure costs per capita controlling for density have a multiple range of 5. In other words road infrastructure costs increase by a higher than proportional rate with falling density.

No allowance is made in the results presented here for differences between flats and houses although some simple sensitivity analysis suggests it would lead to up to a 10% reduction in costs in some areas for electricity connections. The results suggest that while density is a factor influencing infrastructure costs there are clearly additional factors at play for example the lay out of road networks. Nevertheless there is considerable scope for reducing infrastructure costs by higher

City	Case Study	Persons per Hectare (density)	Road Length per person	Road Length per Person as a ratio of Density (×100)	Infrastructure Costs per Person (£)	Infrastructure Costs per Person as ratic of density
Oxford	between	74	2.0	2.70	6278	85
	outer	64	2.8	4.38	8384	131
Sheffield	between	80	2.4	3.00	7581	95
	outer	43	2.9	6.74	6102	142
Leicester	between	89	2.2	2.47	4955	56
	outer	36	3.5	9.72	8911	248
Edinburgh	between	64	2.3	3.59	6510	102
-	outer	31	3.8	12.26	9584	309
Glasgow	between	55	2.5	4.55	6536	119

 Table 7.3
 Road Lengths and Densities in Selected Case Study Neighbourhoods

density development. However, there is some caveats to this conclusion in relation to adapting a city. Much of the road infrastructure and associated services are already in place so greater intensification of land use on brownfield sites it could be argued will only add marginally to costs and if this is instead of development on greenfield sites then the savings are magnified compared to the figures presented. This argument is diluted where infrastructure systems such as water supply and treatment are at full capacity.

Conclusions

This chapter dissects the arguments linking city form and economic sustainability. The fundamental determinants of the elements of urban form are shown to be the outcome of the operation of real estate markets within the city's transport infrastructure and moderated by local planning policies. The housing market as the largest land market and the determinant of population density in particular has a dominant role in urban form and its sustainability. Similarly commercial property market processes determine the spatial distribution of employment. Underpinning the property market are location decisions of firms and households that are based on the associated agglomerations economies and accessibility of individual locations. The spatial structure of cities is also influenced by the wealth, the distribution of incomes and age structures of households through demand influences on the local housing market.

Advocates of high density sustainable cities base their arguments on simple ideas about agglomeration economies and see cities implicitly as almost static phenomenon rather than dynamic entities with spatial land use patterns continuously if gradually changing. More sophisticated models of urban economic sustainability are too abstract to be of practical use. An alternative approach to sustainability that also favours high density cities is the 'smart growth' movement in the USA, but the argument is based narrowly on the high infrastructure costs of sprawl.

It is not clear that the framework of the current debate is fruitful and there is unlikely to be a definitive sustainable economic city form. The approach here is to express the problem as maximising potential urban output or productivity subject to a series of sustainability constraints. These would encompass social, environmental and economic factors. From an economic perspective these constraints include the viability of sectors of the local economy and infrastructure costs. Within the real estate sector this would require an adequate supply of housing for the workforce and their families and sustainable markets. In particular, desired urban forms will need to comprise land use patterns that are economically viable otherwise market forces will mean they will not be stable in the long run. These arguments provide the basis for adapting the existing city form to make it more sustainable.

The empirical research presented suggests some locational flexibility in the commercial and industrial property sectors. While the city centre remains the dominant location for offices and retailing, the analysis suggests that agglomeration economies are no longer focused entirely on central city locations and there are benefits from decentralisation. These patterns suggest that the viability of commercial property market will not constrain different urban forms including decentralised urban systems. However, while decentralised city forms follow market trends it is constrained by higher physical infrastructure costs.

The analysis of the housing market shows that households prefer low density housing and that there appears to be a household life-cycle element to residential location choice. Younger and non-pensioner single households live in the central city areas but move out to suburban locations as they move through the family life cycle. This means that it will be difficult to encourage more concentrated urban forms without significant changes to the underling forces of city housing markets, for example by increasing commuting costs. In addition the patterns of viability in the city housing markets suggest there are major constraints to reshaping local housing markets. To overcome these constraints will require substantial public expenditure costs to engineer a strategic restructuring of price structures. On the other hand in cities such as Oxford, where residential development is severely constrained, it is highly likely that the affordability constraint of sustainability is not being met.

The reshaping of cities will require positive planning policies to achieve a specific more sustainable urban form and must centre on transport infrastructure and redrawing the property market. The need to achieve sustainable local real estate markets means it is essential to understand the operation of these markets. This in turn necessitates developing a policy analysis within a system of functional markets and submarkets and requires the appropriate information systems (Jones, 2002, Jones et al., 2005). The process of adjustment through the establishment of new occupier and investment markets involving new property forms in non-traditional parts of a city is not necessarily a smooth process and can take decades. The analysis suggests that strategies to adapt the city form will require a consensual long

term public policy framework to ensure confidence for private property investment. Sustainable land use markets remain a condition of viability.

References

- Alonso, W. (1964) Location and Land Use: Towards a General Theory of Land Rents, Harvard University Press, Cambridge, MA.
- Alonso, W. (1971) Economics of Urban Size. *Papers of Regional Science Association*, **26**, pp. 67–86.
- Anas, A., Arnott, R. & Small, K. A. (2000) Urban Spatial Structure. In *Readings in Urban Economics: Issues and Public Policy*, (ed. R. W. Wassmer), Blackwell Publishers.
- Barton, H. (2000) Urban form and locality. In Sustainable Communities: The potential for econeighbourhoods, (ed. H. Barton), Earthscan, London.
- Bramley, G. and Evans, M. (2000) Getting the Smaller Picture: Small Area Analysis of Public Sector Expenditure and Incidence in Three English Cities. *Fiscal Studies*, 21(2), pp. 231–267.
- Brueckner, J. (2000) Urban Sprawl: Diagnosis and Remedies. International Regional Science Review, 23(2), pp. 160–171.
- Burchell, R. (2000) *The State of Cities and Sprawl: Bridging the Divide*, US Department of Housing and Urban Development, Washington DC.
- Cheshire, P. and Sheppard, S. (1989) British planning policy and access to housing: some empirical estimates. *Urban Studies*, **26**, pp. 469–485.
- Camagni, R., Capello, R. and Nijkamp, P. (1998) Towards sustainable city policy: an economyenvironment technology nexus. *Ecological Economics*, 24, pp. 103–118.
- Camagni, R., Gibelli, M. C. and Rigamonti, P. (2002) Urban mobility and urban form: the social and environmental costs of different patterns of urban expansion. *Ecological Economics*, 40, pp. 199–206.
- Capello, R. and Camagni, R. (2000) Beyond Optimal City Size: An Evaluation of Alternative Urban Growth Patterns. Urban Studies, 37(9), pp. 1479–1496.
- Cervero, R. (2001) Efficient urbanisation: Economic performance and the shape of the metropolis. *Urban Studies*, **38**(10), pp. 1651–1671.
- Ciccone, A. and Hall, R. E. (1996) Productivity and the Density of Economic Activity. *American Economic Review*, **86**(1), pp. 53–70.
- Dunse, N. and Jones, C. (2005) UK Roads Policy, Accessibility and Industrial Property Rents. In *Planning Policy and Property Markets*, (eds. D. Adams, C. Watkins and M. White), Blackwell, Oxford.
- Evans, A. W. (1972) The Pure Theory of City Size in an Industrial Economy. *Urban Studies*, **9**, pp. 49–77.
- Fujita, M. and Thisse, J-F. (2002) *Economics of Agglomeration: Cities, Industrial Location and Regional Growth*, Cambridge University Press, Cambridge.
- Gordon, P. and Richardson, H.W. (1997) Are Compact Cities a Desirable Planning Goal? *Journal* of the American Planning Association, **63**(1), pp. 95–106.

Henderson, J. (1974) The Sizes and Types of Cities. *American Economic Review*, **64**, pp. 640–656. Jacobs, J. (1969) *The Economy of Cities*, Vintage, New York.

- Jones, C. (2002) The Definition of Housing Market Areas and Strategic Planning. *Urban Studies*, **39**(3), pp. 549–564.
- Jones, C. (2010, forthcoming) Remaking the Monopoly Board: Urban Economic Change and Property Investment. *Urban Studies*, **47**(2).
- Jones, C., Leishman, C. and MacDonald, C. (2009) Sustainable Urban Form and Residential Development Viability. *Environment and Planning A*, **41**(7), pp. 1667–1690.
- Jones, C., Leishman, C. and Watkins, C. (2005) Housing Market Processes, Urban Housing Submarkets and Planning Policy. *Town Planning Review*, 76(2), pp. 215–233.

- Jones, C. and Watkins, C. (1996) Urban Regeneration and Sustainable Markets. *Urban Studies*, **33**(7), pp. 1129–1140.
- Lynch, K. (1981) A Theory of Good City Form, MIT Press, Cambridge, MA.
- McCann, P. (2001) Urban and Regional Economics, Oxford University Press, Oxford.
- Muth, R. (1969) Cities and Housing, University of Chicago Press, Chicago.
- Newman, P. and Kenworthy, J. (1989a) *Cities and Automobile Dependence: A Sourcebook*, Gower, Aldershot.
- Newman, P. and Kenworthy, J. (1989b) Gasoline Consumption and Cities a comparison of US Cities with a global survey. *Journal of the American Planning Association*, 55, pp. 24–37.
- Office of Deputy Prime Minister (ODPM) (2002) *Mixed use development, practice and potential* http://www.odpm.gov.uk/stellent/groups/odpm_planning/documents/pdf/odpm_plan_pdf_ 606215.pdf (as at 17/02/04).
- Orr, A., MacDonald, C., Leishman, C. and Jones, C. (2007) Retailing and the City: an investigation into the relationships between urban form and retailing in Edinburgh. Paper presented at *The Vital City, 10thth Anniversary Conference of the European Urban Research Association*, University of Glasgow, Scotland, September 2007.
- Parr, J. B. (2002) Agglomeration economies: ambiguities and confusions. *Environment and Planning A*, 34(4), pp. 717–731.
- Porter, M. E. (1990) The Competitive Advantage of Nations, Free Press, New York.
- Prud'homme, R. and Lee, C-W. (1999) Size, sprawl, speed and the efficiency of cities. Urban Studies, 36(11), pp. 1849–1858.
- Richardson, H. W. (1973) The Economics of Urban Size, Saxon House, Farnborough, Aldershot.
- Richardson, H. W. and Gordon, P. (1993) Market planning: oxymoron or common sense? *Journal of the American Planning Association*, 59, pp. 347–52.

Chapter 8 Adapting the City

Hildebrand Frey and Samer Bagaeen

Introduction

The main objectives of this chapter are to develop a theoretical underpinning of and a methodology for urban intensification and regeneration that could help transform currently unsustainable urban areas into sustainable urban neighbourhoods and communities. The key task of the research is accordingly to find ways of translating sustainability into strategic concepts of urban restructuring and regeneration. It is argued that, once this key task has been achieved, the other tasks of dealing with urban growth as well as restructuring inner city and urban fringe areas would follow suit.

The chapter considers first the application of indicators of sustainable development to urban regeneration primarily in the UK. It then considers the dynamic to intensify urban neighbourhoods to meet the growing demand for housing while minimising the ecological footprint of built up areas. This leads on to a discussion of the criteria of sustainable intensification. The subsequent research is presented in three stages:

- The generation of an assessment tool for the systematic measurement and evaluation of levels of (un)sustainability of urban areas with the help of threshold and target values.
- The application of the tool for the assessment of the existing values and deficiencies of selected areas of Glasgow and the formulation of regeneration programmes for these areas based on the discrepancies between their existing values and the threshold and target values developed for the assessment tool.
- The development of two and three-dimensional regeneration models for the selected urban areas and the testing of their achievements and viability.

Finally conclusions are drawn on the practical application and the general approach.

H. Frey (⊠)

Department of Architecture, University of Strathclyde, Glasgow, UK

Emergence of Indicators of Sustainable Development

The publication of Agenda 21 at the Earth Summit at Rio de Janeiro in 1992 (UN, 1993) was an important milestone with the search for a common approach to the assessment of the sustainability of regions, nations and cities. There is, however, a problem with Agenda 21: the vagueness of its action programmes. Indicators help establish profiles and trends of social, economic and environmental conditions in a region, nation or city. But each indicator is assessed separately and no mechanism is provided to establish the interdependence of indicators and data. Furthermore, no guidelines are given for the interpretation of the collected statistical data; no threshold values or benchmarks are made available against which current performance values of urban areas can be assessed. The interpretation of the 'goodness' or 'badness' of data and the decision what action to take is therefore left to the value judgement of regional, national and local authorities. Most of them cannot be expected to have any specialist knowledge and they have, therefore, to rely on more or less intelligent guesswork. On that basis, the effect of chosen action programmes on the environment is unpredictable: they may achieve environmental improvements, but they may also do significant and potentially irreversible harm to the environment.

It is one of the goals of the research to develop not only lists of parameters but also threshold and target values for each parameter as far as they can be established. These values allow development and regeneration decisions to be based on the systematic measurement of the differences between existing values of urban areas and threshold and target values believed to lead to sustainable urban form.

Current Practice in Urban Regeneration and Development

Many urban regeneration projects use Local Agenda 21 lists of parameters, but very few are based on a thoroughly worked out set of objectives and targets to be achieved. In comparison with other disciplines generating products for public use and consumption, current planning and urban design seem largely to be working towards a 'product' without classifying clearly what kind of performance qualities it ought to have in order to work efficiently and economically for the user and the public at large and protect the environment from damage. Within the current planning approach in the UK these performance qualities are the result of negotiations of planners with developers and, at least sometimes, communities. An alternative is developing a general framework of target values of sustainability of the 'product' first and then discuss it with communities and stakeholders to achieve a commonly agreed set of goals and benchmarks. There may be quite a few projects, specifically those focusing on the improvement of social housing areas, that involve the 'community' (the people living in the area), the client (a housing association or co-operative), the planning system (planning and building control) and the developer and architect in the discussion of the qualities that the urban area under investigation ought to have and this is appreciated. But even then the discussion is rarely based on predefined, generally valid sustainability targets and deals with a fraction of parameters only that are of importance to the community and other stakeholders rather than developing a holistic approach.

Clearly, the frequent outcomes of such projects are accordingly at best uncertain, at worst counterproductive. It was one of the motivations behind this research project to investigate how the expected properties of the 'product' to be generated – an urban settlement or new housing or a new transport system or new community facilities – can best be specified to represent clear rules and target as to what is to be achieved and what the impact of the 'product' on people, the local economy and the environment might be.

Urban renaissance and Sustainable Urban Neighbourhoods

The Urban Task Force (UTF) in the UK, formed in 1998, sought to establish

... a new vision for urban regeneration founded on the principles of design excellence, social well-being and environmental responsibility within a viable economic and legislative framework (UTF, 1999, internal front cover page).

The report's main recommendation is the return to a design-based approach of urban development, but it also calls for the return to the compact city and the review of urban densities to achieve viable local amenities and public transport. These recommendations were to yield a new approach to the planning, design and building of new urban villages and sustainable communities, generated on the basis of sustainability criteria and target values. Of importance was the development of models of urban capacity: the report makes it clear that viable local amenities and public transport stops in urban neighbourhoods require a threshold population of 7,500 people, given a socially mixed community and a minimum density of 100 persons/ha (UTF, 1999). To achieve this, existing urban areas that do not meet this threshold population size need to be intensified.

The UK government implemented the UTF's recommended programme of Millennium Villages and Sustainable Communities and so it is possible to see the direct impact in community pilot projects carried out on the basis of a sustainable development framework. This provides governments and researchers with a rare chance: if monitoring of development and of outcomes is consistent then it might be possible to generate values backed up with the currently missing empirical evidence that their application will lead to sustainable development. Unfortunately, most of the target values employed in these Urban Villages and Sustainable Communities projects are dealing with matters of energy, water management, sustainable materials, emissions reduction and so forth, matters of concern in the design and detailed design stage of a project which in this research had to be excluded due to time and resource limitations. Targets such as a threshold population size for urban neighbourhoods or villages, dwelling densities, socio-economic profiles of urban communities and other characteristics important for the strategic planning and conception stage are sometimes not and frequently not consistently used.

The Implications of the Growing Demand for Additional Housing

A major driver of change is the growing demand for new housing as a consequence of the shrinking size of households in the UK. In 1961, the average household size was 3.1 persons but by 2001 the figure had shrunk to 2.4 (Office for National Statistics, Spring, 2002). The Barker Review estimates that by 2021 an additional 3.8 million new dwellings would be needed to reduce the trend in house prices to the European average and increase the supply of social housing to meet the needs of new households and reduce the backlog of needs (Barker, 2004).

The question that needs to be asked is at what density housing development ought to be built. It is very unlikely that the promoted minimum density of 30 dwellings/ha (DETR, 2000a) would generate the population density required to support local services and facilities; inhabitants would therefore be dependent on amenities elsewhere and on the car to get there. In addition, low-density development would cause a loss of undeveloped land, particularly in the south-east of England were pressure for new dwellings is greatest. The UTF recommended densities of 100 or better 150 people/ha for urban neighbourhoods to achieve fully viable public transport and local amenities, all accessible on foot (UTF, 1999). At the average UK household size of 2.4 persons in 2001, this equates to a development density of 42 to 63 dwellings/ha. The average household size of 2.2 persons used by Barton et al. (2003) would equate to a density of 45 to 68 dwellings/ha, which is considerably over the densities recommended by UK planning guidance. Dwelling density is, however, only one important factor for new housing development. Other planning guidance (DETR, 2000b) suggests that development, including housing development, should be co-ordinated with public transport planning and provision to improve access to, and promote a more intensive use of, public transport.

The need to tie development with accessibility appears to be a well-supported policy in the UK. The government's focus on transport – signalled by Planning Policy Guidance Note 13 (ibid.), and reinforced by its revision (DETR, 2001) – has encouraged developers to put greater emphasis on the accessibility of public transport and greater collaboration between developers, local transport authorities and transport providers. But there is also considerable resistance to the increase of housing densities. According to a government study (ODPM, 2003) a majority of developers had increased the density of housing applications in response to their guidance (DETR, 2000a), but two-thirds of the developers indicated that they had encountered barriers in raising density: local political concerns (73%), spacing standards (61%) and parking standards (59%). Other findings of this study suggested that higher densities tended to be resisted in:

- lower-density suburbs because they were felt to be inappropriate in terms of design and character;
- weaker market areas where the policy priority is to increase the number of low-density higher-value dwellings;

8 Adapting the City

• some parts of northern regions – where local authorities sought new development to increase the range of type and size of dwellings, and meet modern housing aspirations.

This resistance reflects arguments enunciated particularly by Breheny (1997) that market forces lead to low density housing. Nevertheless there are also urban intensification arguments in order to preserve the countryside. This is frequently regarded as a false argument due to the relatively low percentage of land overall in the UK that is developed and the massive land reserves available for further urban development. Only 16.9% of all UK land is developed, whereas 83.1% of the land is undeveloped and used for agriculture, grazing, forests and woodlands and set aside land (Defra, 2006).

There are, however, two important counter environmental and ecological arguments:

- The necessity to preserve as much of the countryside and biologically productive land as possible in order not only to maintain and strengthen biodiversity, but also to be able to use the land for the production of food, wood and materials to become more self-sufficient and reduce as a result the large ecological footprint of our towns and cities;
- The necessity to reduce car use in the city not only to reduce congestion and to give the public streets and squares back to the people but also to reduce fossil fuel consumption and pollution levels causing global warming.

Urban Intensification

Notwithstanding environmental and ecological arguments, it is indisputable that urban areas still require intensification if their population density were to be too low to support local amenities and public transport because (a) this would require their inhabitants to travel elsewhere to work, get their children to school, go shopping, use entertainment and recreation facilities, and because (b) travel would be car dependent and would generate pollution, contributing to global warming and climate change. Many of UK urban and suburban areas fall into that category.

It is also quite obvious that intensification of urban areas does clearly not automatically reduce vehicular traffic unless:

- first, the achievable population size is large enough and is socially sufficiently mixed to support local services and facilities;
- second, the distances between home and local amenities and public transport stops are walkable;
- third, the intensified area is located along an existing public transport route or, if this is not the case, the installation of such a public transport route becomes viable due to the achievable population size.

If intensification does not achieve these three conditions, it will result in an increase of vehicular transport in and out of intensified urban areas as more people live there but, due to the lack of viable public transport, need the car to travel to work and service outlets outside their living area and this will generate more exported traffic and congestion as well as pollution in more central areas. The research here therefore pursues intensification of areas only if located along existing or achievable new public transport corridors and specifically around public transport stops.

In short, the reduction of car-dependent travel is essential but can only be achieved if urban areas have a large enough socially mixed population to support local amenities and public transport. The arguments discussed in this chapter can now be summarised, and form the basis for the development of the rest of the chapter.

- The use of indicators or parameters of sustainable urban development on their own are not effective unless for each of them threshold and target values are formulated, against which development programmes and projects can be gauged.
- Current UK practice of the development of targets in ad-hoc discussions needs to be modified to use predefined target values of sustainable urban development that set a framework for the discussions between local authorities, clients and communities, developers and the planning and design professions. The target values may be prioritised in response to local conditions and constraints so that over- and under-achievements in different urban areas of cities can be balanced, but they should not be ignored.
- The measurable performance values of millennium villages and sustainable communities projects need to be carefully monitored and documented; feedback may well lead to the formulation of target values based on empirical evidence that is currently lacking.
- The need to accommodate a growing number of smaller households and to provide for a population increase makes new housing essential. Higher-density development inside built-up urban as well as rural settlements will help reduce ecological footprints.
- Intensification of urban areas is, however, only sustainable if (a) the areas to be intensified are located at nodes of existing or new viable public transport links and networks, and if (b) the achievable population size and density is high enough and socially sufficiently mixed to support local services and facilities and public transport that can be accessed on foot.

Developing a Tool for the Measurement of Sustainability

The chapter now develops a tool to establish systemically the intensification of urban neighbourhoods (that are seen as currently unsustainable) based on the threshold and target values that are derived from UN, EU and UK government recommendations, research publications and best practice case studies. It is clearly understood that

only a few of the values recommended by these sources are supported by empirical evidence, specifically those derived from the evaluation of pilot development schemes, and so are in need of being upgraded as soon as such evidence is made available. There is also an absence of social thresholds. But using no threshold and target values is clearly not the better option as the outcomes of such an approach are at best uncertain, at worst counterproductive.

The research focuses on the planning and conception stages of the regeneration of urban areas. Accordingly, only threshold and target values for indicators of sustainable development relevant to these two stages are derived from available sources and through case studies. The limitation of this research project to strategic issues of the planning and conception stage should not, however, be a problem for potential users of the tool developed here. Issues of design of urban neighbourhoods are covered in considerable detail by a growing number of publications, among them Bentley et al. (1985), Barton et al. (2003) and the English Partnerships (2000) to mention but a few.

The relevant key indicators for the planning and conception stage of urban regeneration are those concerned with the operational, social, and economic viability of urban areas that in turn generate the quality of life of those who live and work there. Therefore the development of threshold and target values cannot only respond to housing needs but has to focus on sustainable communities with their own services and facilities and public transport. The smallest planning entity and building block of the city for which targets are developed is therefore what is usually called an 'urban village' or an 'urban neighbourhood', the equivalent of a small market town. There are three categories of sustainability indicators for which values have to be formulated:

- Urban form characteristics of a sustainable urban neighbourhood
 - Access to the minimum required services and facilities and public transport.
 - The minimum required population size and density to support services and facilities and public transport.
 - The size, area and development density of a walkable urban neighbourhood (walking time and distance from edge to central area).
 - Graded densities from edge to centre of the urban neighbourhood.
 - Graded densities of urban neighbourhoods from edge to central location in a town or city and accessibility of core areas of neighbourhoods, districts, towns and city.
- Characteristics of a socially balanced and inclusive urban neighbourhood
 - A balanced population age profile.
 - A balanced mixture of dwelling and tenure types.
 - A balanced mixture of household types and sizes.
 - A socially balanced population expressing itself in a mixture of levels of qualification of people in working age, the health condition of the population, and indicators of deprivation.

- Economic characteristics of a sustainable urban neighbourhood
 - A balanced range of economically active and inactive people of working age.
 - A balanced range of property prices and rent levels as key to a socially inclusive urban neighbourhood.

The recommended values are listed in Tables 8.1, 8.2, 8.3, and 8.4. It is not possible within the framework of this chapter to make reference to all arguments that support these targets.

As noted above, there seems to be a lack of research publications and government guidance notes on threshold and target value proposals of a sustainable population profile. Although the call for socially inclusive urban neighbourhoods and communities is supported by virtually all political institutions, from the UN to the European Council and the UK governments, this seems to be the one of the concepts of sustainability most difficult to describe, let alone to achieve. Part of the reason is that there are enormous differences in the community profiles of existing conditions in rural and urban areas as well as in post-industrial cities and towns and those that had never seriously industrialised. The social target profile depends much on the current conditions of towns and cities and in turn on historic development factors. Social population target profiles need therefore to be developed on the basis of local conditions. For our research on Glasgow the recommended value profiles are predominantly based on statistical analysis of Scotland and Glasgow averages (Table 8.3).

The target value Tables 8.1, 8.2, 8.3 and 8.4 can be moderated by the prioritisation of characteristics and values. It can be anticipated, that local context conditions and constraints do not always allow target values to be achieved in full. Weights given to characteristics and values allow for the expression of priorities and permit the generation of a compromise solution. The scale of prioritising suggested in the target value tables is a weight from 4 to 1 with 4 expressing that a value is absolutely essential and must be achieved in full, and with 1 expressing that a value is not essential and may be compromised.

Application of the Tool for the Measurement of Levels of (Un)sustainability of Urban Areas

This part of the chapter is a short account of the application of the tool for the measurement of levels of (un)sustainability of urban areas. The process involved the investigation of a considerable number of urban areas located at the edge of the city (all selected suburban areas of the core programme) and in intermediate or inner-city locations (all Glasgow communities north and south of the River Clyde between city centre and Clyde Tunnel). Within the framework of this chapter a full account of this investigation is impossible. Therefore one area is selected to illustrate the application of the developed tool and the potential achievements if threshold and target values are rigorously pursued. The selected case is Drumoyne in Glasgow's Govan area.

Indicator	Target Distances	Target Uses	Source
Access to amenities at local hubs and neighbourhood	150–250 m walk local hubs	Primary school Doctor Corner shop Community park	Urban Task Force (1999), p. 31
	400–600 m walk neighbourhood	Public house Group of shops Post office Community office Community centre Access to public transport stop	
		Connectivity to other neighbourhoods, district, town and city centres and beyond A permeable street network An integrated public transport system	Research team; Urban Task Force (1999); Barton et al., 2003
Access to district and town amenities	District 1,500 m 2,000 m 2,000 m 2,000 m 1,500 m 2,000 m	Secondary school District Centre Superstore/larger shops Leisure Centre Playing fields Natural green space (outside district) Access to public transport node	Barton et al., 2003, pp. 96–99
	Town 5,000 m 5,000 m 5,000 m	Health centre Library Sports centre Cultural/entertainment centre Higher education General hospital Access to public transport node	Urban Task Force (1999), p. 31
Access to city amenities	City, up to 20 km	Stadium Cathedral City Hall Key museums Major theatre + public facilities University Regional exhibition centre	Urban Task Force (1999), p. 31

 Table 8.1
 Services and Facilities at Urban Neighbourhood, District, Town and City Cores

	Table 8.2 Built Form C	haracteristic	s of Urban Neighbourhood (NBH)	
Indicator	Threshold Value	Target Valı	ue	Source/comments
Population	7,500	7,500 ¹ NBI 9,000 ² NBI 10,500 ² NF	H around district core H around town core BH around city core	Urban Task Force Report (1999), p. 61; Research Team
Distance edge to centre	670 m	Best Average Worst	530 m 600 m 670 m	Research Team
Total area	140 ha (100%)	Best Average Worst	88 ha 113 ha 140 ha	
Size of housing land	84 ha (60 %)	Best Average Worst	53 ha 68 ha 84 ha	
Size of non-housing land (communal area & space, amenities, workplaces)	56 ha (40 %)	Best Average Worst	35 ha 45 ha 56 ha	
Size of housing land	68 ha (60 %)	53 ha (60 %	(4	
Gross population density	53.6 ppha	Best Average Worst	85 ppha (persons/ha) 66 ppha 54 ppha	
Net population density	89.3 ppha	Best Average Worst	141 ppha (persons/ha) 110 ppha 89 ppha	

172
		Table 8.2 (co	ntinued)	
Indicator	Threshold Value	Target Value		Source/comments
Net dwelling density at 2.1 persons/household	52 dpha	Best 6 Average 5 Worst 4	57 dpha (dwellings/ha) 22 dpha 12 dpha	Glasgow average 2.1 persons per household - needs to be adjusted to local conditions
Graded density of urban quarter edge to centre, near district core		52 dpha at ed 67 dpha at in 82 dpha at co average 67 dj	lge termediate location ore pha	
Graded density of urban quarter edge to centre, near town core		67 dpha at ed 82 dpha at in 97 dpha at co average 82 dj	lge termediate location ore pha	
Graded density of urban quarter edge to centre, near city core		82 dpha at ed 97 dpha at in 112 dpha at c average 97 dj	lge ttermediate location core pha	
Connectivity to other neighbourhoods, district, town and city centre and beyond		A permeable (no cul-de An integrate	street pattern -sacs, all streets connected) d pubic transport system	
Key:				

ha: hectare pphh: persons per household dpha: dwellings per hectare NBH: neighbourhood

Indicator		Target Value	Source/comments
Population age profile	age 0–15 age 16–64 age 65+	20% of population 64% of population 16% of age profile	UK and England average; close to Scotland average
Mixture of dwelling types for graded	Close to district core	32% flats 68% family homes (with garden)	
density in neighbourhoods	Close to town/city core	56% flats 44% family homes (with garden)	
Household sizes		42.0% 1 person households 29.5% 2 person households 14.5% 3 person households 9.5% 4 person households 4.5% 5+ person households	Glasgow average suitable for Govan neighbourhood areas
Mixture of tenure types		39% social housing 61% owner-occupied housing	Target is the Glasgow average to achieve an equitable distribution of households in need of social housing provision
Qualification of people in working age 16–74		 42.0% without qualification 21.0% with level 1 qualification 13.5% with level 2 qualification 6.0% with level 3 qualification 17.0% with level 4 qualification Target is also an intensive education and training programme for those without qualifications 	Target is the Glasgow average to achieve an equitable distribution of underperforming people
Other indicators of poverty and deprivation	% of population wi % of population in % of persons in we SIMD ranking of n	th LLTI come deprived orking age claiming benefits eighbour-hood population	LLTI = limiting long-term ill-health SIMD = Scottish Index of Multiple Deprivation

 Table 8.3
 Social Characteristics of Socially Inclusive Urban Neighbourhood

Indicator	Target Value	Source
Economically active	65%	Scottish average
Economically inactive	35%	
A balanced range of property prices as key of social inclusion	For Glasgow: 39% social housing 25% affordable owner-occupied housing 36% upper market owner-occupied housing	Survey of property prices in the Govan area of Glasgow, 2006; Glasgow average

Table 8.4 Economic Characteristics of an Urban Neighbourhood

The Investigation of the Govan Areas

A preliminary investigation of the Govan areas generates secondary data by using MapInfo, the valuation roll (property taxation register), Census statistics, ArcGIS, and information provided by the city council. Additional primary data on housing conditions, the quality of services and facilities and their location and catchment populations, are collected in a number of area site visits. The data are collated in maps as well as built form, social and economic characteristics sheets. It is not possible to include this material here, but a brief outline of the key characteristics of the four Govan areas is outlined to enable an understanding of the location of Drumoyne in the Govan area, its socio-economic context, and for each of the four areas the current location of services and facilities (marked with green and red circles) (Fig. 8.1).

- Drumoyne, described in more detail below, is one of the four potential neighbourhood areas of Govan, located to the west of Govan Centre and the Helen Street Industrial corridor (compare Fig. 8.1)
- Govan Centre is an area of considerable historical importance and acts as a town centre. Its housing area is, however just a fraction of what is should be, and so is the population, which also shows high levels of deprivation. Most of the local services and facilities, including a shopping centre, are centrally located and access is even. Expansion is hampered by the river north and industry south and is only achievable through relocation of some of that industry and the redevelopment of part of the river front area.
- Ibrox to the east of the central north-south industrial Helen Street Corridor has a fairly deprived population slightly below the threshold value of one neighbourhood but an area almost twice as large as the equivalent target value. However, much of the area accommodates industrial and sports uses and largescale relocation is not a viable option. There is, however, a good chance of developing one target size neighbourhood. With a few exceptions, local services and facilities are dispersed or located at the edge and access is uneven.



Fig. 8.1 Glasgow's central River Clyde Corridor and the urban areas investigated

• Kingston is, of all the Govan areas, the one with the best socio-economic profile and the most consistent higher-density tenemental development, but half of the area is occupied by industry the relocation of which is not a viable option. Due to the high quality and density of development and the reasonably good population profile there is no chance and no need to remodel the area.

It is clear that with regard to threshold and target values there are serious problems in all Govan areas but Kingston. There are, however, also problems with the built form and the land uses in the Govan areas (compare Fig. 8.1): a considerable incoherence of development form and structure; a very high percentage of the land used for industry that fragments the existing housing areas and potential neighbourhoods; a fragmented street network in Govan's central area, specifically the Helen Street Industrial Corridor, that results in the areas west and east of the industrial corridor to be completely disconnected except for an east-west road north and south at a distance of 1,000m, with the industry in between as an impermeable barrier; a lack of pedestrian and vehicular links from Govan to areas north of the River Clyde, specifically in the central and western areas of Govan, which contributes to the isolation of Govan from the areas of Glasgow's west end and to the socio-economic divide of the well-to-do north and the marginalised south of the river.

Investigation also shows that much of the land along the river – at Govan Centre and the areas west of it – is industrial or disused industrial land that has a considerable potential from new mixed-use development. Except for the 1960s

housing estate immediately east of Govan Centre, the river front areas further east are already developed or under development – like the area east of Prince's Dock with the Science Museum, the BBC and ITV Television companies, new executive housing and a new bridge across the Clyde – or is earmarked for development – like the Graving Docks west of Prince's Dock for which a high-density housing scheme has been proposed. It seems a pity that, with the exception of the Govan Centre area itself for which a masterplan has been developed, the potential reuse of other river front areas west of the Graving Docks is not yet investigated.

The key objectives for the overall Govan area are formulated on the basis of this investigation: (a) to remodel all areas such that their potential value profiles are as close to the threshold and target values, (b) to develop all neighbourhoods with their own central services and facilities to achieve even access on foot, and (c) to link all neighbourhood centres with each other by public transport and connect the Drumoyne and Govan Centre areas across the river with Glasgow's west end neighbourhoods.

Detailed Investigation of the Drumoyne Area

Of all areas, Drumoyne to the west (compare Fig. 8.1) shows the most continuous development of housing, but most of it at rather low densities. With 8,240 inhabitants in 2001, the population size is above the threshold value of one neighbourhood, but the total area (excluding the Southern General Hospital area) of 157 ha is almost twice as large as the target areas of two best scenario neighbourhoods with 88 ha each, and this results in a net density of 39 dwellings/ha which is way below the threshold value of 67 dwellings/ha. There is reasonably good access to public transport to the north of the neighbourhood where also most services and facilities are located, but for the southern parts access on foot is difficult. The centre of Drumoyne, where amenities would best be located to achieve even access, is a void: a large secondary school and playing fields and a run-down park have little to offer the local population. But the potential of this area to form a new neighbourhood centre is high. The most problematic characteristic of Drumoyne is the very high level of deprivation.

Table 8.5 shows that Drumoyne's overall area of 157 ha is 65 ha above one best scenario neighbourhood (7,500 people, 88 ha area, 67 dwellings/ha) and 19 ha short of two best scenario neighbourhoods. It is therefore suggested that two neighbourhoods should be formed, one with its centre at the secondary school south of Pirie Park (with an area of about 9 ha) and one with its centre at Govan Road to the north where many shops and community facilities are located but currently struggle for survival. The waterfront area immediately east of the Clyde Tunnel (with and area of about 14 ha) is largely disused and offers the potential of expanding the northern neighbourhood towards the river. The use of Pirie Park and the waterfront area generate 23 additional hectares and a sufficiently large total land area for two neighbourhoods. To that end the number of dwellings, at 2001 level a total of 4,076, has to be extended by a little more than 3,000 new dwellings, around 700 of which

Otal land needed 176 ha Choice: 2 NBHs at 7,500 population each Needed dwellings 7,143 Needed population 15,000	Shortfall –991		Land Available land 157 ha Total land needed 176 ha Add. land required + 19 ha Pirie Park 9.27 ha Riverfront 13.94 ha Total gain 23.21 ha	 Option For 1 NBH area is 17 ha over worst scenario For 2 NBHs area is 19 ha short of best scenario (see Table 8.2) Land available for new build: Pirie Park 9.27 ha Waterfront east of tunnel 13.94 ha Waterfront east of tunnel 13.94 ha Choice: 2 NBHs at 7,500 population each graded density) Developed at 82 dpha (NBH core area graded density) Developed at 112 dpha (City core area graded density) 2 intensified NBHs 	Dwellings Existing dwellings 4,076 Existing dwellings 4,076 Needed dwellings 7,143 Needed dwellings +3,067 Added dwellings +760 Added dwellings +1,561 Added dwellings +1,561 Total dwelling gain +2,321 Shortfall -991	Population at 2.1 pphh Existing population 8,240 Needed population 15,000 Needed population 16,7
Available land 157 ha For 1 NBH area is 17 ha over worst Existing dwellings 4,076 Existing population 8,240 Available land 157 ha scenario Existing population 8,240 For 2 NBHs area is 19 ha short of best scenario (see Table 8.2) Land available for new build: Price Park 9.27 ha • Waterfront east of tunnel 13.94 ha	Available landIS7 haFor 1 NBH area is 17 ha over worstExisting dwellings 4,076Existing population 8,240RenarioFor 2 NBHs area is 19 ha short of bestscenarioExisting dwellings 4,076Existing population 8,240For 2 NBHs area is 19 ha short of bestFor 2 NBHs area is 19 ha short of bestExisting dwellings 4,076Existing population 8,240For 2 NBHs area is 19 ha short of bestFor 2 NBHs area is 19 ha short of bestFor 2 NBHs area is 19 ha short of bestExisting dwellings 4,076Existing population 8,240For 2 NBHs area is 10 ha short of bestPark 9,27 haPark 9,27 haNeeded dwellings 7,143Needed population 15,000Add. land needed 176 haChoice: 2 NBHs at 7,500 population eachNeeded added dwellings 4,067Needed population 15,000Add. land required +19 haChoice: 2 NBHs at 7,500 population eachNeeded dwellings 4,143Needed population 15,000Piric Park 9,27 haDeveloped at 82 dpha (NBH core areaAdded dwellings +1,561Needed added population 15,000Piric Park 9,27 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Needed added population 13,000Riverfront 13,94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Added dwellings +1,561Total gain 23.21 ha2 intensified NBHsTotal dwelling gain +2,321	Available land157 haFor 1 NBH area is 17 ha over worstExisting dwellings 4,076Existing population 8,240For 2 NBHs area is 19 ha short of bestscenarioFor 2 NBHs area is 19 ha short of bestExisting dwellings 4,076Existing population 8,240For 2 NBHs area is 19 ha short of bestscenario (see Table 8.2)Land available for new build:Eand available for new build:Eand available for new build:Total land needed 176 haWaterfront east of tunnel 13.94 haNeeded dwellings 7,143Needed population 15,000Add. land required +19 haChoice: 2 NBHs at 7,500 population cachNeeded dwellings +3,067Needed population 15,000Pirie Park9.27 haDeveloped at 82 dpha (NBH core areaAdded dwellings +3,067Needed population 15,000Pirie Park9.27 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Stored added population 15,000Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Stored added population 15,000Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Stored added population 15,000Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Stored added dmelling s+1,561Total gain23.21 ha2 intensified NBHsTotal dwelling gain +2,321Shortfall -991	Land	Option	Dwellings	Population at 2.1 pphh
	Total land needed 176 haChoice: 2 NBHs at 7,500 population eachNeeded dwellings 7,143Needed population 15,000Add. land required +19 haDeveloped at 82 dpha (NBH core areaAdded dwellings +3,067Needed added population +6,7Pirie Park9.27 haDeveloped at 82 dpha (NBH core areaAdded dwellings +760Needed added population +6,7Pirie Park9.27 haDeveloped at 82 dpha (NBH core areaAdded dwellings +760Needed added population +6,7Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Added dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Added dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Added dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dpha dpha dpha dpha dpha dpha dpha dph	Total land needed 176 haChoice: 2 NBHs at 7,500 population eachNeeded dwellings 7,143Needed population 15,000Add. land required +19 haDeveloped at 82 dpha (NBH core areaAdded dwellings +3,067Needed added population +6,7Pirie Park9.27 haDeveloped at 82 dpha (NBH core areaAdded dwellings +760Needed added population +6,7Pirie Park9.27 haDeveloped at 82 dpha (NBH core areaAdded dwellings +760Needed added population +6,7Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Staded density)Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Staded density)Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Staded density)Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Staded density)Riverfront13.94 haDeveloped at 112 dpha (City core areaAdded dwellings +1,561Staded density)Riverfront23.21 ha2 intensified NBHsTotal gain +2,321Shortfall -991		Land available for new build:Pirie Park 9.27 haWaterfront east of tunnel 13.94 ha		
Total land needed 176 ha Choice: 2 NBHs at 7,500 population each Needed dwellings 7,143 Needed population 15,000	Function 23.21 ha Developed at o 2 upta (NDH Core area Added dwellings +1.561 Riverfront 13.94 ha Developed at 112 dpha (City core area Added dwellings +1.561 Total gain 23.21 ha 2 intensified NBHs Total dwelling gain +2.321	The rate Developed at o 2 upta (NDI COL area Added dwellings + 1,561 Riverfront 13.94 ha Developed at 112 dpha (City core area Added dwellings + 1,561 Riverfront 13.94 ha Developed at 112 dpha (City core area Added dwellings + 1,561 Riverfront 13.94 ha Developed at 112 dpha (City core area Added dwellings + 1,561 Total gain 23.21 ha 2 intensified NBHs Total dwelling gain +2,321 Shortfall -991 1001 1001 1001	Total land needed 176 ha Add. land required +19 ha Dirig Dorb 0 27 hg	Choice: 2 NBHs at 7,500 population each	Needed dwellings 7,143 Needed added dwellings +3,067 Adad Austlings (760)	Needed population 15,000 Needed added population +6,7
Add. land required +19 ha Needed added population +6,0 Needed added population +6,0 Needed added population +6,0	Riverfront 13.94 ha Developed at 112 dpha (City core area Added dwellings +1,561 graded density graded density Total gain 23.21 ha 2 intensified NBHs	Riverfront 13.94 ha Developed at 112 dpha (City core area Added dwellings +1,561 graded density) graded density) Total gain 2.3.21 ha Zotal gain 2.3.21 ha Z intensified NBHs Shortfall -991	FILIC FAIR 9.2/ 114	Developed at 6.2 upita (INDIA core area graded density)	Added dwellings +/00	
Add. land required +19 ha Needed added population +6 ,7 Needed added population +6 ,7 Needed added population +6 ,7 ha Developed at 82 dpha (NBH core area Added dwellings +760 graded density)	Total gain 23.21 ha 2 intensified NBHs Total dwelling gain +2,321	Total gain 23.21 ha 2 intensified NBHs Total dwelling gain +2,321 Shortfall -991 Shortfall -991	Riverfront 13.94 ha	Developed at 112 dpha (City core area graded density)	Added dwellings +1,561	
Add. land required +19 ha Needed added dwellings +3,067 Needed added population +6 ,7 Pirie Park 9.27 ha Developed at 82 dpha (NBH core area Added dwellings +760 graded density) Kiverfront 13.94 ha Developed at 112 dpha (City core area Added dwellings +1,561 graded density)		Shortfall –991	Total gain 23.21 ha	2 intensified NBHs	Total dwelling gain +2,321	

178

ha: hectare pphh: persons per household dpha: dwellings per hectare NBH: neighbourhood

can be built on Pirie Park and around 2,000 on the riverfront site; the rest of new housing will be achieved by developing gap sites, disused plots of land and by replacement of poor quality and largely unused housing.

In 2001, 50% of the 4,076 existing dwellings are social-rented; roughly 50% are private-rented and owner-occupied. To accommodate two neighbourhoods with a population of 7,500 each requires, at 2.1 persons/household (Glasgow average), a total number of about 7,140 dwellings, which means that 3,060 new dwellings have to be build. To achieve an overall target percentage of 39% social-rented dwellings and 61% owner-occupied and private-rented dwellings (the Glasgow average), of the new dwellings 750 should be social-rented, 2,320 should be owner-occupied and private-rented of which 580 dwellings (25%) should be affordable housing. This would seriously improve the socio-economic profile of the population of both Drumovne neighbourhoods (Fig. 8.2), specifically if the building of private sector housing to the magnitude as indicated above would be supported by substantial training and education programmes to get currently long-term economically inactive people back to work. This account clearly demonstrates how strongly threshold and target values guide regeneration briefs and how the potential outcome of regeneration will make a substantial contribution to the formation of two sustainable neighbourhoods.

The detailed investigation of the Govan areas shows that it is possible to achieve two best scenario neighbourhoods in the Drumoyne area, expand Govan Centre to



Fig. 8.2 Computer model of the Drumoyne neighbourhoods as existing and proposed

become a full-size best scenario urban quarter, and restructure the Ibrox area to become a best scenario neighbourhood as well. In addition, two new neighbourhood centres became possible, one in Drumoyne South, one in Ibrox; furthermore the core areas of Drumoyne North and that the Govan Centre area have the potential of being greatly reinforced by enlarging areas and catchment population. This would be a considerable step towards the sustainability of Govan at large, because all neighbourhoods would have their own amenities in walking distance from peoples' front doors but also because these centres would be directly linked by public transport. It has not, however, possible to establish the response of local communities and stakeholders to the proposed restructuring and intensification of the Govan areas and to develop a commonly agreed compromise solution.

One last observation can be made regarding the economic viability of housing development of the scale proposed for Glasgow's Govan area and the question whether there would be opposition by the local communities to generate more social integration and improve the socio-economic profile of the existing communities by suggesting a substantial increase of owner-occupied housing in the area. In Govan there is a good mixture of tenure and dwelling types as well as of old and new build. According to the Register of Sasines for sale in 2006 and the Glasgow City Council 2006 Land Audit, property prices ranged from £3,615 for a first floor flat to £309,995 for a new build sixth floor property at the Garden Festival development. The most expensive property outside the Garden Festival Park development was a second hand non-flatted property that sold for £200,000. Only five years back such high prices in Govan would have been unthinkable. It seems that new development at riverfront locations has wetted the appetite of developers to push further south into Govan areas. Our proposal to link the riverfront development with the neighbourhood centres might therefore have unknowingly followed a trend that seems to have started. This for us has changed our rather pessimistic view on the potential to regenerate Govan's areas at the beginning of the project. If explored with some rigour, this trend would be for Govan perhaps an important lifeline.

Conclusions

As stated at the beginning of this chapter, the main objectives of the research project summarised here were to develop a theoretical underpinning of, and a methodology for, urban intensification that could help transform currently unsustainable urban areas into sustainable urban neighbourhoods and communities. To achieve these objectives, the project develops target and threshold values of sustainable urban settlements and thereby generates a tool to offer important guidance for the development of regeneration and intensification programmes. These values and tool are then rigorously applied to a number of case studies not to generate specific regeneration programmes for particular urban settlements, but to evaluate the potential impact of the established threshold and target values on the form, spatial organisation and structure and the socio-economic profiles of interconnected

8 Adapting the City

and integrated neighbourhoods and districts. The exercise has sought to establish the general significance and effectiveness of the tool. The case study tests helped formulate a number of important conclusions rather clearly:

- The values on which the tool is based offer strong and systematic guidance regarding the characteristics that urban neighbourhoods and communities, and the spatial, physical and socio-economic networks they form, ought to have in order to become sustainable.
- The case studies clearly demonstrate that it is possible to restructure currently unsustainable urban areas into a series of integrated sustainable neighbourhoods, all with their own amenities in walking distance from peoples' front doors and all with their local centres directly linked by public transport, provided the sustainability values, and the tool based on them, are rigorously applied.
- The case studies furthermore show that it is possible to physically and visually link neighbourhood cores with upmarket flagship development projects, and thus achieve a degree of social and spatial integration of the stronger and weaker members of urban society.
- The case study tests, however, make it clear that to achieve targets can be a somewhat painful process and requires strategic and multilateral thinking and the courage not to compromise too soon. As long as the existing conditions of underperforming urban areas, specifically their social and spatial segregation, are considered untouchable no sustainable solutions to current problems will be achieved.
- Most importantly, if rigorously pursued, as the investigation of the Govan areas of Glasgow demonstrates, the value-based tool is capable not only of helping to restructure in terms of sustainability currently underperforming individual urban areas into sustainable urban neighbourhoods, but also of interconnecting such neighbourhoods to form physically, spatially and socio-economically balanced urban districts, towns and cities.
- The need to think laterally and holistically is an important key message gained from the theoretical and case investigations:
 - Individual urban regeneration projects do not generate a balanced neighbourhood unless they are planned and executed within a planning and design framework for the entire neighbourhood.
 - In turn, individual sustainable neighbourhoods do not generate a balanced town or city unless they are planned and executed within a structuring framework for the entire town and city.
 - It is thus evident that sustainable urban regeneration, restructuring and intensification of urban areas demand urban planners and urban designers to generate not only land-use strategies but also formal, spatial, transportation, social and economic strategies for urban neighbourhoods and districts and for the town, city or even city region at large.

It is accordingly pertinent to understand that, first, the pursuit of sustainability threshold and target values, and the development of urban regeneration programmes

for specific urban areas based on them, requires an integrated and coordinated multilateral approach involving not only all stakeholders but also all relevant professional disciplines, and that, second, the generation of such individual programmes requires a town or city-wide framework based on threshold and target values for the town and city at large. Strategic planning and design at urban neighbourhood and district and at town and city level on the basis of a set of threshold and target value, supported by all those involved in and impacted upon by urban development, are thus the precondition for achieving sustainable urban regeneration and development. Local governments as well as professionals and academics dealing with urban regeneration and development need to recognise that they require a much wider range of skills than those they currently have. It is essential to rethink educational, professional and administrative responsibilities and competences and the role of developers, investors, community groups, associations and cooperatives if sustainable urban form and development is to become achievable.

References

- Barker, K. (2004) Review of Housing Supply Delivering Stability: Securing our Future Housing Needs, Final Report – Recommendations. Norwich, HMSO.
- Barton, H., Grant, M. and Guise, R. (2003) Shaping Neighbourhoods A Guide for Health, Sustainability and Vitality. London and New York: Spon Press.
- Bentley, I., Alcock, A., Murrain, P., McGlynn, S. and Smith, G. (1985) *Responsible Environments A manual for designers*. London: The Architectural Press.
- Breheny, M. (1997) Urban compaction: feasibility and acceptability? *Cities*, Vol. 14, No. 4, pp.39–51.
- Department of the Environment, Food and Rural Affairs (DEFRA) (2006) Environmental Protection Land Use and Land Cover, Key Facts about: Land Use and Land Cover, Land by agricultural and other uses: 1998 and 2005. London: Defra.
- Department of the Environment, Transport and the Regions (DETR) (2000a) *Planning Policy Guidance Note 3: Housing, (PPG3).* London: DETR.
- Department of the Environment, Transport and the Regions (DETR) (2000b) *Planning Policy Guidance Note 13: Transport, (PPG13).* London: DETR.
- Department of the Environment, Transport and the Regions (DETR) (2001) *Planning Policy Guidance Note 13: Transport, (PPG13 Revised).* London: DETR.
- English Partnerships (2000) *Urban Design Compendium*. Prepared by Llewelyn-Davies and Alan Baxter & Associates for English Partnerships and the Housing Birchwood: English Partnerships.
- Office of the Deputy Prime Minister (ODPM) (2003), *Delivering Planning Policy for Housing: PPG3 implementation study*. London: HMSO.
- Office for National Statistics (spring 2002) *Labour Force Survey Household Datasets*. Newport. [online] http://www.statistics.gov.uk/.
- United Nations (UN) (1993) *Earth Summit Agenda 21: The UN Programme of Action from Rio.* New York: United Nations.
- Urban Task Force (1999) *Towards an Urban Renaissance: Final Report of the Urban Task Force*. London: E & FN Spon.

Chapter 9 Neighbourhood Design and Sustainable Lifestyles

Katie Williams, Carol Dair and Morag Lindsay

Introduction

This chapter attempts to answer a seemingly simple question: do residents of new housing developments, built according to sustainability principles, behave any more sustainably than the population in general? The research takes thirteen 'sustainable' housing (or predominantly housing with some other uses) schemes in the UK and investigates how sustainably their residents behave. We are only interested in sustainable behaviours that are argued, in planning and urban design theory or policy, to be supported or enabled by the physical design of the schemes. We address this task by comparing our findings with those taken from national surveys and the 'core' case study neighbourhoods' survey. We also give evidence from our study (for ease, termed the 'sustainable behaviours' [SB] study) on residents' self-reported links between their behaviours and the design of their developments.

The chapter has the following structure. First we set out our rationale for undertaking the research, based on an argument that the 'behavioural' aspects of sustainable design are both under-researched and contentious. Second, we set out how the research is undertaken. Third we explain, in more detail, how we define sustainable behaviours, and how we derive our hypothesised links between specific behaviours and the design of housing developments. This is undertaken for three key areas of sustainable behaviour related to neighbourhood-scale design:

- residents' home-based sustainable behaviours; including reducing energy and water consumption, recycling and composting waste and supporting wildlife in gardens
- residents' travel behaviour and car ownership
- residents' 'social sustainability' behaviours; such as social participation and the use of local services, businesses and facilities

K. Williams (⊠)

Department of Planning and Architecture, University of the West of England, Bristol, UK

We then move to the empirical part of the project. We start by describing the sustainable housing schemes that we study, setting out which sustainable design features are present. Then we present some descriptive data about the sample of residents surveyed partly to explore whether any differences in behaviour can be explained by either socio-economic differences or by the fact that residents in this SB survey might have 'self-selected' to live in the new developments because they are already more sympathetic to a more 'environmentally friendly' way of life. We then present our findings on how the residents' behaviour scompare with the population in general in the three categories of behaviour set out above. Finally, we conclude with a discussion of what these analyses can and can not tell us about the extent to which 'sustainable' developments are associated with sustainable lifestyles.

It should be said at the outset that the SB survey was designed to enable further detailed statistical analyses to determine which physical features support specific sustainable behaviours. It is not the purpose of this chapter to present this aspect of the study, but to set out evidence of comparative behaviours and of specific, self-reported, links between actions and design. This is particularly interesting in relation to the core research case studies.

Why Do We Need to Investigate the Effectiveness of New 'Sustainable' Housing Developments?

It is now widely accepted that the ways in which the built environment is planned, constructed and used are unsustainable (Haughton and Hunter, 1994; Williams et al., 2000). For the last twenty years, finding more sustainable ways to develop the built environment has been the focus of much theoretical and practical effort. Alongside changes in strategic spatial planning policy, a number of discrete projects are being built according to sustainability principles (Williams and Lindsay, 2007). We are seeing 'sustainable business parks', 'sustainable holiday villages' and 'sustainable housing schemes'. All of these projects are designed and built with the aim of seeking a balance between environmental, social and economic performance, both now and in the future.

In the UK there are now a number of flagship residential projects that demonstrate some of the latest thinking in sustainable design. Schemes such as the Millennium Village at Greenwich (English Partnerships, 2004) demonstrate recent best practice, and many more like them are being built. These developments contain a number of physical features now commonly associated with sustainable design, such as energy efficient homes, infrastructure to promote walking and cycling, and a range of housing types and sizes. These sorts of schemes are commonly argued to have a number of benefits over 'normal' housing developments, which are often seen as land-rich, car oriented and socially exclusive.

How was the Relationship Between the Built Environment and Sustainable Behaviour Conceptualised?

What all the sustainable schemes described above have in common is an underlying assumption that they will contribute to sustainability in two distinct ways, which we have termed 'technical' and 'behavioural' sustainability (see Fig. 9.1, Williams and Dair, 2007). Technical sustainability means that the technologies, materials or design features used in the development perform effectively and contribute to sustainability in their own right. They fulfil sustainability objectives by virtue of being present in the development, but do not rely for their effectiveness on any specific behaviour by the users of the scheme. For example, using construction materials with low embodied energy ensures environmental benefits without the residents acting in a particular way.

Behavioural sustainability contrasts with technical in that it refers to the sustainable actions of those living, working and enjoying their leisure time in a development. It is argued that some elements of the built environment can enable or support behavioural sustainability (shown as the area of overlap in Fig. 9.1). For example, providing cycle paths and pedestrian routes is argued to encourage people to walk and cycle rather than drive their cars, and providing neighbourhood recycling facilities may encourage people to recycle their household waste. However, these features have no intrinsic sustainability value, unless they are used properly. There are also sustainable behaviours that are not reliant on the physical environment and can be carried out in any given setting (for example, ethical investing, shown in the right hand section in Fig. 9.1).



This chapter concentrates solely on 'behavioural' sustainability supported by the physical environment. We are interested in whether housing schemes built to enable sustainable behaviours actually make any difference to how people act in their daily lives. We recognise that both technical and behavioural aspects of sustainability are important in assessing the success of schemes, but we would argue that the behavioural performance of new 'sustainable' schemes is under-researched (notable exceptions are Lazarus, 2003; Butler, 2004; Mulholland, 2003). Furthermore, where research evidence does exist, it often casts doubt on the extent to which the physical environment can positively affect sustainable behaviour (Breheny 2004; Pett and Guertler, 2004).

Research Methods

The SB survey investigates the extent of sustainable behaviours in thirteen sustainable residential schemes in the UK shown in Fig. 9.2. The developments were chosen to give a spread and range of physical features to be examined. Each development is either solely residential or predominantly housing, with a range of other uses (such as shops, schools etc.). The developments chosen have been occupied for a minimum of two years, to enable behaviour patterns to 'settle'.

In each of the sustainable developments, the behaviour of residents is assessed via a questionnaire, administered to homes, that asks about current actions in a number



Fig. 9.2 The Case Study Locations

of key areas such as travel and energy use. The questionnaire was administered to all households in smaller developments and a sample in larger ones: 659 completed questionnaires were collected: a response rate of 34%. The physical features of the developments are assessed using a sustainability checklist that lists all the elements that could support sustainable behaviour and, potentially, be provided in a scheme. Researchers completed one checklist for each development by analysing architectural plans and drawings, and undertaking site surveys.

Comparison Surveys

In order to establish if the residents of 'sustainable' schemes are behaving any more sustainably than the population in general, nine comparison surveys are used. These are the core research, and eight national surveys, used as appropriate given their coverage (see DEFRA, 2001, 2007; ONS, 2005; 2006; DfT, 2005; Scottish Executive, 2002; and DCLG, 2005).

Understanding How Sustainable Design Features in Housing Schemes Could Support or Enable Sustainable Behaviour

In order to carry out the research, we had to understand how neighbourhood design is argued to enable sustainable behaviours. We developed a research framework, summarising a number of key hypotheses, by undertaking a literature review which covered planning policy statements, design guidance, and empirical and theoretical material on sustainable design and planning (Williams and Dair, 2007). In order to carry out this review, we needed to define some of the key terms being used in the research.

First, we needed a working definition of 'sustainable behaviour'. The one we adopted is developed from the conceptualisation of sustainability used in the global policy context. This sees three objectives: environmental protection (in terms of reducing resource consumption, waste and pollution); social development (equity and justice); and, more controversially, economic growth, being collective goals of societies. If these goals are to be achieved then people need to behave in certain ways. Hence, 'sustainable behaviours' are behaviours by individuals or groups that contribute to these three sustainability objectives. As a setting for sustainable behaviours, various spatial scales of human activity have been targeted, for example, 'sustainable cities', 'sustainable neighbourhoods' and 'sustainable homes' (Williams et al., 2000; Haughton and Hunter, 1994).

We are interested in behaviours related to (but not bounded by) residential neighbourhood contexts. Behaviours relevant at the neighbourhood scale, which could in some way be enabled or supported by the physical environment are identified. Only those behaviours that could contribute to sustainability in a relatively 'mainstream' built environment are included. We do not include behaviours common only to groups of people with particular philosophies, for example on communal-living or low-tech eco-design. The focus is a built environment that supports a contemporary society and economy, and may require some lifestyle adjustments, but not major cultural or practical changes.

Second, we refine our definition of the 'physical' or design features to study. We include any physical feature of the built environment claimed to support sustainable behaviour at a scale from the individual home to the neighbourhood. In terms of home-based technologies we draw a line at features that were included in the building and fitting of a new dwelling. Hence, we include items such as inhouse grey water systems and energy-efficient boilers that were installed when the schemes were built, but exclude features subsequently introduced by the residents, such as energy saving light bulbs. Most physical features are relatively easy to identify and quantify: measures were devised for elements such as cycle paths and recycling facilities. However other features are less tangible and some are more qualitative. For example, in the literature, 'high quality' designs are commonly associated with behaviours that help to develop social capital (see below), yet defining high quality is subjective and complex (Dempsey, 2007). We study both discrete design features, and also specific 'qualities' or 'levels' of provision that are claimed to be required for that feature to contribute to sustainable behaviour.

Which Behaviours Did We Study?

As stated above, the study examines behaviours clustered in three key areas: home-based sustainable behaviours, travel behaviour and car ownership, and social sustainability behaviours. These are the behavioural categories that are most commonly cited in literature, policy and design guidance as being affected by neighbourhood-scale design. The next sections of this chapter set out why each is important, and how each is argued to be supported by neighbourhood design.

Home-Based Sustainable Behaviours

Four sustainable behaviours are considered under this category. These are: reducing energy consumption; using water efficiently; recycling and/or composting waste; and supporting wildlife.

Reducing Home Energy Consumption

It is commonly known that UK households need to reduce their energy use in the home to reduce consumption of finite resources, to cut down on pollution and to help reduce fuel costs. Many professions engaged in housing production have looked to residential design to see if and where features can be incorporated into new housing to support residents to reduce their energy consumption. Within homes, energy efficient heating systems can be fitted and residents need to use the systems properly to make savings. For example, they need to time heaters and heating systems to be on only when someone is at home, to set thermostats on heaters and heating systems to the lowest temperature needed to satisfy their comfort needs, to leave empty rooms unheated (or at a low temperature) and to heat only the water they need.

Using Water Efficiently

Reducing the amount of mains water used in the home is an important aspect of sustainable behaviour for several reasons. It conserves scarce water reserves, it limits abstraction and any consequent environmental damage, and it lowers the amount of waste water discharged which in turn helps prevent flooding. In order to enable or support residents to use less water in the home, several physical features can be provided, such as grey water recycling systems, rainwater recycling systems, garden water butts, and dual flush toilets.

Waste Recycling and Composting

Recycling and composting waste are included in the study because they are significant behaviours in terms of environmental impact (DEFRA, 2005a). A number of physical features of new residential schemes are argued to encourage residents to recycle their waste. Within homes, this includes space to sort waste material at source. In multiple occupancy buildings, recycling bins can be provided in shared utility spaces. Refuse chutes can also be provided to deliver segregated waste straight into bins. Within developments, on-site collection facilities can be provided (Barton et al., 2003; Rao et al., 2000). Provision can also be made for composting organic waste, either in private gardens or in shared space for neighbourhood use (Brownhill and Rao, 2002).

Encouraging Wildlife in Gardens

Globally biodiversity is in decline. In the last 200 years more species have become extinct than at any other time in the last 65 million years (TCPA, 2004). Hence, protecting biodiversity is a central aim in seeking a more sustainable future (DEFRA, 2005b). It has been argued that, in urban areas, biodiversity can make a contribution to the quality of life of residents, workers and visitors and add economic value. Residents can encourage wildlife in their gardens by actions such as providing food (although some argue this interferes with eco-systems), providing 'natural' habitats such as ponds or 'undisturbed' areas, or by using only organic gardening techniques. The physical features in new developments argued to enable householders to do such activities include the provision of private outdoor space in the form of gardens, balconies and roof terraces (TCPA, 2004; Gaston et al., 2003).

Travel Behaviour and Car Ownership

Travel behaviour is one of the most pressing areas of behavioural change required to move society towards a more sustainable future. The predominance of car use, and reduced levels of walking, cycling and public transport usage have created significant environmental and social problems. Here we look at behaviours related to making fewer and shorter journeys using fuel-efficient modes of transport, and car owning.

Making Fewer and Shorter Journeys by Car, and Using More Fuel-Efficient Modes of Transport

Reducing the number and length of trips by fuel inefficient modes of transport (i.e. reducing travel demand, particularly by car) is crucial for a sustainable future as it reduces petrol (and diesel) consumption, and therefore pollutants that affect climate change and air quality. A modal shift away from the car and towards walking, cycling and public transport also has wider benefits: it ensures there are more people 'on the streets' which improves neighbourhood activity and safety, and reduces noise pollution. It also means that public transport services are more likely to be viable. In addition, it could help to improve public health (see DfT, 2004; Transport 2000 Trust, 2003).

Neighbourhood design features claimed to enable residents to make fewer, shorter and less car trips vary in scale and type. At the master planning scale it is argued that high-density developments within existing built up areas can enable most people to live near amenities, facilities and employment and thus reduce the need to travel (DETR, 2000c). Mixed-use developments are advocated for similar reasons (Barton et al., 2003). In addition, the appropriate design of the movement framework is seen as the best way to ensure that car use is limited (DETR, 1998). This means transport networks that are well integrated with the surrounding area, have dedicated, convenient, direct routes for pedestrians and cyclists, and are linked in a grid or deformed grid pattern (rather than a cul-de-sac configuration). They also need to be able to accommodate public transport and offer direct routes to interchanges (DETR and CABE, 2000; Llewelyn-Davies et al., 2000).

Due to the localised nature of short journeys, much importance is attached in design guidance to the detail of the public realm at the neighbourhood scale as a way to encourage walking, cycling and public transport use. A number of elements could contribute to ease of access, legibility, safety and convenience, which are significant in influencing peoples' travel mode choices (Llewelyn-Davies et al., 2000; Civic Trust, 2001; Brownhill and Rao, 2002; Transport 2000 Trust, 2003; DfT, 2003). Such features include signs and maps, adequate seating, and convenient places to park or store a bicycle in homes and at trip ends.

Car Owning

The vast majority of households in the UK (77%) now have access to at least one car. Clearly car ownership and usage rates are related: when people have access

to a car, especially if they have invested in buying and maintaining it, they often feel an incentive to use it, rather than use other modes, such as public transport. Designers of new sustainable residential developments have been keen to dissuade residents from owning one or more cars. Overall, the desire is to create places which do not 'feel' car oriented and that are comfortable to walk and cycle around. But, alongside the positive measures set out above to encourage walking, cycling and public transport, there are more punitive attempts to reduce car ownership, such as reducing the amount of parking or providing limited parking space for a pool of shared cars.

Social Sustainability: Social Participation and the Use of Local Services and Facilities

There are elements of neighbourhood design that are argued in policy and literature to be related to social sustainability (see Chapter Five for a fuller discussion). Here we are interested in the simple concepts of social participation (in community groups, for example) and the use of local services and facilities as indicators of social sustainability. We have picked these out for the reasons set out below.

Social Participation

Social participation is seen as the cornerstone for building and maintaining social capital, and in turn social capital is essential to avoid social exclusion (Social Exclusion Unit, 2001). Research findings indicate that people living in communities with high levels of social capital are more likely to benefit from personal wellbeing, lower crime rates, more empowerment and a higher quality of life than those living in communities where social capital is lacking (Healey and Côté, 2001; Putnam, 2001). Additionally, through positive social interaction, people become more attached to a particular neighbourhood and want to remain there and invest time and energy in maintaining community organisations. In this way, social networks of friends and family are built over time. A sense of 'belonging to' or 'ownership of' a neighbourhood is argued to be a key element in creating sustainable communities (Urban Task Force, 1999).

The relationship between social participation and the physical environment is neither direct nor easy to define, but a number of physical features are argued to support it. An obvious requirement is the presence of community facilities and amenities close to homes (Barton, et al., 2003). In addition, the *quality* of the environment is important (Urban Task Force, 1999; Gehl, 2001). For people to engage with each other they need to feel comfortable and enjoy their neighbourhood. In addition, people need to be able to access space for social interaction easily and efficiently. In this respect the physical features related to the transport behaviours above are also all relevant.

Use of Local Services, Amenities and Businesses

The use of local public services and businesses is argued to support sustainability objectives in a number of ways. First, it supports local economic sustainability: the more people use services, amenities and businesses, the more viable they become, and this in turn creates employment demand. It also supports local supply chains and helps retain money circulating within the local economy for longer periods, with the effect of increasing growth in that area (Dixon and Marston, 2003). There are also benefits in terms of increased vitality of the neighbourhood. There are higher numbers of people in the public realm and this in turn increases feelings of safety and reduces the need for formal security. It also increases opportunities for social interaction and helps guard against isolation and exclusion (Carmona et al., 2001), and makes use of any spare capacity in social provision (DETR, 1998).

As with social participation, the link between using local services, amenities and businesses and the physical environment is not always direct. However, a number of links are commonly made. First, in order to enable people to use local services and businesses, they must be provided nearby. Therefore buildings for services such as surgeries, schools, community centres and commercial activities (e.g. retail and workshops) need to be either provided in new developments, or the developments need to be located near to existing provision. The services and businesses will be more viable if they are located in high density developments with sufficient numbers of people to patronise them. Consideration also needs to be given to accessibility. Hence, the physical features supporting reduced travel demand and supporting the use of public transport, walking and cycling are relevant. High quality urban design, buildings and open spaces are also associated with the regular and frequent use of local facilities (DETR, 2000a and b; Carmona et al., 2001). Research evidence also suggests that the attractiveness of a place has a direct impact on the number of people spending time in an area and using its facilities.

We have now set out the main theoretical and conceptual thinking underpinning the SB study. The next sections of the chapter set out the empirical analysis. We start with a description of the thirteen case study schemes.

The 'Sustainable' Developments

The thirteen 'sustainable' schemes are spread around the UK. The locations are shown in Fig. 9.2. The main characteristics of the developments are set out in Table 9.1. It should be noted that all the information relates to the part of the development that is included within the site boundary for the research (or, in the case of 'uses' in the development or within 500m of the boundary). In some cases this was the whole development, in others part of it.

As Table 9.1 shows, the schemes vary in scale and in the sustainability features they have. Some have almost all features, while others have fewer (in the site survey this information is broken down into a finer grain of measures). This is taken into consideration in the analyses. Another interesting aspect of the schemes is their

 Table 9.1
 Sustainable Developments in Study

						•		•					
Name of 'Sustainable' deve	elopment												
	Grange Farm	Amersham Road	The Waterways	Alpine Close	The Courtyards	Great Notley Garden Village	Greenwich Millennium Village	Ingress Park	Lansdowne Gardens	Newcastle Great Park	Westoe Crown Village	The Staiths South Bank	Cooper Road
General profile data No of units Dwellings per hectare (net) Total no of uses ¹	39 26.0 1	172 27.1 5	291 42.0 2	27 42.0 2	104 32.5 2	265 28.0 3	303 153.0 5	216 32.0 2	215 38.7 2	175 29.1 2	122 87.1 2	159 55.0 3	68 29.9 1
Uses No. parks/play areas No. cafes, pubs, etc	4 0	v v	4 0	4 0	0 0		4 v	4 4	1	4 -	9	4 1	6 4
No of schools ² No of local shops	$\begin{array}{c} 1 \\ 0 \end{array}$		6	6	0 0	1 3	0 0	0 0	4 ω	r 7	8 61	0 0	1 2
Tenure % private homes % RSL homes ³	$\begin{array}{c} 100\\ 0\end{array}$	36 64	87 13	87 13	85 15	89 11	85 15	$\begin{array}{c} 100\\ 0\end{array}$	75 25	100 0	100 0	$\begin{array}{c} 100\\ 0\end{array}$	100
Sustainability features % homes + efficient heating systems ⁴	100	33	33	33	0	0	33	Ξ.	33	0	100	100	33
% homes with sustainable water features % homes with dual flush	100	26 26	0 100	0 100	64 100	0 100	0 100	0 100	14 100	0 100	0 100	0 100	100
WCs % homes with recycling Wildlife habitats % ⁵	100 33	100 33	0 67	0 33	64 33	0 33	15 0	100 33	25 0	100 33	0 0	0 33	0 0

9 Neighbourhood Design and Sustainable Lifestyles

					lable y.	I (conti	nued)						
Name of 'Sustainable' deve	lopment												
	Grange Farm	Amersham Road	The Waterways	Alpine Close	The Courtyards	Great Notley Garden Village	Greenwich Millennium Village	Ingress Park	Lansdowne Gardens	Newcastle Great Park	Westoe Crown Village	The Staiths South Bank	Cooper Road
Quality of development $\frac{\infty}{6}6$	89	66	95	89	95	93	95	98	82	91	93	93	89
Quality of routes $\%^7$ Public transport provision $\%^8$	64 43	61 71	76 43	66 86	71 71	53 57	81 56	63 56	63 57	64 11	68 22	64 33	49 86
Pedestrian routes % ⁹	68	65	85	38	58	50	50	88	10	55	38	88	20
Cycle routes % ¹⁰ Average parking standard per home	100 -	0 2.00	13 1.50	$0 \\ 1.50$	$13 \\ 1.50$	100	33 0.75	67 1.50	0 1.85	13 2.00	$13 \\ 1.50$	$100 \\ 1.00$	$0 \\ 1.94$
Notes: ¹ This is simple a count office, newsagent or foc open space. This count boundary) ² This includes pre-schc ³ RSL: Registered Socii ⁴ This is a composite wo ⁵ This is a composite wo ⁶ This is a composite wo ⁷ This is a composite wo ⁷ This is a composite wo ⁷ This is a composite wo ⁸ This is a composite wo ⁸ This is a composite wo ⁹ This is a composite wo ⁹ This is a composite wo ¹⁰ This is a composite wo ¹¹ The '-' indicates mis	of the m od store), is for use sol, prim: al Landlo eighted s re ad qua eighted s eighted s read qua sighted s read qua	umber of dif shopping ce es in the dev ary and seco ard core based c core based d inty of public iccore based d inks outside core based o secore based o inks outside inks out	ferent uses. intre or high elopment (i and SAP ratii in the site si on assessme on assessme on assessme on e.g.: nun on, e.g.: nun on, e.g.: qui	. Categc . Categc i.e. with i.e. with e develc ngs and urvey: the ngs and urvey: the ngs and the ents of <i>i</i> pment, <i>s</i> pment, <i>s</i> a fitty set a fitty set	rries were: s social space in the boun ppment or w heating syst he preservat a number of a number of bus routes, I bus routes, I bus routes, P a routes, P a number of bus routes, P bus routes, P a number of bus routes, P a num	chools, h dary of tl dary of tl ithin 500 ithin 500 features, features, features, ings and e number number	ealth facilitie blic house, res ne case study m of the boun ags. sting wildlife sting wildlife e.g.: predorr huing bus stops an direct routes t of cycle path	s, place of staurant, c area) and areas, the areas, the nance, litt inant stre d distance o outside of in our of	f worship or c afé'), indoor I nearby (with creation of n er, active fron et pattern, ex the developm the developm cr-access roul	community l leisure/spor in a 500m i ew wildlife ; ew wildlife ; ntage, buildi tent of natur nsport interv tes to outsid	nalls, loca ts facility, adius of 1 areas and ng qualit; ral surveil changes e the deve	ul store (e, park and the devel the mana y, attract llance, le	.g. post I public gement veness, gibility,

194



Fig. 9.3 Millennium Village Greenwich

architectural quality. Some have an 'eco-aesthetic', others are more traditionally designed. In addition, the extent to which the schemes are 'known' or 'marketed' as 'sustainable' varies considerably. Developments such as the Millennium Village in Greenwich, are well known for their 'green' aspirations, whereas in others, the sustainability elements are conveyed more subtly, or not at all. Figures 9.3, 9.4, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, and 9.11 give some images of the developments.



Fig. 9.4 Millennium Village Greenwich

Differences Between the Sustainable Behaviours Study and the Comparison Surveys

Before we move on to the results about behaviours, it is useful to comment on our sample of households and how it compares with the comparison surveys. Overall, the SB sample is quite similar to national averages on most key characteristics, such as household type and size, age, and way in which the homes are owned, except that two of our schemes are occupied wholly by tenants of Registered Social Landlords



Fig. 9.5 Amersham Road, Reading



Fig. 9.6 Alpine Close, Maidenhead

(RSLs). As would be expected, people in our sample have lived in their homes for far shorter a time than average: 40% of our sample had lived in the schemes for less than 2 years. This may affect some behaviours, such as social participation, that take time to develop. We also have a moderate over-representation of higher social



Fig. 9.7 Cooper Road, East Sussex



Fig. 9.8 Great Notley, Essex

classes in our sample: 24% higher managerial and professional, compared with a national average of 13%; and 38% lower managerial and professional, compared with 23% nationally. This may have some bearing on the sustainable behaviours as some have been shown in previous studies to be undertaken more by some socio-economic groups than others.



Fig. 9.9 Ingress Park, Kent



Fig. 9.10 Ingress Park, Kent

In terms of variations in urban form or design between the SB study and the samples, clearly our sample has far more sustainability features than either the core study or the national comparisons. There is also a wider range of densities in the core case study neighbourhoods (from 24.5 dph to 270.5 dph) than in the SB sample (26



Fig. 9.11 The Staiths, Gateshead

to 153 dph). The national comparator surveys cover both rural and urban areas, so some differences may be explained by these different contexts. Where data is broken down by area type in comparison surveys, this is mentioned.

Notwithstanding these variations, it should be stressed that it is impossible to control for all socio-economic or urban form variations in this type of research. No samples will ever be identical save one or two key built form variables: comparative built environment research always faces this problem. However, undertaking the comparisons presented here, using carefully selected descriptive data, has value in assessing the question posed at the outset about the relative behaviours of residents in 'sustainable' schemes. The inclusion of the self-reported links with the physical environment helps to shed further light on the behaviours.

Residents' Attitudes Towards, and Knowledge of, Sustainable and Environmental Issues

The first comparative task undertaken is to see if the residents of the SB schemes are more sympathetic to, or knowledgeable of, sustainability issues than the population as a whole. This is to determine any predispositions to behaving sustainably that may not be linked to the physical environment. First, residents were asked if they had heard of the term 'sustainable development': 64% of SB respondents had heard of the term, compared with only 32% in the DEFRA survey (2001). The SB study



How concerned are you about the environment in general?

Fig. 9.12 The proportions concerned about the environment in the SB and DEFRA surveys

then asked how concerned residents are about the environment (Fig. 9.12). Here, we found no significant difference between the SB study and the national average. This perhaps suggests that our sample is more knowledgeable, but not more concerned, than the population as a whole.

We also asked our sample if they feel they, or other people, need to change their behaviour so that other people can enjoy a good quality of life and environment in the future (Fig. 9.13). Overall, the SB sample residents have a stronger feeling that both 'themselves' and 'most people' need to change, but both surveys reveal a strong feeling that as well as personal change, other people need to do their bit.

Finally, we asked respondents in the SB study what had been the most important factors when choosing their home (Fig. 9.14). It is interesting that around a third of respondents state that energy and water efficiency was important (34% and 28%), and around 40% state that the quality of the development and of local facilities were factors. One would expect aspects such as size and type of home to be highly rated, but this set of responses shows quite a high degree of conscious prioritising of sustainability features when choosing a new home. Clearly, the high prioritising of parking space (48%) goes against this, but is consistent with most studies of requirements for new housing, particularly in higher social classes.



Do you, and other people need to change your/their way of life so other people can continue to enjoy a good quality of life and the environment?

Fig. 9.13 The proportions who see the need for change to improve quality of life and the environment in the SB and DEFRA surveys Note: SS: Scottish Survey



Important reasons for chossing your home

Fig. 9.14 Reasons for choosing a home in sustainable developments

Findings on Sustainable Behaviours

Home Energy Use and Water Efficiency

To see if the SB respondents are efficient in their use of energy and water in the home we asked them about their behaviours in a few precise areas that are also included in national surveys: turning off lights in empty rooms: heating rooms only when they are in them; and taking showers instead of baths. For all three behaviours, the SB sample is significantly more active than the population as a whole, as evidenced by a comparison with the DEFRA Survey 2007 (89% turn off lights, 56% only heat required rooms and 74% shower instead of bathe) (Fig. 9.15).

We then asked the SB sample if living in an energy or water efficient home had made them more cautious about how they used these resources. Figure 9.16 shows the responses only from those who are actually living in energy or water efficient homes in the sample. It shows a positive impact on behaviour, with the majority saying it has made them more cautious about energy (56%) and water use (62%), slightly fewer saying it has made no difference, and only a very small percentage saying they are now less cautious. This indicates a direct impact on behaviour due to the design of homes.



Energy and water efficiency behaviours in the home

Fig. 9.15 Energy and water efficiency behaviour recorded in the SB and DEFRA Surveys Note: The wording in the DEFRA and SB studies is slightly different, but the data are comparable



Has your energy/water efficient home encouraged you to be ...?

Fig. 9.16 Influence of energy/water efficient home on efficiency behaviour in sustainable developments

Waste Recycling and Composting

We then compared the percentage of the SB study that regularly recycle or compost their waste with national surveys (here we have used the DEFRA surveys from 2001 and 2007 as they illustrate that, in general, these behaviours are becoming more common) (Fig. 9.17). In terms of recycling, the SB study has a higher percentage of recyclers than the 2001 survey, and slightly fewer than the 2007 study. Overall though, almost 79% of SB respondents regularly recycle waste compared with 84% nationally. In terms of composting, the SB study performs less well than both DEFRA surveys. This could be partially due to the fact that composting rates are generally far higher in rural areas (48% compared with an average of 36%), but still does not explain the very low rates. Open-ended responses from the questionnaire suggest reasons for not composting are related to wanting to keep gardens pristine, or not having enough outdoor space.

We asked our sample which recycling facilities they used regularly, to see if local provision of facilities were being utilised (Fig. 9.18). Respondents could tick any facility they used on a regular basis. 55% used kerbside collections, 54% used local facilities, and 37% used facilities in their own homes.



Regular recycling or composting of waste

Fig. 9.17 Regular Recycling or Composting of Waste recorded in the SB and DEFRA Surveys Note: The DEFRA data relate to paper recycling only, the SB study does not differentiate



Which recycling facilities do you use regularly?

Fig. 9.18 Recycling facilities used regularly in sustainable developments

Encouraging Wildlife

In terms of encouraging wildlife, we compare our findings (Fig. 9.19) with some from the core survey. Interestingly, we find that in the SB study far fewer people provided food for wildlife (30% compared with 52%), and fewer maintained ponds than in the core study (4% compared with 8%). The core data is a useful comparison for this behaviour, because both studies are largely of urban populations, and the data shown here are only for respondents in both surveys that have access to private outdoor space. Reasons given in the SB survey for not providing food or a pond are that people liked their gardens to be tidy, they find wildlife a nuisance, and that they do not have enough space.



Supporting wildlife: do you provide food for wildlife (animals and birds)? Or have a pond?

Fig. 9.19 Support for wildlife recorded in the SB and core surveys

Making Fewer and Shorter Journeys by Car, and Using More Fuel-Efficient Modes of Transport

Although, as set out above, numerous claims are made about the travel impacts of new forms of development, it is very difficult to make comparisons because local context is so critical. Hence, here we have chosen the simple measure of



Mode of travel to main place of work or study

Fig. 9.20 Mode of Travel to Work recorded in the SB, Core, Census and DEFRA Surveys Note: the DEFRA, 2007 reference makes use of data from the Labour Force Survey (ONS, 2006)

mode of travel to main place of work. Clearly this does not cover travel for non-work uses, nor does it address frequency (which is addressed to some extent below) but it is the most readily comparable measure with wider data sets and is therefore useful. Figure 9.20 shows that fewer people walk to work in the SB study (9%) than in any comparable survey; this is at odds with the theory and policy advice on sustainable housing schemes. Fewer people drive to work than national averages (60% compared with 68% nationally), but more do so than in the core survey. Cycling is also slightly higher than national averages, but less than the core. However, public transport use is higher than all national comparisons and the core survey (24%), signifying perhaps some success in integrating public transport facilities. But, it should be noted that this result is slightly skewed by large numbers of respondents at Greenwich Millennium Village using the tube and bus services.

We asked the respondents who regularly walked, cycled or used public transport (for any trips, not just work) whether any of the common design aspects thought to encourage these modes of travel were important in their choices. We also asked them about non-design related public transport provision such as frequency of services. Just under 40% of people said that convenient pedestrian routes, well lit routes, and direct routes to local facilities were important (Fig. 9.21). The most commonly cited influence was the frequency of regular bus services (45%). We have to remember



Do the following encourage you to walk, cycle or use public transport?

Fig. 9.21 Factors that encourage People to Walk, Cycle or Use Public Transport

for this chart though, that not all the features listed are present in all schemes, so the relative benefit may actually be higher.

Car Owning

We asked SB respondents how many cars they had access to. The hypothesis is that the SB study residents will own fewer cars than the population as a whole. In contrast to this, our sample had the lowest percentage of car-free households in any of the surveys (13% compared with 32% in the core survey) (Fig. 9.22). The SB survey also had the highest percentages owning one and two cars. This result could be explained partially by the proportion of higher social classes in the study, who tend to own more cars, but considering the mainly urban locations and the limited parking, this is an unexpected result.

We asked those in the SB study who did not have access to a car if any of the 'stick' or 'carrot' measures designed to discourage car ownership had been important in their decision. We found that these measures had a negligible impact (Fig. 9.23). A lack of parking was hardly ever reported as a disincentive (5%), even though many of our schemes have limited allocations by recent standards. Positive 'carrots' such as good public transport facilities were only seen as important by 13% of the population. However, it may be that as most of our schemes had at least one parking space per household, the question would have been more revealing if it had asked about owning more than one car.


Number of cars or vans owned, or available for use by, your household?

Fig. 9.22 Number of Cars owned/available recorded in the SB, Core, National Travel and DEFRA Surveys



Are any of the following important to you in deciding not to own, or have regular use of, a car or van?

Fig. 9.23 Factors influencing Car ownership

Social Participation

As explained above, social participation is a complex concept. Here, we have chosen a simple measure of 'participation in local community or neighbourhood groups' as an indicator. This leaves out any measure of individualised and less formal social participation. The SB study results for participation are very similarly to those in the core study: the numbers involved are very low, with only 10% of respondents in the SB survey, and 13% in the core survey regularly taking part. Whilst there is no direct national comparator, the DEFRA 2007 survey records that half of the population (50%) had been involved in a social activity in their local area in the two weeks previous to the survey. This count allows for a wider definition of social activity, and does not directly imply 'regular' participation, but the percentage still seems significantly higher than the core and SB survey to raise the question of whether these rates are particularly low.

Use of Local Services

As with social participation, use of local services and facilities is a particularly difficult behaviour to measure meaningfully in a comparative context, as so much relies on the extent and provision of services. Hence, we have chosen to look not at absolute numbers of trips to facilities, but at the most commonly recorded frequencies of use to see if the SB residents use local facilities more frequently than



Frequency of use of local services

Fig. 9.24 Use of local services

those in the core survey. SB residents were asked about their use of facilities both 'in' their development, and 'outside their development but in the nearby area'. This distinction is obviously not relevant for the core sample, hence the differentiation is not made. Also, some of the categories of facilities are framed slightly differently in the two surveys. Hence, the data shown is presented exactly as it was collected from the respective surveys.

The facility that is most commonly used on a daily basis is the corner shop/convenience store in the core survey (Fig. 9.24). The local shops in the SB survey both inside the development and nearby are the next most commonly used on a daily and weekly basis. Community and sports facilities are also used relatively regularly (weekly) with cafes, pubs etc. used less frequently.

Conclusions

We began this chapter by asking the seemingly simple question: do residents of new housing developments, built according to sustainability principles, behave any more sustainably than the population in general? We have answered this by looking at key sustainable behaviours identified in policy and literature as being linked to neighbourhood scale design. This has been done through the use of descriptive data from the SB survey and comparable national surveys.

Overall, our answers are mixed, but more negative than positive. The residents of the 'sustainable' schemes only seem to behave more sustainably than the rest of the population in home-based resource efficiency behaviours, such as water and energy use. Results for recycling and frequency of use of local facilities are about the same as national comparisons. For most other behaviours, such as travel to work by car, owning (or having access to) a car, social participation, encouraging wildlife, and composting they behave less sustainably than the population in general. For around a third of residents energy and water efficiency is important in choosing their home, and many feel that living in a more sustainable house has either affected their resource consumption behaviour positively, or at least not changed it. Just under half of residents who regularly walked, cycled and used public transport feel supported in doing so by elements of neighbourhood design, but hardly any are dissuaded from owning a car. What we see, in general, are residents who are more knowledgeable about sustainability issues than the general population, but not necessarily more concerned or 'active'. However, as with all research of this type, the devil is in the detail and these results need further unpacking to give a more critical picture.

First, we need to consider the fact that these settlements are all new. As we stated above, many residents have only recently moved to them. This may have an impact on certain behaviours in comparison with more established places (unfortunately no comparable studies of 'normal' new developments exist). Behaviours related to social participation, for example, traditionally develop over time. Similarly, we find that behaviours like outdoor composting and encouraging wildlife in gardens are often unpopular because people feel their gardens are 'new and pristine' and want to keep them like this. Many also said they do not have enough space in the new development: a consequence of higher density policies. Second, not all the case studies have all the sustainability features (or only have them in small degrees), so in some cases it could be seen as 'unfair' to judge the case study in terms of consequential behaviours. Where appropriate in the analysis we present results via subsets (for example by selecting samples only of energy and water efficient homes), but in some instances this has not been undertaken as we are attempting a general comparison at this stage. To a certain extent we are interested in the totality of sustainable behaviours and also the extent to which they 'add up' to form sustainable lifestyles in these new settlements.

Third, as has been mentioned throughout the analysis, the SB survey does not have a representative social profile: it has higher proportions of managerial/professional residents than would be expected. This could be affecting the results: some negatively, for example, for issues like car ownership where we would expect higher rates, yet we would also expect higher rates of, for example, recycling and energy efficiency, but these results are more mixed. We should also say that in the case studies that were populated by RSL tenants some differences in both opinions and behaviours could be noted and these also require further analysis.

This comparative research has proved useful in setting our results alongside national data, but we are now undertaking further statistical analysis to examine in detail the relationships between the specific behaviours and individual elements of physical design to determine if any relationships exist at this level, and what the nature of these relationships may be. During this analysis we will test for the impact of socio-economic and other contextual variables, and examine additional behaviours, such as travel for other uses.

As a footnote, it is interesting to stand back from this research and contemplate the value of trying to build housing schemes to support sustainable lifestyles at the present time. Although evidence of sustainable behaviours does not seem strong at present in the SB survey, it could be argued that the real benefit of the schemes studied is that they provide built environments that *can* support more sustainable lifestyles if and when people are ready to take them up. Increases in fuel costs, for example, see people reduce their car use and start walking, cycling and using public transport instead. The SB schemes allow residents to make this transition easily, unlike many 'normal' schemes where residents are locked into car use. It would be interesting to revisit these schemes in ten years.

Acknowledgements We would like to thank all the participants in our surveys and our team of field researchers. We would also like to thank the developers and planners who provided technical information on the schemes. The project was guided by an active steering group, and its input is also much appreciated.

References

Barton, H., Grant, M. and Guise, R. (2003) *Shaping Neighbourhoods for Health and Sustainability*, Spon, London.

Breheny, M. (2004) Sustainable settlements and jobs-housing balance, in *Urban Sprawl in Western Europe and the United States*, (eds. H. W. Richardson and C-H. Bee), Ashgate, Aldershot.

- Brownhill, D. and Rao, S. (2002) A Sustainability Checklist for Developments: A Common Framework for Developers and Local Authorities, Construction Research Communication Ltd, London.
- Butler, S. (2004) Green Voices and Choices: Residents Views of Environmental Housing and Lifestyles, www.sustainablehomes.co.uk
- Carmona., de Magalhães, C., Edwards, M., Auer, B. and Manasseh, S. (2001) *The Value of Urban Design*, report by Bartlett School of Planning for Commission for Architecture and the Built Environment and Department of the Environment Transport and the Regions, Thomas Telford, London.
- Civic Trust (2001) Position Statement: Walking in Towns and Cities, www.civictrust. org.uk/policy%20and%20campaigns/positions/walking.shtml
- Department for Communities and Local Government (DCLG) (2005) *English House Condition Survey*, Stationery Office, London.
- Department of the Environment, Food and Rural Affairs (DEFRA) (2007) Survey of Public Attitudes and Behaviours Towards the Environment, www.defra.gov/environment/statistics
- Department of the Environment, Food and Rural Affairs (DEFRA) (2005a) *E-Digest of Environmental Statistics: Waste and Recycling*, Department of Environment Food and Rural Affairs, www.defra.gov.uk/environment/statistics/waste/index.htm
- Department of the Environment, Food and Rural Affairs (DEFRA) (2005b) *Environmental Information Tables – Biodiversity*, Department of Environment Food and Rural Affairs, www.sustainable-development.gov.uk/key/environmental/environmental2.htm
- Department of the Environment, Food and Rural Affairs (DEFRA) (2001) Survey of Public Attitudes to Quality of Life and the Environment, www.defra.gov/environment/statistics
- Dempsey, N. (2007) Are high quality neighbourhoods socially sustainable? Methodological challenges of unpacking multidimensional concepts, *European Urban Research Association* (*EURA*) 10th Anniversary Conference: The Vital City, University of Glasgow, 12th-14th September, 2007.
- Department of the Environment, Transport and the Regions (DETR) (1998) *Places, Streets and Movement: A companion Guide to Design Bulletin 32: Residential Roads and Footpaths,* Department of the Environment Transport and the Regions, DETR, London.
- Department of the Environment, Transport and the Regions (DETR) (2000a) *By Design: Urban Design in the Planning System: Towards Better Practice* Department of the Environment Transport and the Regions, Thomas Telford Publishing, London.
- Department of the Environment, Transport and the Regions (DETR) (2000b) *Our Towns and Cities: the Future Delivering the Urban Renaissance* Department of Environment, Transport and the Regions, Stationery Office, London.
- Department of the Environment, Transport and the Regions (DETR) (2000c) *Planning Policy Guidance Note No.3: Housing* Department of the Environment Transport and the Regions, HMSO, London.
- Department of the Environment Transport and the Regions and the Commission for Architecture and the Built Environment (DETR and CABE) (2000) By Design: Urban Design in the Planning System: Towards Better Practice, DETR and CABE, Thomas Telford Publishing, London.
- Department for Transport (DfT) (2005) *The National Travel Survey*, Transport Statistics and DfT, www.dft.gov.uk/162259
- Department for Transport (DfT) (2003) On the Move: By Foot a Discussion Paper, Department for Transport, London.
- Department for Transport (DfT) (2004) *The Future of Transport: A Network for 2030, White paper on transport presented to Parliament by the Secretary of State for Transport* July 2004 -Cm 6234, The Stationery Office, London.
- Dixon, T. and Marston, A. (2003) Mixed Use Urban Regeneration at Brindley Place, Birmingham and Gunwharf Quays, Portsmouth: An Assessment of the Impact on Local and National Economies, Report for British Property Federation, The College of Estate Management, Reading.

English Partnerships (2004) Greenwich Peninsula, English Partnerships, London

- Gaston, K. J., Smith, R. M., Thompson, K., and Warren, P. H. (2003) Urban domestic gardens (II): experimental tests of methods for increasing biodiversity. *Biodiversity and Conservation*, **00**, pp. 1–19.
- Gehl, J. (2001) *Life Between Buildings: Using Public Space Fourth Edition*, Arkitektens Forlag: The Danish Architectural Press, Copenhagen.
- Haughton, G. and Hunter, C. (1994) Sustainable Cities, Jessica Kingsley, London.
- Healey, T. and Côté, S. (2001) *The Well-being of Nations: the Role of Human and Social Capital*, Organisation for Economic and Co-operation and Development, Paris.
- Lazarus, N. (2003) Beddington Zero (Fossil) Energy Development: Toolkit for Carbon Neutral Developments- part II, Bioregional Development Group, Wallington Surrey.
- Llewelyn-Davies in association with Alan Baxter Associates (2000)*Urban Design Compendium*, English Partnerships, London.
- Mulholland, H. (2003) Perceptions of Privacy and Density in Housing, research undertaken by Mulholland Research and Consulting, Design for Homes Popular Housing Research, London.
- Office of National Statistics (ONS) (2005 and 2006) The Labour Force Survey, www.statistics.gov.uk
- Pett, J. and Guertler, P. (2004) User Behaviour in Energy Efficient Homes Association for the Conservation of Energy, Phase 2 Report Number G01-14284, www.ukace.org/ research/behaviour/index.htm
- Putnam, R. D. (2001) Bowling Alone, Touchstone, New York
- Rao, S., Yates, A., Brownhill, D. and Howard, N. (2000) Ecohomes: the Environmental Rating for Homes, BRE Ltd and CRC Ltd, London
- Scottish Executive (2002) Public Attitudes to the Environment in Scotland, Scottish Executive and National Statistics Publications, www.scotland.gov.uk/publications/2002/11/15864/14272
- Social Exclusion Unit (2001) *Preventing Social Exclusion: a Report by the Social Exclusion Unit*, Cabinet Office, London.
- Town and Country Planning Association (TCPA) (2004) Biodiversity by Design: A Guide for Sustainable Communities, TCPA, London.
- Transport 2000 Trust (2003) Walking the Way Ahead: Good Practice, Transport 2000 and the Department for Transport, London.
- Urban Task Force (1999) Towards an Urban Renaissance: Final Report, HMSO, London.
- Williams, K, (ed.) (2005) Spatial Planning, Urban Form and Sustainable Transport, Ashgate, Aldershot.
- Williams, K., Burton, E. and Jenks, M. (eds.) (2000) Achieving Sustainable Urban Form, E & FN Spon, London.
- Williams, K. and Dair, C. (2007) A Framework of sustainable behaviours that can be enabled through the design of neighbourhood-scale developments. *Sustainable Development*, **15**, pp. 160–173.
- Williams, K. and Lindsay, M. (2007) The Extent and Nature of Sustainable Building in England: an Analysis of Progress. *Planning Theory and Practice*, 8(1), pp. 31–51.

Chapter 10 Ecological and Psychological Value of Urban Green Space

Katherine N. Irvine, Richard A. Fuller, Patrick Devine-Wright, Jamie Tratalos, Sarah R. Payne, Philip H. Warren, Kevin J. Lomas and Kevin J. Gaston

Introduction

In urban environments, perhaps more so than in any other setting, people and nature must coexist in close, and sometimes uncomfortable, proximity. With half of the world's human population living in cities and a continued decline of biodiversity in the wider landscape, urban nature plays an increasingly important role in creating cities that are both ecologically and socially sustainable. However, understanding the value of urban green spaces as a resource requires an integration of several, rarely overlapping, approaches to evaluating and managing these places.

Ecologically, urban green spaces form significant components of regional and national biodiversity conservation networks. However, few urban ecologists have explicitly studied human interactions with biodiversity which hampers a full understanding of urban ecosystems in two important ways. First, many urban biodiversity patterns arise in response to, and are maintained specifically by, repeated human activity. Therefore, understanding urban ecosystems must entail knowledge of human motivations and responses in relation to biodiversity. Second, human interventions not only degrade but can also improve urban ecosystems. The progress of conservation efforts in cities frequently depends on decisions made by individual householders, yet ecologists often lack the tools to engage with human communities to understand and encourage their involvement.

From a social-psychological perspective urban nature is an important component of quality of life for urban residents. Researchers have generally focused on benefits gained from "nearby nature", often measured as proximity to or amount of green space, or even a window view. The biological components as typically measured by ecologists (e.g. species richness, vegetation characteristics) are often subsumed into a single entity, such as the "greenness" of the urban landscape. However, all green space is clearly not equal, and there is emerging evidence that treating it as such will mask important responses by people to specific components of biodiversity.

K.N. Irvine (⊠)

Institute of Energy and Sustainable Development, De Montfort University, Leicester, UK

Also, other, more direct human interactions with urban biodiversity have received scant attention, such as wildlife gardening, feeding wild birds, damage to habitats and disturbance to wildlife. Whether properties of the natural environment itself can influence the strength of these interactions is unknown. To more fully integrate urban green space into human well-being requires an understanding of the biological form and function in these spaces.

The literatures on urban green spaces rarely cite one another, and there are few examples of collaborative research projects as typically divergent language is used, different data are collected, and questions are framed in different ways. This chapter considers the inter-relationship between biodiversity and human well-being in public and private urban green space, the links between the distribution of people and biodiversity in the city, and the potential trade-offs between managing green spaces for people and for biodiversity. We begin with a review of relevant literature from several research perspectives, then provide examples of people-nature interactions drawn principally from our collaborative research between ecologists and environmental psychologists, and, lastly, identify key management and policy implications.

People and Urban Nature

Human Impacts on Urban Nature

The process of urbanization dramatically transforms natural landscapes. Specific elements of urban form that impact patterns of biodiversity include the type of development, the amount of impervious surface, distance to city centre, tree cover and housing density. The impacts of such factors on the spatial distribution of biodiversity across urban landscapes have been amply reviewed (e.g. Niemelä, 1999; McKinney, 2002; Pauleit et al., 2005; Garden et al., 2006). Importantly, increasing urban intensity does not always lead to a simple linear decline in the number of species (species richness) or the number of individual plants and animals (abundance), and the biological response varies markedly depending on the group of organisms under study (Chace and Walsh, 2006; McKinney, 2008).

Many features of a city result from large-scale land use planning decisions. The consequences of decisions made by individual landowners are not so obvious. Many householders have land attached to their property, and the results of many thousands of individual decisions on how that land is managed can sum to a significant effect at a landscape scale. Physical garden features such as trees, shrubs, bird feeding stations, compost heaps, ponds, bird baths and nest boxes provide specific resources for urban biodiversity and can occur at high densities across the urban landscape (Gaston et al., 2005, 2007). In the UK, it has been estimated that more than 60% of households with a garden feed wild birds (DEFRA, 2002), and that 60,000 tonnes of food are presented annually to birds (Glue, 2006). Moreover, garden size, tree cover, the amount of shrubs and the species composition of vegetation are correlated

with invertebrate richness and abundance (Smith et al., 2006a, 2006b), and garden characteristics predict the composition of bird communities (Chamberlain et al., 2004).

A sample of 70 gardens within the three case study neighbourhoods in Sheffield (inner, between, outer) had on average 33% of their area covered with impervious surfaces (Tratalos et al., 2007a), and there are concerns that gardens across the UK are increasingly being paved over, leading to elevated storm water run-off and a decline in biodiversity value (Royal Horticultural Society, 2006). Reductions in garden size, resulting from an increase in housing density through infill development are also associated with a decline in tree cover and habitat heterogeneity (Pauleit et al., 2005; Smith et al., 2005).

Human factors can have a marked effect on how urban green spaces, both public and private, are used – and abused. For example, in Sheffield it has been found that access to public green space is poorer for more affluent sectors of society (Barbosa et al., 2007). Additionally, the proportion of the human population feeding wild birds declines as socio-economic deprivation increases (Fuller et al., 2008), and the popularity of wildlife gardening varies with householder age and employment status, although these effects depend on spatial scale and cultural context (Lepczyk et al., 2004; Fuller et al., in press). Disturbance and damage to habitats can result from heavy recreational, criminal and waste disposal use, leading to increased openness and poor vegetation quality (Moran, 1984; Matlack, 1993).

Urban Nature Impacts on People

People living in cities tend to be less healthy than their rural counterparts (e.g. Verheij, 1995). Natural environments can offset demands associated with city living, such as exposure to noise, crowding and traffic exhaust (e.g. Freeman and Stansfield, 1998; Hunt et al., 2000). While environmental exposure often emphasizes the potential for negative effects (asthma, cancer, etc; e.g. McMichael, 2001), the idea that nature can be beneficial has a long history. Gardens were at one time part of hospitals (Gerlach-Spriggs et al., 1998), and women and children regularly left the city for country homes during the pre- and early-industrial age. In early nineteenth century England there was a call for "breathing zones" around London (MacDougall, 1980); the value of urban green spaces was further recognised during the Victorian era with the creation of large, publicly funded city parks designed to provide places for recreation and to encourage social interaction (Conway, 2000).

Research supports this intuitive belief of a beneficial relationship between contact with nature and quality of life (for review see e.g. Rohde and Kendle, 1997; Irvine and Warber, 2002). Multiple dimensions of health respond positively to the availability of nearby nature including objective measures of the physiological effects of stress (e.g. Ulrich et al., 1991; Parsons et al., 1998), self-reported sensitivity to stress (Stigsdotter and Grahn, 2004), surgical recovery time (Ulrich, 1984), mental fatigue (e.g. Kuo, 2001), cognitive functioning in children (e.g. Wells,

2002) including attention-deficit/hyperactivity disorder (Kuo and Taylor, 2004), mood (e.g. Hull and Michael, 1995; Shibata and Suzuki, 2002), self-discipline (Taylor et al., 2002) and opportunities for reflection (e.g. Herzog et al., 1997). The availability of nearby nature is shown to promote social interaction and a sense of community (Coley et al., 1997; Kim and Kaplan, 2004) as well as reduce aggressive behavior and crime (Kuo and Sullivan, 2001a, 2001b). There is also evidence that people form emotional attachments with natural areas (e.g. Ryan, 2005) and that these spaces contribute to personal identity (Bernardini and Irvine, 2007).

Several recent studies have explored urban green space as a factor in understanding population-level health; de Vries et al. (2003) and Maas et al. (2006) find positive relationships between the amount of green space in a neighbourhood and self-reported health, with stronger effects among the elderly, less affluent socioeconomic groups and women at home. Greater longevity has been found among the elderly in Japan with access to "walkable green space" (easy to walk in and filled with "greenery"; Takano et al., 2002) and the availability of a garden appears to offset the negative health effects among public housing residents (Macintyre et al., 2003). In the light of these findings, public policy in the UK has recently emphasized the need for high quality green space as an additional component of urban form (ODPM, 2002, 2006).

The benefits of urban nature are experienced widely (by adults and children, at work or home, in healthcare and neighbourhood settings) and can be gained from a variety of different types of "nature", including wilderness, residential gardens or managed parks, trees and grass around an apartment building, and potted plants. Natural elements, particularly water features, large trees or woodlands are consistently preferred to built ones often irrespective of culture or nationality (e.g. Kaplan and Kaplan, 1989; Herzog et al., 2000). Yet there has been little study of the direct effects of specific biological components of the landscape on human quality of life (Brown and Grant, 2005).

Interactions Between People and Urban Nature

The foregoing discussion highlights the prevalence of intradisciplinary research into people-urban nature interactions. This section presents findings from an interdisciplinary collaboration on the two-way interactions between people and nature – in both public and private green space – using examples from the CityForm's case study neighbourhoods survey, an intensive study of Sheffield green space, and population-level data linked with GIS spatial analysis.

Public Green Space

Patterns of Use and Visit Motivations

Public green spaces (often referred to as open space or parks) have been the focal point for much of the work on interactions between people and urban nature (Speirs,

2003; Balram and Dragicevic, 2005; Pincetl and Gearin, 2005). Research on the use and design of public green space has typically focused on its amenity function, crime reduction, and ways to encourage physical activity and social interaction (Dunnett et al., 2002; Commission for Architecture and Built Environment [CABE Space], 2005, 2007). Variation in the use of these spaces also affects the quantity and quality of interactions that urban residents experience with nature. Viewed from this perspective, urban parks, quite often (although not always) located close to where people live and work, become important arenas for contact between people and nature.

Within the UK there are about 27,000 urban parks and in England alone some 33 million people make more than 2.5 billion visits a year. These designated areas, which form the bulk of publicly provided (thus publicly accessible) green space, compose about 14% of total urban space (Dunnett et al., 2002). Within the city of Sheffield alone the 87 municipal parks attract over 25 million visits per year (Beer, 2005). Varying in purpose, facilities provided and catchment area, these spaces also differ in the amount and type of natural features present (e.g. grass, woodland), as well as their intrinsic biodiversity value. While studies suggest that just knowing these spaces are available can facilitate well-being (Bell et al., 2004), our research focused on actual visits, emphasizing the interplay between people and biological diversity with respect to benefits and management possibilities for both. To this end, using ecological surveys, interviews with 312 park users, and direct observation of usage and visitors, we studied 15 parks in Sheffield that varied in size and biological diversity and were located between the city centre and the western suburbs (see Fuller et al., 2007 for further details).

While a wide range of people make use of parks, demographic differences do exist. For example, women may be less likely to visit woodlands alone (e.g. Burgess, 1998), and a study of green space in the East Midlands region of the UK found that they are less well used by women, individuals with disabilities, and those from ethnic minorities (Bell et al., 2004). Sheffield park users were predominantly of European ethnicity, and comprised roughly equal numbers of men and women, individuals and groups as well as a wide age range (Table 10.1).

Common among the reasons for visiting urban green spaces are sport, relaxation, socializing, entertaining children, dog walking and to "be in nature" (e.g. Ulrich and Addoms, 1981; Burgess et al., 1988; Garvin and Berens, 1997; Chiesura, 2004; Dines et al., 2006). Results from CityForm's case study neighbourhoods survey show recreation to be the most frequent reason for using local green space (Table 10.2; see Gaston et al., 2007 for details). This result is confirmed by direct observation of people using Sheffield parks (Table 10.3); the most frequently mentioned motivation among interview participants is walking (walking the dog, taking a stroll, or walking en route to another destination). Exercise and sport (e.g. cricket, football, skateboarding) are also commonly cited reasons for visits with recurrent reference to bringing children to play. Other activities include having a meal, reading, sitting, photography and socialising. As might be expected some places are visited specifically because of their facilities (e.g. play area), the type of space (e.g. cemetery) or because it is the closest space.

	Total people Gender			Age (years)				Group size		
		Male	Female	<10	10–16	Adult <60	Adult >60	Single	Couple	>2
Pre-Holiday										
Weekend (1400–1500)	1823	52	48	16	9	71	4	25	41	33
Weekday lunch (1300–1400)	1040	47	53	10	6	81	3	47	34	19
Weekday post-lunch (1400–1500)	569	51	50	16	4	76	5	42	40	18
Holiday										
Weekend (1400–1500)	895	59	41	21	13	52	14	26	42	32
Weekday lunch (1300–1400)	332	53	47	30	18	46	6	35	29	36
Weekday post-lunch (1400–1500)	431	52	48	31	25	36	8	32	25	43

Table 10.1 Visitor composition across 15 public green spaces in Sheffield

Values for gender and age are the percentage of all people of known gender and age that were male/female or in the different age groups. Values for group size are the percentage of all groups of known size that were made up of singles, couples and groups of three or more people.

Visit Motivation	Number of responses	Use at least occasionally (%)	Never visit (%)	
Recreation	4056	72	28	
Be in natural environment	3872	69	31	
See local wildlife	3849	55	45	
Meet friends/family	3959	53	47	
Feed ducks	3621	36	64	
Children to play	3714	36	64	
Sport	3831	33	67	
Walk dog	3622	16	84	

Table 10.2 Use by Urban Residents of Local Neighbourhood Green Spaces

Values in the second column are the number of responses to each individual question. Individuals reporting no access to green space are excluded.

functions of these spaces are as places for peace and quiet, relaxation, rest and taking a break. Clearly, there is potential for these latter kinds of functions to be linked intimately to the quality and extent of green space within urban parks.

While wanting to be in nature is consistently reported as a reason to visit public green space (e.g. Hayward and Weitzer, 1984; de Groot and van den Born, 2003), little is known about which components of the natural environment are particularly important, and whether this is related to biodiversity. Responses to

	Total people	Lie/sit	Walk	Walk dog	Child oriented	Sport	Exercise (run/cycle)	Café	Feed ducks	Other
Pre-holiday										
Weekend	1706	50	22	3	10	7	2	4	0	2
Weekday lunch	1018	58	24	3	6	2	1	5	0	1
Weekday post-lunch	825	46	31	3	6	8	1	3	0	2
Holiday										
Weekend	543	17	39	3	14	7	5	6	1	7
Weekday lunch	320	23	37	7	12	2	4	9	1	6
Weekday post-lunch	417	16	32	5	23	6	3	4	0	11

Table 10.3 Activities of visitors to 15 public Green Spaces in Sheffield

Values are the percentages of all people where activity was recorded that engaged in each activity. See Table 10.1 for observational time periods.

CityForm's case study neighbourhoods survey indicate that 69% use local parks to "be in a natural environment", 55% to see local wildlife, and 36% to feed ducks (Table 10.2). Similarly, Sheffield park interviewees strongly endorse the importance of nature to the park experience, and at least two-thirds said that the diversity of flora and fauna is valuable. Interestingly, when asked to give specific reasons for visiting, explanations rarely refer to flora and fauna directly (those that did included greenery, feeding ducks, collecting seeds, looking at birds/squirrels), although broader constructions of nature are mentioned (e.g. fresh air, being outside, peace and quiet, open space, topography). This suggests that specific elements of biodiversity value (e.g. species richness, habitat heterogeneity) may not be directly perceived as important, yet the combination of these components into a natural scene is part of the reason why green spaces are used.

Sheffield park interviewees commonly reported feeling relaxed, refreshed, calm and peaceful after leaving green spaces. Interview responses also indicated that visits to green space facilitate the ability to reflect, engender strong emotional attachments and support both feelings of uniqueness as well as a connection to one's past experiences. This reflects two of the more intangible purposes of parks identified in recent UK policy initiatives, that of contemplation and a sense of place (CABE Space, 2005). Moreover, the degree of psychological benefit gained increased with park area, the variety of habitats present in the park, and plant species richness (Fuller et al., 2007). This suggests that biological complexity is directly related to the quality of the urban green space experience.

A less frequently studied component of urban green spaces is the "soundscape" (acoustic environment) and its quality. In three Sheffield parks, located in the three case study neighbourhoods (inner, between, outer), the sounds most frequently mentioned by visitors are those of natural origin (e.g. bird song, dogs, wind in the trees). Vegetation buffers could have reduced the prevalence of mechanical sounds

(e.g. traffic, building ventilation, construction; Irvine et al., 2009). Mechanical sounds are consistently rated less pleasant than natural sounds both within green spaces (e.g. Payne et al., 2007) and in public squares (Yang and Kang, 2005) and the soundscape has been shown to affect opportunities for reflection (Payne et al., 2007).

Biodiversity Management

A traditional view of biodiversity management in urban green spaces might emphasize the exclusion of people to allow vegetative growth and reduce disturbance to specific species of flora and fauna (Sorace, 2001). Parks and remnant fragments of original vegetation that have been managed in this way do tend to support a greater diversity of species than the urban landscape in general (Jokimäki and Suhonen, 1993; Hadidian et al., 1997). The variety of habitats available and the presence of particular park features such as rough grass are associated with elevated species richness (Chamberlain et al., 2007). Species richness is positively correlated with park area, and biological communities within parks are structured in a way that mirrors that found in more natural environments (Fernández-Juricic, 2001; Cornelis and Hermy, 2004; Fernández-Juricic et al., 2005). However, the emerging evidence for a positive relationship between biodiversity value and benefits to psychological well-being, along with the importance that green space users attach to an experience of nature, suggests, at the very least, that management for people and biodiversity should be considered simultaneously, and even that carefully planned management to enhance biodiversity value will also benefit park users, in a win-win scenario.

Managing for biodiversity necessarily raises a number of issues given that recreation is, and rightly so, an important use of parks. Viewing the function of parks as an amenity for people has historically led to provision of large formal spaces (e.g. botanical gardens) that include sizable expanses of mown grass (amenity turf) for sport and recreation, and open spaces with few trees and shrubs (Hunziker et al., 2007). There is also a large and thriving literature suggesting that this "urban savannah" design is fundamentally preferred (Kaplan and Kaplan, 1989). This presents a clear challenge in terms of maintaining biological diversity. Additionally, given the range and intensity of activities undertaken in these spaces, there could be concerns over constraints that biodiversity management might place on usage.

These concerns are apparent in comments from Sheffield interviewees in response to a question about the use of rough grassy areas or fences as management tools for the "welfare of plants and animals". Specific activities mentioned that could be negatively impacted included sport, sitting, walking and children playing, with additional comments concerning aesthetics (e.g. it would look neglected, fences are ugly), the purpose of parks (e.g. parks are for people, parks should be accessible), and safety (e.g. things could hide in the grass, children could hurt themselves on fences).

Yet just as many interviewees are not opposed to the use of fences or rough grass areas, seeing the potential benefits (e.g. educational opportunity, people doing something for nature), and, among those opposed, most provide suggestions for circumstances under which such management could coexist with human use. Crucial to that coexistence is the degree to which recreational use coincides with the distribution of biodiversity. Our study of Sheffield park usage included observation of the distribution of people across a number of sub-areas within each park. These sub-areas were determined based on dominant land cover type – amenity planting (e.g. flower beds), building, grassland, scrub (e.g. shrubs, bushes), impervious surface (e.g. paved paths), water and woodland. Usage of sub-areas declined as the amount of woodland and grassland increased (Fig. 10.1a). Conversely, coverage by



Fig. 10.1 Relationships between the Number of People using Sub-areas within 15 Public Green Spaces in Sheffield and the Proportion of that Area Covered by (a) Woodland and (b) Impervious Surface. Values of woodland and impervious surface coverage were divided into deciles for presentation. There was only one data point with a value for sealed surface of 1.36–2.74, so the point is omitted from (b). Error bars are \pm One Standard Error

impervious surfaces was positively correlated with frequency of use (Fig. 10.1b). This suggests that habitats associated with elevated biodiversity, such as woodland and rough grassland, might be less heavily impacted by recreational use than turfed and paved areas, which are more suited for amenity use. Crucially, these more complex natural features are among the motivations for visiting, despite the fact they are not always directly used.

The foregoing suggests that the strategic development of areas specifically managed for biodiversity could in fact enhance the green space experience without negatively affecting use of the space as a whole. While these may vary culturally and regionally, acceptable circumstances mentioned by Sheffield park users include (i) the use of small patches, (ii) locating areas along the edges of fields, on hills, in already existing woodlands or along unpaved paths, (iii) providing an educational component, and (iv) making it look orderly and purposeful (e.g. the use of signs, a mown strip of grass at edge). The use of such "cultural cues to care" can offset the perceived "messiness" of more naturalistic management (Nassauer, 1995, 2004).

The development of more natural, less manicured landscapes also raises issues of safety. There is evidence that a high density of vegetation in urban parks leads to negative perceptions of safety (Schroeder and Anderson, 1984; Bixler and Floyd, 1997). Our work suggests a more complex relationship. Across the fifteen parks in Sheffield the perception that a park contained unsafe places is significantly positively correlated with the amount of shrubs present (shrub layer volume) per hectare. Yet the relationship with the amount of tree cover (tree canopy volume) per hectare is non-significant, as were relationships between actually feeling safe and both vegetation measures (Table 10.4). While vegetation is mentioned by interviewees as a feature contributing to making a place appear unsafe, it is not necessarily the presence of vegetation per se, but also that these areas were secluded, less well lit and offering hiding places for people (e.g. walkways under trees, woodlands, overgrown bushes).

Vegetation might also influence perceptions of safety by affecting the ability to see long distances, or ease of locomotion (e.g. Nasar et al., 1993; Nasar and Jones, 1997; Forsyth, 2003). This, however, contradicts research within the landscape preference literature involving the concept of "mystery"; vegetation structure and structural designs that obscure the view explains why some landscapes are well-liked (e.g. Herzog and Miller, 1998). This highlights the need for a more refined understanding of the perceptions of safety in urban green space and

 Table 10.4
 Correlations between Perceptions of Safety and Vegetation across 15 Public Green

 Spaces in Sheffield

	Shrub layer	volume (m ³ /ha)	Tree canopy	volume (m ³ /ha)	
There are unsafe places I feel safe	r _s 0.63 0.47	p 0.01 0.08	r _s 0.36 0.23	p 0.19 0.41	

Safety items measured on a 5-point Likert scale from strongly disagree to strongly agree

researchers have begun to explore the relationship among naturalistic designs, safety, preference and use. For example, Bjerke et al. (2006) find a preference for more densely vegetated parks among middle-aged, educated individuals as well as among those with an interest in wildlife and a more pro-environmental attitude. Jim and Chen (2006) note a similar preference for more naturalistic design in China, while Özgüner and Kendle (2006) demonstrate that both the naturalistic and the more manicured designs are equally appreciated among the public in Sheffield. In contrast, signs of neglect may also influence perceptions of safety. Indeed, park interviewees in Sheffield mention signs of neglect, just as much as vegetation, as contributing to places feeling unsafe. These comments often are related to the people and their activities (e.g. anti-social behaviour), poor maintenance (e.g. cracked hardscape under children's play areas, dilapidated buildings), and litter.

Private Green Space

Many urban residents have access to a parcel of private land around their home, an additional avenue for contact with nature. Because the land is privately owned, people make independent decisions about how to manage their garden, even if the decision is to do nothing and let the garden grow "wild", or to pave the area over entirely and treat it continually with herbicides (both extremes observed during fieldwork in Sheffield). Coupled with the fact that gardens cover around a quarter of the area within a typical UK city (Loram et al., 2007), these decisions must inevitably sum to influence urban nature at a landscape scale across the city.

Gardens and Well-Being

Having a view over gardens has been shown to increase residents' satisfaction with the community and to contribute to neighbourhood satisfaction (Kaplan, 1985, 2001), and having physical access to a garden positively influences self-reported health (Macintyre et al., 2003; de Vries et al., 2003). Gardens are used in a variety of ways, as places to socialize, play, retreat from the world, gain pleasure from horticulture, watch nature and act as venues for household projects or for self-expression (Bhatti and Church; 2001, 2004). They also appear valuable for the psychological processes of self-esteem, self-efficacy and personal identity (ARCWIS 2002; Bernardini and Irvine, 2007). Among gardeners, peace and quiet, sustained engagement with the natural environment, and stress reduction are cited as important benefits (Kaplan, 1973; Catanzaro and Ekanem, 2004). Gardens also connect people with their personal history; daily or seasonal cycles may bring reminders of the past (Francis, 1988; Sime, 1993). Other benefits derived include a sense of accomplishment, and the joy of anticipation that comes through the planning, planting and harvesting of one's garden (Catanzaro and Ekanem, 2004).

Wildlife Gardening

At one extreme of the continuum of possible garden management decisions, one might explicitly attempt to attract wild plant and animal species into the garden. Providing resources in this way for birds and other animals is a popular activity across much of the world. For example, between one-fifth and one-third of households in Europe, North America and Australia provide supplementary food for wild birds (Clergeau et al., 1997; Rollinson et al., 2003; Lepczyk et al., 2004), and in the United States alone some 52 million people frequently feed garden birds (United States Fish and Wildlife Service, 2001). As well as benefiting the birds themselves, pleasure in observing birds is presumably one of the motivations for engaging in this kind of activity, although we are aware of no published studies on this issue. Collectively, such activities are typically referred to as "wildlife gardening", which can broadly be defined as any actions conducted in private gardens to increase their suitability for wildlife. It thus encompasses both a general approach to garden management and specific provision of resources for wildlife. Areas might be left to grow as a wild flower meadow, weeds might be tolerated, or supplementary resources such as feeding stations or nest boxes for birds, a compost heap or log pile to encourage invertebrates, or a wildlife pond might be established. If the occurrence of these features is sufficiently widespread across the urban landscape, this may represent a substantial resource for, and substantial effect on, urban biodiversity levels.

Evidence that wildlife gardening increases richness and abundance of invertebrates is equivocal but the effect on birds is somewhat clearer. Overall bird abundance, and the abundance of selected urban specialist birds in Sheffield, had a strong positive relationship with the density of bird feeding stations (Fuller et al., 2008, Fuller et al., in press), hinting intriguingly that the provision of food for birds can influence the structure of urban bird communities at a landscape scale.

Incidence of wildlife gardening varies with urban form. Across CityForm's case study neighbourhoods 56% of households with access to outside space provide one or more "wildlife-friendly" garden features (Gaston et al., 2007). The occurrence of such garden features, and the proportion of households providing food for birds and other animals, are positively associated with the amount of garden coverage in the local area and average garden size, but is independent of household density, proportion of households in the professional socioeconomic group AB, and the amount of coverage by non-garden green space (Gaston et al., 2007).

Spatial Configuration of Biodiversity and People in the Landscape

As well as people modifying and responding to biodiversity at local scales in public green spaces and private domestic gardens, decisions by local authorities can dictate city-wide patterns of built form and conversely green space coverage (Kinzig et al., 2005). Aspects of biodiversity within such green spaces, for example vegetation structure or habitat heterogeneity, will, in turn, impact the types of human

activities that are conducted in them. Human activity and management patterns will again modify the pattern of biodiversity (e.g. through trampling, seed dispersal, bird feeding). Thus, while urban form can strongly influence the distribution of biodiversity, the resulting pattern of biodiversity also constrains and modifies human access to nature.

In urban areas, coverage by green space (be they parcels of countryside, parks, gardens, road verges or other patches of vegetated surface) is an important correlate of biodiversity value, both in terms of species richness and sometimes abundance. For many groups of organisms, species richness declines as coverage by green space declines. Furthermore, the influence of green space on biodiversity can act at different spatial scales, both locally and regionally (Clergeau et al., 2001; Melles et al., 2003). In other words, the level of biodiversity may depend not only on the extent of green space at that locality, but also on the extent of green space in the wider region in which the locality is embedded.

Within Sheffield, coverage by green space is correlated with the diversity of both birds and plants. The proportion of land cover comprising green space in the 100 m radius around a series of survey points across the city shows a strong positive association with bird species richness, although the explanatory power was rather low (Fig. 10.2a). Bird abundance initially increases as the amount of green space increases but eventually declines in areas with very high green space coverage (Fig. 10.2b). The initial rise is probably associated with an increase in habitat heterogeneity in the early stages of urbanizing an area (Tratalos et al., 2007b), whereas continuously-vegetated areas (e.g. woodland habitats) tend to support homogeneous communities of native species. There is a strong positive relationship between green space coverage and native plant species richness (Fig. 10.2c), and a hump-shaped relationship between green space coverage and non-native plant species richness (Fig. 10.2d).

The rather low explanatory power of many of the foregoing relationships suggests that although green space is clearly associated with the distribution of biodiversity, there are many other factors at play. Research is beginning to identify fascinating predictive relationships between socioeconomic variables and patterns of urban biodiversity (Collins et al., 2000; Hope et al., 2003; Martin et al., 2004). In Sheffield, bird species richness varied significantly across 47 neighbourhood types (Fuller et al., 2008). One important study, conducted across the Central Arizona-Phoenix region in the United States, reveals that in addition to elevation and current and former land use, spatial variation in plant diversity is best explained by family income and age of housing (Hope et al., 2003). Such relationships presumably reflect either subtle habitat characteristics not easily measured by standard habitat parameters, or variation in the way in which different kinds of urban areas are managed by residents and local authorities (Kinzig et al., 2005).

While spatial variation in human society might impact strongly on the distribution of biodiversity, for example though variation in how land is managed, there are clear interactions in the opposite direction. Sheffield alone supports over 600,000 birds, equating to 1.18 birds per person (Fuller et al., 2009), so urban areas are potentially important arenas for contact between people and wildlife.



Fig. 10.2 Relationships between Green Space Coverage around a Series of Sampling Locations in Sheffield and (**a**) Breeding Bird Species Richness, (**b**) Breeding Bird Density, (**c**) Native Plant Richness and (**d**) Alien Plant Richness. Green space was measured in 100 m buffers around survey locations. Error bars are 95% confidence intervals

Urban biodiversity can be framed as a quality of life indicator, promoting enhanced psychological and physiological well-being. For example, the UK government has established five biodiversity indicators "to ensure that urban areas contribute fully to the goals of biodiversity conservation and enhance the quality of life of people who live there" (DEFRA, 2002). Within Sheffield there is a negative relationship between bird species richness and the level of deprivation among neighbourhoods, as measured by the Index of Multiple Deprivation (Fig. 10.3); less privileged sectors of society have lower levels of biodiversity around the places where they live.

Although access to green space is an important issue in urban planning, rather little is known about actual levels of such access across the human urban population (Wray et al., 2005). This is despite the fact that regulatory agencies provide explicit guidance to safeguard access to green space in urban areas. The European Environment Agency (EEA) recommends that people should have access to green space within 15 minutes walking distance (roughly equivalent to 900 m), a condition that appears to be met for several of Europe's smaller cities (Stanners and Bourdeau, 1995). More stringently a UK government agency recommends that "people living in towns and cities should have an accessible natural green space less than 300 m



Fig. 10.3 Relationship between Local Levels of Deprivation and Avian Species Richness in Sheffield. Data points are for eleven neighbourhood types and represent mean bird species richness derived from point transect surveys in summer 2004

from home" (Wray et al., 2005). In Sheffield, 64% of households are located more than 300 m from the nearest public green space and 72% are more than 300 m along the road network from the nearest municipal park (Barbosa et al., 2007). Some households are more than 20 times this distance. Additionally, proximity varies significantly across society, with households in wealthier neighbourhood types tending to live further from public green space than households in less affluent neighbourhoods. While wealthier neighbourhoods might have greater access to gardens, the two types of green space play very different roles (Kellett, 1982; Bernardini and Irvine, 2007), and the degree to which they can substitute for one another is unclear (Barbosa et al., 2007).

In terms of urban planning, access to green space depends closely on where and how that space is located across the landscape. An uneven distribution might lead to the kinds of social inequalities described above. For a given total area of green space, the spatial configuration of the component patches is also likely to play a role in its biodiversity value (Donnelly and Marzluff, 2006). Urban environments are highly fragmented, with habitat patches of various sizes being isolated from each other by the road network and buildings. Despite this, the degree to which urban habitat patches are isolated from one another shows no general relationship with patch species richness (e.g. Bastin and Thomas, 1999; Fernández-Juricic, 2000; Gibb and Hochuli, 2002). In part, this may result from links between habitat patches formed by interstitial vegetation within the urban matrix (most notably gardens). Additionally, some mobile organisms, such as birds, may be able to disperse among widely separated patches of suitable habitat. While there appears to be no strong general effect of patch connectivity in urban environments, larger patches do typically contain higher species richness, and some native species only occur in such patches (e.g. Dickman, 1987; Mörtberg and Wallentinus, 2000; Park and Lee, 2000). Ecological surveys of gardens in Sheffield reveal that on average those

which are larger contain greater numbers of plant species, although small gardens can nonetheless be very species rich (Smith et al., 2006a).

Overall, the amount of green space is the single most important predictor of biodiversity levels in a city. Thus, for example, while enhancing connectivity can be important, maximizing the amount of vegetated space within cities should always feature strongly in planning for urban biodiversity (Rebele, 1994). The same can be said for the provision of green spaces as places for people to engage with the natural environment (Ward Thompson, 2002; Gomez and Salvador, 2006). Of course, conservation of biodiversity is not and should not always be the key driver of green space management within cities. In some cases, increased urban density and the subsequent isolation of small city green spaces may reduce pressure elsewhere in the landscape (Merrill 2004).

Conclusion

Urban green space has been studied extensively from both ecological and psychological perspectives with little communication between the groups of disciplines involved. We suggest that a fuller understanding of urban green space depends on considering simultaneously factors such as the amount of green space, density of housing, socio-economic-demographic composition and distribution of local human populations as well as the spatial configuration, quality and type of green space available. While clearly there is still much to learn about the interactions among humans, green space and biodiversity within cities here, we close this chapter by outlining the management and policy implications of what we do know, and identify the most pressing research challenges for the future.

While a large body of literature has documented positive effects of human contact with nature, and is useful for developing general approaches to urban planning, more specific knowledge about the magnitude or enduring quality of responses by human well-being to the presence of natural features in the urban landscape would enable a more quantitative approach to siting and managing green spaces, both public and private. The degree to which these two radically different types of green space can substitute for one another is unclear. Our data suggest they facilitate quality of life through different routes. Similarly, assessments of the quality of urban green spaces typically emphasize tidiness, the absence of litter and graffiti, and the presence of well-maintained facilities. Our demonstration that levels of biodiversity are positively linked to human well-being adds another angle to the debate over what constitutes quality in urban green space. We must now begin to understand the mechanism by which biodiversity in a landscape promotes human well-being.

The designation and management of green space is typically not strategic, frequently resulting in a collection of land parcels which, for various reasons, have not been developed for other purposes. This can lead to an ad hoc approach to green space planning and provision. However, designing spaces to meet the needs of all

users or establishing uniform criteria to be met may ultimately be detrimental for both biological and psycho-social benefits. We suggest a need to recognize the value and importance of local differences, creating a mosaic of opportunities for different kinds of interactions with green space (Thwaites et al., 2005). Given the importance of geographic proximity for green space usage and that access varies across society, explicit spatial planning is required for this kind of strategy to work. Yet the tools to predict the consequences of alternative plans for biodiversity and human well-being are limited and the extent to which access to green space equates with access to biodiversity is currently unknown. Given that psychological benefits increase with biodiversity levels, it is imperative that biodiversity measures be incorporated into audits of green space access for urban residents. Understanding these issues is central to designing cities that improve the experience of nature while also addressing biodiversity conservation.

Broad public participation is vital for the creation of public green spaces that address these multiple issues of quality. Working at local scales and providing meaningful opportunities for involvement may not only facilitate acceptance of what necessarily may be radically different types of spaces but could generate solutions that are more extensive and economical than otherwise might be possible (Irvine and Kaplan, 2001). An additional intriguing arena for public involvement is the private residential urban garden where multiple individual management decisions can result in landscape scale effects. This raises the fascinating possibility that, for example, garden bird feeding could be explicitly harnessed to influence the conservation status of a particular species that occurs within urban areas in order to achieve regional biodiversity targets. Despite the prevalence of private gardens in many modern cities, surprisingly little is known about the motivations for particular garden management decisions or the propensity to engage in "wildlife-friendly" activities.

From a policy perspective, both the conservation of urban biodiversity and the enhancement of public health depend on a better understanding of the interactions between people and urban nature. There is growing realization that urban green spaces are valuable resources for creating sustainable cities. These places are not only a contribution to achieving biodiversity conservation objectives but they form one of the main routes through which urban residents can benefit from nature. The interactions between people and urban nature are surprisingly finely tuned, with feedbacks occurring in both directions. Management strategies must therefore be sought that complement both the ecological and the human value of urban green space. Because of the highly synergistic quality of the relationship between these two types of benefits, the needs of management for biodiversity value and human well-being may not always indicate divergent solutions.

Acknowledgements We are grateful for generous assistance, discussion and/or comments from P. Armsworth, K. Baker, C. Bernardini, S. Herbert, P. Johnson, K. Kirwan, G. Mill, J. Vulliamy, R. Wall, and S. Warber, as well as to all the participants who gave of their time. Ordnance Survey kindly provided MasterMap data under license to the CityForm consortium. K.J.G. holds a Royal Society-Wolfson Research Merit Award.

References

- ARCWIS (2002) Perth Domestic Water-use Study: Household Appliance Ownership and Community Attitudinal Analysis 1999–2000, Australian Research Centre for Water in Society, CSIRO, Perth, Australia.
- Balram, S. and Dragicevic, S. (2005) Attitudes toward urban green spaces: integrating questionnaire survey and collaborative GIS techniques to improve attitude measurements. *Landscape and Urban Planning*, **71**(2-4), pp. 147–162.
- Barbosa, O., Tratalos, J.A., Armsworth, P.R., Davies, R.G., Fuller, R.A., Johnson, P. and Gaston, K.J. (2007) Who benefits from access to green space? A case study from Sheffield, UK. *Landscape and Urban Planning*, 83(2-3), pp. 187–195.
- Bastin, L. and Thomas, C.D. (1999) The distribution of plant species in urban vegetation fragments. Landscape Ecology, 14(5), pp. 493–507.
- Beer, A. (2005) The green structure of Sheffield. In *Green structure and urban planning. Final report of COST Action C1*, (eds. A.C. Werquin, B. Duhem, G. Lindholm, B. Oppermann, S. Pauleit and S. Tjallingii), COST Programme, Brussels.
- Bell, S., Morris, N., Findlay, C., Travlou, P., Gooch, D. Gregory, G. and Ward Thompson, C. (2004) Nature for People: The Importance of Green Spaces to East Midlands Communities, English Nature Research Report No. 567. English Nature, Peterborough.
- Bernardini, C. and Irvine, K.N. (2007) The 'nature' of urban sustainability: Private or public greenspaces? Sustainable Development and Planning III, 102, pp. 661–674.
- Bhatti, M. and Church, A. (2001) Cultivating natures: Homes and gardens in late modernity. Sociology, 35(2), pp. 365–383.
- Bhatti, M. and Church, A. (2004) Home, the culture of nature and meanings of gardens in late modernity. *Housing Studies*, **19**(1), pp. 37–51.
- Bixler, R.D. and Floyd, M.F. (1997) Nature is scary, disgusting and uncomfortable. *Environment* and Behavior, **29**(4), pp. 443–467.
- Bjerke, R., Østdahla, T., Thraneb, C. and Trumseb, E. (2006) Vegetation density in urban parks and perceived appropriateness for recreation. *Urban Forestry and Urban Greening*, **5**(1), pp. 35–44.
- Brown, C. and Grant, M. (2005) Biodiversity and human health: What role for nature in healthy urban planning? *Built Environment*, **31**(4), pp. 326–338.
- Burgess, J. (1998) But is it worth taking the risk? How women negotiate access to urban woodland: A case study. In *New Frontiers of Space, Bodies and Gender*, (ed. R. Ainley), Routledge, London.
- Burgess, J., Harrison, D.M. and Limb, M. (1988) People, parks and the urban green: A study of popular meanings and values for open spaces in the city. *Urban Studies*, 25(6), pp. 455–573.
- Catanzaro, C. and Ekanem, E. (2004) Home gardeners value stress reduction and interaction with nature. Acta Horticulturae, 639, pp. 269–275.
- Chace, J.F. and Walsh, J.J. (2006) Urban effects on native avifauna: a review. *Landscape and Urban Planning*, **74**(1), pp. 46–69.
- Chamberlain, D.E., Cannon, A.R. and Toms, M.P. (2004) Associations of garden birds with gradients in garden habitat and local habitat. *Ecography*, **27**(5), pp. 589–600.
- Chamberlain, D.E., Gough, S., Vaughan, H., Vickery, J.A. and Appleton, G.F. (2007) Determinants of bird species richness in public green spaces. *Bird Study*, 54(1), pp. 87–97.
- Chiesura, A. (2004) The role of parks for the sustainable city. *Landscape and Urban Planning*, **68**(1), pp. 129–138.
- Clergeau, P., Sauvage, A., Lemoine, A., Marchand, J.-P., Dubs, F. and Mennechez, G. (1997) Quels oiseaux dans la ville? *Les Annales de la Recherche Urbaine*, 74, pp. 119–130.
- Clergeau, P., Jokimäki, J. and Savard, J.-P. L. (2001) Are urban bird communities influenced by the bird diversity of adjacent landscapes? *Journal of Applied Ecology*, **38**(5), pp. 1122–1134.
- Coley, R.L., Sullivan, W.C. and Kuo, F.E. (1997) Where does community grow?: The social context created by nature in urban public housing. *Environment and Behavior*, 29(4), pp. 468–494.

- Collins, J.P., Kinzig, A., Grimm, N.B., Fagan, W.F., Hope, D., Wu, J. and Borer, E.T. (2000) A new urban ecology. *American Scientist*, 88(5), pp. 416–425.
- Commission for Architecture and Built Environment (CABE) (2005) *Does Money Grow on Trees?* Commission for Architecture and Built Environment, London.
- Commission for Architecture and Built Environment (CABE) (2007) Decent parks? Decent behaviour? The Link Between the Quality of Parks and User Behaviour. Commission for Architecture and Built Environment, London.
- Conway, H. (2000) Parks and people: The social functions. In *The Regulation of Public Parks*, (eds. J. Woodstra and K. Fieldhouse), E. and F.N. Spon, London.
- Cornelis, J. and Hermy, M. (2004) Biodiversity relationships in urban and suburban parks in Flanders. *Landscape and Urban Planning*, **69**(4), pp. 385–401.
- de Groot, W.T. and van den Born, R.J.G. (2003) Visions of nature and landscape type preferences: An exploration in the Netherlands. *Landscape and Urban Planning*, **63**(3), pp. 127–138.
- de Vries, S., Verheij, R.A., Groenewegen, P.P. and Spreeuwenberg, P. (2003) Natural environments–healthy environments? An exploratory analysis of the relationship between greenspace and health. *Environment and Planning A*, 35(10), pp. 1717–1731.
- Department of the Environment, Food and Rural Affairs (DEFRA) (2002) Working with the Grain of Nature. Defra Publications, London.
- Dickman, C.R. (1987) Habitat fragmentation and vertebrate species richness in an urban environment. *Journal of Applied Ecology*, 24(2), pp. 337–351.
- Dines, N., Cattell, F., Gesler, W. and Curtis, S. (2006) Public Spaces, Social Relations and Wellbeing in East London, The Policy Press, Bristol.
- Donnelly, R. and Marzluff, J.M. (2006) Relative importance of habitat quantity, structure, and spatial pattern to birds in urbanizing environments. *Urban Ecosystems*, 9(2), pp. 99–117.
- Dunnett, N., Swanwick, C. and Woolley, H. (2002) Improving Urban Parks, Play Areas and Green Spaces, Department for Transport, Local Government and the Regions, London
- Fernández-Juricic, E. (2000) Avifaunal use of wooded streets in an urban landscape. Conservation Biology, 14(2), pp. 513–521.
- Fernández-Juricic, E. (2001) Avian spatial segregation at edges and interiors of urban parks in Madrid, Spain. *Biodiversity and Conservation*, **10**(8), pp. 1303–1316.
- Fernández-Juricic, E., Poston, R., de Collibus, K., Morgan, T., Bastain, B., Martin, C., Jones, K. and Tremino, R. (2005) Microhabitat selection and singing behavior patterns of male house finches (*Carpodacus mexicanus*) in urban parks in a heavily urbanized landscape in the western U.S. Urban Habitats, **3**, pp. 49–69.
- Forsyth, A. (2003) People and urban green areas: perceptions and use. Design Brief Number 4, University of Minnesota, Design Centre for American Urban Landscapes, Minnesota.
- Francis, M. (1988) The garden in the mind and in the heart. In *Looking Back to the Future*, (eds. H. van Hoorgdalem, N.L. Prack, Th.J.M. van der Voordt and H.B.R. van Wegen), IAPS 10/1988 Proceedings, Vol. 2. Delft University Press, Delft.
- Freeman, H.L. and Stansfield, S.A. (1998) Psychosocial effects of urban environments, noise and crowding. In *Environment and Mental Health*, (ed. A. Lundberg). Lawrence Erlbaum, London.
- Fuller, R.A., Irvine, K.N., Devine-Wright, P., Warren, P.H. and Gaston, K.J. (2007) Psychological benefits of greenspace increase with biodiversity. *Biology Letters*, 3(4), pp. 390–394.
- Fuller, R.A., Warren, P.H., Armsworth, P.R., Barbosa, O. and Gaston, K.J. (2008) Garden bird feeding predicts the structure of urban avian assemblages. *Diversity and Distributions*, 14(1), pp. 131–137.
- Fuller, R.A., Irvine, K.N., Davies, Z.G., Armsworth, P.R. and Gaston, K.J. (in press) Interactions between people and birds in urban landscapes. *Studies in Avian Biology*.
- Fuller, R.A., Tratalos, J. & Gaston, K.J. (2009) How many birds are there in a city of half a million people? *Diversity and Distributions*, 15(2), pp. 328–337.
- Garden, J., McAlpine, C., Peterson, A., Jones, D. and Possingham, H.P. (2006) Review of the ecology of Australian urban fauna: A focus on spatially explicit processes. *Austral Ecology*, 31(2), pp. 126–148.

- Garvin, A. and Berens, G. (1997) Urban Parks and Open Space, Urban Land Institute, Washington D.C.
- Gaston, K.J., Warren, P.H., Thompson, K. and Smith, R.M. (2005) Urban domestic gardens (IV): The extent of the resource and its associated features. *Biodiversity and Conservation*, **14**(14), pp. 3327–2249.
- Gaston, K.J., Fuller, R.A., Loram, A., MacDonald, C., Power, S. and Dempsey, N. (2007) Urban domestic gardens (XI): Variation in urban wildlife gardening in the United Kingdom. *Biodiversity and Conservation*, 16(11), pp. 3227–3238.
- Gerlach-Spriggs, N., Kaufman, R.E. and Warner, S.B. Jr. (1998) *Restorative Gardens: The Healing Landscape*, Yale University Press, New Haven.
- Gibb, H. and Hochuli, D.F. (2002) Habitat fragmentation in an urban environment: Large and small fragments support different arthropod assemblages. *Biological Conservation*, **106**(1), pp. 91–100.
- Glue, D. (2006) Variety at winter bird tables. Bird Populations, 7, pp. 212–215.
- Gomez, F. and Salvador, P. (2006) A proposal for green planning in cities. *International Journal of Sustainable Development*, **1**(1), pp. 91–109.
- Hadidian, J., Sauer, J., Swarth, C., Hanly, P., Droege, S., Williams, C., Huff, J. and Didden, G. (1997) A citywide breeding bird survey for Washington, D.C. *Urban Ecosystems*, **1**(2), pp. 87–102.
- Hayward, G. and Weitzer, W.H. (1984) The public's image of urban parks. *Urban Ecology*, **8**, pp. 243–268.
- Herzog, T.R., Black, A.M., Fountaine, K.A. and Knotts, D.J. (1997) Reflection and attentional recovery as distinctive benefits of restorative environments. *Journal of Environmental Psychology*, **17**(2), pp. 165–170.
- Herzog, T.R. and Miller, E.J. (1998) The role of mystery in perceived danger and environmental preference. *Environment and Behavior*, **30**(4), pp. 429–449.
- Herzog, T.R., Herbert, E.J., Kaplan, R. and Crooks, C.L. (2000) Cultural and developmental comparisons of landscape perceptions and preferences. *Environment and Behavior*, **32**(3), pp. 323–337.
- Hope, D., Gries, C., Zhu, W., Fagan, W.F., Redman, C.L., Grimm, N.B., Nelson, A.L., Martin, C. and Kinzig, A. (2003) Socioeconomics drive urban plant diversity. *Proceedings of the National Academy of Sciences*, **100**(15), pp. 8788–8792.
- Hull, R.B. and Michael, S.E. (1995) Nature-based recreation, mood change, and stress restoration. *Leisure Sciences*, 17(1), pp. 1–14.
- Hunt, R., Falce, C., Crombie, H., Morton, S. and Walton, E. (2000) *Health update Environment and Health: Air Pollution*, Health Education Authority, London.
- Hunziker, M., Buchecker, M. and Hartig, T. (2007) Space and place Two aspects of the humanlandscape relationship. In A Changing World: Challenges for Landscape Research, (eds. F. Kienast, O. Wildi and S. Ghosh). Springer, Dordrecht.
- Irvine, K.N. and Kaplan, S. (2001) Coping with change: The small experiment as a strategic approach to environmental sustainability. *Environmental Management*, 28(6), pp. 713–725.
- Irvine, K.N. and Warber, S.L. (2002) Greening healthcare: Practicing as if the natural environment really mattered. *Alternative Therapies in Health and Medicine*, **8**(5), pp. 76–83.
- Irvine, K.N., Devine-Wright, P., Payne, S.R., Fuller, R.A. Krausse, B. and Gaston, K.J. (2009) Green space, soundscape and urban sustainability: An interdisciplinary, empirical study. *Local Environment*, 14(2), pp. 155–172.
- Jim, C.Y. and Chen, W.Y. (2006) Impacts of urban environmental elements on residential housing prices in Guangzhou (China). *Landscape and Urban Planning*, 78(4), pp. 422–434.
- Jokimäki, J. and Suhonen, J. (1993) Effects of urbanization on the breeding bird species richness in Finland: A biogeographical comparison. *Ornis Fennica*, 70(2), pp. 71–77.
- Kaplan, R. (1973) Some psychological benefits of gardening. *Environment and Behavior*, 5(2), pp. 145–162.
- Kaplan, R. (1985) Nature at the doorstep: Residential satisfaction and the nearby environment. *Journal of Architectural and Planning Research*, 2(2), pp. 115–127.

- Kaplan, R. (2001) The nature of the view from home: Psychological benefits. *Environment and Behavior*, 33(4), pp. 507–542.
- Kaplan, R. and Kaplan, S. (1989) The Experience of Nature: A Psychological Perspective, Cambridge University Press, New York.
- Kellett, J.E. (1982) The private garden in England and Wales. *Landscape Planning*, **9**(2), pp. 105–123.
- Kim, J. and Kaplan, R. (2004) Physical and psychological factors in sense of community: New Urbanist Kentlands and nearby Orchard Village. *Environment and Behavior*, 36(3), pp. 313–340.
- Kinzig, A., Warren, P., Martin, C., Hope, D. and Katti, M. (2005) The effects of human socioeconomic status and cultural characteristics on urban patterns of biodiversity. *Ecology* and Society, 10(1), p.23.
- Kuo, F.E. (2001) Coping with poverty: Impacts of environment and attention in the inner city. *Environment and Behavior*, 33(1), pp. 5–34.
- Kuo, F.E. and Sullivan, W.C. (2001a) Environment and crime in the inner city. Does vegetation reduce crime? *Environment and Behavior*, **33**(3), pp. 343–367.
- Kuo, F.E. and Sullivan, W.C. (2001b) Aggression and violence in the inner city: Effects of environment via mental fatigue. *Environment and Behavior*, 33(4), pp. 543–571.
- Kuo, F.E. and Taylor, A.F. (2004) A potential natural treatment for Attention-Deficit / Hyperactivity Disorder: Evidence from a national study. *American Journal of Public Health*, **94**(9), pp. 1580–1586.
- Lepczyk, C.A., Mertig, A.G. and Liu, J. (2004) Assessing landowner activities related to birds across rural-to-urban landscapes. *Environmental Management*, 33(1), pp. 110–125.
- Loram, A., Tratalos, J., Warren, P.H. and Gaston, K.J. (2007) Urban domestic gardens (X): the extent and structure of the resource in five major cities. *Landscape Ecology*, **22**(4), pp. 601–615.
- Maas, J., Verheij, R.A., Groenewegen, P.P. and de Vries, S. (2006) Green space, urbanity, and health: how strong is the relation? *Journal of Epidemiology and Community Health*, 60(7), pp. 587–592.
- MacDougall, E.B. (ed.) (1980) John Claudius Loudon and the early Nineteenth Century in Great Britain, Dumbarton Oaks Trustees for Harvard University, Washington, D.C.
- Macintyre, S., Ellaway, A., Hiscock, R., Kearns, A., Der, G. and McKay, L. (2003) What features of the home and the area might help to explain observed relationships between housing tenure and health? Evidence from the west of Scotland. *Health and Place*, **9**(3), pp. 207–218.
- Martin, C.A., Warren, P.S. and Kinzig, A.P. (2004) Neighborhood socioeconomic status is a useful predictor of perennial landscape vegetation in residential neighborhoods and embedded small parks of Phoenix, AZ. Landscape and Urban Planning, 69(4), pp. 355–368.
- Matlack, G.R. (1993) Sociological edge effects: Spatial distribution of human impact in suburban forest fragments. *Environmental Management*, 17(6), pp. 829–835.
- McKinney, M.L. (2002) Urbanization, biodiversity, and conservation. *BioScience*, **52**(10), pp. 883–890.
- McKinney, M.L. (2008) Effects of urbanization on species richness: A review of plants and animals. Urban Ecosystems, 11(2), pp. 161–176.
- McMichael, T. (2001) Human Frontiers, Environments and Disease: Past Patterns, Uncertain Futures, University Press, Cambridge.
- Melles, S., Glenn, S. and Martin, K. (2003) Urban bird diversity and landscape complexity: species-environment associations along a multiscale habitat gradient. *Conservation Ecology*, 7(1), p.5.
- Merrill, S.B. (2004) The role of open space in urban planning. Conservation Biology, 18(2), p.294.
- Moran, M.A. (1984) Influence of adjacent land use on understory vegetation of New York forests. Urban Ecology, 8(4), pp. 329–340.
- Mörtberg, U. and Wallentinus, H.-G. (2000) Red-listed forest bird species in an urban environment—assessment of green space corridors. *Landscape and Urban Planning*, 50(4), pp. 215–226.

- Nasar, J. L., Fisher, B., and Grannis, M. (1993) Proximate physical cues to fear of crime. *Landscape and Urban Planning*, 26(1-4), pp. 161–178.
- Nasar, J. L. and Jones, K. M. (1997) Landscapes of fear and stress. *Environment and Behavior*, 29(3), pp. 291–323.
- Nassauer, J.I. (1995) Messy ecosystems, orderly frames. Landscape Journal, 14(2), pp. 161–170.
- Nassauer, J.I. (2004) Monitoring the success of metropolitan wetland restorations: Cultural sustainability and ecological function. *Wetlands*, **24**(4), pp. 756–765.
- Niemelä, J. (1999) Ecology and urban planning. *Biodiversity and Conservation*, **8**(1), pp. 119–131.
- Office of the Deputy Prime Minister (ODPM) (2002) *Living places: Cleaner, Safer, Greener*, The Stationery Office, London.
- Office of the Deputy Prime Minister (ODPM) (2006) *Enhancing Urban Green Space*, The Stationery Office, London.
- Özgüner, H. and Kendle, A.D. (2006) Public attitudes towards naturalistic versus designed landscapes in the city of Sheffield (UK). Landscape and Urban Planning, 74(2), pp. 139–157.
- Park, C.R. and Lee, W.S. (2000) Relationship between species composition and area in breeding birds of urban woods in Seoul, Korea. *Landscape and Urban Planning*, 51(1), pp. 29–36.
- Parsons, R., Tassinary, L.G., Ulrich, R.S., Hebl, M.R. and Grossman-Alexander, M. (1998) The view from the road: Implications for stress recovery and immunization. *Journal of Environmental Psychology*, 18(2), pp. 113–140.
- Pauleit, S., Ennos, R. and Golding, Y. (2005) Modeling the environmental impacts of urban land use and land cover change – a study in Merseyside, UK. *Landscape and Urban Planning*, 71(2-4), pp. 295–310.
- Payne, S.R., Devine-Wright, P. and Irvine, K.N. (2007) Peoples' perception and classification of sounds heard in urban parks: semantics, affect and restoration. Presentation at the 36th International Congress on *Noise Control Engineering (Inter-Noise 2007)*, Istanbul, Turkey.
- Pincetl, S. and Gearin, E. (2005) The reinvention of public green space. Urban Geography, 26(5), pp. 365–384.
- Rebele, F. (1994) Urban ecology and special features of urban ecosystems. *Global Ecology and Biogeography Letters*, **4**(6), pp. 173–187.
- Rohde, C.L.E. and Kendle, A.D. (1997) *Human well being, natural landscapes and wildlife in urban areas*, English Nature Science Report no. 22. English Nature, Peterborough, UK.
- Rollinson, D.J., O'Leary, R. and Jones, D.N. (2003) The practice of wildlife feeding in suburban Brisbane. *Corella*, 27(2), pp. 52–58.
- Royal Horticultural Society (2006) Front Gardens: Are we Parking on our Gardens? Do driveways Cause Flooding? Royal Horticultural Society, Woking, UK.
- Ryan, R. (2005) Exploring the effects of environmental experience on attachment to urban natural areas. *Environment and Behavior*, **37**(1), pp. 3–42.
- Schroeder, J.W. and Anderson, L.M. (1984) Perception of personal safety in urban recreation sites. *Journal of Leisure Research*, 16(2), pp. 178–194.
- Shibata, S. and Suzuki, N. (2002) Effects of the foliage plant on task performance and mood. *Journal of Environmental Psychology*, 22(3), pp. 265–272.
- Sime, J. (1993) What makes a house a home: the garden? In *Housing: Design, Research, Education*, (eds. M. Bulos and N. Teymur), Avebury, Aldershot, UK.
- Smith, R.M., Gaston, K.J., Warren, P.H. and Thompson, K. (2005) Urban domestic gardens (V): relationships between landcover composition, housing and landscape. *Landscape Ecology*, 20(2), pp. 235–253.
- Smith, R.M., Warren, P.H., Thompson, K. and Gaston, K.J. (2006a) Urban domestic gardens (VI): environmental correlates of invertebrate species richness. *Biodiversity and Conservation*, 15(8), pp. 2415–2438.
- Smith, R.M., Gaston, K.J., Warren, P.H. and Thompson, K. (2006b) Urban domestic gardens (VIII): environmental correlates of invertebrate abundance. *Biodiversity and Conservation*, 15(8), pp. 2515–2545.
- Sorace, A. (2001) Value to wildlife of urban-agricultural parks: A case study from Rome urban area. *Environmental Management*, **28**(4), pp. 547–560.

- Speirs, L.J. (2003) Sustainable planning: the value of green space. *Sustainable Planning And Development*, **6**, pp. 337–346.
- Stanners, D. and Bourdeau, P. (1995) The Urban Environment. In Europe's Environment: The Dobříš Assessment, (eds. D. Stanners and P. Bourdeau), European Environment Agency, Copenhagen.
- Stigsdotter, U.A. and Grahn, P. (2004) A garden at your doorstep may reduce stress Private gardens as restorative environments in the city. Proceedings of the *Open Space-People Space Conference*, 27–29th October 2004, Edinburgh, Scotland. Paper 00015.
- Takano, T., Nakamura, K., and Watanabe, M. (2002) Urban residential environments and senior citizens' longevity in megacity areas. The importance of walkable green spaces. *Journal of Epidemiology and Community Health*, 56(12), pp. 913–18.
- Taylor, A.F., Kuo, F.E. and Sullivan, W.C. (2002) Views of nature and self-discipline: Evidence from inner city children. *Journal of Environmental Psychology*, 22(1-2), pp. 49–63.
- Thwaites, K., Helleur, E. and Simkins, I. M. (2005) Restorative urban open space: Exploring the spatial configuration of human emotional fulfilment in urban open space. *Landscape Research*, 30(4), pp. 525–547.
- Tratalos, J., Fuller, R.A., Warren, P.H., Davies, R.G. and Gaston, K.J. (2007a) Urban form, biodiversity potential and ecosystem services. *Landscape and Urban Planning*, 83(4), pp. 308–317.
- Tratalos, J., Fuller, R.A., Evans, K.L., Davies, R.G., Newson, S.E., Greenwood, J.J.D. and Gaston, K.J. (2007b) Bird densities are associated with household densities. *Global Change Biology*, 13(8), pp. 1685–1695.
- Ulrich, R.S. (1984) View through a window may influence recovery from surgery. *Science*, **224**(4647), pp. 420–421.
- Ulrich, R.S. and Addoms, D.L. (1981) Psychological and recreational benefits of a residential park. *Journal of Leisure Research*, **13**(1), pp. 43–65.
- Ulrich, R.S., Simons, R.F., Losito, B.D., Fiorito, E., Miles, M.A. and Zelson, M. (1991) Stress recovery during exposure to natural and urban environments. *Journal of Environmental Psychology*, **11**(3), pp. 201–230.
- United States Fish and Wildlife Service. (2001) National Survey of Fishing, Hunting and Wildlife Associated Recreation, United States Government Printing Office, Washington D.C.
- Verheij, R.A. (1995) Explaining urban-rural variations in health: a review of interactions between individual and environment. *Social Science and Medicine*, **42**(6), pp. 923–935.
- Ward Thompson, C. (2002) Urban open space in the 21st century. *Landscape and Urban Planning*, **60**(2), pp. 59–72.
- Wells, N.M. (2002) At home with nature: the effects of nearby nature on children's cognitive functioning. *Environment and Behavior*, **32**(6), pp. 775–596.
- Wray, S., Hay, J., Walker, H. and Staff, R. (2005) Audit of the Towns, Cities and Development Workstream of the England Biodiversity Strategy, English Nature research report number 652. English Nature, Peterborough.
- Yang, W. and Kang, J. (2005) Soundscape and sound preferences in urban squares: A case study in Sheffield. *Journal of Urban Design*, **10**(1), pp. 61–80.

Chapter 11 Complementarities and Contradictions

Colin Jones, Mike Jenks and Glen Bramley

Introduction

The final chapter addresses the key question which brings together the research presented in the book: *to what extent and in what ways does urban form affect sustainability?* It begins by recounting the elements of urban form and the oft repeated mantras so prevalent in policy. It summarises the relationships between urban form elements and the dimensions – social acceptability, energy use, travel and mobility, ecology and biodiversity and economic viability – set out in earlier chapters. The lessons drawn from some key overarching sustainability issues – the potential impact of building sustainable developments on lifestyles, the use of open space and adapting the city are then examined. The next sections identify the essential links between sustainability and elements of urban form and the important messages for sustainability policies identified in the earlier chapters. The chapter concludes with some ideas about rethinking sustainability.

There is an increasingly intense debate in policy and practice about sustainability and a key issue is to what extent the adaptation of the physical form of cities and the way people live in them and travel around them can improve it. To date many of the dominant arguments about urban sustainability policy have been put forward in simple black and white terms, couched as the compact urban form versus dispersed urban form or urban sprawl. Viewed from this perspective it is perhaps not surprising that 'compact city' arguments have been more attractive to governments and sustainability policies have focused on increasing the density of urban development, improving public transport, ensuring a mix of uses and containing sprawl. Despite this widespread adoption of these policies, the evidence base supporting it is very limited.

The book has sought to get beneath the veneer of the sustainability debate by, first, looking in depth at the issues and second, to present the evidence from a study that has been designed to assess individual components of the arguments.

C. Jones (⊠)

School of the Built Environment, Heriot-Watt University, Edinburgh, UK

The analysis has attempted to unpack some of the complexities of the relationships between urban form and sustainability.

The task is not straightforward as the measurement of urban form is not easy and sustainability itself is an elusive concept widely open to interpretation. Despite the apparent simplicity of policy prescriptions, the concept of urban sustainability has arguably become increasingly complex. The issue can be as wide as a city's ecological footprint, influenced not just simply by the physical urban form of a city but also the transport network, the nature of the wider (national) urban system and links between the urban areas. This could include factors such as the degree of inter-urban commuting and freight distribution patterns, as well as the waste disposal process and air pollution. However, this research has viewed sustainability from a defined perspective – it draws a line and defines and tests the key arguments in theory and literature that have been used to link urban form to sustainability (Figure 11.1).

Unlike many other texts on this subject, the core of the book looks at the sustainability of cities from the environmental, social and economic perspectives. There are a series of potential contradictions and complementarities as social acceptability, environmental concerns and economic forces vie for hegemony. The planner's challenging task is to address and resolve the tensions from this triangle of potential conflicts. On the one hand cities are the centres for wealth generation, the consumption of goods and services and the hubs of the economy. On the other hand they are consumers of land and the primary polluter or producer of waste. At



Fig. 11.1 The relationships between elements of urban form and sustainability dimensions researched

the same time cities are where the majority of the population lives and so need to be places in which they are happy to live. Weighing up these different factors is not simple – there are different languages and metrics applied by professional specialists and there are also divergent interests and time horizons from different groups of the community that create barriers to a commonality of views.

This chapter explores and attempts to draw some conclusions about these apparent conflicts based on a structured review of empirical evidence. As a first step it is useful to have a reminder of urban form and this is followed by a review of the different dimensions of sustainability and summary of the findings.

Urban Form

Urban form can be considered at a number of different spatial scales and can be measured in a number of different ways. In this research it is characterized in terms of five elements – the pattern of land use, accessibility defined by transport infrastructure, density, housing/building characteristics and urban layout. Each of these, to a degree, overlap and it is difficult to completely isolate individual components.

The core research in this book is based on five UK cities. Within these cities, case studies were selected comprising neighbourhoods located in the inner, middle and outer areas, effectively representing a slice through each city. These offer a range of different types of British city, and have some similarities to other cities in a wider world context. They are cities that historically grew around a central core, and have a mono-centric urban form, with central business and commercial cores, higher density housing in the inner core with lower densities moving out towards the peripheries. In general terms they are at the lower point in the global spectrum of urban densities (although relatively high by UK standards) and are also less dominated by their central cores than, for example, US cities. The analysis of street networks reveals a broadly consistent negative gradient from the city centre in terms of their density and complexity, interconnectedness and degree of sprawl. Gross population densities follow a similar spatial structure.

This internal physical spatial structure is mirrored by the housing tenure and demographic patterns. Younger people with few children live primarily in the higher density inner areas while older households and families are predominantly in low density suburban areas. Private rented housing is focused in the inner areas and outer rings are often mainly owner occupied. Social housing in British cities is spatially concentrated and located mainly in inner areas but with some larger peripheral estates. The spatial structure of each of the five cities therefore aligns closely to the more 'traditional' mono-centric city form and there is a strong relationship between physical urban form of neighbourhoods and socio-economic-demographic characteristics of households. Nevertheless, there were also some decentralising trends apparent.

Core Sustainability Dimensions

Transport: Travel and Mobility

The adverse environmental and energy consumption consequences of the dominant use of private cars in cities are perhaps the most fundamental source of demands for more sustainable urban policies. The increasing use of the car has contributed to the widespread suburbanisation of cities and urban sprawl. The process is self-reinforcing as new low density residential areas necessitate the use of a car for commuting, accessibility to shops, schools and virtually every facet of everyday life. It is a significant worldwide issue and has intensified calls for sustainable transport policies based on the promotion of public transport. As a result many have argued that cities can be designed to create urban neighbourhoods with local services nearby in such a way as to stimulate the use of public transport, walking and cycling.

This approach faces a number of major hurdles. There is the obvious travel flexibility provided by a car that means that it is often chosen over other convenient transport modes. In the absence of constraints such as road pricing or parking restrictions at key destinations the car is likely to continue to be the principal method of travel to most activities for those who have access to one. Indeed people often choose to travel further distances by car to access a 'better' range of shopping than use the more limited local options. Even if urban neighbourhood design could stimulate greater use of local amenities, it is probable that the savings generated will only be used for wider travel. In addition not all urban travel is determined by reference to a person's home and surrounding neighbourhood, so that for example shopping and leisure travel patterns may be linked to commuting and work location. Overall the assumption that redesigning urban form can bring about a substantial change in travel behaviour is open to question. A key issue is to what extent residential location choice is the consequence of household travel preferences. Chapter 3 addresses this question and investigates the effectiveness of urban form as a means to induce more sustainable travel patterns. In particular the chapter contributes to our understanding of the underlying relationships by exploring the extent to which the availability of cars, trips made and distance travelled by a household is influenced by urban form.

Chapter 3 finds that traditional urban forms characterised by moderately high densities of housing, mixed land-uses, proximity to public transport and grid-pattern road layouts are definitely linked with lower levels of car availability which in turn are associated with lower trip frequencies and shorter travel distances. Overall car ownership levels are found to increase with decreasing population density and increasing distance from city centre. There are a number of explanations for these phenomena. Localities with high population density can support a greater range of local services and facilities while at the same time usually offering good public transport services. In the highest density areas, limited parking supply and regulatory control can also be expected to play a role in deflating car demand. However, self-selection of residential location on the basis of travel preferences

is *not* found to be a major influence on car travel. Household income is a strong influence on car ownership, although this is lessened in higher density areas; it also influences trip making behaviour but has no measurable effect on distance travelled.

A distinction can be drawn between travel within neighbourhoods and commuting. Local travel is influenced by the frequency of use of a number of key services/facilities and these are found to decline with distance from home. There remain a number of limitations to these conclusions. For example, residents with lower levels of access may organise their use of services more efficiently than those living in close proximity to local services, or may be more likely to use services located close to their workplace.

The relationship between travel and urban form is therefore not simple. Generally, urban form is not found to influence the number of work-day journeys of the employed, the number of trips for work and non-work journeys and the additional distance required to undertake work travel over and above the shortest route from home to work. At the neighbourhood level, too, the relationships between travel and use of local services is clouded and the analysis suggests that (re)designing a neighbourhood per se will not necessarily bring substantial change to travel behaviour. Other measures will be needed to secure a fully 'sustainable' shift in travel behaviour, for example relating to the higher taxation and pricing of fuel, increased regulation and stronger direct management of travel demand. Nevertheless, the research does provide some support for basic propositions linking density with car ownership and use.

Environment: Ecology and Biodiversity

The growth of the urban landscape across the world has a substantial impact on the natural habitat and hence on the disruption of ecosystems and biodiversity. Much has been written on these transformations but very little focus has been placed on how the precise geography of towns and cities can and does shape and moderate these processes within their boundaries. Given the inevitability of an increasing proportion of the world's population living in urban areas it is an important sustainability policy area. A key variable is the spatial pattern of green spaces in a city in terms not only of its extent but also its composition. The road network in particular configures green spaces and potentially forms a barrier (or a conduit) for dispersal of animals and plants.

Chapter 4 assesses relationships between urban form and green space extent, ecosystem service provision and biodiversity. In this context urban form is best measured in terms of population density, the patterns of coverage of different land use types, and the degree of connectivity of different patches of land cover. The analysis is based primarily, although not exclusively, from empirical studies of these issues undertaken within the five case study cities.

Looking first at biodiversity, higher urban density is found to be strongly associated with a reduction in total green space coverage, and to influence the connectivity of vegetated patches across the townscape. Increased population density has implications for essential elements of the local ecosystem that are mediated by green space. These encompass the regulation of water and temperature regimes, carbon sequestration and the provision of pest control and pollinators across the urban landscape. One striking relationship between biodiversity and density is given by the incidence of bird species. Levels of bird species richness showed a hump-shaped relationship with housing density, rising initially as density increased, but then declining sharply at highly urbanized locations. In addition, reductions in the scale and quality of green space through higher densities lead to substantial restrictions on recreation within urban areas and access to an experience of nature, especially for children.

An understanding of the distribution of ecosystem elements and the associated pattern of biodiversity across the urban landscape is crucial for predicting the consequences of policies designed to increase urban density. There is a clear message from the evidence that there is a decline in ecosystem functions and biodiversity potential with increasing urban density at the higher end of the range. Nevertheless the analysis suggests that there are opportunities for policies that are designed to improve the environmental and ecological performance of urban areas for any given level of urban density. But this will require further research.

There are more worrying issues linked to high densities, notably observed reductions in numbers of species, even those expected to be most able to live in and exploit urban environments. The result is a difficult ecological trade-off between using more green fields for new development and accommodating more people into existing cities. Greater urban management is required to minimise the ecological impact of urban change. Spatially explicit area selection exercises could help identify areas that are crucial to maintaining an effective ecosystem function, and those areas that might efficiently be used for high density residential developments. A better understanding of this trade-off is urgently required if we are to plan for increasingly dense urban forms and ensure that declines in green space and biodiversity will not lead to an substantially impaired ecosystem function, and by implication the provision of ecosystem services to the majority of the human population. Such declines would also degrade human experiences of nature such that contact with the natural world will be diminished in both quantity and quality. There are no easy answers to the sustainable ecological dimension of cities.

Social Acceptability

A sustainable city must be a place where people want to live and work. Closer examination suggests that social acceptability comprises two broad concepts – social equity and the sustainability of the local community. The social dimension of sustainability therefore incorporates issues of social justice, social inclusion, social capital and social cohesion. Social equity can be defined in terms of ease of access

to local services, facilities and opportunities. A community which is sustainable displays high levels of what is formally described as social capital and/or social cohesion that can be translated as pride in and attachment to the locality, good social interaction, safety/trust and stability. These attributes offer residents what can be summarised as a good 'quality of life'. Overall social sustainability is reflected in high levels of satisfaction with home and neighbourhood, and an appreciation of the local environment.

In Chapter 5 elements of social sustainability are measured and their relationships with urban form are quantified. For most aspects of sustainability of community (particularly pride/attachment, stability, neighbourhood and home satisfaction, and perceived environmental quality) lower density suburbs appear 'best'. These aspects of the social dimension challenge the 'compact city' orthodoxy, but there are some counter-balancing benefits of compactness in the equity aspect of social sustainability, particularly access to services. The complexity of the relationship with density is reinforced by the finding that social interaction is best at medium densities, while some aspects are neutral (e.g. community participation).

Some care has to be taken in the interpretation of these relationships as they are also partly the result of factors not directly linked to urban form, such as housing tenure and the social composition of neighbourhoods. In general, disadvantages of compactness are more marginal once socio-demographic characteristics of residents are controlled for. Poverty is often more important than urban form – who lives where, and whether they are able to choose where they live, matters. However, accessibility to key services, including a supermarket, within the neighbourhood are identified as very important for different groups of residents such as the unemployed, older people and young families and play a significant role in social and community life.

The relationship between open space and social sustainability is not straight forward. The use and social benefits derived from open and public spaces are not just based on design but also dependent on the maintenance and supervision of these spaces. Perceptions of safety within open spaces are crucial to their use and linked to maintenance. However, management solutions are not necessarily easy and shared communal gardens and spaces in higher density flats can in particular be problematic.

The social dimension of urban sustainability cannot be linked to a precise urban form, although lower density neighbourhoods have a significant advantage in quality of life and community. Good quality, easily accessible services and facilities in a neighbourhood are important, and are promoted by higher density, confirming the findings of the analysis of travel in Chapter 3. Services will also reflect public provision policies and local planning. Everyday management of urban space, rather than its spatial pattern and scale, is a crucial contributor to social sustainability. Overall social sustainability is not just dependent on neighbourhood physical characteristics and urban form but also a function of urban management and limited in particular by the extent and concentration of poverty within a city.
Environment: Energy Use

A key imperative of the drive toward a sustainable future is the reduction of energy consumption and a high proportion of the energy consumption of cities is linked to buildings. Long term policies can be aimed at adapting the existing housing stock to improve their energy conservation and promote carbon neutral new housing. A wider question is to what extent domestic energy use is influenced by the type of housing and the built form. Arguments in favour of high density development in the form of terraced housing or flats have emphasised a potential reduction in transport (energy) use, and evidence from Chapter 3 provides some support for this view albeit with significant qualifications. It is possible that these types of housing also lower domestic energy consumption to some degree compared with semi-detached and detached houses.

Chapter 6 assesses this proposition but the analysis on domestic energy use finds only a weak relationship with built urban form. Residential energy use is determined less by house type and more by the level of occupancy within a home, and the type of heating and domestic appliances used. Lifestyles and demographics therefore influence energy consumption more than building type, so urban form is of only marginal importance. Thus the connection with urban form is an indirect association with occupancy (particularly number of bedrooms) – the smaller the house the less energy used, and the smaller the house the more likely it is to be part of a higher density urban form. There is also a slight suggestion that the organisation of a city's economy and urban form in shaping commuting may impact on energy consumption via home working.

Economic Viability

The environmental and social dimensions are at the centre of sustainability concerns for cities but economics is also a key influence on why cities exist in the form that they do. The urban economy is a fundamental influence on sustainability, as any policy solution will find it difficult to work counter to spatial market forces in the long term. Sustainability policies have to balance wealth creation, economic performance and the spatial pattern of economic activity determined primarily by markets (albeit usually subject to planning) with meeting social and environmental objectives.

It is in this context that Chapter 7 assesses the arguments and examines to what extent the spatial economy constrains change. The crux of the debate has centred on normative models of high density cities versus dispersed communities. Proponents of the former view base their case on the benefits from magnified agglomeration economies in a compact city and/or on the high infrastructure costs of sprawl. The chapter finds that these high density arguments are based on too simplistic a concept of agglomeration economies which does not take into account cities as dynamic entities with spatial land use patterns subject to change. The urban dispersal 'alternative', partly the inevitable outcome of market forces, is

found to have the draw-back that the existing longstanding urban dispersal trends have substantially increased commuting distances and travel to work, as well as public/shared infrastructure costs. These 'externality' effects are not considered by individual market decision-makers in their individual decisions to decentralise.

There is no single urban form that satisfies economic sustainability uniquely. Despite this, spatial economic forces are still a key to sustainable cities in that land use patterns must be economically viable; otherwise market forces will mean they will not be stable in the long run. An adequate supply of housing for the workforce and their families is also necessary to ensure affordable housing.

The evidence suggests a potential for adapting the existing city form to make it more economically sustainable. The commercial and industrial property sectors demonstrate some locational flexibility and imply that the viability of these land use markets will not constrain different urban forms, including decentralised urban systems. However, the analysis of the housing market shows that many households prefer low density housing and that there appears to be a household life-cycle element to residential location choice. Younger and non-pensioner single households live in the central city areas but move out to suburban locations as they move through the family life cycle. This means that it will be difficult to encourage more concentrated urban forms without significant changes to the underlying forces of city housing markets. In addition the patterns of viability in the city housing markets suggest there are major constraints to reshaping local housing markets, particularly where concentrated poverty makes market-led redevelopment unviable.

Some Key Sustainability Issues

While the individual dimensions of the sustainability city have been examined, it is evident that these overlap. The social dimension incorporates social equity and the use of local services which in turn are linked to their economic viability. An important contributor to environmental sustainability is the amount and location of green space but this is similarly a major contributor to the quality of life and social sustainability. The extent of green space is also a function of the operation of the housing market and the relative profitability of different housing types. Three chapters examine these inter-relationships focusing on practical issues to improve sustainable urban living – increasing neighbourhood density, providing sustainable developments for sustainable lifestyles, and the provision and management of green space.

Intensifying Neighbourhood Density

A major theme of many protagonists for change is the need for higher residential densities to enhance sustainability. As the analysis above demonstrates these arguments are not entirely founded on empirical analysis. Nevertheless Chapter 8

accepts this paradigm and seeks to develop a theoretical underpinning of, and a methodology for, urban intensification and regeneration. It transforms indicators of sustainability into strategic concepts of urban restructuring and regeneration and applies them to an inner city area of Glasgow. The core of this research is generation of an assessment tool for the systematic measurement and evaluation of levels of (un)sustainability of urban areas with the help of threshold and target values. The output of this analysis in the form of a land use plan is then presented using two and three-dimensional models for the selected neighbourhoods, which are then tested for viability.

The detailed investigation of Govan, a working class area of Glasgow, shows that a systematic tool can be applied to restructure cities into a series of sustainable neighbourhoods, all with their own amenities within walking distance of peoples' front doors with local centres directly linked by public transport. However, the outcomes of the analysis also demonstrate that the achievement of intensification targets can be a rather painful process, which will require both strategic and multilateral thinking and the courage not to compromise too soon given the substantial upheaval involved.

Sustainable Developments and Sustainable Behaviour

In addition to sustainability strategies for the restructuring of neighbourhoods is the building of developments designed to be sustainable. Chapter 9 addresses a seemingly simple question: do residents of new housing developments, built according to sustainability principles, behave any more sustainably than the population in general? The research examines three elements of sustainable behaviour related to neighbourhood-scale design:

- residents' home-based sustainable behaviours; including reducing energy and water consumption, recycling and composting waste and supporting wildlife in gardens
- · residents' travel behaviour and car ownership
- residents' 'social sustainability' behaviours; such as social participation and the use of local services, businesses and facilities

These elements encompass all aspects of domestic sustainability behaviour. The analysis is based on a survey of households living in thirteen 'sustainable' developments and the findings are benchmarked against more general national surveys.

The findings are, perhaps surprisingly, more negative than positive. Households living in the 'sustainable' developments only appear to behave more sustainably than the rest of the population with regard to home-based resource efficiency activities, such as water and energy use. Behaviour toward recycling and frequency of use of local facilities are equivalent to national benchmarks. More importantly when it comes to other activities, such as travel to work by car, owning (or having access to) a car, social participation, encouraging wildlife, and composting they behave less sustainably than the population in general. While living in these developments supported positively a more sustainable lifestyle it does not make the residents take further steps in the process. Although the residents are more knowledgeable about sustainability issues than the general population, this is not necessarily translated into positive activity.

It is possible that the fact that these developments are new, works against a social sustainability indicator such as social interaction because households may not have had time to put roots down. The nature of high density may also militate against such activities as outdoor composting and encouraging wildlife in gardens. However, the residents in these sustainable developments have a social profile biased toward higher proportions of managerial/professional households, and this could act negatively against reducing car use but positively on other attitudes toward sustainability. The analysis suggests that while sustainable developments can provide the built environment to support sustainable lifestyles they do not necessarily lead to a change in behaviour.

Ecological and Psychological Benefits of Urban Green Space

Public green space in cities is generally designed and managed to support the recreational activities of people with little or no reference to the ecological benefits. At the same time from a social-psychological perspective urban nature is an important if under-recognised component of quality of life for urban residents. Similarly urban public and private green spaces are an understated ecological resource and represent significant components of regional and national biodiversity conservation networks. In addition urban biodiversity patterns arise in response to, and are maintained specifically by, repeated human activity. Human interventions can not only degrade but can also improve urban ecosystems. Chapter 10 argues that there are benefits to managing urban green spaces better from an ecological perspective that will also increase the benefits to people.

The analysis examines biodiversity and human recreation activities in private and public green space. Proximity to green space is a strong selection factor in park usage and proximity has been shown to play a role in levels of physical activity and self-reported health. This research suggests that access to green space varies across social groups. However, the extent to which access to green space equates with access to biodiversity is currently unknown, and will depend on how the biodiversity value of urban green spaces is distributed across the urban landscape. Given that the psychological benefits of green spaces increase with their biodiversity, there is a need to incorporate biodiversity measures into an audit of access to green space for urban residents.

The research on private gardens demonstrates that individual decisions made by landowners can result in large scale effects on environmental conditions. This raises the intriguing possibility that garden bird feeding could be galvanised to influence the conservation status of a particular species occurring within urban areas. Urban environments support nationally important populations of some species, yet in many cases conservation biology has yet to incorporate the urban environment into conservation planning exercises. The degree to which people-nature interactions can be harnessed to achieve national biodiversity targets remains unknown. Such interactions might also contribute to targets relating to urban liveability. Given that private gardens cover such a large proportion of many modern cities, surprisingly little is known about motivations and drivers of private garden management. From a policy perspective, both the conservation of urban biodiversity and the enhancement of public health depend on a better understanding of this issue

Public green space and private gardens might, to some extent, provide alternatives for contact with nature in urban settings. Indeed, in Sheffield there is a negative correlation between the extent of public green space and private garden space across the city. However, the degree to which these different kinds of green space can substitute for one another to support biodiversity is unclear as these two very frequently play different roles. For example, public green space can promote community integration while social interactions in gardens are focused around a private social network. The persistence of privately provided green space may also be less secure as gardens are built over during infill development.

Overall urban green spaces are valuable for both biodiversity and for people, and management solutions must complement both values. Because of the highly synergistic nature of the relationship between these two types of benefits, the needs of management for biodiversity value and human well-being need not always indicate divergent solutions. To discover and implement such solutions will require systems of implementation that can actively trade off ecological value and benefits to human well-being. Urban design and management strategies must therefore be sought that complement both the ecological and the human value of urban green space.

Urban Form and Sustainability

Having examined how the different dimensions of sustainability link to elements of urban form, this section suggests how some relationships between them work. As noted earlier, elements of urban form overlap, and for simplicity the general focus here is on density in its broadest sense as a key physical variable. It is a useful vehicle to summarise our findings and also represents the primary variable at the centre of much debate about the shape of sustainable cities. However, some caution must also be applied as there are various measures of density, and related urban form elements do not reduce to a single dimensional variable.

The conclusions from the preceding chapters can be summarised in the following points:

• Green space in total, and its connectivity, is reduced at higher urban densities, which affects the level of ecosystem services provided as well as opportunities for recreation and other benefits to human health and wellbeing.

- 11 Complementarities and Contradictions
- There are positive relationships between biodiversity and human well-being across both public and private green spaces within the city, so indirectly relating wellbeing to urban form in conjunction with the above point
- One key measure of biodiversity, bird species richness, shows a hump-shaped relationship with housing density, initially increasing but declining sharply in more highly urbanized sites.
- Road and associated infrastructure costs increase by a higher than proportional rate with falling density.
- Energy use in buildings is only weakly linked to the built urban form, being determined primarily by socio-demographics and lifestyles.
- There is some flexibility in the intra-urban location of commercial and industrial land uses. While the city centre remains the dominant location for offices and retailing there are benefits from decentralisation. These patterns suggest that the economics of these land uses will not constrain the establishment of compact or dispersed urban forms.
- Residents' sense of pride and attachment to neighbourhood, their sense of safety, their rating of environmental quality, satisfaction with the home and residential stability are all greater in lower density neighbourhoods, controlling for a wide range of other influences.
- Social interaction and friendliness and participation in local collective activities tend to be highest in areas of medium density, after allowing for socio-demographics.
- Use of local services is greater in areas of higher density and better accessibility.
- Car ownership, particularly the propensity of higher income households to have two cars, is markedly less in higher density areas allowing for other factors. Limited parking supply and regulatory control in high density neighbourhoods can also be expected to play a role in reducing car demand.
- Travel distances in general and by car tend to be less in central and higher density areas, and where employment locations are closer, and these tendencies will reinforce the car ownership effect. These effects are not simply due to selection effects but appear to be causally related to density and location.
- Private residential choices in general favour low density suburbs but there is a household life-cycle element to residential location choice. Younger and non-pensioner single households live in the central city areas but move out to suburban locations as they move through the family life cycle.
- Gardens emerge strongly as a positive feature for both individual preference satisfaction and social/public goods, including biodiversity and ecosystem services.

The interpretation of some of these relationships between urban form and dimensions of sustainability must be treated with some care. In studying these links, for example between density and social outcomes it is essential to control for the influence of other intervening variables, as these factors can have significant influences and affect conclusions. For example, poverty in particular is a key influence on the social sustainability of a neighbourhood. Lifestyles, including the use of cars, are not necessarily determined by urban physical form, although they may be influenced and constrained by it. Consumption in its broadest sense is more broadly determined by incomes and social class. This is reflected in the patterns of the spatial viability of new housing development being more dependent on the socio-economic status of a neighbourhood than any particular feature of urban form.

The implication of the above findings is that there are indeed complementarities and contradictions, and that there will be trade-offs between them. The detailed findings that measured each aspect give a clear understanding of the magnitude or significance of each. Thus this research challenges the simple compact city thesis in that, for example, the social dimension of sustainability is compromised by high density living while there are no clear economic benefits to this urban form beyond the greater viability of local services. Very high density cities also have significant negative biodiversity and ecosystem effects. These negative effects mean that from an environmental perspective there are ecological pluses and minuses of different urban forms within a sub-regional context and that minimising the built up urban area through high densities/maximising the surrounding green space may not be the optimum solution.

Towards Urban Sustainability

The research presented in this book queries the implicit direct role of physical determinism on human behaviour in many of the sustainability arguments. Creating high density cities does not necessarily reduce car travel. Building sustainable developments does not mean that the residents have sustainable lifestyles. Simply (re)designing a series of 'sustainable' neighbourhoods is not a sufficient condition for the sustainable city. Wider socio-economic-demographic influences can be more dominant influences. In policy terms, these point towards other measures, for example the use of social marketing via media and promotion, tax incentives, regulations and so forth.

The relationships between household behaviour, lifestyles and attitudes and urban form is complex, although there is a clear family life cycle relationship. While high density living may be socially acceptable to some for others, particularly families, suburban living appeals as a compromise between providing space and the availability of services. Not only is there no single optimum density for a neighbourhood but market forces will ensure that density will vary with location, shaping any planned urban form in the medium to long term. Sustainable planning policy and guidance cannot ignore these forces.

The influences on the elements of urban form are argued to be the outcome of the operation of real estate markets within a framework set by a city's transport infrastructure and moderated by local planning policies. They reflect a myriad of private choices, whether by individual households seeking a home (balancing affordability and access against a wish for more space) or by developers looking at the optimal mix for developments within constraints set by planning. However, these decisions also have pervasive external or 'public' effects: each new development changes the outlook for existing residents, reduces or changes green space, increases congestion, and has consequences for urban sustainability. There are inevitably contrasts between the private and public perspective on urban form that planning policy has to manage.

Policies toward housing as the dominant land use in cities are key to urban sustainability. The evidence on commercial and industrial land uses is that they are relatively agnostic with regard to specific urban forms. The housing market and household attitudes toward and satisfaction from particular built forms are keys to the sustainability of cities. This requires not only market viability (derived in part from household preferences) but also a sufficient supply of housing that is affordable. Cities are dynamic entities it is not sufficient for urban planning to stifle physical development simply to create a compact city; the city-region must also be able to accommodate any urban growth in a sustainable manner. Our findings suggest a range of densities should be accommodated, and this will require a holistic approach that encompasses the provision of social and physical infrastructure to ensure a sustainable solution.

An important aspect of sustainable urban policy (and shaping urban development) is changing the transport infrastructure and public/private travel costs. To date explicit sustainable transport policies primarily aimed at switching people from private to public transport have not been very effective; stronger price incentives or demand management measures may be necessary.

Achieving sustainability may require fundamental change to existing neighbourhoods and cities, and where necessary it has to be accepted at the outset as a long term process. The substantial upheaval that is often necessary within a neighbourhood to establish more intensive land use and higher densities may engender public opposition. Any reshaping of cities requires major adjustments to the local real estate markets and may require substantial public expenditure costs. Such strategies to adapt the city form will require a consensual long term public policy framework to ensure confidence for private property investment.

Thus, policy emphasis may be better placed on improved management of cities, especially of neighbourhoods, infrastructure and green space. The message is that sustainability is not a simple concept and hence policies to address it will be diffuse and vary by neighbourhood. There may be scope for greater intensification in some areas that have underused space, while in others the reverse may be true. Neighbourhood strategies need to be bespoke rather than be subject to a generalised 'one size fits all' approach.

A Wider Perspective

The evidence presented in the book questions some of the established views on sustainability and urban form. The research presented in this book is set within a UK context, so how far can the lessons of urban sustainability be taken?

The cities investigated here are relatively high income, with modest population growth, or even decline. Within them there are variations in wealth and poverty, and densities, although varied, are neither very low, nor very high by international standards. The five cities show that lower density areas tend to be associated more with higher income groups with higher levels of satisfaction, and that high density is more associated with poorer people. This may be fairly typical of most cities in the UK, although there are exceptions. For example, some of the wealthiest areas of London or Edinburgh have high density urban forms, and there are many examples of impoverished low density estates on the periphery of the UK's larger cities (including a couple of examples within our sample).

If the frame of reference is widened, and cities are considered in other parts of the world, then again the issues, or at least balance between them changes. In cities of burgeoning size and complexity, with rapid growth and extremes of poverty, such as Mumbai with some 50% of its population living in slums, the balance, management of growth and development will require different solutions. Nevertheless, the principles expounded in this research and the forces identified here are of use in all urban contexts. First, it has identified the key elements of urban form and the potential relationships with the three key dimensions of sustainability. Second, it has shown how this can be measured and understood. Third, it has indicated where potential trade-offs might be found, albeit in the context of Western cities.

A city form is neither wholly sustainable nor unsustainable, rather it is a question of degree. Further, the extent of sustainability depends on which dimension of sustainability is considered. The complexity of the various dimensional trade-offs in terms of different metrics, value judgements, essential or acceptable standards and location specific factors, and the weighting between them means that policy decisions aimed at improving sustainability are not necessarily easy.¹

The scale and context of any such comparison or assessment has also to be considered. A city/region as a whole may be more or less sustainable, but so also may be different neighbourhoods within a given city. A city is a system of parts and different parts are needed for the whole to function – a city needs its centre, its suburbs, and so forth, and different demographic groups or businesses need different types of place to function most effectively. Furthermore, cities are not static phenomena so that any such scoring would need to consider a trajectory or change or only represent a guide to future policy.

This discussion suggests that, given the dynamic nature of cities and in the absence of an single uniquely sustainable urban form, sustainability policies should not seek as a long term goal to create a definitive urban form. A single such form does not exist, but instead, as we have shown, there are a number of avenues down which more sustainable forms can be achieved. It also indicates that simply examining the sustainability dimensions of a core city is not enough, and that it is necessary to consider at least the city region and probably the urban system with a sub-region.

¹ The use of multi-criteria evaluation is one way forward for this general problem, e.g., see: Munda G (2006) 'Social multi-criteria evaluation for urban sustainability policies', *Land Use Policy*, 23, 1, 86–94.

A more incremental and varied approach to urban sustainability implies the development of a set of guiding principles to adapt and manage the existing city forms and to plan urban development. It accepts that sustainability policies have to start with the cities and towns that already exist, and planning is often about incremental change. Given the complementarities and contradictions within the dimensions of urban sustainability, this implies examining strategic sustainability options for a particular city in a holistic manner. Such strategies could encompass a range of generic urban development or management scenarios including: intensification of some neighbourhoods within the existing built-up area of the city; integrative green space planning, incremental expansion at the edge of the existing urban area; corridor developments with intensive nodes oriented to transport network changes; reconfiguration of transport infrastructure linked to expansion of existing satellite towns or one or more new settlements. The list is only indicative but it would also be intrinsic to these sustainability options to integrate the principal infrastructural investments which would accompany them. Beneath these strategic level options there are many micro issues about the constraints on individual new developments, adaptations to the existing built environment and the integration of land uses.

Conclusions

There are no simple messages from this book. There are no easy sets of urban form sustainability rules or forms for policy makers, there are many trade-offs, and often conflicting ones at that. Indeed much of the existing debate is to a certain extent misplaced, as it is of little value to polarise the issues into either the compact city or the dispersed urban form. Both exist side by side, and it is the balance between them, and the way in which they can be enhanced, merged, connected and made more sustainable that matters. Policies to improve urban sustainability must be more bespoke. Urban form does affect sustainability but the relationships are subtle and they can be overstated. Many influences on urban sustainability are not the consequence of a city's form but exogenously determined by the socio-economic-demographic characteristics of the local population or non-spatial factors such as the price of energy.

The book does show that there is scope for re-engineering existing cities. It suggests a sensitive management of urban intensification that is targeted on key urban areas, especially related to accessible public transport. It means that where higher densities are achieved, the management and incorporation of both public and private green space needs to be maintained, perhaps through green corridors and smaller, but useable private gardens. It suggests that, where family homes are to be built, forms need to be designed to discourage urban sprawl at the same time as fulfilling family aspirations, and some of the medium density schemes illustrated in Chapter 9 could provide a model. It suggests that a holistic approach to urban management is desirable. The research here has shown that there is a need to manage

the trade-offs between, for example, density in relation to biodiversity, social acceptability, transport, and the real estate market. A balance could be achieved if it is accepted that densities will vary from the lower (not lowest) to the acceptably high, and that it is the transport links between them, and overall cumulative effect on sustainability that might be managed to accommodate growth.

It is a universally accepted truth that research tends to raise more questions than answers. This book is no exception. It has uncovered some new and surprising relationships. It has given both comfort and pain to advocates of the compact city. It has shown that it is possible to measure something as complex a city in relation to sustainability. It has identified some key aspects where trade-offs can be identified, and which can provide a basis for policies for the sustainability. But, in the end we have to conclude that there are no simple answers or clear relationships between urban form and the dimensions of the sustainable city. Instead we offer strong pointers for policy makers on strategies to improve the sustainability of cities.

Appendix A MCA Spatial Analysis

The aim of the spatial network analysis applied in this research is to:

- identity type of layout (e.g. gridded, tree-like etc.);
- numerically express the level of compactness and complexity of the layout.

The total length of streets in a neighbourhood along with node density and sprawl can express the compactness of layout, for example, more accurately mathematically speaking than a simple measure of the residential density of land area. Multiple Centrality Analysis (MCA) provides an objective way of consistently identifying elements such as streets and junctions as well as a set of scale variables that distinguish the spatial and network characteristics of an urban layout.

Fundamental to MCA is the representation of spatial relationships in a city or neighbourhoods as a primal graph and then the calculation of a centrality space (point) by its location in the network (betweenness) or by its physical distance to all other spaces in the network (closeness). Thus a space with high 'betweenness' value indicates a strategically central space that it is crucial to a large number of short routes¹ between two other spaces. While betweenness and closeness values are useful in distinguishing relative centrality or compactness of individual spaces within a neighbourhood, measures such as cost, efficiency, meshedness and fractal (sprawl) dimensions and complexity distinguish the structural characteristics (size, type, shape) of the spatial network of a whole neighbourhood or sub area.

Two indicators of the structural characteristics of the street layout of a neighbourhood, taken from the range detailed in Table A.1, have been used in the book. First, node density is calculated as the number of street intersections per hectare. Second, a composite 'complexity index' is derived to systematically identify types of layouts such as grid-like or tree like patterns (see Fig. A.1). The relationship between the index and the layout typology was tested by a number of schematic layouts before validating them with real cases from the study areas.

The complexity index is constructed in two steps based on assessments of the 'efficiency' of a spatial network that is determined by reference to the choices of routes available. Choices of routes within a network can be calculated using two MCA efficiency indicators: ('network efficiency' and 'MST efficiency': for details see Cardillo et al., 2000²). Network efficiency calculates the efficiency of the actual

Measurement	Description	Examples of aspects/features measured			
Indicators of spatial characteristics					
Betweenness Centrality (individual space, neighbourhood & Sub-area)	Strategic spaces that are located on the short routes between a pair of spaces in the network	Spaces within the city and neighbourhoods			
Closeness Centrality (individual space, neighbourhood & Sub-area)	The proximity of a space to all other spaces using shortest path	All possible routes from A to B Average distance of all paths between A and B			
Indicators of structural characteristics					
Node density	Number of street junctions per hectare	Street junctions (e.g. T-junctions, culs-de-sac)			
Cost of network	The total physical length of the spaces	Size of network			
Efficiency of network (Neighbourhood & Sub-area)	The evenness of distribution and connectedness of a network	The ratio of proximity of two spaces compared with the virtual distance (i.e. 'as the crow flies' distance)			
Note on Cost and efficiency (Neighbourhood & Sub-area)	Efficiency and cost of network indicates the layout's compactness MST (Minimum Spanning Tree) used to compare above with most efficient layout	Cost of network + node density MST: spatial network is reduced to a tree pattern by removing choices of routes before calculating the cost and efficiency			
Meshedness (Neighbourhood & Sub-area)	 Meshedness of whole network calculates the complexity of a network 	Number of circular routes in the network			
Fractal dimension (Neighbourhood & Sub-area)	 The sprawl of a network measured by taking into account both physical area and the network 	Measure a regular sized box to cover the perimeter of the network			
Complexity index (Neighbourhood & Sub-area)	 Index can indicate the layout pattern (i.e. more tree-like or grid-like) 	Ratio of MST to Network cost			

Table A.1 Indicators of urban layout

network, while MST (Minimum Spanning Tree) efficiency calculates the efficiency of the network once all the possible choices of routes have been reduced to a minimum. These two indicators are then combined to give the 'complexity index' which is the ratio of efficiency values calculated by the minimum spanning tree (MST) to network method. A high value ratio (where there is a greater difference



sac) the pattern

Fig. A.1 Urban layouts and the 'cost' and 'efficiency' of their networks

between the two values) indicates fewer choices of routes, and a more tree-like pattern and a less efficient network. A low value ratio (where there is a lower difference between the two values) indicates a higher number of routes, a more grid-like pattern and a more efficient network. Fig. A.1 shows some schematic urban layouts to illustrate this assessment.

Notes

- 1. For more information on MCA, see www.humanspacelab.com
- 2. Cardillo, A., Scellato, S., Latora, V. and Porta, S. (2006) Structural properties of planar graphs of urban street patterns, *Physical Review E*, **73**: 66–107

Appendix B Overview of Case Study Neighbourhoods

The analysis presented is based on the statistics listed in Chapter 2, Table 2.6 and Table B.1.

Inner Neighbourhoods

The highest densities are in the inner neighbourhoods of Edinburgh and Glasgow where traditional tenements and flats are prevalent. Both of these neighbourhoods have net densities above 225 dwellings/ha. A very high proportion of buildings in these neighbourhoods are more than four storeys in height. Although a traditional form of housing in Scotland, there are now few families living in these tenemented neighbourhoods, which tend to be dominated by small households with one or two adults. Aside from tenements, the inner case studies in the English cities are very similar with lots of tall buildings, lots of flats and few families. The main exception to this is Oxford, where only half of homes are flats and there is a significant proportion of terraced housing. Patterns of tenure vary greatly between the inner case study neighbourhoods. In both Leicester and Sheffield more than 70% of homes are rented, with social housing making up more than 50% of housing stock. Edinburgh and Glasgow's inner neighbourhoods have more of a balance between owner occupation and rented accommodation - with owner occupation accounting for around 50% of housing stock and private renting for 25% of housing stock. Oxford is also exceptional here with little social housing in the inner case study neighbourhood and just over 30% privately rented.

As might be expected, residential buildings are only part of the building stock in these inner neighbourhoods. In the neighbourhoods studied, around 15-20% of the land in the neighbourhood is covered by non-residential buildings – although the Edinburgh inner neighbourhood is notable for having a much smaller proportion of land covered by non-residential buildings (6%). These buildings are mainly used for retail, with office uses also occupying a significant proportion of non-residential properties. The inner neighbourhood of Glasgow is the main exception to this rule, with a far higher proportion of offices than retail uses. This is largely explained by the location of the case study neighbourhood which coincides with the commercial rather than retail centre of the city.

	Urban Form	
Edinburgh Inner	Density	Gross density 92.3 Net density 270.5
(2956 lia)	Land use	 5% of buildings are mixed use 14% of the area is covered by residential buildings, 11% by residential gardens and 6% by non-residential buildings. 24% is grace space
	Public transport Layout Building characteristics	 Mainly buses/ 34% residents do not own a car Compact, with grid and cul-de-sac form 78% of buildings are between 4 and 6 storeys 13% have access to a private garden; 73% access to a shared garden
Glasgow Inner (3694 ha)	Density	Gross density 34.7 Net density 226.2
	Land use	14% of the area covered by residential buildings; 1% by residential gardens; and 21% by non-residential buildings: 29% is green space
	Public transport	Buses and train station within case study area/ 37% residents do not own a car
	Layout Building characteristics	 Deformed compact grid 62% of buildings between 4 and 6 storeys, and 13% more than 6 storeys 8% have access to a private garden; 28% access to a shared garden.
Leicester Inner (3635 ha)	Density	Gross density 15.7 Net density127.1
	Land use	7% of area is covered by residential buildings, 4% by residential gardens and 21% by non-residential buildings: 27% is green space
	Public transport Layout Building characteristics	 Mainly buses/ 86% residents do not own a car Deformed wheel, radial 34% of buildings between 4 and 6 storeys, and 28% more than 6 storeys 10% have access to a private garden; 25% access to a shared garden.
Oxford Inner (3363 ha)	Density	Gross density 30.6 Net density 83.7
	Land use	10% of the area is covered by residential buildings, 10% by residential gardens and 15% by non-residential buildings: 31% is green space
	Public transport	Mainly buses (with main bus station) and train station close to case study/ 59% residents do not own a car

 Table B.1
 Some Physical Characteristics of the Neighbourhoods

	Urban Form	
	Layout Building characteristics	Crucifix form, with small blocks at the centre 35% of buildings are 2 storeys and 37% are 3 storeys 43% have access to a private garden; 25% access to a shared garden.
Sheffield Inner (4027 ha)	Density	Gross density 22.8 Net density 116.9
	Land use	14% of buildings are mixed use, 6% of the area is covered by residential buildings, 10% by residential gardens and 9% by non-residential buildings; 38% is green space
	Public transport	Buses and trams with train station outside 400 m buffer zone/ 81% residents do not own a car
	Layout	Deformed compact grid
	Building	35% of buildings are 3 storeys 27% are more than 6
	characteristics	storeys 22% have access to a private garden; 20% access to a shared garden.
Edinburgh Between (4107 ha)	Density	Gross density 37.9 Net density 69.5
	Land use	14% of the area is covered by residential buildings and 31% by residential gardens; 24% is green space
	Public transport	Mainly buses/ 26% residents do not own a car
	Layout	Predominantly gridded, not orthogonal
	Building characteristics	29% of buildings between 4 and 6 storeys56% have access to a private garden; 45% access to a shared garden.
Glasgow	Density	Gross density 33.1
Between (9863 ha)		Net density 68.4
	Land use	14% of the area is covered by residential buildings and 40% by residential gardens 16% is green space
	Public transport	Buses and train station within case study area/ 15% residents do not own a car
	Layout	Deformed grid with some compact grids
	Building characteristics	36% of buildings are 3 storeys and 47% are 4 to 6 storeys
		31% have access to a private garden; 66% access to a shared garden.
Leicester Between (2066 ha)	Density	Gross density 48.2 Net density 79.9
	Land use	22% of the land is covered by residential buildings and37% by residential gardens<i>9% is green space</i>

Table B.1 (continued)

	Urban Form	
	Public transport Layout Building characteristics	Mainly buses/ 37% residents do not own a car Deformed grid 79% of buildings are 2 storeys 71% have access to a private garden; 8% access to a shared garden.
Oxford Between (4049 ha)	Density	Gross density 39.9 Net density 80.8
	Land use	14% of the area is covered by residential buildings and 33% by residential gardens 22% is green space
	Public transport	Relatively poor provision of bus routes due to proximity to city centre. Train station is close to case study/ 50% residents do not own a car
	Layout	Elongated deformed grid with compact grid within
	Building	47% of buildings are 2 storeys, 30% are 3 storeys
	characteristics	57% have access to a private garden; 25% access to a shared garden.
Sheffield	Density	Gross density 37.5
Between		Net density 59.1
(4964 ha)		
	Land use	16% of the area is covered by residential buildings and 47% by residential gardens
	Public transport	Mainly buses with tram stops close to case study/ 47% residents do not own a car
	Layout	Deformed compact grid
	Building	79% of buildings are 2 storeys
	characteristics	75% have access to a private garden; 8% access to a shared garden.
Edinburgh Outer (17227 ha)	Density	Gross density 18.3 Net density 26.6
	Land use	10% of the area is covered by residential buildings and 34% by residential gardens
	Public transport	Mainly buses/ 9% residents do not own a car
	Lavout	Compact super grid
	Building	32% of buildings are single storey and 60% are 2 storey
	characteristics	88% have access to a private garden; 13% access to a shared garden.
Glasgow Outer (6456 ha)	Density	Gross density 8.2 Net density 46.3
	Land use	2% of the area is covered by residential buildings and 8% by residential gardens 77% is green space
	Public transport	Mainly buses with train station outside case study area/ 19% residents do not own a car

Table B.1 (continued)

	Urban Form	
	Layout Building characteristics	Clustered dispersed culs-de-sac 84% of buildings are 2 storeys 86% have access to a private garden; 4% access to a shared garden.
Leicester Outer (4998 ha)	Density	Gross density 17.2 Net Density 24.5
	Land use	9% of the area is covered by residential buildings and 54% by residential gardens 17% is green space
	Public transport	Mainly buses/ 19% residents do not own a car
	Layout	Very deformed grid with culs-de-sac
	Building	96% of buildings are 2 storeys
	characteristics	97% have access to a private garden; 1% access to a shared garden.
Oxford	Density	Gross density 30.8
Outer	,	Net density 62.8
(4449 ha)		
	Land use	11% of the area is covered by residential buildings; and 30% by residential gardens
	Public transport	Buses only/46% residents do not own a car
	Lavout	Predominantly culs-de-sac
	Building	76% of buildings are 2 storeys
	characteristics	80% have access to a private garden; 12% access to a shared garden.
Sheffield Outer (8660 ha)	Density	Gross density 18.6 Net density 26.5
	Land use	11% of the area is covered by residential buildings and 57% by residential gardens 13% is open space
	Public transport	Buses only/ 22% residents do not own a car
	Layout	Curvilinear with culs-de-sac
	Building	79% of buildings are 2 storeys
	characteristics	84% have access to a private garden; 11% access to a shared garden.

Table B.1 (continued)

Between Neighbourhoods

As a group the between case study neighbourhoods are characterised by their diversity. Net residential density ranges from 60 to 80 dwellings/ha. Two and three storey homes are common, although in both Glasgow and Edinburgh around one third of homes are between 4 and 6 storeys (e.g. tenements), and the vast majority of residents have access to a private or shared garden. Flats dominate the Glasgow between neighbourhood making up 80% of housing stock, while terraced houses dominate in Sheffield. The other between neighbourhoods have a more mixed

housing stock, and tend to include a variety of terraced, detached and semi-detached houses as well as flats. Tenure mix is also variable, ranging from almost no social renting in the Leicester case study neighbourhood (lowest) to around one-quarter in the Edinburgh case study neighbourhood (highest). As might be expected, the variety in housing type and tenure is reflected in the resident population. Retired and family households both feature here, although they usually account for no more than half of households.

Non-residential land uses also vary considerably amongst this group. While retail is the most important non-residential land use in all of the 5 between neighbourhoods, both Glasgow and Oxford are distinguished by a significant proportion of office buildings. Generally between neighbourhoods are also included some industrial buildings, which are not so plentiful in either inner or outer case study neighbourhoods.

Outer Neighbourhoods

The outer case study neighbourhoods fall into two groups: Sheffield, Leicester and Edinburgh typically have net densities around 25 or 26 dph. Glasgow and Oxford on the other hand have much higher residential densities of 46 and 63 dph respectively. In the lower density suburbs, the housing stock is almost exclusively 2 storeys in height, with some single storey housing (bungalows) in the Edinburgh neighbourhood. Residential gardens account for a significant proportion of land cover in these neighbourhoods. This is particularly true in the outer neighbourhoods of the English cities studied. Here, around 50% of land is taken up by residential gardens *in addition to* open green space. The outer case study neighbourhoods are also notable for the small proportion of rented accommodation – although Oxford is also exceptional here, with 40% of homes owned by social landlords and 10% in shared ownership.

Non-residential buildings in the outer case study neighbourhoods are more varied in their character than in the inner and between neighbourhoods. No clear pattern is evident here. Retail uses dominate in the Edinburgh and Sheffield outer neighbourhoods, while warehouses are dominant in Leicester and industrial uses in Glasgow and Oxford.

Index

Note: The letters 'f', 'n' and 't' following the locators refer to figures, notes and tables respectively

A

Accessibility/transport infrastructure, 146-147 impact on, 147 'pedshed,' 24 transport system, 24 Acioly, C., 4 Additional housing, demand for, 166-167 average household size, 166 Barker review, 166 dwelling density, 166 environmental/ecological arguments, 167 resistance, 166-167 Addoms, D.L., 219 Affordability constraint, 151, 160 Agglomeration economies, 146, 148-151, 160, 246 See also Economics of city forms Akbari, H., 90 Alberti, M., 91 Allen-Wardell, G., 90 Alonso, W., 146, 149 Alpine Close, Maidenhead, 197f Amersham Road, Reading, 197f Anas, A., 149 Anderson, L.M., 224 Anderson, W.P., 21 Anthropogenically-perturbed ecosystems, 90 Area profiles, case study, 32–33 CityForm research, 33 empirical analysis in five British cities small number of neighbourhoods, 32 Arnold, C.L., 86 Atkinson, R., 109-110 Attention-deficit/hyperactivity disorder, 218 Attwell, K., 6 Audirac, I., 111

Avendaño, V.L.G., 90 Aydinalp-Koksal, M.V., 137

B

Badoe, D.A., 7-8, 55 Bagaeen, S., 163-182 Bailey, N., 109 Baillie, S.R., 94 Baker, K., 129–142 Baker, N., 84 Balram, S., 219 Banister, D., 55, 60 Barbosa, O., 92–93, 217, 229 Barker review, additional housing, 166 Barton, H., 110, 149, 166, 169, 171, 189-191 Basantani, M., 5 Bastin, L., 76, 80, 229 Bed spaces per hectare, 27 Beer, A., 78, 219 Begg, I., 10 Behaviours car owning, 208-209 factors influencing car ownership, 209f number of cars owned/available, 209, 209f encouraging wildlife, 206, 206f factors, encouraging to walk/cycle/use public transport, 208f home-based sustainable behaviours, 188 efficient, use of water, 189 encouraging wildlife in gardens, 189 home energy consumption, reduction of, 188–189 waste recycling and composting, 189 home energy use and water efficiency, 203-204, 203f-204f

local services/amenities/businesses, use of, 192 links, common, 192 shorter journeys, fuel-efficient modes of transport, 206-208, 207f social participation, 191, 210 travel behaviour and car ownership car owning, 190-191 parking space for pool of shared cars, 191 shorter journeys, fuel-efficient modes of transport, 190 walking/cycling/public transport use, 190 use of local services, 210-211, 210f waste recycling and composting, 204-205, 205f Bell. S., 219 Bentley, I., 169 Berens, G., 219 Bernardini, C., 218, 225 Bhat, C., 55, 57, 225 Bhatti, M., 225 Bibby, R., 87 Bicycle, 35, 190 Bierwagen, B.G., 81 Biesmeijer, J.C., 90 Biodiversity and environment, 75-98 See also Environment and biodiversity and people, spatial configuration, 226-230 Central Arizona-Phoenix region in U.S., 227 coverage by green space, 227 ecological surveys of gardens, 229-230 EEA, 228 green space management, 230 hump-shaped relationship, 227 Index of Multiple Deprivation, 228 patch species richness, 229 pattern of biodiversity, 227 UK government agency, 228-229 urban matrix, 229 vegetation structure or habitat heterogeneity, 226–227 woodland habitats, 227 and urban form, 93-97 housing density, 95-96, 96f relationship between noise levels/impervious surfaces, 97f relationships between housing density, 96f

species richness of taxonomic groups, 93-94 surfaces for Sheffield of birds, 95f urban noise, 96 Biodiversity value, elements of, 221 Bixler, R.D., 224 Bjerke, R., 225 Boardman, B., 142 Boarnet, M.G., 4, 54-55, 60 Bogota, 4 Bolund, P., 6, 83 Bonaiuto, M., 6 Borooah, V.K., 58 Botanical gardens (large formal spaces), 222 Bourdeau, P., 92, 228 Bowers, K., 109 Bowman, R., 95 Bramley, G., 10, 21-49, 105-124, 158, 239-256 "Breathing zones," London, 217 BREDEM, see UK Building Research Establishment Domestic Energy Model (BREDEM) Breheny, M., 4, 9, 23, 110, 167, 186 Bridge, G., 111 The Bristol Accord, 107 Brown, C., 21-49, 105-124, 218 Brownhill, D., 189–190 Brueckner, J., 150 Brundtland, definition, 145 Bryman, A., 122 Buchmann, S.L., 90 Buffer zone, 27, 263t Building fabric aspects, 129, 131 Building regulations, 25, 131 Built environment and sustainable behaviour, relationship between, 185-186 Bunker, R., 110 Burgess, J., 7, 219 Burgess, R., 4 Burton, E., 2, 8, 24, 106, 108, 110-111 Butler, S., 186

С

CABE Space, *see* Commission for Architecture and Built Environment (CABE Space) Cabeza, M., 98 Cadenasso, M.L., 76 Calthorpe, P., 4, 110 Camagni, R., 149 Cameron, E., 91 Cannon, A.R., 6, 94

Capello, R., 149 'Capital,' broad classes of, 106 economic, 106 environmental, 106 social, 106 Carbon emissions, 87, 90 Carbon pools, 87, 89-90 Carbon sequestration, 84, 87-90, 244 carbon emissions, 90 carbon pools, 87, 89-90 distribution of circumferences of trees across urban Sheffield, 89f patterns in tree species richness and size, 88 significance of, 90 surfaces for Sheffield of trees, 88f urban forest management, 89 Cardillo, A., 257, 259 n2 Carmona, M., 9, 192 Carreiro, M.M., 90 Car use, traditional urban forms in reducing, 54-55 Catanzaro, C., 225 CBD, see Central business district (CBD) CEC, see Commission of the European Communities (CEC) Central Arizona-Phoenix region in U.S, 227 Central business district (CBD), 21, 147, 158 Cervero, R., 4, 7, 55, 150 Chace, J.F., 75, 94, 216 Chamberlain, D.E., 217, 222 Characklis, G.W., 87 Chen, W.Y., 84, 225 Cheptou, P.O., 90-91 Cheshire, P., 10, 150 Chiesura, A., 219 Choice tasks, housing, 157 Church, A., 225 Churchman, A., 23 Chu, Y-L., 57 Ciccone, A., 149 Cities, characteristics of, 35 Edinburgh, Scotland capital administrative/financial services centre, 35 Glasgow and Sheffield, 35 lowest proportion of private renters, 35 traditional industrial cities, 35 Leicester/Oxford economic structure/thriving ethnic minority community, 35 tourist and business centre, 35 City, adaptation

additional housing, demand for, 166-167 case study tests, 181-182 indicators of sustainable development urban regeneration and development, current practice, 164-165 urban renaissance/urban neighbourhoods, 165 tool application, levels of (un)sustainability measurement, 170 Drumoyne area investigation, 177-180 Govan areas investigation, 175-177 tool development of sustainability measurement, 168-170 urban intensification, 167-168 See also Drumoyne area; Govan areas City form economics of accessibility/transport infrastructure, 146-147 characteristics of built environment, 147 - 148density, 28t, 147 land use patterns, 146 infrastructure costs, 157-159 road lengths and densities, 158, 159t urban infrastructure costs, types, 157 See also Economics of city forms CityForm's case study neighbourhoods survey, 218 - 219Clausen, C.P., 91 Clergeau, P., 95, 226–227 CNU, see Congress of the New Urbanism (CNU) Cognitive functioning in children, 217-218 Coley, R.L., 218 Collie, M., 110 Collins, J.P., 227 Colton, R.D., 140 Commercial space, 147, 154 Commission for Architecture and Built Environment (CABE Space), 76, 122, 219, 221 Commission of the European Communities (CEC), 2, 106, 110-111 'Compact city,' 1, 8, 14, 110, 124, 239, 245 Compactness or sprawl, degree of, 37 Complexity index, 257-258 Congress of the New Urbanism (CNU), 110 Constraints-based accessibility, 55, 66 Construction costs model, 155 Conway, H., 217 Cook, E.A., 81 Coombe, D., 8

Cooper Road, East Sussex, 198f Cornelis, J., 222 Côté, S., 191 Cowan, R., 25 Crane, D.E., 87 Crane, R., 4, 54–55 *Crepis sancta:* Asteraceae, 91 Crisp, V., 6 Crookston, M., 9 "Cultural cues to care," 224 Curitiba, 4, 17 Curtis, S., 91 Cutsinger, J., 26

D

Dair, C., 183-212 Darlow, A., 110 Dataset, 26, 56-57, 60, 72, 120, 131, 138, 140 - 142Davies, C., 2, 81 Davies, R.G., 75-98 Dawson, R., 2 Dehnen-Schmutz, K., 96 Dempsey, N., 1, 5, 21-49, 105-124, 188 Density, urban element English housing policy, 23 measuring, 27 bed spaces per hectare, 27 DPH, 27 habitable rooms per hectare, 27 pph, 27 residential, definition, 27 planning policy and practices, 23 Trafalgar Square in London, 23 UK policy makers, 23 Department of Business Enterprise and Regulatory Reform, 130f, 137 Department of the Environment, Transport and Regions (DETR), 1-2, 7, 24, 54, 106, 110, 131, 166, 190, 192 DETR, see Department of the Environment, Transport and Regions (DETR) Devine-Wright, P., 215-231 De Vries, S., 76, 218, 225 Diamond, H.L., 111 Dichotomous variables, 138 See also Dataset Dickman, C.R., 229 Dicks, S.E., 75 Digital elevation models, 130 Dimensions, core sustainability economic viability, 246-247 environment, ecology and biodiversity, 243-244

environment, energy use, 246 social acceptability, 244-245 transport, travel and mobility, 242-243 Dimensions of sustainable city, future city 2 CityForm research, structure of, 13 economic viability methodology, 11-12 researching sustainable city, 10 global significance, issue of, 4-5 urban form and claims to sustainability, 1 - 3Dines, N., 219 Dixon, T., 192 Domestic energy consumption, 131-134, 142, 246 fabric, 131-132 dwelling's energy efficiency, 132 household appliances use, 133-134 "mood" lighting, 134 models, 129, 131, 134f occupancy, 132-133 owner-occupancy, type, 132 Dongtan, 5 Donnelly, R., 229 DPH, see Dwellings per hectare (DPH) Dragicevic, S., 219 Drumoyne area, 177-180 total area. 177 two best scenario neighbourhoods, 177 Duany, A., 7, 54, 110 Duhme, F., 6, 76, 78, 81 Dunnett, N., 76, 219 Dunse, N., 146 Dwellings per hectare (DPH), 27, 28t, 35-36, 118, 121f, 131, 173t, 178t, 193t Dwelling types, 113, 129-132, 132f, 135-138, 174t, 180

E

East Midlands region of UK, 219 'Eco-aesthetic,' 195 Ecological surveys of gardens, 229–230 Economics of city forms accessibility/transport infrastructure, 146–147 characteristics of built environment, 147–148 building maintenance, 148 influencing factors, 147 density, 28t, 147 CBD, 147 land use patterns, 146 demand factors, 146 Economic viability, 3, 9-10, 12, 15, 145-161, 169, 180, 246-247 economics of city form, 145-148 accessibility/transport infrastructure. 146-147 characteristics of built environment. 147 - 148density, 28t, 147 land use patterns, 146 of individual land uses, 151 commercial and retail change, 153-154 housing, 154-157 offices, 151-152 retailing, 152-153 infrastructure costs and city form, 157-159 methodology, 11-12 researching sustainable city, 10 urban form, economic performance and sustainability, 148-151 Ecosystem services and urban form carbon sequestration, 87-90 pollination, 90-91 provision of pest control, 91-92 recreation, 92-93 temperature regulation, 84-86 diurnal temperature range, 84 water regulation, 86-87 See also Individual Eco-towns, 2, 5 EEA, see European Environment Agency (EEA) Effland, W.R., 90 Ekanem, E., 225 Elements of urban form, 8, 14, 21-49, 22f, 56-57, 63, 110, 145, 153, 216, 239, 250, 252, 254 accessibility/transport infrastructure, 24 measuring, 28-29 GIS-generated land use map, 29f characteristics of cities, 35-36 density, 23 housing and building characteristics, 25 land use, 23-24 measuring density, 27-28 measuring housing/building characteristics, 29 - 31indicators of accessibility, 31t indicators of housing/building characteristics, 31t indicators of land use, 30t measuring land use, 27-28 indicators of density, 28t measuring layout, 32

measuring urban form, 26-27 neighbourhoods, 36-38 characteristics, 37 lavouts, 37-38 non-physical aspects, 21 physical characteristics of cities, 35-36 density and housing types, 36t profiles, case study area location of the case study cities, 34f MCA-generated map of Leicester, 33f 'urban form,' description, 21 urban layout, 25 Elkin, T., 106 Energy use analysis results, 138-140, 139t-140t controls type, 140 glazing type, 138 TRV, 140 clusters, 130f, 138-139t modelling domestic energy consumption fabric, 131-132 household appliances use, 133-134 occupancy, 132-133 and urban form, relationship, 130-131 physical form of buildings, 130 variation in, research method cluster analysis, 137 data analysis procedure, 137-138 design of survey instrument, 136-137 energy consumption, 137, 139t home energy efficiency questionnaires, 136 multiple regression analysis, 137-138 samples selection, 135-136, 135f-136f simple regression, 137 Environmental/ecological arguments, 167 Environmental exposure, 217 Environment and biodiversity, 75-98 biodiversity/urban form, 93-97 relationship between noise levels/impervious surfaces, 97f relationships between housing density, 96f surfaces for Sheffield of birds, 95f carbon sequestration, 87-90 distribution of circumferences of trees across urban Sheffield, 89f surfaces for Sheffield of trees, 88f ecosystem services carbon sequestration, 87-90 pollination, 90-91 provision of pest control, 91-92 recreation, 92-93

temperature regulation, 84-86 water regulation, 86-87 green space, 78-83 distribution, 81 environmental surfaces for Sheffield, 82f green space in Sheffield, 80f proportions of green space, 79-80, 79f relationship between date of first urbanization and proportions, 83f size, 80 total green space coverage, 78t pest control, provision, 91-92 holly leaf-miner demographics, 92t pollination, 90-91 recreation, 92-93 satellite image of Sheffield, 77f temperature regulation, 84-86 relationships between cover by impervious surface, 86f temperature surfaces across Sheffield, 85f water regulation, 86-87 Eremeeva, N.I., 90 European Commission, 2, 53-54 European Environment Agency (EEA), 228 European Union (EU), 53-54, 107, 168 Evans, A.W., 10, 149, 158 Evapotranspiration, 6, 87 'Everyday eight' local neighbourhood services, 24 Ewing, R., 26, 55, 110

F

Faeth, S.H., 91, 95 Falk, N., 8 Farthing, S., 24 Ferguson, N., 53-73 Fernández-Juricic, E., 222, 229 Finkel, S.E., 57 Fisher, T., 109 Flint, J., 110 Floyd, M.F., 224 Forest management, urban, 89 Forrest, R., 108 Forsyth, A., 224 Francis, M.H., 225 Freeman, H.L., 111, 217 Frey, H., 163-182 Fujita, M., 149 Fuller, R.A., 75-98, 215-231

G

Garden feed wild birds, 216 Garling, T., 6 Garvin, A., 219 Gas consumption, 138, 140-141, 141f Gaston, K.J., 75-98, 189, 215-231 Gearin, E., 219 Gehl, J., 25, 191 Geographical Information System (GIS), 11, 26-27, 29f, 63, 67, 135-136, 218 based platform, 27 Geography zone system, 56 Gerlach-Spriggs, N., 93, 217 Germann-Chiari, C., 92 Ghazoul, J., 90 Gibb, H., 229 Gibbons, J., 86 Gilbert, O.L., 6 Gill, S.E., 87 Gio, J.Y., 55, 57 GIS, see Geographical Information System (GIS) Glasgow's Govan area, 170, 180 Gleeson, B., 4 Global positioning system (GPS), 26, 30t Global significance, issue of, 4-5 Glue, D., 216 Glynn, T., 111 Golubiewski, N.E., 87 Gomez, F., 230 'A good quality of life', 107 Gordon, P., 111, 150 Govan areas characteristics, 175-177 objectives, 177 primary data generation, 175 secondary data generation, 175 GPS, see Global positioning system (GPS) Graham, S., 4 Grahn, P., 93, 217 Grant, J., 23 Grant, M., 218 'Gravity model', 152-153 See also Retailing Gray, D.E., 26 Great Notley, Essex, 198f Green belts, 150, 154 Green space and urban form, 78-83 distribution, 81 environmental surfaces for Sheffield, 82f gardens, purely green space, 78 high resolution vector mapping, 78 large patches of internal green space

building/housing density, 80 Edinburgh/Oxford, 79 public parks/playing fields, 80 Peak District National Park, 78 proportions of green space, 79-80, 79f relationship between date of first urbanization and proportions, 83f in Sheffield, 80f size, 80 small patches of green space domestic gardens/roadside verges, 80 Glasgow/Leicester, 79-80 total green space coverage, 78t urban area, definition, 78 Green space, ecological/psychological value, 215-231 collaborative research projects, 216 coverage, 76, 79, 86, 227 "greenness" of urban landscape, 215 interactions between people/urban nature, 218 - 230biodiversity/people, spatial configuration, 226-230 private green space, 225-226 public green space, 218-225 people/urban nature, 216-218 human impacts on urban nature, 216-217 people, impacts of urban nature on, 217-218 regional/national biodiversity conservation networks, 215 urban ecosystems, 215 urban residents, 215 Gregory, R.D., 94 'Grey' water systems, 3 Gross population densities, 37 Grove, J.M., 6 Guertler, P., 186 Guy, S., 5, 10

H

Habitable rooms per hectare, 27 Habitat heterogeneity, 217, 221, 226–227 Hadidian, J., 222 Handy, S.L., 1, 7, 55, 72, 93 Hanson, J., 25 Haughton, G., 110, 184, 187 Hayward, G., 220 Healey, P., 24 Healey, T., 191 Heat loss standards, 131 Hedgcock, D., 106 Henderson, J., 148 Henry, J.A., 75 Hermy, M., 222 Herzog, T.R., 218, 224 Hey, D., 76 High-rise city centre apartment dwellers, 25 Hillier, B., 25 Hirschfield, A., 109 Hochuli, D.F., 229 Holmes, C., 25 Home-based sustainable behaviours, 188 efficient, use of water, 189 encouraging wildlife in gardens, 189 home energy consumption, reduction of, 188-189 waste recycling and composting, 189 Hope, D., 227 Hope, S., 56 Household car availability, 57-59 neo-traditional developments, 57 ordered logit model for cars available to households, 58t for change in available cars on moving house, 59t Housing analysis steps, 154 (building) characteristics, 25 high-rise city centre apartment dwellers, 25 household questionnaire, 29 'lifetime homes' model, 25 measuring, 29-31 non-domestic buildings, 29 constituent attribute, 155 densities, 156 household characteristics, 157 house prices model, 155 spatial price structures, 154 Huberman, A.M., 122 Hull, R.B., 218 Human impacts on urban nature, 216-217 city features, 216 factors on spatial distribution, 216 species richness, 216 varying biological response, 216 human factors socio-economic deprivation, 217 wildlife gardening popularity, 217 invertebrate richness, 216-217 physical garden features, 216 Sheffield, neighbourhoods, 217 tree cover/habitat heterogeneity, 217

urbanization process, 216 Hunhammar, S., 6, 83 Hunter, C., 110, 184, 187 Hunt, R., 217 Hunziker, M., 222

I

IMD, see Indices of Multiple Deprivation (IMD) 'IMD Geog Access', 116-117, 117t Index of Multiple Deprivation, 174t, 228-229, 229f Indicators, sustainable development goals of research, 164 socially balanced and inclusive urban neighbourhood, 169 urban neighbourhood, characteristics economic, 170 urban form, 169 urban regeneration and development, current practice, 164-165 urban renaissance and sustainable urban neighbourhoods, 165 Indices of Multiple Deprivation (IMD), 113, 121 Individual land uses, economic viability of, 151 commercial and retail change, 153-154 new property forms, 154 housing, 154-157 analysis steps, 154 constituent attribute, 155 densities, 156 household characteristics, 157 house prices model, 155 spatial price structures, 154 offices, 151-152 retailing, 152-153 'gravity model', 152 Industrial age, pre/early, 217 Ingress Park, Kent, 186f, 193t-194t, 199f Integrated elements, 25-26 compatibility/statistical analyses, 26 pedestrian/cycling/public transport network, 26 Interactions between people/urban nature, 218 - 230biodiversity/people, spatial configuration, 226-230 green space private, 225-226 public, 218-225 Intrinsic biodiversity value, 219

Irvine, K.N., 215–231 Iverson, L.R., 81

J

Jackson, L.E., 92 Jacobs, J., 148 Jenerette, G.D., 84 Jenks, M, 1–15, 21–49, 110, 239–256 Jim, C.Y., 225 Jo, H.-K., 87 Johnson, E.A., 75 Jokimäki, J., 222 Jones, C., 1–15, 21–49, 145–161, 224, 239–256 Jorgensen, A., 92 Julian, D., 111

K

Kang, J., 222 Kaplan, R., 218, 222, 225, 231 Kaplan, S., 218, 222, 231 Katz, P., 2, 110 Kaye, J.P., 87 Kearns, A., 90, 108t Kellett, J.E., 229 Kendle, A.D., 217, 225 Khattak, A.J., 57 Kim, J., 218 Kintrea, K., 109 Kinzig, A., 6, 226-227 Klein, A.-M., 90 Klemens, M.W., 75 Knight, C., 110 Kowarik, I., 94 Kozak, D., 4 Krizek, K.J., 55 Krueger, R.A., 122 Kuo, F.E., 76, 217-218 Kwan, M-P., 55

L

Lacy, R., 6 Lafrance, G., 130 Lambregts, B., 4 Land Registry, 155 Land use, 23–24 description, 23 'everyday eight' local neighbourhood services, 24 'locally-unwanted land uses', 23–24 measuring, 27–28 'buffer zone', 27 GIS-based platform, 27

Index

NLUD, 27 primary data collection, 27 Langdon, P., 7 Larivièr. I., 130 Latora, V., 259 n2 Lavout measuring, 32 MCA, street network systems, 32 spatial network analysis, 32 urban historical development/planning/building regulations, 25 Lazarus, N., 186 Least squares regression, 114 Lee, C.-W., 148 Lee, W., 53-73 Lee, W.S., 229 Leicester subsample, 138 Leishman, C., 145-161 Lepczyk, C.A., 93, 217, 226 'Lifetime homes' model, 25 Lighting energy consumption, 130 Lindsay, M., 183-212 Littlefair, P.J., 131 Liu, S., 24 Livingstone, M., 109 Llewelyn-Davies, 1-2, 190 Local Agenda 21, 133, 164 'Locally-unwanted land uses', 23-24 Local services/amenities/businesses, use of, 192 links, common, 192 Lomas, K.J., 129-142, 215-231 London, 23, 89,217, 254 Loram, A., 225 Lorenz, K., 87, 90 Low, N., 4 Low-tech eco-design/communal-living, 188 Lynch, K., 145

Μ

Maas, J., 218 MacDonald, C., 145–161 MacDougall, E.B., 217 Macintyre, S., 93, 218, 225 March, L., 130 Mardaljevic, J., 25 Margo, J., 123 Marston, A., 192 Martin, C.A., 92, 227 Martin, L., 130 Marvin, S., 4, 5 Marzluff, J.M., 94, 229 Masdar, 5 Masnavi, M.-R., 9 Mason, C.F., 94 Matlack, G.R., 217 MCA, see Multiple Centrality Assessment (MCA) MCA spatial analysis, 257-259 McCann, P., 149 McKinney, M.L., 75, 94, 216 McMichael, T., 217 McPherson, E.G., 87, 90 Measurement, urban form, 26-27 GIS, 26 GPS, 26 PDAs, 26 prototype software, 26 'snapshot' approach, 26 street-by-street basis, 26 two-pronged process of data collection, 26 valuation roll and ordnance survey data, 26 Melles, S., 227 Mental fatigue, 217 Merrill, S.B., 230 Michael, S.E., 218 Miles, M.B., 122 Millennium Ecosystem Assessment, 93 Millennium village, 2, 165, 168 at Greenwich, 184, 195f-196f Miller, E.J., 7-8, 55, 224 Miller, H.J., 55 Mobility and travel car use, traditional urban forms in reducing, 54-55 distance travelled by car in Edinburgh (non-work day), 70f by car in Glasgow (work/non-work day), 69f multiple linear regression analyses/models, 67-68, 68t ordered logit model for change, 71t Travel Diary, 67 household car availability, 57-59 neo-traditional developments, 57 ordered logit model, 58t ordered logit model for change, 59t new urbanists in the US, 54 research approach analysis, 56-57 data, 56 multivariate statistical models, 57 trip generation

trip frequency, 60–63 use of local amenities, 63-65 urban sprawl, 53 Urban Villages Campaign, 54 Modelling domestic energy consumption fabric, 131-132 dwelling's energy efficiency, 132 household appliances use, 133–134 "mood" lighting, 134 occupancy, 132-133 owner-occupancy, type, 132 Moilanen, A., 98 Monk, S., 10 Moran, M.A., 217 Morgan, J., 109, 111 Mörtberg, U., 229 Mown grass (amenity turf), 222 Mulholland, H., 186 Multichotomous variables, 138 See also Dataset Multiple Centrality Assessment (MCA), 11, 32, 37, 113 street network systems, 32 Mumbai, 5 Mumford, K., 109 Municipal parks, 219 Muth, R., 147

N

Nabhan, G.P., 90 Nasar, J.L., 224 Nassauer, J.I., 224 Næss, P., 60, 62, 66-67 National Land Use Database (NLUD), 27 Natural environments, 92, 216-217, 220-222, 225, 230 Neighbourhood design and lifestyles Alpine Close, Maidenhead, 197f Amersham Road, Reading, 197f behaviours home-based sustainable behaviours, 188 social participation, 191 travel behaviour and car ownership, 190-191 use of local services, 192 See also Behaviours built environment and sustainable behaviour, relationship, 185-186 'technical' and 'behavioural' sustainability, contrast, 185, 185f comparison surveys, 187 Cooper Road, East Sussex, 198f

differences, behaviours study and comparison surveys, 196-200 effectiveness of new 'sustainable' housing developments, 184 findings on sustainable behaviours car owning, 208-209 cars owned/available, 209 encouraging wildlife, 206, 206f factors, encouraging to walk/cycle/use public transport, 208f home energy use and water efficiency, 203-204, 203f-204f short journeys, fuel-efficient modes of transport, 206-208, 207f social participation, 210 use of local services, 210-211, 210f waste recycling and composting, 204-205, 205f Great Notley, Essex, 198f Ingress Park, Kent, 199f neighbourhood-scale design, 183 research methods, case study locations, 186-187, 186f results, 212 residents' attitudes, 200-202 environment concerns, 201, 201f need to improve quality of life and environment, 202f reasons for choosing home, 202f The Staiths, Gateshead, 200f sustainable design features in housing schemes, 187-188 low-tech eco-design/communal-living, 188 'physical' or design features, 188 sustainability in global policy context, objectives, 187 'sustainable' developments, 192-196, 193t-194t Neighbourhoods, 36–37 case study, 261-266 between. 265-266 indicators of layout, 258t inner. 261 outer, 266 physical characteristics, 262t-265t urban layout, cost' and 'efficiency' of networks, 259t characteristics, 37 consistent spatial demographic patterns, 37 gross population densities, 37 private rented housing, 37

social housing in British cities, 37 layouts, 37-38 compact and sprawling neighbourhoods, 38f 'fractal' dimension of street patterns, 37 MCA analysis, 37 spectrum of complexity index scores, 38f -scale design, 248 and urban renaissance, 165 millennium villages/sustainable communities, 165 targets, 165 UTF, 165 Nelessen, A.C., 111 Newton, P., 25 New Urbanists in US, 54 Niemelä, J., 75, 81, 94, 216 NLUD, see National Land Use Database (NLUD) (non) linear programming problem, 150 Noonan, P.F., 111 Nowak, D.J., 87

0

ODPM, *see* Office of the Deputy Prime Minister (ODPM) Office of the Deputy Prime Minister (ODPM), 2, 25, 27, 106–107, 110, 149, 166, 218 Oliver, G., 2 'Optimal' city, 149 Organic gardening techniques, 189 Orr, A., 145–161 Owers, R., 2 Owner-occupancy, 132–133, 133f Oxford university, 35 Özgüner, H., 225

P

Panerai, P., 54
Parr, J.B., 146, 149
Parsons, R., 6, 217
Partecke, J., 86
Pataki, D.E., 83, 87
Patch species richness, 229
Pauleit, S., 6, 76, 78, 81, 216–217
Payne, L., 7
Payne, S.R., 215–231
PDAs, *see* Personal digital assistants (PDAs)
'Pedshed,' 24
Penn, A., 25
People, urban nature impacts on, 217–218 benefits of urban nature, 218

"breathing zones," London, 217 environmental exposure, 217 natural environments, 217 physiological effects of stress, measures, 217 pre- and early-industrial age, 217 social interaction/sense of community, 218 "walkable green space," Japan, 218 Perfect linear system, 38 Perkins, A., 130 Personal digital assistants (PDAs), 26 Persons per hectare (pph), 27, 36, 61t, 68t, 71t, 159t Pest control, provision of, 91-92 holly leaf-miner Phytomyza ilicis Curtis, 91 demographics, 92f, 92t Pett, J., 186 Physical infrastructure costs, 158, 160 Pickett, S.T.A., 75-76, 84, 90 Pierson, J., 108 Pilot survey, design of survey instrument, 136 Pincetl, S., 219 Pirie, G.H., 55 Planning and conception stages, city key indicators, 169 threshold and target values increase, 169 Planning applications, 155 Planning Policy Guidance Note 13, 166 Planning Policy Statement 3, 131 Plater-Zyberk, E., 7, 110 Polese, M., 106 Policies, housing, 253 Pollination, 84, 90-91 pollinator faunas, 90 Porta, S., 21-49 Porter, M.E., 148 Pouyat, R.V., 90 Power, S., 105–124 pph, see Persons per hectare (pph) Pressey, R.L., 98 Private green space, 43t, 45t, 218, 225–226, 249, 251, 255 gardens/well-being, 225 psychological processes, 225 wildlife gardening, 226 bird feeding stations, 226 garden management/specific provision of resources for wildlife, 226 richness and abundance of invertebrates, 226 "wildlife-friendly" garden features, 226 Private rented housing, 37, 39, 241 Psychological processes, 225

self-esteem/self-efficacy/personal identity, 225 Public green space biodiversity management, 222-225 concept of "mystery," 224 correlations between perceptions of safety/vegetation across, 224t "cultural cues to care," 224 large formal spaces (botanical gardens), 222 mown grass (amenity turf), 222 negatively impacted activities, 222 Sheffield park usage, 223 shrub layer volume per hectare, 224 species richness, 222 tree canopy volume per hectare, 224 "urban savannah" design, 222 "welfare of plants and animals," 222 woodland and rough grassland, 224 Public green space, 30t, 92-93, 217-226, 228-229, 231, 249-250 uses/motivations, 218-222 broader constructions of nature, 221CABE Space, 219 CityForm's case study neighbourhoods survey, 219 East Midlands region of UK, 219 ecological surveys, 219 elements of biodiversity value, 221 flora and fauna, 221 intrinsic biodiversity value, 219 mechanical sounds, 221 municipal parks, 219 natural environment, 221 Sheffield park, 219 Sheffield park interviewees, 221 "soundscape" (acoustic environment), 221 total urban space, 219 urban residents of local neighbourhood green spaces, 220t Public sector investment, 146 Putnam, R.D., 108t, 191

Q

'Quality of life,' good, 124, 245 Quantitative relationships, social acceptability 'pride/attachment' factor, 120 stability/mobility criterion, 114 Quasim, M., 76 Quigley, M.F., 88

R

Raman, S., 21-49 Rao, S., 189-190 Rapoport, A., 23 'Rateable value,' 152 spatial structure, 152t Ratti C., 130 Rebele, F., 75, 230 Recreation, 30t, 84, 92-93, 112-113, 167, 217, 219-220t, 222-224, 244, 249-250 domestic garden and human health, 93 Registered Social Landlords (RSLs), 132, 196-197 Residents' attitudes, 200-202 environment concerns, 201, 201f need to improve quality of life and environment, 202f reasons for choosing home, 202f Resistance to higher densities lower-density suburb, 166 northern regions, 167 weaker market areas, 166 Retailing, 15, 146, 152-153, 160, 251 'gravity model,' 152 level of shopping trips, 153 Revill, J., 2 Richardson, H.W., 111, 149-150 Rodgers, M.O., 84-85 Rohde, C.L.E., 217 Rollinson, D.J., 226 Rowntree, R.A., 87 Roy, D.B., 94, 217 RSLs, see Registered Social Landlords (RSLs) Rudlin, D., 8 Ryan, R., 218 Rylatt, M., 129-142

S

Salvador, P., 230 Sarmiento, S., 60 Satellite image of Sheffield, 77f Savard, J-P.L., 6 Scellato, S., 259 n2 Schoon, N., 24 Schroeder, J.W., 224 Scottish Household Survey (SHS), 56, 58, 60, 66, 67 Seeland, K., 92 SEH, *see* Survey of English Housing (SEH) Selman, P., 106 Shaftoe, H., 110 Shanghai, 5 Shaw, S-H., 55

Index

Shay, E., 57 Sheffield city, 76 distinct urban character, 76 public green spaces visitor composition, 220t visitors activities, 221t relationships between green space coverage, 228f local levels of deprivation/avian species richness, 229f people using sub-areas within public green spaces, 223f steel manufacturing industry, 76 westward sprawl of city boundary/increased population growth, 76 Sheffield park, 219 interviewees, 221 usage, 223 Sheppard, S., 10, 150 Shibata, S., 218 Shochat, E., 95 Shore, W.B., 111 Shorrock, L.D., 132 Shrub layer volume per hectare, 224 SHS, see Scottish Household Survey (SHS) Sime, J., 225 Simmel, G., 111 Simmie, J., 10 Simmonds, D., 8 Skjaeveland, O., 6 Smith, G., 81, 85, 89-90, 96, 217, 230 Smith, R.M., 81, 85, 89-90, 96, 217, 230 'Snapshot' approach, 26 Social acceptability, 105 definition, 105-111 broad classes of 'capital,' 106 equity, 106 'Quality of Life Counts,' 106 (sense of) community, 106 social capital, cohesion and inclusion, 107-108 social equity issues, 108 social sustainability/urban form, relationship, 110-111 urbanity, 106 household survey, 105 measurement, 111-113 composite measures, 112 qualitative findings, 121-124 community/attachment, 123-124 local services/spaces, 122-123 quantitative relationships, 114-121 area effects, 120-121

descriptive patterns, 114 regression modelling results, 114-118 urban form relationships, 118-120 urban form variables, individual impacts of, 117f Social acceptability, quantitative relationships 'pride/attachment' factor, 120 stability/mobility criterion, 114 Social capital, social cohesion/social inclusion social exclusion, 107 comparison, 108t outcome by location/density, 115t social networks, 107 Social composition factors, 111 Social equity, 8, 106, 108, 110, 244, 247 issues, 108 facilities, 108 opportunities, 108 services, 108 Social exclusion, 107-108, 108f, 191 Social housing in British cities, 37, 39, 241 Social infrastructure costs, 157-158 Social sustainability/urban form, relationship, 110-111 'compact city' vs. 'sprawl,' 110 Socio-economic-demographic composition, 230 Solar gain heating energy consumption, 130-131 Sorace, A., 222 "Soundscape" (acoustic environment), 221-222 Sounds, mechanical vs. natural, 221 Sound theory, application of, 8 Spatial network analysis, 32 Species richness, 88, 93-94, 97, 215-216, 221-222, 251 alien plant, 96 bee. 90 bird. 227–229, 229f, 244 of breeding birds, 94, 95f–96f, 228f native plant, 94, 95f-96f, 227 patch, 229 positively correlated with park area, 222 Speirs, L.J., 218 Spicer, J.I., 93 Sprawling layout of Glasgow, 38 Staiths, Gateshead, 200f Stanners, D., 92, 228 Stansfield, S.A., 217 Stead, D., 7, 57, 66 Steemers K., 131 Stigsdotter, U.A., 93, 217

Stone, B. Jr., 84, 85 Street patterns, 'fractal' dimension of, 37 Stren, R., 106 Stubbs, M., 2 Suhonen, J., 222 Sukopp, H., 96 Sullivan, W.C., 76, 218 Surgical recovery time, 217 Survey instrument, design of, 136 street-by-street basis, 26 Survey of English Housing (SEH), 113, 120-121 Sushchev, D.V., 90 Sustainability issues developments and sustainable behaviour, 248 - 249ecological and psychological benefits of urban green space, 249-250 intensifying neighbourhood density, 247 - 248Sustainability measurement tool application, 170 Drumoyne area, detailed investigation, 177 - 180Govan areas, investigation, 175–177 tool development, 168-170 indicators, categories, 169-170 planning and conception stages, 169 Sustainable behaviours (SB), 15, 183-189, 196-200, 212, 248-249 built environment and, 185 definition, 187 developments and, 248-249 differences between study and comparison surveys, 196-200 elements of, 248 findings on, 203-211 home-based, 188-189 survey, 186-187 sustainable design features and, 187-188 See also Behaviours 'Sustainable business parks,' 184 Sustainable city, meaning, 1 researching dimensions of, 6-12 economy, see Economic viability environmental sustainability, 6-7 social benefits, 8-9 transport, 7-8 Sustainable Communities Plan, 107 Sustainable community, 165, 168–169, 191 crime/anti-social behaviour, 107

'friendly and cooperative communities,' 107 'identity and belonging,' 107 leisure and cultural opportunities, 107 'tolerance of difference,' 107 Sustainable design features in housing schemes, 187-188 'Sustainable holiday villages,' 184 'Sustainable housing schemes,' effects of, 184 Sustainable urban neighbourhood economic characteristics, 170 urban form characteristics, 169 See also Indicators, sustainable development Suzuki, N., 218 Syme, G., 7

Т

Takano, T., 218 Talen, E., 24, 110 Taylor, A.F., 218 TCRP, see Transit Cooperative Research Program (TCRP) 'Technical' and 'behavioural' sustainability, contrast, 185 Temperature, 84-86 ecological consequences on change in, 85-86 relationships between cover by impervious surface, 86f surfaces across Sheffield, 85f Thermostatic radiator valves (TRV), 140 Thisse, J-F., 149 Thomas, C.D., 76, 80, 229 Thompson-Fawcett, M., 54 Thompson, K., 54, 230 Thorington, K.K., 95 Thwaites, K., 237 Tinsley, D.J., 6–7 Titheridge, H., 57 Total urban space, 219 Trafalgar Square in London, 23 Train, K., 67 Tram network, Sheffield, 35 Transit Cooperative Research Program (TCRP), 111 Transit oriented development, 4 Transmilenio, 4 Transport infrastructure, 14, 22-23, 22f, 28-29, 31t, 148-149, 151, 160, 240-241, 240f, 252-253, 255 accessibility, 24, 146-147 measuring accessibility and, 28-29

Index

private, 31t public, 23, 29, 31, 31t Tratalos, J., 75-98, 215-231 Travel behaviour car ownership owning car, 190-191 parking space for pool of shared cars, 191 shorter journeys, fuel-efficient modes of transport, 190 walking/cycling/public transport use, 190 distance travelled by car in Edinburgh (non-work day), 70f by car in Glasgow (work/non-work day), 69f multiple linear regression analyses/models, 67-68t ordered logit model for change in car travel, 71t Travel Diary, 67 See also Mobility and travel Travel Diary, 60, 66-67 Tree canopy volume per hectare, 224 Trip generation trip frequency, 60-63 household income, 62 number of available cars, effect of, 63 ordered logit model for trip frequency, 61, 61t-62t SHS travel diary, 60 use of local amenities, 63-65 frequency of use of classes of local amenity, 64t See also Mobility and travel Trotter, C., 5 TRV, see Thermostatic radiator valves (TRV) Tyrväinen, 6

U

UK Building Research Establishment Domestic Energy Model (BREDEM), 131 UK government agency, 228–229 Ulrich, R.S., 6, 92, 217, 219 Underground system, Glasgow, 35 United States Fish and Wildlife Service, 226 Urban ecosystems, 215, 249 infrastructure costs, types, 157 intensification, 163, 167–168 advantages, 167 goals of researches, 164

research stages, 163 vehicular traffic reduction, 167 matrix, 229 nature human impacts on, 216-217 impacts on people, 217-218 interactions between people and, 218 people and, 215-216 social-psychological perspective, 215 noise European robins Erithacus rubecula, 96 relationship between noise levels and impervious surfaces, 97 (public) space, management of, 124 regeneration and development, 164-165 residents, 215 savannah, 222 sprawl, 53 sustainability, 241, 250-253 Urban form, economic performance, 148-151 affordability constraint, 151 arguments, 149 benefits, 148 environments, 149 'optimal' city, 149 retailing, relationships, 152 role of density, land use, 148-149 Urban form elements and sustainability dimensions, relationship, 240f Urbanization, 78, 81, 83-85, 87, 90-91, 93-94, 98.216 ecology transformation, 75 effects of, 75-76 green space built environment, 76 cover/urban density, 76 impervious surface, 76 nature of urban form, 76 Urban Task Force (UTF), 1-2, 4, 8, 24, 54, 165, 171t-172t, 191 Urban Villages, 165, 169 campaign, 54 Urban White Paper, 1 UTF, see Urban Task Force (UTF) Utley, J.I., 132

V

Van Diepen, A., 9 Vegetation structure, 75, 224, 226 Vehicular traffic reduction, 167 Verheij, R.A., 217 Victorian era, 217 Vu, T., 6

W

"Walkable green space," Japan, 218 Wallentinus, H.-G., 229 Walsh, J.J., 75, 94, 216 Walters, S., 2 Warber, S.L., 217 Ward Thompson, C., 230 Warren, P.H., 75-98, 215-231 Water regulation, 84, 86-87 evapotranspiration, 87 Watkins, D., 105-124, 145-161 Watkins, R., 6-7, 10 Watson, J.E.M., 81 Webster-Brown J.G., 87 Weitzer, W.H., 220 "Welfare of plants and animals," 222 Wells, N.M., 217 Whitehead, C., 10 Whitford, V., 76, 87 Whyte, W.H., 6 Wiesner, M.R., 87 Wildlife gardening, 216-217, 226 Williams, K., 1, 9, 21, 110, 183–212 Winter, J., 24, 84, 85f-86f

Wirth, L., 111 Wittig, R., 94 Wong, N.H., 84 Woodford, G., 27 Woodland and rough grassland, 224 Woodland habitats, 227 Woods, L., 53–73 Woolever, C., 109 Wray, S., 228 Wu, J., 86 Wurzel, A., 96

Y

Yang, W., 222 Yannis, S., 6 Yiftachel, O., 106 Yli-Pelkonen, V., 81

Z

Zhang, X.Y., 86 Zhu, X., 24 Zifou, M., 111 Zonneveld, W., 4